

Towards an African perspective of urban sustainability: evidence from Ghana

Kwasi Peprah Anarfi

A thesis submitted in partial fulfilment of the requirements of Bournemouth University for the degree of Doctor of Philosophy

December 2019

Copyright

This copy of the thesis has been supplied on condition that anyone who consults it is understood to recognise that its copyright rests with its author and due acknowledgement must always be made of the use of any material contained in, or derived from, this thesis.

List of publications

Based on abridged versions of Chapter Five and Chapter Six in this thesis, the following manuscripts have been published or are under review.

1. Anarfi, K., Hill, R.A. and Shiel, C., 2020. Highlighting the sustainability implications of urbanisation: a comparative analysis of two urban areas in Ghana. *Land*, 9 (9), 300. <u>https://doi.org/10.3390/land9090300</u>

2. Anarfi, K., Shiel, C., Hill, R.A., 2020. Assessing how sustainability is promoted in Ghana's National Urban Policy and Action Plan. *Sustainability*, 12(8), 7556. <u>https://doi.org/10.3390/su12187556</u>

3. Anarfi, K., Shiel, C. and Hill, R.A. Citizens' perspectives of urban sustainability: a comparative study of two urban areas in Ghana. Under review.

Abstract

With over half of the world's population living in cities, rapid urbanisation has significant implications for urban sustainable development. While urban areas drive national economies and provide opportunities for improved standards of living, they also face significant sustainable development challenges which include socio-economic inequalities, increasing poverty and informal settlements, urban sprawl, natural resource depletion and environmental degradation, and climate change impacts. Therefore, assessing the interactions between urbanisation and sustainable development at the local level is fundamentally important for guiding more sustainable urban development. There is, however, limited empirical research and understanding of the interactions between urbanisation and sustainable development in Sub-Saharan Africa (SSA) due to a paucity of data on urban areas. This paucity of data is further exacerbated by the lack of a theoretical framework for evaluating the interactions between urbanisation and sustainable development due to the disparate nature of the two processes. To address the challenges highlighted above, this study applies a Driver-Pressure-State-Impact-Response (DPSIR) conceptual framework to explore the relationship between urbanisation and sustainable development in the context of Ghana, in order to make recommendations for urban sustainability, and to develop a conceptual model that could facilitate decision-making for the transition to sustainability in Ghanaian urban areas.

This study adopted a mixed-method approach which involved collecting and analysing qualitative and quantitative data. A case study strategy was adopted to derive data from both secondary and primary research, with Ghana as the overarching case study. Beneath the Ghana case study were sub-case studies which explored the interactions between urbanisation and sustainable development comparatively in two urban areas (Kumasi and Obuasi), and sustainable city development in Ghana (Appolonia City project). In order to achieve the aim of the study, published data were used to assess overall sustainable development and urbanisation patterns in Ghana. In doing so, sustainable development in Ghana was assessed based on four primary dimensions of sustainable development: (i) safeguarding long-term ecological sustainability; (ii) satisfying basic needs; (iii) promoting inter-generational equity, and (iv) promoting intra-generational equity. The following processes were followed in order to assess the interactions between urbanisation and sustainable development at the local urban level in Kumasi and Obuasi: (i) spatio-temporal analysis of land cover change was undertaken using remotely-sensed satellite data to assess landscape urbanisation, with published data used to assess demographic urbanisation; (ii) data from the literature were used to assess sustainable development based on selected socio-economic indicators which covered aspects of life expectancy, education and standard of living; and (iii) a citizen perception survey (n=624) was conducted to explore the subjective experiences of respondents in the context of urbanisation and sustainable development. To explore policy response to urbanisation and sustainable development in Ghana, the country's National Urban Policy (NUP) and Action Plan (AP) were analysed to ascertain alignment with sustainability. To assess sustainable city development (as a response to sustainable urbanisation) in Ghana, the Appolonia City project was examined to assess how it contributes to a sustainable urban form.

The findings of the study have confirmed the unsustainability of overall development and the rapid pace of urbanisation in Ghana. In Kumasi and Obuasi, the findings have confirmed that landscape urbanisation has led to significant urban expansion at the expense of natural vegetation, and consistent with national patterns of urbanisation, demographic urbanisation in the two urban areas has been rapid. As a result, the population density in Kumasi decreased from 11,491 people / km² in 1984 to 4,135 people / km² in 2019, compared with Obuasi where the population density increased from 2,569 people / km^2 in 1984 to 3,529 people / km². The findings have confirmed that urbanisation has contributed to improvements in some aspects of sustainable development in Kumasi and Obuasi compared with overall national performance. For example, 86% and 68% of citizens in Kumasi and Obuasi respectively had access to improved sanitation in 2017, compared with the national coverage of 15%. The findings have also established the importance of citizens' subjective experience in the context of urban sustainability and have highlighted the need to consider group differences (including gender and economic groups) in urban sustainability decision-making. In general, the findings on Kumasi and Obuasi showed that development in the two areas was unsustainable when the trade-offs between positive impacts and challenges are considered. The findings showed that Ghana's NUP provides useful initiatives to manage urbanisation in Ghana. However, in the current state, the contribution of the NUP to sustainability in Ghana's urban areas is limited. The findings showed that while the Appolonia City project contributes to sustainable city development through innovative features such as mixed land-use and solar panel designs, the overall design of Appolonia City undermines diversity and wider urban sustainability by potentially creating an elite enclave. The findings confirmed that the transition to a trajectory of sustainability in Ghana's urban areas is in a nascent stage (if a transition exists at all). The study proposes that a radical shift in policy and practice at the institutional and community levels is required to manage urbanisation at the local level into more sustainable outcomes. A DPSIR-based conceptual model is proposed to facilitate decision-making for a transition to a trajectory of urban sustainability. Overall, the findings of the study generate direct implications for policymakers in Ghana and also provide lessons for the wider SSA region.

Acknowledgement

To God be the glory for blessings accorded to me throughout this PhD journey.

I wish to extend my sincere gratitude to my supervisors Professor Chris Shiel and Professor Ross Hill for the expertise, constructive criticism, and guidance offered, as well as the patience with which they supervised this research.

My deepest gratitude and love go to my parents, Nana Anarfi Kokortoh (of blessed memory) and Theresa, for giving me the best available educational foundation, and for the support and love. I love you Pamela, Michael, Nana, and Delphina. I also extend my thanks to my siblings and family in England, Ghana, and elsewhere. I am grateful to all my friends for their love, support, and patience. Finally, to the uncountable organisations and individuals who contributed in diverse ways towards the success of this research, I say 'God bless you'.

Declaration

This thesis is submitted in partial fulfilment of the requirements for the degree of Doctor of Philosophy (Geography) at Bournemouth University, United Kingdom. I declare that this thesis is based on my original work except for quotations and citations which have been duly acknowledged. I also declare that this thesis has not been previously or concurrently submitted, either in whole or in part, for any other qualification at Bournemouth University or other institutions.

Kwasi Peprah Anarfi December 2019

Table of Contents	vii
List of Tables	XV
List of Figures	xvii
Abbreviations and acronyms	xix
Chapter One	1
1. Introduction and overview of research	1
1.1. Introduction	1
1.2 Background	3
1.2.1 Urbanisation: trends, contributory factors and impacts	3
1.2.2 Urbanisation in Africa.	5
1.2.3 Progress towards sustainable development in Africa.	8
1.2.4 Towards a sustainable world: sustainable urbanisation as a pathway to urban sustainability.	10
1.2.5 Concluding comments on background.	11
1.3 Motivation for this research	12
1.4 Research aim, research questions and specific objectives	13
1.4.1 Research questions.	13
1.4.2 Specific objectives.	13
1.5 Structure of the thesis	14
Chapter Two	17
2. Literature Review	17
2.1 Introduction	17
2.2 Urbanisation and urban policy	17
2.2.1 Urbanisation	17
2.2.2 What is 'urban'?	18
2.2.3 Global urbanisation	18
2.2.4 Urbanisation and globalisation.	20
2.2.5 Urban policy and development.	23
2.3 Sustainable development and sustainable urban transformation	26
2.3.1 Sustainable development and sustainability.	26
2.3.2 Urban sustainable development / urban sustainability.	28
2.3.3 Approaches to urban sustainability	29

Table of Contents

2.3.3.1 Smart City	
2.3.3.2 Eco-Cities	
2.3.3.3 Concluding comments on approaches to urban sustainability	
2.3.4 Sustainability Transitions.	
2.3.4.1 Transitions	
2.3.4.2 Sustainability Transitions	41
2.3.4.3 Concluding comments on sustainability transitions.	45
2.3.5 Sustainable urban transformation.	45
2.4 Urban sustainability assessment	48
2.4.1 Sustainability indicators (SI).	48
2.4.1.1 Development of sustainability indicators	49
2.4.1.2 Selection of sustainability indicators.	51
2.4.1.3 Challenges to the use of sustainability indicators.	53
2.4.2 Modelling sustainability.	55
2.4.2.1 Urban sustainability modelling.	58
2.4.2.2 Geospatial applications and urban sustainability	58
2.5 Sustainable development, urbanisation, and the urban citizen	60
2.6 Towards a framework for understanding urbanisation and sustainable develop	oment64
2.6.1 Theories of urbanisation and development	64
2.6.1.1 Modernisation and urbanisation	64
2.6.1.2 Dependency and urbanisation.	65
2.6.1.3 Urban bias and urbanisation	67
2.6.2 Utopianism and sustainable development.	69
2.6.3 Natural capital and sustainable development.	70
2.6.4 Equity and sustainable development.	72
2.6.5 Human well-being and sustainable development.	72
2.6.6 Socio-ecological systems theory.	74
2.6.7 Towards a conceptual framework for understanding the inter-relationship urbanisation and sustainable development in Ghana	s between
2.6.7.1 Theoretical assumptions	76
2.6.7.2 DPSIR conceptual framework for understanding urbanisation and sus development in Ghana.	stainable 77
Chapter Three	80
3. Data and Methods	80

3.1 Introduction3.2 Methodological choices	80
3.3.1 Questionnaire administration	83
3.3.2 Observation	84
3.3.3 Documentary analysis	84
3.3.4 Critical and systematic review of literature	85
3.4 Ethics and integrity	85
3.4.1 Ethical issues in the questionnaire survey.	86
3.5 Methodological choices specific to research objectives	87
3.5.1 Methods for Objective One.	
3.5.2 Methods for Objective Two	89
3.5.3 Methods for Objective Three	91
3.5.3.1 Study site selection.	91
3.5.3.2 Demographic and landscape urbanisation.	92
3.5.3.3 Assessment of sustainable development in study sites	98
3.5.3.4 Face-to-face survey of citizens perceptions in study sites	99
3.5.4 Methods for Objective Four.	103
3.5.5 Methods for Objective Five	106
3.5.6 Methods for Objective Six	106
Chapter Four	108
4. Urbanisation and sustainable development in Ghana: a critical analysis	108
4.1 Introduction	108
4.2 Review of urban challenges and key factors that mediate urban development	t in Ghana
	109
4.2.1 Urban challenges.	109
4.2.1.1 Environmental degradation.	109
4.2.1.2 Unemployment and informality	110
4.2.1.3 Poverty and inequality.	111
4.2.1.4 Housing	112
4.2.1.5 Transport	113
4.2.2 Selected key factors that influence urban development in Ghana	115
4.2.2.1 Politics.	115
4.2.2.2 Land tenure.	117

1223 Spatial and urban planning	118
4.2.2.5 Spatial and urban planning.	110
4.2.2.4 How pointies and fand tenure affect urban planning and development	119
4.5 Sustainable development in Ghana	121
4.3.1 An overview of progress towards sustainable development in Gnana	121
4.3.2 An assessment of sustainable development in Ghana.	123
4.3.2.1 Human Development Index (HDI) for Ghana.	123
4.3.2.2 Inequality-Adjusted HDI (IHDI) for Ghana.	125
4.3.2.3 Ecological Footprint for Ghana.	125
4.3.2.4 Gini coefficient scores for Ghana.	126
4.3.2.5 Proportion of renewable energy in total primary energy in Ghana	127
4.3.2.6 Sustainable Development Space (SDS): ecological sustainability and ba needs.	isic 128
4.3.2.7 Sustainable Development Space: ecological sustainability and intra- generational equity	129
4.3.2.8 Sustainable Development Space: ecological sustainability and inter- generational equity.	129
4.3.2.9 Sustainable Development Space: basic needs and intra-generational equ	ity. 130
4.3.2.10 Sustainable Development Space: basic needs and inter-generational eq	luity.
	131
4.3.2.11 Sustainable Development Space: inter-generational equity and intra-	
generational equity	132
4.3.3 Discussion	133
4.3.3.1 Sustainable Development Space: ecological sustainability and basic nee	ds133
4.3.3.2 Sustainable Development Space: ecological sustainability and intra- generational equity.	134
4.3.3.3 Sustainable Development Space: ecological sustainability and inter- generational equity	136
4.3.3.4 Sustainable Development Space: basic needs and intra-generational equipment	uity.137
4.3.3.5 Sustainable Development Space: basic needs and inter-generational equ	ity. 138
4.3.3.6 Sustainable Development Space: inter-generational equity and intra- generational equity	139
4.3.4 The state of sustainable development in Ghana.	140
4.4 Urbanisation in Ghana	142
4.4.1 An overview of urbanisation in Ghana	142
4.4.1.1 Urbanisation and mining in Ghana.	144
-	

4.4.2 An assessment of urbanisation in Ghana	146
4.4.2.1 Urbanisation trends.	146
4.4.2.2 Urbanisation patterns in the major urban areas in Ghana, 1970-2010	146
4.4.2.3 Rural-urban migration and natural population growth.	147
4.4.2.4 Urbanisation and economic growth.	149
4.4.3 Discussion	149
4.4.3.1 Trends in urbanisation and economic growth	149
4.4.3.2 Driving forces of urbanisation in Ghana	151
4.4.4 Sustainability implications of urbanisation in Ghana	154
Chapter Five	159
5. Urbanisation and sustainable development in Ghana: comparative case studies ir and Obuasi.	1 Kumasi 159
5.1 Introduction	159
5.2 The sustainability implications of landscape and demographic urbanisation in k and Obuasi	Kumasi 159
5.2.1 LULC Characterisation	159
5.2.2 Landscape urbanisation: Kumasi vs Obuasi	168
5.2.3 Demographic urbanisation: Kumasi vs Obuasi.	171
5.2.4 Population density in Kumasi and Obuasi	173
5.2.5 Discussion	173
5.2.5.1 Demographic urbanisation in Kumasi and Obuasi.	173
5.2.5.2 Sustainability implications of demographic urbanisation in Kumasi and Obuasi.	1 175
5.2.5.3 Landscape urbanisation in Kumasi and Obuasi.	177
5.2.5.4 Sustainability implications of landscape urbanisation	181
5.2.6 Section summary.	184
5.3 An assessment of the sustainability implications of urbanisation in Kumasi and based on selected socio-economic indicators	Obuasi
5.3.1 Education and human development in Kumasi and Obuasi	186
5.3.1.1 Literacy levels.	186
5.3.1.2 Pupil-teacher ratio in Kumasi and Obuasi.	187
5.3.1.3 Gender Parity Index in Kumasi and Obuasi.	188
5.3.2 Life expectancy and human well-being in Kumasi and Obuasi	188
5.3.2.1 Crude Death Rate (CDR).	189

5.3.2.2 Maternal Mortality Rate (MMR) in Kumasi and Obuasi	189
5.3.3 Living standard in Kumasi and Obuasi.	190
5.3.3.1 Access to improved sanitation in Kumasi and Obuasi	190
5.3.3.2 Sustainable access to safe water sources in Kumasi and Obuasi	190
5.3.4 Discussion	191
5.3.4.1 Education and human well-being in Kumasi and Obuasi: implications for sustainability.	191
5.3.4.2 Life expectancy and human development in Kumasi and Obuasi: implication for sustainability	ions 193
5.3.4.3 Living standards in Kumasi and Obuasi: implications of access to improve water and sanitation in Kumasi and Obuasi for sustainability	ed 194
5.3.5 Section summary.	195
5.4 An analysis of citizens' perceptions on dimensions of and attitudes towards urban sustainability in Kumasi and Obuasi	196
5.4.1 Demographic summary	196
5.4.2 Perceptions of the dimensions of sustainability and urbanisation	198
5.4.2.1 Citizens' perceptions of economic sustainability statements	198
5.4.2.2 Citizens' perceptions of social sustainability statements	199
5.4.2.3 Citizens' perceptions of environmental sustainability statements	200
5.4.2.4 Citizens' perceptions of urbanisation statements	203
5.4.3 Attitudes towards urban sustainability.	204
5.4.4 Group differences in sustainability perception.	206
5.4.4.1 Gender differences in citizens' perceptions of sustainability statements	206
5.4.4.2 Sustainability perception by economic (employment) status.	208
5.4.4.3 Sustainability perception by age (years) of respondents	211
5.4.4.4 Sustainability perception by level of education of respondents	213
5.4.5 General comments by citizens.	215
5.4.6 Discussion	215
5.4.6.1 Perceptions of sustainability and urbanisation dimensions.	215
5.4.6.2 Group differences in citizens' responses to sustainability statements	219
5.4.6.3 Willingness to engage in pro-sustainability practices	221
5.4.6.4. General inference from citizens survey.	222
5.4.7 Section summary.	224
5.5 Integrated Discussion	225

Chapter Six	227
6. National Urban Policy and sustainability in Ghana: a critical analysis	227
6.1 Introduction	227
6.2 Overview of National Urban Policy (NUP)	228
6.3 Assessment of urban sustainability in development and urban policy	229
6.4 Evaluating the policy approach to sustainability	231
6.4.1 Urban economies.	231
6.4.2 Fiscal sustainability.	232
6.4.3 Inclusivity and quality of life	233
6.4.4 Natural environment and resources.	234
6.4.5 Climate action and resilience	235
6.4.6 Governance and integrated planning.	236
6.5 Challenges to sustainability operationalisation in Ghana's NUP	237
6.5.1 Distributional focus and integration of sustainability dimensions.	238
6.5.2 Sustainability rhetoric	240
6.5.3 Policy coordination	241
6.5.4 Political barriers and the role of metropolitan and municipal authorities.	243
Chapter Seven	246
7. Sustainable city development and the transition to urban sustainability in Ghana	246
7.1 Introduction	246
7.2 Sustainable city development in Ghana: a case study of Appolonia City project	247
7.2.1 An overview.	247
7.2.2 A sustainability evaluation of Appolonia City.	249
7.2.3 Appolonia City: A sustainable city?	257
7.3 Barriers to sustainable city development in Ghana	258
7.4 Towards a transition to urban sustainability in Ghanaian urban areas	260
7.4.1 Towards a transition to a trajectory of sustainability in Ghana's urban areas	260
7.4.1.1 Policy and organisational context.	262
7.4.1.2 Individual and Community contexts	268
7.4.2 Conceptual model for adaptive response towards urban sustainability transition	n in
Ghana	269
Chapter Eight	274
8. Conclusions	274

8.2 Summary of critical issues and main conclusions	275
8.2.1 The extent to which knowledge of urbanisation and sustainable developmen be organised within a DPSIR conceptual research framework model	nt can 275
8.2.2 An overview of urbanisation and sustainable development in the Ghanaian o	context. 276
8.2.3 The extent and depth of interactions between urbanisation and sustainable development at the local urban level in Ghana	277
8.2.4 How does policy interpret and apply the principles of sustainability in order manage contemporary and future urbanisation towards sustainability?	to 281
8.2.5 To what extent can innovative approaches to sustainable city development be adapted to shape sustainable urbanisation within the African context?	be 281
8.2.6 Towards a trajectory of sustainability in Ghanaian urban areas: recommenda and a conceptual model to facilitate evidence-based decision-making	ations 282
8.3 Policy implications and recommendations	
8.4 Re-assessment of sustainability transitions and sustainable urban transformation	
8.5 An assessment of the validity of this research	
8.6 Limitations of the study and suggestions for further research	
8.7 What is the key contribution to knowledge of this research study?	292
References	295
Appendices	369
Appendix 1: Summary of key studies consulted in establishing indicators for sustair assessment.	nability 369
Appendix 2: Questionnaire for field survey.	371
Appendix 3: Selected responses from 'further comments' section of questionnaire	375
Appendix 4: Confusion matrices for accuracy assessment of classified land cover m	aps.377
Appendix 5: LULC change matrices for Kumasi and Obuasi, 1986–2018.	
Appendix 6: Maps showing Built-up vs Non-built-up land cover in Kumasi and Ob	uasi.384
Appendix 7: Statistics showing which economic groups were causing group different Kruskal Wallis H test for Kumasi.	nces in 388
Appendix 8: Statistics showing which economic groups were causing group different Kruskal Wallis H test for Obuasi.	nces in 391
Appendix 9: Statistics showing which age groups were causing significant interaction Kruskal Wallis H test for Kumasi.	ons in 394

List of Tables

Table 1.1: Global urban population distribution	4
Table 2.1: Definitions and assumed benefits of the factors that contribute to a sustainable	
urban form	32
Table 2.2: Inventory of selected 'sustainable city' projects in SSA	34
Table 2.3: Characteristics of indicators.	51
Table 2.4: Criteria for indicator selection.	53
Table 3.1: Dimensions, indicators, and suggested 2030 threshold values for sustainable	
development	90
Table 3.2: Source of data for dimensions of urbanisation in Ghana	91
Table 3.3: Characteristics of satellite sensors and images	94
Table 3.4: Land cover classes analysed.	95
Table 3.5: Number reference points (RPs) in the land cover class types used for training a	ind
accuracy assessment for Kumasi, 1986 - 2018.	97
Table 3.6: Number of reference points (RPs) in the land cover class types used for trainin	g
and accuracy assessment for Obuasi, 1986 - 2018.	98
Table 3.7: HDI dimensions, indicators used and source of data	99
Table 3.8: Breakdown of questionnaires administered.	102
Table 3.9: Enabling dimensions and associated goals.	105
Table 3.10: Outcome dimensions and associated goals	105
Table 3.11: Sustainable urban form matrix for an eco-city.	106
Table 4.1: Component values of Ghana's HDI, 1990 - 2017.	124
Table 4.2: Ghana's IDHI for 2017.	125
Table 4.3: Urban population, urbanisation level and urbanisation rates in Ghana compared	d
with World figures.	
Table 4.4: Contribution of rural-urban migration and natural population growth to	
urbanisation in Ghana.	
Table 4.5: GNI per capita vs urbanisation levels in Ghana, 1970-2010	.149
Table 5.1: LULC characteristics in Kumasi (1986 - 2018)	
Table 5.12: Distribution of responses for perceptions on economic sustainability statement	nts.
	.198
Table 5.14: Distribution of responses for perceptions on social sustainability statements	.200
Table 5.15: Distributions of responses for perceptions on environmental sustainability	
statements	202
Table 5.16: Distribution of responses for perceptions on urbanisation statements.	204
Table 5.17: Mann-Whitney U test results for responses in Kumasi	207
Table 5.18: Mann-Whitney U test results for responses in Obuasi.	208
Table 5.19: Kruskal-Wallis H test results for group differences (economic) in responses in	n
Kumasi.	.209
Table 5.20: Kruskal-Wallis H test results for group differences (economic) in responses in	n
Obuasi.	.210
Table 5.21: Kruskal-Wallis H test results for differences in responses based on age (years	5)
group in Kumasi	.212

Table 5.22: Effect size for age group differences in perceptions in Kumasi for Q17	212
Table 5.23: Kruskal-Wallis H test results for differences in responses based on age group	in
Obuasi.	213
Table 5.24: Kruskal-Wallis H test results for differences in responses based on level of	
education in Kumasi.	214
Table 5.25: Kruskal-Wallis H test results for differences in responses based on level of	
education in Obuasi	214
Table 6.1: Evaluation matrix for NUP and AP based on CPI sub-dimensions	230
Table 6.2: Planning models captured in Ghana's National Urban Policy.	237
Table 7.1: Characteristics of the Appolonia City project.	249
Table 7.2: Evaluation matrix for Appolonia City based on Jabareen's (2006) sustainable	
urban form matrix	250

List of Figures

Figure 1.1: Overall thesis structure and integration of chapters.	16
Figure 2.1: Characteristics of a smart city.	35
Figure 2.2: Phases of a transition	41
Figure 2.3: Intersection of the pillars of sustainability.	57
Figure 2.4: DPSIR-based conceptual framework applied to this study	79
Figure 3.1: Map of Ghana showing Kumasi, Obuasi, and Appolonia City	81
Figure 4.1: Human Development Index for Ghana, 1990-2018.	124
Figure 4.2: Ecological Footprint and Bio-capacity statistics for Ghana, 1965-2016	126
Figure 4.3: Ghana's Gini coefficient values, 1987-2016.	127
Figure 4.4: Proportion of renewable energy in total primary energy in Ghana, 2005-201	7127
Figure 4.5: The sustainable development space (SDS) for ecological footprint and huma	an
development in Ghana.	128
Figure 4.6: The sustainable development space (SDS) for ecological footprint and Gini	
coefficient.	129
Figure 4.7: The sustainable development space (SDS) for ecological footprint and renew	vable
energy	130
Figure 4.8: The sustainable development space (SDS) for Human Development Index a	nd
Gini coefficient.	131
Figure 4.9: The sustainable development space (SDS) for Human Development Index a	nd
renewable energy.	132
Figure 4.10: The sustainable development space (SDS) for Gini coefficient and renewal	ble
energy	133
Figure 4.11: Urbanisation patterns in the major urban areas in Ghana, 1970 - 2010	147
Figure 5.1: Land cover classification of Kumasi, 1986.	160
Figure 5.2: Land cover classification of Kumasi, 2000.	161
Figure 5.3: Land cover classification of Kumasi, 2010.	162
Figure 5.4: Land cover classification of Kumasi, 2018.	163
Figure 5.5: Land cover classification of Obuasi, 1986.	165
Figure 5.6: Land cover classification of Obuasi, 2000.	165
Figure 5.7: Land cover classification of Obuasi, 2010.	166
Figure 5.8: Land cover classification of Obuasi, 2018.	167
Figure 5.9: Land cover change detection map for Kumasi, 1986 - 2018	170
Figure 5.10: Land cover change detection map for Obuasi, 1986 - 2018	171
Figure 5.11: Migration statistics for Kumasi and Obuasi, 2010.	172
Figure 5.12: Literacy levels of the population aged 11 years and older in Kumasi and O	buasi.
	187
Figure 5.13: Attitudes towards urban sustainability practices in Kumasi	205
Figure 5.14: Attitudes towards urban sustainability practices in Obuasi	206
Figure 7.1: Map of Appolonia city showing land-use mix.	251
Figure 7.2: Image showing 'The Oxford' types of houses under construction in Appolor	nia
City	253
Figure 7.3: Image showing the housing conditions in a nearby Appolonia village	253

Figure 7.4: Image showing basic school in Appolonia village before and after renovat	tion. 254
Figure 7.5: Image showing transport infrastructure in Appolonia City	255
Figure 7.6: Image showing an Appolonia house fitted with solar panels	256
Figure 7.7: Adaptive DPSIR-based conceptual model for urban sustainability transition	on in
Ghana.	273

Abbreviations and acronyms

AIDS	Acquired Immune Deficiency Syndrome
AOI	Area of Interest
AP	Action Plan
BC	Before Christ
BRT	Bus Rapid Transit
CBD	Central Business District
CDR	Crude Death Rate
CDRF	China Development Research Foundation
СРІ	City Prosperity Index
CSR	Corporate Social Responsibility
DESD	Decade of Education for Sustainable Development
DETR	Department of the Environment, Transport and the Regions (UK)
DFID	Department for International Development (UK)
DPSIR	Driver- Pressure-State-Impact- Response
EEA	European Environment Agency
ECOWAS	Economic Community of West African States
EF	Ecological Footprint
EKC	Environmental Kuznets Curve
EPA	Environmental Protection Agency (Ghana)
EPI	Environmental Performance Index
ESD	Education for Sustainable Development
ETM	Enhanced Thematic Mapper
FCUBE	Free Compulsory Universal Basic Education
FDI	Foreign Direct Investment
GCIF	Global City Indicators Facility
GDP	Gross Domestic Product
GHG	Greenhouse Gas
GNI	Gross National Income
GOG	Government of Ghana
GPI	Gender Parity Index
GPS	Global Position System
GPRS	Ghana Poverty Reduction Strategy

Ghana Statistical Service
Human Development Index
Human Immunodeficiency Viruses
Human Well-Being
Information and Communications Technology
Integrated Environmental Health Impact Assessment System
International Monetary Fund
Kwame Nkrumah University of Science and Technology
Land Cover Change
Land Use / Land Cover
Land Use and Spatial Planning
Millennium Development Goals
Ministry of Local Government and Rural Development (Ghana)
Ministry of Land and Natural Resources (Ghana)
Multi-level Perspective
Metropolitan, Municipal, and District Assemblies (Ghana)
Maternal Mortality Rate
Multi-National Corporation
Ministry of Transport (Ghana)
Ministry of Roads and Highways (Ghana)
National Environmental Sanitation Strategy and Action Plan (Ghana)
National Research Council (USA)
National Urban Policy (Ghana)
Organisation for Economic Co-operation and Development
Operational Land Imager
Pupil-Teacher Ratio
Renewable Energy
Reference Point
Structural Adjustment Programme
Sustainable City
Sustainable Development
Sustainable Development Goals
Sustainable Development Space
Socio-Ecological System

SI	Sustainability Indicators
SNM	Strategic Niche Management
SSA	Sub-Saharan Africa
SSTEC	Sino-Singapore Tianjin Eco-City
SUT	Sustainable Urban Transformations
TIRS	Thermal Infra-Red Scanner
TIS	Technical Innovations Systems
TM	Transition Management
TM	Thematic Mapper
TNI	Transnational Institute
UN	United Nations
UNCSD	United Nations Commission on Sustainable Development
UNCSD	United Nations Conference on Sustainable Development
UNDP	United Nations Development Programme
UNDESA	United Nations Department of Economic and Social Affairs
UNECA	United Nations Economic Commission for Africa
UNEP	United Nations Environment Programme
UNICEF	United Nations International Children's Emergency Fund
U.S.	United States
USF	Urban Sustainability Framework
USGS	United States Geological Survey
WCED	World Commission on Environment and Development
WHO	World Health Organisation
WWF	World Wide Fund for Nature

Chapter One

1. Introduction and overview of research

1.1. Introduction

Over half of the world's population resides in urban areas (Dodman et al. 2013), and 90% of future global population growth is expected to occur in the urban areas of less-developed countries (United Nations [UN] 2014). Rapid urbanisation, arguably the most important socio-economic process of the 20th and 21st centuries, poses intractable challenges to the achievement of sustainable development (Allen 2009; Jaeger et al. 2010; Zhou et al. 2015). These challenges, which include natural resource depletion, climate change impacts, socio-economic inequalities, urban sprawl, etc., exist because the capacities of local and national governments to plan and manage the demographic transition in an efficient, equitable, and sustainable manner are generally exceeded and overwhelmed by the rapid pace of the urbanisation process (Biswas 2006). The impacts of existing structural challenges, which are exacerbated by poverty and unemployment, also mean that less-developed countries will struggle to meet the demands and difficulties of rapid urbanisation (Peter and Swilling 2012). However, the ramifications of this complex urbanisation process, especially in the global South, are still poorly understood.

The inter-linkages between sustainability, human well-being, and the function of urban areas have also attracted attention as the implications of their convergence in an era of rapid urbanisation is complex and poorly understood (Holden et al. 2008; Sassen 2011). There is a consensus that urban sustainability is pivotal for global sustainability (Bugliarello 2006; Walsh et al. 2006; Allen 2009; Fang et al. 2016) and recognition that research to explore the interactions between urbanisation and sustainability, in order to enhance understanding of the (un)sustainability of rapid urbanisation, is necessary.

For Sub-Saharan Africa (SSA), ongoing and future urbanisation requires the formulation of sustainability strategies to address present and future challenges, and to secure the integrity of urban systems. For example, as a region which relies heavily on the extraction of mineral resources to fuel national development, there is a potential risk that Africa's path to growth will fail to align with urbanisation and sustainable development, as it is believed that

resource-based growth is not naturally urban-oriented (United Nations Economic Commission for Africa [UNECA] 2012a; Turok and McGranahan 2013). Thus, from a resource-economy perspective, strategic efforts will be required in order to maximise the developmental benefits of Africa's urbanisation (Turok and McGranahan 2013) and to enhance sustainable development in African urban areas. However, the quest for urban sustainability strategies in Africa is undermined by an inherent paucity of data on urban systems (Smit and Parnell 2012), a situation made worse by the politically-charged nature of urbanisation in most SSA cities (Pieterse 2008; McGranahan et al. 2009; Jones and Corbridge 2010; Parnell and Simon 2010). Therefore, there is a need for fresh-thinking in order to understand the synergies and tensions between the urbanisation process, the resultant demographic and economic shifts and their implications for sustainable development, in a way that takes the unique conditions in SSA into account.

At present, empirical evidence on the extent and nature of the interactions between the processes of urbanisation and sustainable development in Africa is limited. Consequently, the trans-disciplinary nature of sustainable development as a concept means that the lack of a comprehensive framework of evaluation (Jabareen 2008) potentially undermines an understanding of the complex inter-relationships between the sustainable development and urbanisation processes. Therefore, this study aims to holistically re-embed an understanding of the dynamics of contemporary urbanisation within the context of sustainable development, with a view to making recommendations for a transition towards urban sustainability in Africa. To achieve this, a targeted case study in an African country, that investigates the characteristics of the urbanisation process and how it interacts with the sustainable development process, can hopefully provide a clear understanding of the general trend of (un)sustainability of urbanisation in African cities. An empirical study is presented hereafter to expand scientific knowledge which is organised mainly around the trends of urbanisation and the operationalisation of sustainable development as a concept in both policy and praxis in Ghana. Thus, this chapter highlights the context of the study with an explication of the background to the research topic. The motivation for this study is then presented. Following this, an outline of the research aim, specific objectives, and the research questions for this study are presented. The chapter concludes with an outline of the remaining chapters in this thesis.

1.2 Background

1.2.1 Urbanisation: trends, contributory factors and impacts.

Urbanisation is not a new phenomenon, but has over the last two decades re-emerged in critical international debates due to the rapid rate at which it is occurring and influencing change at local, regional, and global spatial levels. It has been defined by Cobbinah et al. (2015a, as cited in Cobinnah and Niminga-Beka 2017, p.1) as a "demographic, ecological, sociological and economic phenomenon that concentrates population in urban areas and has the potential to either fuel or impede growth and development of these areas". The main factors driving urbanisation or urban expansion include, but are not limited to, natural population increase, rural-urban migration, reclassification of rural areas as urban areas, and when people move to urban areas from other (often smaller or less prosperous) urban areas or from abroad (Sporrek 1985; McGranahan et al. 2009; Jedwab et al. 2017). Natural population increase refers to the excess population as a function of the rates of birth and death (Cobinnah et al. 2015a) and may result in congested urban areas (Jedwab et al. 2017). Rural-urban migration is explained through the notions of 'rural-push' and 'urban-pull' factors (Jedwab et al. 2017). Rural-push factors, in the African case, could be the result of poverty due to landlessness, natural disasters, and conflicts (Barrios et al. 2006; da Mata et al. 2007; Henderson et al. 2013). Urban-pull factors may affect migration flow through industrialisation and associated wage incentives (Lucas 2004; Henderson et al. 2013), urbanbiased government policies which boost utility provision (Davis and Henderson 2003), and the provision of goods and services through resource rent spending in natural resource exporting economies (Gollin et al. 2013; Jedwab 2013; Cavalcanti et al. 2014). All these factors may draw populations into urban areas. Furthermore, the rate of urbanisation increases when the boundaries of a previously rural area are expanded and classified as urban (Satterthwaite et al. 2010).

Empirical projections have estimated the world's urban population at about 6 billion people by the year 2050 (see Table 1.1). Out of the total, developing countries are expected to be home to about 5 billion urban citizens, with the remainder housed in developed countries. In the global South, SSA and Asia are projected to dominate future urban growth with over 55% of their respective populations living in urban areas by 2050, an increase on the 2010 levels of between 40% to 45%. In absolute numbers, Asia is expected to have about 3.3 billion urban residents at an average annual urbanisation rate of 1.46% by 2050, while SSA will house about 1.3 billion urban citizens at an average annual urbanisation rate of 2.87% (Cobbinah and Aboagye 2017). The projected figures suggest a substantial demographic shift which will have profound socio-economic transformation and ecological impacts, especially in less-developed regions of the world where urban growth is usually uncoordinated, fragmented, and originates from multiple nuclei (Duh et al. 2008).

	Population ('000)				Urbanisation Level (%)				Urbanisation rates (%)		
Region	1950	2000	2010	2050	1950	2000	2010	2050	1950-	2000-	2010-
									2000	2010	2050
World	745,495	2,858,632	3,558,578	6,252,175	29.44	46.69	51.60	67.18	2.69	2.19	1.41
Developed Countries	441,845	881,344	957,251	1,127,222	54.47	74.14	77.45	85.93	1.38	0.83	0.41
Developing Countries	303,650	1,977,289	2,601,326	5,124,953	17.64	40.08	45.96	64.11	3.75	2.74	1.70
Asia	245,052	1,392,232	1,847,733	3,309,694	17.46	37.44	44.37	64.36	3.47	2.83	1.46
Europe	280,602	514,545	536,611	591,041	51.27	70.80	72.69	82.17	1.21	0.42	0.24
Latin America and the Caribbean	69,264	393,619	465,246	650,479	42.38	75.49	78.84	86.62	3.48	1.67	0.84
Northern America	109,667	247,911	282,480	395,985	63.90	79.13	81.99	88.61	1.63	1.31	0.84
Oceania	7,907	21,924	25,857	40,346	62.38	70.43	70.66	73.05	2.04	1.65	1.11
Africa	33,004	288,402	400,651	1,264,629	14.36	35.56	39.19	57.70	4.34	3.29	2.87

Table 1.1: Global urban population distribution.

Source: Cobbinah and Aboagye 2017 as adapted from Cobinnah et al. 2015 a/b, UNDESA/PD 2012.

Rapid urbanisation can act as a double-edged sword for human well-being, serving as both a blessing and a curse. If managed well, urbanisation can potentially be a springboard to prosperity (especially in less-developed countries) through increased opportunities in life, as well as through social and cultural freedoms for the urban populace (Earle 2016). For example, urbanisation is associated with improved living standards and increased access to utilities such as health facilities and employment opportunities (Gong et al. 2012; Turok and McGranahan 2013). Evidence from research suggests that uncoordinated rapid urbanisation

produces negative environmental and social impacts and contributes to global environmental change (see Bengtsson et al. 2006; Grimmond 2007; Parnell et al. 2007). For example, urbanisation is linked to energy emissions (with CO₂ emissions being prominent) through agricultural modernisation, construction activities, transportation systems, as well as increased residential energy use (Ouyang and Lin 2017). An estimated 71% of energy-related greenhouse gas (GHG) emissions have been attributed to cities (Hoornweg et al. 2011). This implies that unplanned and rapid urbanisation could lead to increased GHG emissions and thereby contribute to climate change. Urbanisation also alters the urban socio-economic status and social structures, and has been linked to the spread of both communicable and non-communicable diseases (Gong et al. 2012).

1.2.2 Urbanisation in Africa.

Described evocatively as an 'urban revolution' by Grant (2015), the past few decades have seen an unprecedented increase in the scale and scope of urbanisation in SSA. This 'urban revolution' started on the back of decades of structural adjustment policies and economic stagnation in African countries and resulted in an extensive movement of people to cities in search of better opportunities. The potentially pernicious effects of rapid urbanisation on urban environments in Africa, including resource depletion and pollution due to the intensity of anthropogenic activities and population density, have also been captured in the literature by some scholars (see, for example, Bao and Fang 2012; Cobbinah et al. 2015a). With Africa expected to accommodate about 1.3 billion people of the global urban population in 2050 (UNDESA / PD 2012), it has been predicted that the continent faces potential challenges of increasing vulnerability and insecurity, effects of climate change, energy-related problems and general ecological destruction (World Economic and Social Survey 2013; Darkwah and Cobbinah 2014; Cobbinah et al. 2015a).

A striking feature of urbanisation and urban development in SSA since the 20th century has been the influence of colonialism. Scholars such as Iliffe (2007) and Fox (2012) have discussed how colonialism promoted urbanisation and urban growth through the introduction of institutions and technologies which facilitated improvements in mortality rates and food security. However, as Fox (2012) further argues, the colonial legacy of underinvestment resulted in the lack of physical and economic capacities to enable SSA cities to accommodate the influx of people attracted to the urban areas in the post-independence era. The unique

political and economic factors associated with colonialism also ensured that any form of development (as would drive urbanisation) was based on Western models and primarily aimed at serving the interests of colonial masters (Boon 2009).

Urbanisation is seen as pivotal to the economic development processes in countries as it triggers a shift from a rural agrarian society to an urban-industrial one (Davis and Henderson 2003). With industrialisation as a driving factor (Schnore 1961), urbanisation is inextricably linked to the economic development process (Davis and Golden 1954). Extant literature on urbanisation in Africa (for example, Fox 2011), however, point to a historically weaker relationship between urbanisation and economic development than witnessed elsewhere in the world. Put simply, it is argued that SSA is experiencing 'urbanisation without development' as the rate of urbanisation is not commensurate with the level of industrialisation (Turok 2013). This diagnosis is borne out of the reality that developmental growth in the majority of countries in SSA is mostly based on the extraction of natural resources including oil, gas, and mineral ores (Turok and McGranahan 2013). Between 2000 and 2008, natural resource extraction and related spending are estimated to have contributed to about 30% of Africa's Gross Domestic Product (GDP) growth (Mckinsey and Company 2010). However, the interplay between resource extraction and the associated urbanisation process has been largely ignored (Bryceson and Mackinnon 2012; Gough et al. 2013).

More relevant in the SSA context is the reality that urbanisation challenges in SSA are coupled with and aggravated by the issues of poverty and increasing informality¹ in the urban space (Afranie and Ahiable 2011; UN-Habitat 2014). To an extent, the UN-Habitat (2008, p.106) once described Africa's urbanisation process as a "widespread poverty-driven economic survival strategy". Furthermore, a World Bank Report (2000, p.130) noted that "Africa's pattern of 'urbanisation without growth' is in part the result of distorted incentives that encouraged migrants to exploit subsidies rather than in response to opportunities for more productive employment". Contrary to other less-developed regions where the majority of urban poor reside in smaller cities, Coulombe and Lanjouw (2013) show that the majority of urban poor in Africa reside in its big cities. In fact, a greater proportion of the landscape

¹ As a by-product of poverty, informality, in this context, refers to the range of businesses, individuals, and activities that accrue benefits by producing goods and services outside established legal and regulatory frameworks (de Soto 1989; Bruton et al. 2012).

and demographic change in Africa has taken place in the form of slums, and in unplanned shanty towns where poverty is pervasive and access to essential services is limited (Pauleit et al. 2018). Over half of the urban population in Africa live in slums (i.e. developments that are usually not captured by urban development plans) that remain outside the coverage of municipal services (Cities Alliance 2008). The proliferation of slums in African cities alters the configuration of urban landscapes, and the appalling conditions that prevail in such places are an affirmation of the inability of the cities to cope with the pace of urbanisation (Locke and Henley 2016). By using poverty as a measure, the literature has attributed a reduction in poverty levels across the world to the economic growth that is derived from the agglomeration effects of urbanisation. For example, according to a study on new evidence for the urbanisation of poverty by Ravallion et al. (2007, p.27), "the urbanisation process has played a quantitatively positive role in overall poverty reduction by providing new opportunities to rural out-migrants (some of whom escape poverty in the process) and through the second-round impact of urbanisation on the living standards of those who remain in rural areas". While not denying the economic benefits due to urbanisation in Africa, some studies have argued that the process has not resulted in wholesale economic gains as witnessed in developed economies in the past (Chang 2009). For example, according to an international study on urbanisation and poverty in Africa, "there is generally an unequivocal (positive) correlation between urbanisation and economic development and growth, but in Africa, this appears not to apply" (Kamete et al. 2001; as cited in Njoh 2003, p.169). This stance was reinforced by the study of Ravillion et al. (2007, p.27) whose position on Africa, based on their study, was that: "the pattern of falling poverty overall is far less evident in Sub-Saharan Africa, where the population (including the poor) has been urbanising, yet with little reduction in aggregate poverty". As a result, some writers have described the continent's urbanisation as demographic, rather than economic (Gould 1998; Songsore 2003; Songsore 2009). Demographic in the sense that it has prematurely resulted in a significant increase in the number of urban centres and people in response to push factors, such as rural droughts, conflicts, and falling agricultural prices, rather than economic agglomeration effects (Commission for Africa 2005). Others (for example, Njoh 2003; Kessides 2007) have contested the notion that the SSA urbanisation process does not deliver economic development. For example, Njoh (2003) established a positive relationship between urbanisation and economic growth in 15 out of the 24 SSA countries in her study. Others have, however, taken a balanced view. For example, according to a study that modelled the impact of accelerated urbanisation in Africa on economic growth, poverty, and migration by

Dorosh and Thurlow (2012, p.24), the authors concluded that "...urbanisation and agglomeration economies are important sources of economic growth and might well be a driver of long-term structural transformation in Africa. However, over the short-term, investing in major cities does little to address national poverty". What can be inferred from the studies above is that while agglomeration from the urbanisation process in SSA delivers benefits, the nature (i.e. pace) of the process means that, in the short-term, the economic benefits are modest and the impact is limited (Turok and McGranahan 2013). Accordingly, from a demographic perspective, Ferguson (1999) has suggested that population urbanisation should be conceptualised as two distinct and independent processes of natural population growth and migration, for which the likelihood of either becoming the most important component of urbanisation is primarily determined by the prevailing economic conditions. In the context of migration, Potts (2016, p. 261) has also argued that "migration rates go up when [urban] employment is being generated but are not sustained in the face of strong negative changes in urban economies, incomes and condition; they change in logical ways and according to the basic precepts of standard neo-classical migration theory". Thus, these findings on the variable relationship between economic growth and urbanisation in SSA provide a context for a deeper understanding of the practical realities of the interaction between urbanisation and economic growth.

The discussion above is an indication that a salient feature of Africa's urbanisation is the incidence of poverty, which has far-reaching implications for achieving sustainable development. While urbanisation has brought new economic, political, and cultural opportunities for the urban populace in SSA to enhance their livelihoods, this transformational process has also been accompanied by social challenges and threats to the ecological integrity of urban environments. Despite these trends, the consequences of this complex urbanisation process in SSA are still poorly understood, and surprisingly little systematic comparative research has been devoted to understanding the dynamics of the relationship between urbanisation and sustainable development in a holistic manner. This requires conducting research in order to help gather policy-relevant data, experiences, and knowledge resources which could potentially lead to the formulation of much-needed interventions.

1.2.3 Progress towards sustainable development in Africa.

Sustainable development (SD), as a concept, emerged in the 1980s (du Pisani 2006) as a point of reference for a new development paradigm that attempts to rethink development by incorporating environmental concerns in the pursuit of socio-economic development (Hopwood et al. 2005). From the Brundtland Report, SD is the process of pursuing *"development that meets the needs of the present generation, without compromising the ability of future generations to meet their own needs"* (World Commission on Environment and Development [WCED] 1987, p.8). The above description of SD unequivocally suggests that its achievement will involve a complex mix of social, economic, cultural, political, legal, and environmental actions (Ait-kadi 2016). The United Nations Conference on Sustainable Development (UNCSD) in Rio de Janeiro in 1992 marked the start of the global adoption of sustainable development as a concept and introduced AGENDA 21 as a way of incorporating and implementing SD in countries. The subsequent years have seen the introduction of the Millennium Development Goals (MDGs) in 2000, and the Sustainable Development Goals (SDGs) in 2015, with both providing sets of goals, based on which countries formulated and enshrined SD principles in policies and legislation.

Narrowing the focus from the global level to Africa, it has to be understood that the notion of development on the continent can best be described as 'problematic' as it has been shaped by colonial legacies which resulted in dysfunctional boundaries, exploitation, skewed development, and underdevelopment; situations which were further exacerbated in the postcolonial era by authoritarian regimes which mostly mismanaged and promoted elite-capture of State resources, thereby leading to pervasive poverty for the masses (Cloete 2015). The prospect of achieving sustainable development in the region is, thus, influenced by a range of issues including general socio-economic characteristics, cultural, and regional considerations. A report by the United Nations Economic Commission on Africa in 2012 reflected on Africa's mixed progress towards sustainable development. For example, the report (see UNECA 2012b) notes that there has been significant economic growth performance in African countries as over half of the top ten fastest growing economies in the world during the decade preceding the report were from Africa. At the same time, one statistic in the report emphasised the pervasiveness of poverty in Africa as the number of poor people in the region rose from 268 million to 306 million in the decade preceding the report (i.e. 2001-2011). Inequality in wealth distribution increased as African countries ranked poorly in the United Nations Development Programme (UNDP)'s adjusted Human Development Index (HDI). Universal primary education targets were on course to be reached, although improvements in

the quality of education were critical in some areas, and there were significant efforts aimed at empowering women. Food security and hunger challenges are critical, as food prices at the time of the report were higher than at any time since 1984. Also, unemployment in Africa is still an issue as the report showed that over 60% of the youth were unemployed, with the majority living on less than U.S.\$ 2 a day. Improvements have been made concerning access to water and sanitation. Significant progress has been made to reduce maternal mortality and the fight against diseases such as malaria, tuberculosis, and HIV/AIDS. Furthermore, Africa is not a significant emitter of GHGs and, thus, contributes little to ongoing climate change (Schwerhoff and Sy 2017). However, Africa's rapidly growing population will increase energy consumption and thus requires development strategies that do not disregard the environment, but rather contribute to effective mitigation of climate change and its impacts (Schwerhoff and Sy 2017).

Despite ongoing international and national efforts, SD remains elusive for Africa due to less-favourable conditions and extensive developmental backlogs (Cloete 2015). Given the reality that nations in Africa are among the least-developed in the world (Freitas 2013), and the continent has experienced massive population growth (Cobbinah et al. 2015a), the question thus arises: is the evidence of contemporary trajectories of development practice consistent with the principles of sustainable development in Africa? There is a need to ensure that developmental strategies are aligned to the principles of SD, and this will require systematic monitoring, measurement, and evaluation of issues relevant to SD. However, like many other less-developed regions, the stark reality for Africa has been the lack of reliable and quality data, making the idea of effective sustainable development performance evaluation at best, rhetoric (Cloete 2015).

1.2.4 Towards a sustainable world: sustainable urbanisation as a pathway to urban sustainability.

The trends and impacts of the rapid urbanisation process have implications for sustainable development (Wu 2014). On the one hand, there is the notion of urban sustainability which denotes a desirable state of an urban society which has been developed to meet the needs of people in a way that avoids unacceptable social and environmental impacts (Hamilton et al. 2002). On the other hand, is the notion of sustainable urbanisation which characterises the processes through which the earlier-described desirable state can be reached (Zhang 2016).

Recognised as an essential part of SD (UN-Habitat / DFID 2002), sustainable urbanisation has been defined as a dynamic process of urban development within which environmental, social, economic, and governance issues are given equal attention (Shen et al. 2011). Essentially, urban sustainability is considered as an intersection between the notions of sustainability and urbanisation (Bugliarello 2006) and can be viewed from the utopian lens of a 'sustainable city', which according to Whitehead (2003, p.1188) "represents an economic space within which the social, economic and ecological contradictions of capitalism are being managed and strategically addressed". Accordingly, there is a renewed understanding among stakeholders, including governments and international organisations, on the centrality of urban areas towards the goal of a sustainable global future. This is reflected in the recently-agreed SDGs, as the 11th of the 17 goals formulated aims to "make cities and human settlements inclusive, safe, resilient and sustainable" (United Nations 2015, para.1). For SSA (one of the poorest regions in the world [UNECA 2012b]), achieving SD will require the dynamic and rapid urbanisation process to be consistent with the principles of sustainable development.

1.2.5 Concluding comments on background.

The exigency for sustainable urbanisation in SSA is necessitated by the rate and scale of current and projected urbanisation in the region. Urbanisation is expected to challenge sustainable development as pressure is exerted on immediate and surrounding environments and the locus of poverty shifts to expanding and new urban areas. The rapid and irreversible pace of urbanisation in SSA requires effective and pragmatic policy responses to enable the continent to capitalise on this demographic shift, rather than create a pernicious trap for future generations in SSA. While stakeholders, including governments, international organisations, and individuals, agree that sustainable urbanisation in the context of sustainable city development is desirable, there is limited empirical evidence to guide this shift, or to suggest the form it should take, especially for geographic regions such as SSA. The literature reviewed in Chapter Two is replete with evidence of the sustainability strategies and approaches implemented in cities across the globe (mainly in more-developed nations). Differences in economic, political, social, and environmental conditions across countries, however, mean that it would be difficult to replicate strategies from one country to another. In essence, 'one size fits all' solutions to sustainability offer little or no flexibility to cities of the global South where resources for development are often limited. A fundamental

argument of this study, therefore, is that the myriad of approaches has not coalesced into solutions that fully consider the place-specific conditions in regions such as SSA, where poverty is prevalent and institutional structures are weak. As an initial step, it is critical to conduct a place-specific examination on the state of contemporary urbanisation and the state of (un)sustainability in African urban centres in order to help identify solution-oriented interventions for transformational change.

1.3 Motivation for this research

The motivation for this study is premised on the following. First, global urbanisation is proceeding at a rapid pace and it is creating intractable challenges for sustainable development in urban areas (Allen 2009; Zhou et al. 2015). Second, in Sub-Saharan Africa, the situation is exacerbated by the reality that urbanisation is proceeding at a pace that exceeds the capacities of local and national governments to plan and manage the demographic transition in an efficient, equitable, and sustainable manner (Biswas 2006). Third, there is a generally-poor understanding of the interactions between urbanisation and sustainable development in SSA due to the paucity of data on African urban areas (Smit and Parnell 2012). Thus, the main motivation for this study is premised on a desire to understand and produce new knowledge on the current patterns of urbanisation and the characteristics of the sustainable development process in the urban areas of SSA. It is based on the realities outlined above that this study intends to explore evidence from Ghana in order to explicate the interactions between urbanisation and sustainable development in the SSA context. Ghana provides an important case study for two main sets of reasons. First, urbanisation is proceeding rapidly in Ghana and has resulted in over half of the population residing in urban areas since 2010, a situation that has created challenges for achieving sustainable development in Ghana's urban areas. At an annual growth rate of 3.4%, rapid urbanisation in Ghana has complicated the government's capacity to provide basic infrastructural and social services (Farvacque-Vitkovic et al. 2008; Okeke 2014). Second, the country has successfully formulated and implemented a National Urban Policy (NUP) to guide sustainable urbanisation in Ghana's urban areas. Essentially, the expectation is that recommendations can be drawn from this study in order to inform policy-making and to identify solutions for transformational change in order to ensure that urbanisation proceeds within a trajectory of positive sustainable development.

1.4 Research aim, research questions and specific objectives

A keystone of this research is to provide evidence-based proposals to facilitate urban sustainability transitions that emanate from the findings of the analyses of place-specific areas in SSA, and which take into context the developmental characteristics and conditions prevailing in these areas. Thus, the main aim of this study is to explore the relationship between urbanisation and sustainable development in the context of Ghana, in order to make recommendations for urban sustainability, and to develop a conceptual model that could facilitate decision-making for the transition to sustainability.

1.4.1 Research questions.

In order to realise the research aim, the following research questions are posed:

(a) What are the dynamics of the interactions between urbanisation and sustainable development at the national and local urban levels in Ghana, and how can the interactions be assessed?

(b) What are the policy and development responses to rapid urbanisation in Ghana, and how do the responses align with the visions and principles of sustainability?

(c) How can decision-making for a transition to a trajectory of urban sustainability in Ghana be facilitated?

1.4.2 Specific objectives.

(1) To apply a conceptual framework that organises the interactions between urbanisation and sustainable development in an integrative and holistic manner.

(2) To evaluate the urbanisation trends and the sustainable development situation through an overarching case study of Ghana.

(3) To provide a deeper understanding of the Ghanaian situation through a comparative urban sub-case-study (Kumasi and Obuasi) that specifically: (i) examines the patterns of landscape and demographic urbanisation and the implications for sustainable development; (ii)

evaluates the sustainable development situation based on recognised indicators from the literature; and (iii) explores urbanisation and sustainable development in Ghana based on the subjective experience (perceptions) of citizens.

(4) To examine the sustainability construct and operationalisation in Ghana's urban policies in order to determine their consistency with the principles of sustainable development.

(5) To examine sustainable city development in Ghana through a sub-case-study (Appolonia satellite city) in order to establish the extent to which it aligns with sustainability as a concept.

(6) To suggest recommendations for urban sustainability transition, and to develop a conceptual model that could facilitate and enhance evidence-based decision-making for the transition to urban sustainability in Ghana, and by extension, SSA.

1.5 Structure of the thesis

Following this introductory chapter, the remainder of this thesis comprises a further seven chapters (Figure 1.1). These are briefly outlined as follows:

Chapter Two provides a critical review of the existing literature to guide this research. The first five sections of the chapter discuss relevant literature related to urbanisation and sustainable development in a way that establishes the area of the research and the research gaps addressed in this thesis. The sixth section introduces relevant theories and concepts that provide a foundation upon which a conceptual research framework for understanding urbanisation and sustainable development in Ghana is proposed.

Chapter Three presents the broad methodology and specific aspects of the methods used to answer the research questions. The philosophy underpinning this research (i.e. pragmatism) is highlighted. This is followed by a discussion of the research strategy and the methodological choices made. Other issues including ethics and data collection are discussed.

Chapter Four is the first of four chapters which present and discuss the findings. Chapter Four is the overarching case study in this thesis, and it critically assesses overall sustainable
development and urbanisation in Ghana. The first part of this chapter presents an overview of urban challenges, as well as key factors that influence urban development in Ghana. The second part considers the overall sustainable development situation in Ghana as consistent with the sustainable development principles of providing basic needs, as well as the fairness of development expressed in terms of inter-generational and intra-generational equity. The third part of the chapter considers the overall urbanisation situation in Ghana, including driving forces, and links them with issues such as economic growth. This chapter concludes by discussing the sustainability implications of contemporary urbanisation in Ghana.

Chapter Five presents the findings and discussions on a sub-case-study that comparatively explores the interactions between urbanisation and sustainable development in two urban areas of Ghana (Kumasi and Obuasi). The first section presents findings and discussion on demographic and landscape urbanisation and the implications for sustainable development in the two urban areas. The second section presents findings and discussion on the assessment of sustainable development based on selected socio-economic indicators in the two urban areas. The third section presents findings and discussion on the subjective experience of citizens in the context of urbanisation and sustainable development, which is explored through a quantitative survey of the perceptions of citizens.

Chapter Six is a critical review of the National Urban Policy and Action Plan which have been formulated and promoted to manage urbanisation in a way that leads to the sustainable development of Ghana's urban areas. The chapter considers sustainability within the documents reviewed and identifies and discusses the challenges to the operationalisation of sustainability in the policy documents.

Chapter Seven presents the findings and discussions on a sub-case-study (Appolonia City) that explores sustainable city development in Ghana. It is worth noting that the Appolonia City sub-case-study was not part of initial study design. However, after considering the initial findings from the comparative case-study on two urban areas and the urban policy analysis, and recognising the importance of sustainable city projects as evident in the literature, it became necessary, therefore, to explore whether Appolonia City is a demonstrable exemplar of practical efforts being made to promote urban sustainability in Ghana. Furthermore, Chapter Seven identifies barriers to sustainable city development in Ghana. It presents discussions on strategies for urban sustainability transition in Ghana. It concludes by

proposing a conceptual model that could facilitate evidence-based decision-making for the transition to urban sustainability in Ghana and by extension, SSA.

Chapter Eight provides conclusions and recommendations for the study. It discusses the extent to which the research questions have been answered and how the primary aim and objectives have been fulfilled. The contributions of this study to knowledge are outlined, limitations are outlined, and suggestions for future research are made.



Figure 1.1: Overall thesis structure and integration of chapters.

Chapter Two

2. Literature Review

2.1 Introduction

This chapter examines key literature and theories related to the main themes of the study in order to enable an understanding of the complexity of the interactions between urbanisation and sustainable development to be developed, with a view to providing an appropriate research framework. The remainder of this chapter proceeds in five sections. The first four sections provide a review of literature with an expectation that it will help to identify gaps in research related to the urbanisation and sustainable development processes, especially in the SSA context. The fifth section addresses scholarly and methodological issues in order to support the focus of this study and provide the foundations based on which a research framework is proposed.

2.2 Urbanisation and urban policy

2.2.1 Urbanisation.

Defining urbanisation from various perspectives and applying such definitions in an appropriate way is important in order to gain a holistic understanding of the process and its role as one of the central themes in this research. In statistical terms, urbanisation can be described as the proportion of a country's population living in well-defined urban areas (Gough et al. 2013). From a demographic perspective, urbanisation has been described as an increasing share of a population that dwells in urban areas (Poston and Bouvier 2010). From a land use perspective, urbanisation has been described as a process whereby land is converted from a previous state into an urban built-up area (Fang et al. 2016). Thus, a presumption within this thesis is that the inexorable process of urbanisation is as much a social process as it is a spatial, physical or economic process due to its overarching influence on factors such as societal organisation, as well as demographic and landscape characteristics. Others have also described urbanisation as a transformation of the population in a region from a rural to an industrial way of living (Firman et al. 2007). At the very fundamental level, what is meant by the word 'urbanisation' is the level of urbanity in a geographical region.

2.2.2 What is 'urban'?

There is no universal definition of what makes a place urban. Determining what makes an urban area mainly involves using criteria which are based on either an economic or a geographic approach (OECD and CDRF 2010). In a geographic approach, density is considered as the main criterion for urbanity, where population and housing are the units of analysis. An economic approach uses thresholds of the agricultural workforce in an administrative unit. Some scholars have, however, defined 'urban' in ways that cut across both the economic and geographic spheres. According to Albala-Bertrand (2003, p. 75), an urban area is "a set of infrastructures, other structures, and buildings that create an environment to serve a population living within a relatively small and confined geographic area". Depietri et al. (2012) have identified three classifications of urban areas based on (i) population size (where a threshold number is set, for example, > 2000), (ii) agricultural workforce (for example, the proportion of population engaged in agriculture rather than nonagricultural activity), and (iii) administration (for example, the type of local government). Nevertheless, what constitutes an urban area differs among countries, and it is determined by the approach or definition adopted by a particular country. In general, it is recognised that comprehensive international comparisons of urbanisation rates can be difficult to achieve while 'urban' definitions remain country-specific.

2.2.3 Global urbanisation.

Seto et al. (2013) have discussed the variations in the global pattern of urbanisation. Citing the examples of India, Russia, and Brazil, they argue that there is no single 'urban transition' as the rates and periods of urbanisation, land use patterns, and environmental conditions sharply differ between and within countries. For instance, Brazil reached 36% urbanisation in 1950, while India's urbanised proportion was only 31% in 2011 (Government of India 2011). In an intra-country context, Russia's urbanisation depicts one of disproportionate urban primacy as the process resulted in the creation of huge urban areas (Becker et al. 2012). From a temporal perspective, Seto et al. (2013) observe that contemporary global urbanisation varies tremendously from past urbanisation in ways that include scale, rate, and shifting geographies of urbanisation. For example, the scale of urbanisation in contemporary times is enormous as there are over twenty cities in the world with populations of ten million or more, when there was none in 1900. The scale of urbanisation leads to the creation of a hierarchy of

cities where differences in city sizes influence the available opportunities for improved life (Tselios 2014). Scott and Storper (2003, p.581) reinforce this idea as they suggest that "large cities are locomotives of the national economies within which they are situated, in that they are the sites of dense masses of inter-related economic activities and offer a wider and better quality of the producer services that are essential to technological innovation". Thus, as McCann and Shefer (2004) have noted, large cities (for example, capital cities and metropolitan regions) are shaped by urbanisation economies, while small cities are shaped by localisation economies. The factors above make large cities net gainers of resources, including institutional and human resources (Kim and Rowe 2012; Fielding 1989). Concerning the rate of urbanisation; while it took all of history until 1960 for the global urban population to reach one billion, it only took another twenty-six years for the population to double to two billion. In the spatial context, the past decades have seen a shift in the geography of urbanisation from the developed nations to developing nations, where 90% of future urban growth is expected to occur (United Nations 2014). By pulling these various strands together, it can be recognised that urbanisation has an indelible mark on major world systems. Nevertheless, there is a paucity of vital information on urban areas and processes of urbanisation (Smit and Parnell 2012; Seto et al. 2013). Smit and Parnell (2012) in their research on urban sustainability and human health from an African perspective highlight the limited data on African cities, compared to other regions, where the impacts of rapid urban growth might be less severe. Thus, although past research on urbanisation is relevant, the shift in the locus of urbanisation to less-developed countries, especially in Africa, necessitates additional studies.

As a dynamic and multi-dimensional process, urbanisation is influenced by complex changes in the environmental, socio-cultural, economic, political, and physical systems of a geographic region, usually transcending through different spatial scales (Ernstson et al. 2010; Swyngedouw and Heynen 2003). According to the World Bank (2009), it is nearly impossible for any country to develop without the growth of its cities and urban areas. This is because urban areas are estimated to account for about 80% of global GDP (McKinsey Global Institute 2012). However, urbanisation has been branded a socio-economic threat, especially for developing nations, based on arguments that sustainable development does not go hand in hand with urbanisation (Misilu et al. 2010; UNDESA / PD 2012; Cobbinah et al. 2015a). For SSA, Bryceson et al. (2009) note that ongoing urbanisation has provided a significant impetus to growth and development. For example, a country such as Ghana has seen some form of transformational growth driven by urbanisation (World Bank 2015). However, such a situation is not widespread across the region. Consequently, although most discussions about ongoing urbanisation have been about how to mitigate the negative impacts of urbanisation, others have shifted focus on how urbanisation can propel urban socioeconomic growth when the effects and benefits of agglomeration, such as financial and human capital availability, are maximised (Page 2012; Miller 2014; Locke and Henley 2016). In most cases, however, agglomeration due to urbanisation may not necessarily result in significant economic growth as the benefits may be eroded by other negative externalities, such as conflicts, environmental degradation, and higher living costs (Locke and Henley 2016).

2.2.4 Urbanisation and globalisation.

Perceived as a logical result of capitalism, globalisation has resulted in increased interexchange of goods, services, and capital amongst countries, and is especially facilitated by technological advancements (van Meerhaeghe 2012). A consensus amongst scholars is that urban areas are not stand-alone units but exist as part of an international development milieu (Macotullio 2001) and serve as nodes of globalisation infrastructure (Zhang 2016). Zhang (2016) has cited the example of Mumbai as part of a global real estate development network that attracts investors from distant cities such as London and Bogota. Furthermore, Friedmann (2005) has argued that the world is flat. In this sense, the survival of urban areas, in particular, rapidly growing ones, is impacted by forces that exist well beyond their borders. Therefore, as argued by Held (1992, p.192-202), cities have fewer options than to *"assimilate and integrate with the developing new world order for their survival"*.

An impact of globalisation is the permeation of neo-liberal ethos in the policy frameworks of countries, especially in developing countries (Todaro and Smith 2006; Jones 2010). According to Obeng-Odoom (2012, p. 89), "the political precept of neo-liberalism is that, while all essential collective decisions need to include the constitutional State, State intervention in the social and economic life should generally be minimal. Rather, the role of the State should be primarily to guarantee and ensure freedom of individual economic agents". Therefore, neo-liberalism tends to create and widen markets in all aspects of human life (Jessop 2002). According to Lee and Zhu (2006), the impact of neo-liberalism on the urbanisation process cuts across physical structures and social relationships. Critically, as

Brenner and Theodore (2002, p.15-16) have argued, the uptake of neo-liberal policies "tends to subject the majority of the population to the power of market forces while preserving the protection for the strongest". They further suggest that neo-liberal urban projects tend to 'creatively destroy' existing institutional arrangements by initiating market-oriented reforms and creating new infrastructure for market-oriented economic growth. Thus, the neo-liberal posture assumed by the ideological actors of the globalisation concept, like the Bretton Woods institutions and their progenies, has insidious influences on urban economies. In the context of urbanisation, this is explained through the notion of urban entrepreneurialism. Advanced by Harvey (1989), urban entrepreneurialism refers to the neo-liberalisation of urban governance in a way that results in the shift from urban governance forms that focus on the provision of services to an entrepreneurial form through which pro-growth strategies are pursued in partnership with private actors (OECD 2007; Lauermann 2017). According to OECD (2007), entrepreneurialism is an exploitation of market mechanisms in order to achieve public goals with minimal public intervention. The emergence of this form of urban governance, as stimulated by globalisation, is a double-edged sword as it may lead to the growth and regeneration of urban areas (Wang and Tomaney 2019), but at the same time create pernicious effects that deepen existing divisions in urban areas (OECD 2007). Here, the ability of governments to provide effective governance in an era of reduced public control due to the marketisation influence of globalisation, and its corollary urban entrepreneurialism, becomes questionable (Wu 2001). The pernicious implications of globalisation-induced urban entrepreneurialism can be explained in the context of urban social exclusion where social differentiations perceived in the form of wide-ranging inequalities in wealth, freedoms, etc., hinder the development of urban areas (Young et al. 2006). In this case, some scholars, for example, Mishra (1999), have noted that the neoliberal realignment in urban areas promoted by globalisation has never guaranteed social integration, cohesion or the survival of the social groups which its market mechanisms tend to marginalise. Of practical relevance, the impact of the neo-liberal instruments is demonstrated with the concept of 'land'. The flow of foreign direct investments (FDIs) as a result of globalisation tends to result in 'control grabbing' which has been defined by the Transnational Institute (TNI) as the "capturing of power to control land and other resources like water, minerals or forests, in order to control the benefits of its use" (The Transnational Institute, as cited in Leon 2015, p.262). Subsequently, under neo-liberal conditions, urban land is subject to speculative private acquisition, and landowners and controllers will seek to extract the maximum benefits from land such that the production of public goods, sometimes,

becomes a subordinate issue (Leon 2015). In other words, as Le Gales (2016, p.164) points out, *"land would be completely privatised, with public spaces at the mercy of private individuals. Collective conceptions of public goods would be eradicated"*. The implication of the scenario outlined above is that the rights and interests of ordinary urban citizens are potentially eroded and subject to the discretions of private sector actors and the government (Lee and Zhu 2006). Essentially, as Pacione (2007, p.248) argues, *"the goal of sustainability is not an integral component of market capitalism and will inevitably encounter opposition from entrenched interests"*.

A potentially important manifestation of globalisation is the emergence of a hierarchy of cities that serve as nodes of the global economy. Accordingly, some studies have focused on these 'world cities' and their roles in the world economy (for example, Knox and Taylor 1995; Short and Kim 1999; Sassen 2000; Short et al. 2000). Other scholars have also noted a Western elitist bias to the globalisation literature (for example Grant and Nijman 2002; van der Merwe 2004). Grant and Nijman (2002) have studied globalisation in the context of the changing geographies of developing countries. Citing the examples of Accra and Mumbai, they show that the integration of developing world cities into the global political economy is spatially fragmented. Within the urban and international economics literature, scholars have noted a lack of empirical examinations on the effects of globalisation on the urbanisation process (Gozgor and Kablamaci 2015). For example, research conducted by Nitsch (2006) found no indication of decreased urbanisation after the liberalisation of trade. Others (for example, Ades and Glaser 1995; Davis and Henderson 2003) have also attributed the urbanisation process to political factors including democratisation, rather than economic globalisation.

For SSA, the concept of globalisation has been viewed from a historical perspective as one that confirms its urbanisation experience (Njoh 2005). According to the literature (for example, Winters 1983; Hull 1976; Davidson 1970), towns in SSA had made above standard technological, social, and commercial progress which pre-dated the intrusion of European and Arabian powers. At that time, as Coquery-Vidrovitch (1991, p.21) argues, there was no need to exaggerate or minimise the urbanisation process in Africa because *"its relatively restricted development fitted well within the demographic conditions and modes of production and long distance economic relations of mostly rural and trading societies"*. Subsequently, the trans-Atlantic slave trade, together with colonialism and neo-colonialism,

eroded the influence of African cities on global trade (Njoh 2005). Njoh (2005) asserts that the contemporary process of globalisation had left African cities on the margins. The inability of African cities to actively participate in and benefit from the globalisation process could be attributed to inadequate public infrastructure and services, including water and energy shortages, inadequate transport, and communication infrastructure (Halfani 1996; Mattingly 1999). These inadequacies discouraged industrialisation in African cities as it pushed up the cost of business rapidly, thereby affecting their ability to compete internationally (Njoh 2005). Nevertheless, with increasing globalisation, not only is rapid urbanisation a massive opportunity to provide value to national development in SSA by realising prospects for socioeconomic growth and environmental sustainability, it presents itself as an urgent challenge to devise innovative ways of solving the problems associated with the phenomenon (Van Vliet 2002). It is essential to decipher what globalisation means for cities of the developing world and how they can be actively involved in the process. As both urbanisation and globalisation affect economic growth and national development, and are affected by identical challenges (for example, inadequate public infrastructure), their possible interaction must be carefully considered in any related empirical examination.

2.2.5 Urban policy and development.

Achieving the United Nations' Sustainable Development Goals (SDGs) towards making urban areas and human settlements inclusive, resilient and sustainable, will, to a large extent, depend on how these places are managed. Given the current and projected future urbanisation in Africa, there is a renewed focus on the critical and complex issues facing governments and planners in capturing the benefits while managing the risks of such urban transition. The UN-Habitat (2016a) prescribes National Urban Policies to guide the development of cities on the premise that they will result in: (i) the identification of urban development priorities; (ii) the provision of guidance on the future of development; (iii) the better coordination and guidance of actions; and (iv) increased and more coordinated investments. On a more practical note, and in the context of sustainability, urban policies have increasingly been used as instruments through which cities have tried to revitalise and reorganise their systems in ways consistent with sustainability principles (Portney and Berry 2010). The literature is replete with evidence on research carried out on urban sustainability policies. For example, Portney and Cutler (2010) in an explorative study on the pursuit of sustainability in American cities examined thirteen cities based solely on their urban policies and concluded that only seven of

the cities were on a path to sustainability. Others (for example, Portney 2003; Lubell et al. 2009) have also studied the level of sustainability commitment and activity in urban policies across cities in the United States. The general inference made from these studies is that the extent to which sustainability principles are incorporated in the urban policies of American cities, more often than not, depends on the respective local socio-economic and demographic contexts. However, the consensus within the sustainability literature is that achieving sustainability through policy requires the integration and coupling of social, economic, and environmental dimensions in policy initiatives in a way that considers inter-generational and intra-generational equity (Pelletier 2010; Vojnovic 2014).

Subsequently, in addition to their scarcity (Turok 2015), the formulation and implementation of urban policies in Africa over the past few decades have been described as haphazard (Pillay 2008). This raises serious concerns about the capacity of urban stakeholders to design and deliver essential elements of well-functioning urban areas for a continent that is only halfway through a problematic and risk-replete urban transition (Parnell et al. 2009; Bogaert 2011; Buckley and Kallergis 2014; Cheru 2014; UN-Habitat 2014). Several reasons are usually adduced for the lack of urban-centric policies in African nations. These include weak institutions, inadequate financial resources, out-dated regulations, and development agency bias (Bryceson et al. 2009; Turok and Parnell 2009; Bekker and Therborn 2012; Fox 2014).

In the context of sustainability operationalisation in urban policies, while the formulation of urban policies is evidence that cities are taking sustainability seriously, studies also show that there are challenges to the operationalisation of sustainability principles in these policies. From a planning perspective, Albrechts (2010, p.1115) has noted that a key challenge to the sustainability operationalisation in urban policies *"is to develop an approach to planning that will make these ideas and concepts "travel" and that will translate them into an array of practice arena, which in turn will transform these arenas themselves, rather than merely being absorbed within them"*. In this regard, based on his evaluation of climate action in urban policies of over 35 small and large cities, Wheeler (2008, p.488) concluded that the majority of plans *"lack the strong actions and political and institutional commitment needed to mitigate emissions*". The inference made from the author's finding is that, in the absence of commitment to translate the sustainability features of plans into concrete actions, the sustainability intentions of these policies are nothing but rhetoric. Furthermore, Davidson and Arman (2014) have evaluated the sustainability construct in urban policy and metropolitan

strategies in Australia. They concluded that the dominance of urban politics and neo-liberal urbanism hinders the potential of urban policies to stimulate transformative change in the context of sustainability, as in most cases, the priorities of political actors shape the potential of these policies to deliver transformative sustainability change. From a general perspective, Peters (2018) has highlighted the advantages and challenges of coordination for the realisation of policy objectives. Peters (2018) shows that the benefits of policy coordination include: dealing with cross-cutting problems; capturing changing demands; reducing duplication and contradictions among departments; and mitigating 'displacement' tendencies where an actor takes solo decisions without consultation that creates problems for another actor. Furthermore, Peters (2018) highlights reasons for failure in policy coordination to include: power play in government that leads to inadequate information sharing; and, in the context of performance objectives, where actors focus on their targets at the expense of collective goals. This means that the potential of urban policies to deliver sustainable outcomes is also a function of the level of coordination among actors (McGuirk 2012). The above, among other sources of evidence, cast doubts about whether the formulation of urban policies alone is enough evidence to show that cities are taking sustainability seriously (Portney 2003) and warrants an exploration of their sustainability constructs and operationalisation.

From the point of view of sustainable urbanisation, policy initiatives that will direct urban development on a local level are possible solutions for managing future challenges in the transition towards urban sustainability. Especially for the cities of Sub-Saharan Africa where urbanisation is rapidly taking place and weak urban management is pervasive (Smit and Parnell 2012), it has been suggested that NUPs are critical to harnessing the gains of urban growth and expansion (UN-Habitat 2012). A response to the international demand for delivering better outcomes in urban areas has led to the proliferation of national urban policies on the African continent (for example, Ghana's National Urban Policy in 2012). However, the scarcity of national urban policies, or where they exist, the haphazard implementation of NUPs, mean that there is little understanding about how they provide improved information and decision-support tools to adequately deal with the urban sustainability challenges of a rapidly urbanising region. This will require an examination of existing NUPs in order to determine whether their enthusiasm for an invigorated urban development agenda offers hope for transformative change in the context of sustainability

and adequately addresses the forces of ecological transition that are unleashed by the rapid rates of urbanisation in SSA as much as it does for socio-economic forces.

2.3 Sustainable development and sustainable urban transformation

2.3.1 Sustainable development and sustainability.

Sustainable Development (SD) has emerged as a leading paradigm of development (Stock et al. 2017) as popularised by the Brundtland Commission's publication of *Our Common Future* (WCED 1987). In this landmark publication, the Commission defined SD as "development that meets the needs of the present generation, without compromising the ability of future generations to meet their own needs" (WCED 1987, p.8). In the years leading up to Our Common Future, the United Nations Conference on Human Environment in 1972 brought together world leaders and scientists to discuss ways to conserve and enhance the environment for humans. Post-Brundtland, the 1992 Earth Summit in Rio de Janeiro led to several agreements including AGENDA 21, which sought to help countries on ways to integrate SD in their national strategies. Following that was the World Summit on Sustainable Development in Johannesburg, 2002, which expanded on the first Earth Summit and strengthened commitment towards achieving SD; and the UN Conference on Sustainable Development in Rio de Janeiro, 2012, which aimed at ways of achieving a 'green economy'² by providing clear and practical measures of implementing SD. In addition to the fora mentioned above, the recently-expired eight Millennium Development Goals were launched at the start of the 21st century with an overall aim of tackling poverty in all of its many dimensions. The most recent global SD effort has seen the launch of the Sustainable Development Goals which replaced and expand on the MDGs with the aim of transforming the world through 17 initiatives by 2030.

Despite the widespread uptake of SD across the globe (Barkemeyer et al. 2014), its definition remains highly contested. This is not surprising given the range of opinions and schools of thoughts on what it means to develop sustainably. The abstract and contested nature of SD and sustainability as concepts mean that alternative terminologies have evolved in creating and describing linkages among economic, environmental, and social aspects of development.

² Green Economy is one that results in improved human well-being and social equity, while significantly reducing environmental risks and ecological scarcities (UNEP 2016).

As a result, many dozens, if not hundreds, of definitions have appeared since the Brundtland Commission first defined SD (Kates et al. 2005). Scholars have offered further definitions of SD, and by the mid-1990s, over 70 definitions had been identified (Kirkby et al. 1995). Used alongside, and sometimes in place of SD, is the term 'sustainability' which has emerged to describe futuristic thinking where there is a focus on balancing economic, social, and environmental concerns when making decisions concerning the well-being of society (Jeronen 2013). Governments tend to use the term 'sustainable development' as it encapsulates developmental goals of ameliorating living standards, while scholars and international Non-Governmental Organisations (NGOs) tend to use 'sustainability' as it sets the context of living within the limits of the earth. Nevertheless, a common and overarching view held by people and scholars in different fields, governments, and international agencies is the notion of inter-generational equity, which has persisted until today. Even at the level of implementation, differences exist in how developed and less-developed countries perceive the concept. For instance, developed countries tend to stress the 'sustainable' aspect of SD while less-developed countries tend to stress the 'development' aspect (Zhang et al. 2018). Such discrepancies affect the uptake of SD. For example, in less-developed countries where basic needs are increasingly unmet (WCED 1987), meeting such needs, even at a pernicious cost to the environment, might appear justified (Redclift 2005). The scenario above is succinctly expressed by the Brundtland Commission (WCED 1987, p.72) as they stated that "those who are poor and hungry will often destroy their immediate environment in order to survive: they will cut down forests; their livestock will overgraze grasslands; they will overuse marginal land; and in growing numbers they will crowd into congested cities. The cumulative effect of these changes is so far-reaching as to make poverty itself a major global scourge". Thus, as a prominent economist and sustainability scholar, Daly (1991, p. 248) once argued, a "lack of a precise definition of the term 'sustainable development' is not all bad. It has allowed a considerable consensus to evolve in support of the idea that it is both morally and economically wrong to treat the world as a business in liquidation". Furthermore, the openness and malleability of SD's definition mean that it is adapted to deal with diverse challenges across space and time (Kates 2005). The intricacies of SD as a concept, and its persistence on national and international agendas are, thus, summed up by Lafferty (2004, p.26) who suggests that SD "is now like 'democracy': it is universally desired, diversely understood, extremely difficult to achieve, and won't go away". Therefore, drawing predominantly, but not exclusively, on the Brundtland Commission's definition of sustainable development, this thesis will try to present SD as a concept that aims at balancing economic growth, human well-being, and ecological protection, and is underpinned by a desire to provide equal opportunities among and between generations. In addition, the conflation of sustainable development and sustainability as concepts means that they will be used interchangeably in advancing arguments in this thesis.

2.3.2 Urban sustainable development / urban sustainability.

Kofi Annan, a former General Secretary of the United Nations, once said: "the future of humanity lies in cities....Weak cities will almost certainly act as a brake on national development. Strong cities can be a key factor enabling a country to thrive in the global economy" (United Nations 2005, para. 8). As urban areas continue to grow at a higher rate than any other land-use type (Antrop 2000; Hansen et al. 2005), the proportion of the global human population living in urban areas is expected to rise to 70% by 2050 (United Nations 2008), representing over 6 billion of the projected 9 billion people. From a resource-use perspective, over 75% of the planet's resources are consumed by the global urban areas in achieving sustainable development is reflected in the notion that they serve as battlegrounds on which the battle for sustainability will be won or lost (Clark 2003; Gosling 2006).

Although the United Nations and other world bodies have sought to drive global urban sustainable development through various international fora and guidelines such as the SDGs, the contested nature of SD makes it too simplistic to assume that urban areas in developing and developed countries will undergo similar processes of transformation. On a local to a global scale, pursuing a balanced developmental agenda between and among cities reflects the more extensive sustainable development challenges because urban areas form part of broader physiographic and politico-administrative regions which may extend beyond national boundaries (Simon 2010). Taken together, the underpinning motive of achieving intergenerational equity in the quality of life and ecological footprints could be realised if urban development is carried out in a way such that there is a balanced mix of social, economic, and environmental needs. However, what is meant by urban sustainability? Like other derivatives of SD, there is no standard definition of urban sustainability. However, in its simplest terms, urban sustainability has been defined as *"the process of developing a built environment that meets people's needs whilst avoiding unacceptable social or environmental impacts"*

(Hamilton et al. 2002, p.1). From an ecological perspective, Wu (2014, p. 213) has defined urban sustainability as an "adaptive process of facilitating and maintaining a virtual cycle between ecosystem services and human well-being through concerted ecological, economic, and social actions in response to changes within and beyond the urban landscape".

As shown earlier in this thesis, despite the varied views and definitions of SD, it has had a profound political impact and attracted notable international interest. Thus, in practical response, many programmes, standards, and legislation have been introduced at different spatial levels by governments and international agencies, giving material substance to SD. However, despite the multitude of global and local-level initiatives aimed at making urban areas sustainable, the complexity of urban challenges means that conceptualising progress in the transition towards sustainable cities is problematic. Williams (2010) has observed that there is a conflicting picture of what sustainable cities are supposed to be, and what has happened in reality. The author points out that the popularisation of sustainable development, the infusion of sustainability into urban policies and exemplary initiatives seen around the world in the last three decades, evidences a favourable shift towards making urban areas sustainable. However, the author argues that although the sustainable city concept might have great appeal, it is complex and intangible, so the reality on the ground might be different to and contradict the generally favourable picture painted. For example, the author notes the proliferation of massive construction projects in the global North in ways that disregard any notions of sustainability but are nevertheless heralded by the public and officials. A different scenario presents itself in the global South as increased urbanisation and population growth result in more serious urban challenges in the face of scarcer resources. As Bulkeley and Betsill (2005, p. 42) put it, "despite ... near universal recognition that sustainable cities ... are a desirable policy goal, there is less certainty about what this might mean in practice". Furthermore, Childers et al. (2014) have also argued that there is no ultimate state to characterise a sustainable city per se, as they view sustainability as a process and not an endpoint. Rather, they argue that the goal of urban sustainability is multi-faceted with continuously changing targets. Thus, cities may pursue this goal by following more (or less) sustainable trajectories.

2.3.3 Approaches to urban sustainability.

As the world has become more 'urban', the inter-linkages among sustainability, human wellbeing, and the function of urban areas have gained scholarly and political attention (Holden et al. 2008; Sassen 2011). Subsequently, the fuzziness of sustainability as a concept tends to make its constitution and development rather an open question (Jabareen 2008). As a result, urban sustainability has been viewed from the utopian lens of a 'sustainable city' (SC) and in its widest interpretation; it does not only focus on environmental, economic, and social dimensions, but emphasises the involvement and participation of urban citizens in the pursuit of sustainability (Bugliarello 2006; Munier 2007). The 'sustainable city' concept is also viewed as an automatic derivative of the sustainable development in the context of urban development (de Jong et al. 2015). In essence, the SC concept elaborates on the tenets of sustainability discourse and practice, placing it firmly within the context of urbanisation and development.

Accordingly, several new city categories have emerged in urban policy discourse and practice aimed at capturing and conceptualising key aspects of urban sustainability. Although not exhaustive, de Jong et al. (2015) have identified these new city categories to include sustainable cities, green cities, liveable cities, digital cities, intelligent cities, eco-cities, resilient cities, smart cities, and low carbon cities, etc. According to Caprotti (2018), the greater majority of visions for new and future cities are dominated by conceptualisations of highly technical urban futures, cities as economic spaces and cities as efficient spaces. In the particular context of urban efficiency, Caprotti (2018) suggests that while conceptualisations of efficiency in cities may lead to efficient public services in future cities (for example, transport), such conceptualisations may lead to citizens who are able to promote efficiency in urban life being characterised as 'worthwhile citizens', while citizens who are not in a position to promote efficient acceleration of urban life may be discriminated against and potentially controlled. Furthermore, Jabareen (2006) has outlined the characteristics of a sustainable urban form to include factors such as density, compactness, mixed land-use, diversity, greening, passive solar design and sustainable transport. Consequently, Bosehans and Walker (2020) have provided a summary of the definitions and assumed benefits of Jabareen's (2006) sustainable urban form factors (shown in Table 2.1). In the context of sustainable city implementation in SSA, van Noorloos and Kloosterboer (2018) have conducted research on the contested future of urbanisation in Africa. A key output of their research was producing an inventory of current and planned new cities in Africa (Table 2.2), as evidence of the mainstreaming of sustainable city concepts in SSA. From Table 2.2, it is

seen that the mainly 'new city projects' in Africa have been branded by their proponents as 'smart', 'eco', and 'satellite' cities. They exist in a variety of forms including newly-built cities, retrofitted urban spaces, as well as extensions or in-fills of existing cities (see Joss 2011a).

While the city categories identified by scholars (for example, de Jong et al. 2015; van Noorloos and Kloosterboer 2018) may vary in their conceptualisation of urban sustainability, they promote normative visions of a 'sustainable city' / urban sustainability, and they are fundamentally guided in principle by the synergies among ecological, economic, and social sustainability in order to "*foster economic growth and development through urban activities that reduce negative environmental externalities, the impact on natural resources and the pressure on ecosystem services*" (Hammer et al. 2011, p.34). A few of them will be considered in the following paragraphs as relevant to the research objectives. The review of these selected approaches will enable generalisations of the broader range of approaches of interest to urban sustainability.

Table 2.1: Definitions and assumed benefits of the factors that contribute to a sustainable urban form.

Factor	Definition and Benefits						
City	<i>Definition</i> : Measurable geographical space that a city consumes: the less						
compactness	space consumed, the more compact the city (Zhao et al. 2014).						
	Benefits: Enables high-density residential, industry and transportation						
	developments (Ye et al. 2015), may reduce travel distances (Hamiduddin						
	2018).						
Density	Definition: Ratio of people or dwelling units to land (high-density						
	development is often achieved by vertical expansion through mid- and hig						
	rise building projects; e.g., Koziatek et al. 2016).						
	Benefits: While low-density development may be related to increased GHC						
	emissions (Baur et al. 2013), high-density development allows more people						
	to live and interact in a narrowly defined space.						
Mixed land	Definition: Horizontal mixed land use – that is, an agglomeration of						
use	multiple land uses including residential, commercial, industrial, communal,						
	institutional and recreational on a district level – is a key criterion for						
	compact city development (Abdullahi et al. 2018).						
	Benefits: Lower (central-) agglomerative forces may lead to a greater						
	dispersion of employment, shorter commute distances and reduce						
	congestion (Wheaton 2004).						
Diversity	Definition: Different housing types and densities, varying household sizes						
(social	and incomes, as well as a broad representation of different age groups and						
mixing)	cultures (Jabareen 2006).						
	Benefits: Lowers travel demand due to access to the same local resources,						
	shapes the social and cultural context of the city and may prevent the						
	segregation of population segments (Fincher and Iveson 2008; Fincher et al.						
	2014; Holloway et al. 2012).						
Greening	Definition: Presence of natural green space such as urban parks or						
	community gardens for recreation and leisure						
	Benefits: Urban green space provides numerous social and public health						
	benefits (Hartig et al. 2014; Wolch et al. 2014), may increase people's sense						
	of safety in inner-city neighbourhoods (Kuo et al. 1998) and improves the						

	likelihood of being exposed to biodiversity (Strohbach et al. 2009).					
Passive	Definition: Construction of "building and mechanical systems that are used					
solar design	to convert the sun's energy to useful thermal energy for the heating and					
	cooling of buildings" (Anderson and Michal 1978, p.58). Can be influenced					
	by the density and type of buildings, building design, the use of water and					
	vegetation etc.					
	Benefits: Minimises losses of energy (e.g., heat storage), but also maximizes					
	the gains of passive energy (e.g., solar panels), and may create a more					
	appealing and pleasant urban microclimate (Vallati et al. 2015).					
Sustainable	Definition: Enabling movement by sustainable means of transport such as					
transport	walking, cycling or public transport through pedestrian-friendly					
	neighbourhood design, traffic calming, good street connectivity, high					
	density and a high degree of mixed land use that reduces the physical					
	separation of activities (Ewing and Cervero 2010; Jabareen 2006).					
	Benefits: Conserves energy and resources and promotes physical activity in					
	a multitude of ways (e.g., Handy et al. 2002; Heath et al. 2006).					

Purpose of city	New capital city	Lower and middle- class housing	High class housing and comm. dev	Mixed-use: technology / innovation / residential	Special Economic Zones				
	Examples								
New-build	Egypt: Cairo	Ethiopia: Addis Ababa	Ghana (Accra): Appolonia City; Nigeria (Abuja): Jigna.	Morocco: King Mohammed VI Green; SouthAfrica (Johannesburg): Steyn City.	Mauritius (Port Louis): Jinfei; Nigeria (Lagos): Lekki.				
Urban expansion (In-fills) Retrofit		Ethionia	DRC (Kinshasa): La Cite du Fleuve; Kenya (Eldoret): Segoit.	Rwanda: Kigali 2020 inner	Mozambique:				
Kenom		restructuring.		city restructuring; Mozambique; Beira restructuring.	Palma Development.				

Table 2.2: Inventory of selected 'sustainable city' projects in SSA

Source: Reproduced from van Noorloos and Kloosterboer (2018, p.1231-1232), and adapted from Joss (2011a). Note: bold font = built or some elements operational; regular font = planned / announced.

2.3.3.1 Smart City.

The genesis of the smart city concept can be traced to the era of the Smart Growth Movement in the 1990s (Neirotti et al. 2014). Promoted as a new form of SD, it has attracted significant scholarly attention, yet with an inexplicit description (Hollands 2008; Vanolo 2013). The smart city concept was borne out of the parallel desires of cities to achieve urban sustainability while remaining competitive. The concept has gained traction in urban spheres

as cities look for innovative ways of managing the increasing population numbers and physical expansion. Although used in various contexts and sectors, a definition of the smart city concept which significantly situates it in the urban sustainability discourse is provided by Sta (2017, p.410) as a modern city that "uses smart information infrastructure (contains perfect data) to ensure the sustainability and the competitiveness of the different urban functions by integrating different dimensions of urban development and investments in order to reduce the environmental impact and to improve the quality of citizens' lives". A working paper on smart cities sponsored by the United States Department of Energy suggests that the vision of "smart cities is the urban centre of the future, made safe, secure, environmentally green and efficient because all structures.... are interfaced with computerised systems" (Hall et al. 2000, p.1). The idea of a smart city has been fundamentally enhanced by the rapid evolution in computing and information and communications technology (ICT) in the last five decades which is known to have had profound effects on all dimensions of society (Kramers et al. 2014). Therefore, at the core of the smart city concept is the opportunity to leverage technology in order to address the challenges that cities face in a 'smarter' way (Sujata et al. 2016). That is to say, the concept offers an opportunity to innovatively reassess how cities are planned and developed through the amalgamation of ICT and sustainability principles (Höjer and Wangel 2015). The literature shows that although the concept does not holistically describe a city with particular attributes, its different models indicate the characteristics that make up an ideal smart city (see Figure 2.1) with individual components covering aspects such as natural resources, participation, quality of life, competitiveness; transport and ICT; and social and human capital (Giffinger et al. 2007; Batlle-Montserrat et al. 2016).



Figure 2.1: Characteristics of a smart city.

Source: Giffinger et al. (2007); Batlle-Montserrat et al. (2016).

In their research into the development of smart cities, Sujata et al. (2016) have also identified the pillars of a smart city initiative to include social, management, economy, legal, technology, and sustainability aspects. The aforementioned pillars, they argue, are crucial if the smart city concept is to be holistically appreciated. From the 'social' perspective, they observe that any smart city initiative should enable the active participation of citizens in its design, governance, and management through channels of communication made available by ICT. For example, in addition to traditional modes of communication, contemporary modes such as social media afford citizens easy access to city data and plans, thereby producing well-educated and informed citizens who can influence the failure or success of a city project. Regarding 'management', the authors maintain that a shift away from traditional governance to e-governance will enhance the efficiency and effectiveness of urban administration as ICT provides the platform to effectively identify and incorporate the needs of different target groups, thereby improving decision-making, policy-making, and public governance. A smart city must be legally compliant and provide innovative ways of tackling regulatory issues. The authors further suggest that a smart city must exploit the rapid evolution of technology in order to facilitate real-time decision-making. On sustainability, the authors highlight the need for a smart city to "become more competitive for capacity, opportunity and investment by providing an enhanced quality of life, as well as reducing the environmental consequences of urban life by decreasing its carbon footprint" (Sujata et al. 2016, p.5). The expected benefits of a smart city approach to urban development have been identified by Sta (2017, p.411) to include:

(a) "Making new administrations accessible to residents and suburbanites, for example, constant direction on how best to use various transportation modalities.

(b) Enhancing business ventures through the distribution of continuous information on the operation of city administrations".

Despite the perceived benefits of the smart city concept, some of the literature shows that integrating ICT in different urban settings could potentially undermine the functionality of an urban system. Among the broad range of potential problems are issues of confidentiality, where every move of citizens connected to urban systems are monitored (Smart Cities 2013), the prohibitive costs of installing digital infrastructure (Anthopolous and Fitsilis 2010), and potential forms of exclusion due to inaccessibility to ICT infrastructure (Davis 2010). By

using China as an example, Caprotti and Liu (2019) argue that the mainstreaming of digital technologies in urban systems could potentially redefine citizenship for existing citizens. They suggest that China's social credit systems (SCS) which are designed to reward 'good behaviour' and penalise 'unacceptable behaviour' represent a direct governmental interference in urban living as the behaviour of citizens are judged by what is considered as socially acceptable or not. Overall, Sta (2017) succinctly describes the challenges of the smart city approach as a problem with 'information infrastructure' where prevailing imperfections can have potentially detrimental effects on urban services and decision-making.

2.3.3.2 Eco-Cities.

Eco-cities represent a conceptual model of urban development which is premised on a potential urban future in which there is compatibility between urban growth and ecological processes (Chang and Sheppard 2013). A concept introduced by Richard Register in 1987, he defined an eco-city as "an urban environmental system in which input (of resources) and output (of waste) are minimised" (Register 1987, p.6). In the years following his definition, the concept became more widespread, and according to Roseland (1997), instead of assigning a single definition to eco-city, it rather represents a constellation of ideas that cuts across concepts including urban planning, socio-economic development, housing and transportation, all aimed at making cities sustainable.

Joss (2010) has given a historical account of the emergence of eco-cities. He identified three phases of eco-city development, the first of which was the period between the 1980s and 1990s when real examples of eco-cities were rare. The Earth Summit in 1992 and the subsequent adoption of Agenda 21 typified the second phase which led to the development of eco-city exemplars, including Germany's Freiburg District, New Zealand's Waitakere, and Brazil's Curitiba. The third and current phase, which started at the turn of the 21st century, has seen the mainstreaming of the eco-city concept as a global way of addressing sustainability challenges underpinned by rapid rates of urbanisation and increasing focus on climate change. Thus, as of 2011, there were more than 178 eco-city initiatives around the world (Joss et al. 2011). A critical factor that has driven the proliferation of eco-city projects has been the involvement of the private sector with wide-ranging inputs which include: project initiation and planning facilitation; financial, technical, and structural inputs (Joss 2011b).

Focusing on the global South, many countries, especially in Asia, have embraced the development of eco-cities in response to the global call for sustainability. For example, China (seen as a leader in eco-city development in Asia [Wu 2012]) has embarked on large-scale construction of eco-cities in order to address rapid urbanisation and environmental degradation (Caprotti 2014). Accordingly, in China, Caprotti et al. (2015) have assessed the wider sustainability characteristics of the Sino-Singapore Tianjin Eco-city (SSTEC) project. They argue that the SSTEC project is *"discursively constructed as ecologically beneficial for its inhabitants rather than for the broader socio-environmental landscape"* (Caprotti et al. 2015, p.495). This is because while the SSTEC project is dominated by visions, designs, and technologies that could potentially lead to the creation of an *"exclusive 'green', 'safe' and 'comfortable' urban area"*, the extent to which these visions and technologies cater for social sustainability in SSTEC is limited (Caprotti et al. 2015, p.511). The optimal features of an eco-city have been identified by Premalatha et al. (2013, p.2) based on earlier literature (Roseland 1997, 2001; Urban Ecology 1996) which require that eco-cities should:

- (a) "have land use priorities such that it creates compact, diverse, green and safe mixeduse communities around public transportation facilities;
- (b) have transportation priorities such that it will discourage driving and emphasise access by proximity;
- (c) restore damaged urban environments;
- (d) create affordable, safe, convenient and economically mixed housing;
- (e) nurture social justice and create improved opportunities for the underprivileged.
- (f) support local agriculture, urban greening and community gardening;
- *(g) promote recycling and resource conservation while reducing pollution and hazardous waste;*
- (h) support ecologically sound economic activities while discouraging hazardous and polluting ones;
- *(i)* promote simple lifestyles and discourage excessive consumption of material goods; and
- *(j) increase public awareness of the local environment and bioregion through educational and outreach activities*".

2.3.3.3 Concluding comments on approaches to urban sustainability.

The discussion on approaches to urban sustainability presented above suggests that dealing simultaneously with the rapid wave of urbanisation and managing existing development will

require a paradigm shift, which, Suzuki et al. (2010) argue, must be fundamentally guided by the systematic adoption of best practices across the globe. Therefore, the discussion above has potentially created a better understanding of the respective approaches and how they contribute to the current debate on urban sustainability. Furthermore, because these approaches to urban sustainability have been developed and implemented mostly in the context of developed economies, their potential application elsewhere requires that attention is paid to the specifics of the context. As a place-dependent notion (Pow and Neo 2013), urban sustainability in a particular city will be challenging to realise based on generalisations. Particularly for this research, as rapid urbanisation is expected to continue in SSA, new strategies for urban sustainability are necessary. At this point, a key question that emerges is how do cities move towards the path of achieving urban sustainability? This directs the focus of the literature review to the concepts of transitions and transformations, as will be discussed in the next sub-section.

2.3.4 Sustainability Transitions.

2.3.4.1 Transitions.

Since the advent of SD, there has been increasing scholarly attention given to ways of realising the transition towards sustainable futures. Thus, the last few decades have seen the conception of transition studies aimed at grasping and influencing transitions (Loorbach and Rotmans 2006). Transitions have been defined by Rotmans and Loorbach (2009, p.85) as *"radical, structural changes of societal (sub) systems"*. The process essentially entails extensive changes that cut across various dimensions of society including: *"technological, material, organisational, institutional, political, economic, and socio-cultural"* (Markard et al. 2012, p. 957); and results in interactions among different actors, institutions, material artefacts, and knowledge (Geels 2004; Markard 2011; Markard et al. 2012). Transitions manifest within societal systems, which are described as parts of *"a society to which a function can be attributed"* (Frantzeskaki and de Haan 2009, p.594).

Rotmans et al. 2000 (as cited in Lachman 2013, p.2) have summarised the characteristics of transitions to include the following:

(a) "they inhibit developments that take place within economic, technological, political, environmental and social spheres that affect each other;

(b) transitions involve various actors from different groups;

(c) transitions are radical shifts (in scope) from one configuration to another;
(d) because of the multiple developments that are intertwined, the multi-actor nature and the existence of radical shifts, transitions have an inherent complexity and uncertainty; and
(e) complexity and uncertainty add to the fact that transitions are long-term processes".

In order to conceptualise transitions, it is important to understand the functional components within and outside a societal system that shape how the system meets the needs of society. Frantzeskaki and de Haan (2009) have identified these functional components to include regimes, niches, niche-regimes, and landscapes. Taking each briefly, a regime has been conceptualised as a dominant functioning of a societal system which directs it along an established pathway of development (Frantzeskaki and de Haan 2009; Markard et al. 2012). An example of a regime is the mobility system in the Western world, which is dominated by fossil-fuel based personal mobility. A niche represents a protected space within which a deviant or novel functional sub-system prevails without pressure from an incumbent regime (Frantzeskaki and de Haan 2009; Markard et al. 2012). Compared to regimes, niches are less powerful and require support (be it political or financial) to progress. Niches, therefore, manifest as sub-systems, which foster the growth of innovations which can replace existing regimes. An example of a niche is low carbon advocacy. A landscape can be described as the environment of a societal system and includes everything that surrounds it (Frantzeskaki and de Haan 2009; Markard et al. 2012). Forces within the landscape can exert pressures which potentially create avenues for niches to grow and replace existing regimes (Markard et al. 2012). A landscape is exemplified by the imposition of a particular political ideology by a government. A niche-regime is a more powerful and competitive alternative functional subsystem which can co-exist with an incumbent regime for a long time (Frantzeskaki and de Haan 2009). An example of a niche-regime can be the emergence of low emission electric vehicles which are expected to co-exist with fossil-fuel based vehicles.

For a transition to occur, there must be an interplay among the previously-mentioned functional subsystems (Raven et al. 2010). From the transition literature, this interplay may take place in phases, namely: a pre-development phase, a take-off phase, an acceleration phase, and a stabilisation phase (Rotmans et al. 2000), as shown in Figure 2.2. First, there is a pre-development phase where no visible change occurs in the state of the system. This is followed by a take-off phase where the status-quo begins to change. An acceleration phase follows where there is a radical and visible accumulation of changes in the socio-cultural,

economic, ecological, and institutional dimensions of the system reacting with one another. Finally, there is a stabilisation phase where the rate of change decreases and a new dynamic equilibrium is reached.



Figure 2.2: Phases of a transition.

Source: Rotmans et al. 2000 (as cited in Rotmans and Kemp, 2003).

2.3.4.2 Sustainability Transitions.

In the context of sustainability, the notion of transition is increasingly used to characterise a "radical transformation towards a sustainable society as a response to a number of persistent problems confronting contemporary modern societies" (Grin et al. 2010, p. 1). Thus, from the view of sustainability transition scholars, resolving deeply-rooted problems of existing societal systems, for example, in areas such as energy, climate, water, transport, etc., will require a shift to a new system in which sustainability issues are at the front and centre

(Vob et al. 2009). For example, the inherent threat of climate change on existing sociotechnical systems resulting from the unsustainable manner of production and consumption patterns cannot be remedied with 'end of pipe' solutions, but rather necessitates a shift towards more sustainable systems (Raven and Verbong 2009; Lachman 2013). Essentially, the transition to sustainability is expected to interrupt and destabilise contemporary sociotechnical structures and nurture substitute systems that emerge to replace the vacuum created by structural change (VoB et al. 2009).

From the foregoing, it can be inferred that the notion of a sustainability transition is a longterm one. According to the literature, a sustainability transition could potentially span decades or generations before being achieved (Kemp and Loorbach 2003; Alkemade et al. 2011). Within the transition literature, scholars have identified some characteristic attributes of sustainability transitions that make their attainment long-term rather than short-term (Elzen et al. 2004; Geels 2007; Stirling 2007). First, the contested nature of sustainability results in diverse views about the preferred pathways to sustainability, the appropriateness of prescribed policies and the efficacy of proposed sustainability solutions (Van Geenhuizen and Ye 2014). This is further compounded by the goal-oriented nature of sustainability transitions, which often results in parallel postures from the private and public-sector perspectives. For example, Van Geenhuizen and Ye (2014) have argued that private actors are less-incentivised to promote sustainability transitions as they tend to behave in a freeriding manner because they see the goal of sustainability as a collective good which is beyond their responsibilities. However, the public sector is expected to support the collective good by stimulating economic condition changes and supporting 'green' niches (Smith and Raven 2012). Second, solutions to sustainability issues do not offer instant benefits to society and are sometimes more expensive to achieve compared to conventional ways of doing things (Van Geenhuizen and Ye 2014). This implies that for a transition to materialise, there must be changes in all facets of a societal system. However, these changes will likely not occur without a competing clash between those with vested interests and those propagating the new way of doing things. A transition towards sustainability is often unpredictable and not achieved quickly because of the complexities and uncertainties associated with the process (York and Venkatamaran 2010; Rotmans and Loorbach 2009; Grin et al. 2010; Van Geenhuizen and Thissen 2002). This may be due to less clear cause-and-effect relations, unintended and adverse effects of policy prescriptions, as well as unexpected changes (Van Geenhuzen and Ye 2014).

The concept of sustainability transitions has resulted in a multitude of conceptualisations which account for the processes of societal system transitions and ways of achieving long-term goals (Markard et al. 2012). Four theoretical framings, which by no means provide exhaustive perspectives, but have gained prominence in transition literature include: transition management (Rotmans et al. 2000); strategic niche management (Mejia et al. 2018); multi-level perspective on socio-technical transitions (Mejia et al. 2018); and technical innovation systems (Bergek et al. 2008). An explication of these four theoretical framings of sustainability transition is provided in the next paragraphs.

(a) Transition Management.

Transition Management (TM) evolved out of the analyses of knowledge on the theory of socio-technical systems (Kemp 1994; Geels 2001), complexity theory (Rotmans 2005), evolutionary economics (Kemp 1994; Mulder et al. 1999) and integrated assessment of long-term changes in society (Rotmans and van Vries 1997). From the literature, remedying the sustainability problems of existing societal systems will need a *"new generation of governance approaches"* with a *"long-term orientation"* as well as policies that are *"flexible, adaptive and reflexive"* and *"deliberation, probing, experimentation and learning"* (Van den Bergh et al. 2011, p.13). Loorbach and Rotmans (2010) view transition management as a form of meta-governance which encompasses stimulating actors and activities to compete against dominant forces and practices by identifying ways of influencing and coordinating the interactions between actors and activities. Critically, the TM approach tries to create a new vision for SD, particularly through miniature projects where various actors (stakeholders) interact to find solutions that address pertinent sustainability problems (Loorbach and Rotmans 2006).

In their paper which examined the practice of transition management, Loorbach and Rotmans (2010, p.245) advocate the need for an *"experimental, innovative, multi-disciplinary and participative form of governance"* like transition management, as they view it as one that is on the same wavelength with the demands of society, research, and policy. They stress the dynamics of 'power and people', which they believe is central to future debates on transition management. Citing the case of the Netherlands, their research showed that the case for transformative change had gained momentum, with the transition in most sectors accelerating towards a structural regime change. However, they argue that the challenge for transition management hinges on engaging regime actors and stimulating societal pressure such that

new societal regimes are co-created from the newly emerging niches and innovative regime actors. Picking up on this, critics of TM argue that it provides few opportunities for inclusive participation within transition arenas (Avelino 2009). In a study of transition management in the context of a transport project, Avelino (2009) highlights the lack of involvement of weak stakeholders. The author argues that 'arenas of transition' are dominated by incumbents which resulted in asymmetric power relations, where the weak stakeholders were hardly heard.

(b) Strategic Niche Management.

The Strategic Niche Management (SNM) approach grew out of the attempt to infuse evolutionary economics with insights from constructive science and technological studies, as exemplified by the work of Nelson and Winter (1982). Schot and Geels (2008) view SNM as a concept whose development was underpinned by the need to find ways of dealing with the challenges of managing sustainable innovations and transitions. Thus, a core idea behind the SNM concept is that niches can be deliberately created and supported in order to induce a shift in regime (Kemp et al. 1998; Hoogma et al. 2002). Through the lens of the SNM approach, the literature (see Kemp et al. 1998; Raven 2012; Smith and Raven 2012; Seyfang and Longhurst 2014) has identified six processes key to niche structuration and development. Mejia et al. (2018, p.220) have summarised these key processes to include: "1) Visions and expectations are negotiated and articulated by a growing number of actors. 2) A network of different stakeholders takes shape, increasing resources. 3) A shared learning process among actors takes place. 4) There are intermediary organisations and actors who carry localised knowledge to other localities, promoting and strengthening institutional practices among the niche. 5) There is evidence of niche, regime and landscape dynamics. 6) Niche protection not only serves as a shield, but also prompts innovation development and empowers actors". They argue that the processes mentioned above will optimally allow a niche to develop and stabilise in a way that increases its probability of causing change at the system level.

(c) Multi-Level Perspective.

A useful framework that explicitly describes sustainability transition is the Multi-Level Perspective (MLP). At the core of the MLP approach is the idea that transitions are products of the interactions existing between processes at various levels, which Schot and Geels (2008) have identified as niches, regimes and landscapes.

(d) Technical Innovations Systems.

The technical innovation systems (TIS) concept conceptualises transitions by focusing on "actor constellations, networks and institutional structures and primarily seeks to analyse potentially radically new socio-technical configurations that cross-cut established sectoral and spatial delimitations" (Coenen et al. 2012, p.969). A core assumption of the TIS approach, therefore, is that new technologies can emerge and potentially contribute to future sustainability.

2.3.4.3 Concluding comments on sustainability transitions.

The preceding paragraphs in sub-section 2.3.4 have, thus, elaborated on how transitions evolve and the prominent approaches that have emerged to give more explicit meanings and understanding of the complex and multi-faceted changes that may characterise a potential shift to sustainability. However, as Coenen et al. (2012) argue, the bulk of research on sustainability transitions have focused on sectors of societal systems including health care, water systems, energy, housing, etc., at the expense of a spatial perspective. This means that there is limited knowledge of the locations where the actual transitions take place, "and the socio-spatial relations and dynamics within which transitions evolve" (Coenen et al. 2012, p.969). Again, as Foliente et al. (2007) argue, the fragmented response to the vision of making urban areas more sustainable, as manifested through atomised strategies such as 'sustainable energy futures' and 'sustainable water futures', will continue without crosslinkages to mutually reinforcing factors, such as resource use, technological and environmental impacts, socio-economic factors, among others, unless a common goal or destination is found. The fragmented nature of sustainability solutions means that they are spatially limited, one dimensional, and often lack an integrative consideration of economic, social, and environmental factors (Foliente et al. 2007). Nevertheless, as arenas where multiple transitions co-evolve, there is little empirical work that focuses holistically on cities (Hodson and Marvin 2010). Therefore, in the context of this research, a point of departure will be based on the notion that although an urban sustainability transition might not relate to sustainable urban structures and environments (Ernst et al. 2016), all the processes of fundamental change which manifest in a transition and lead to the realisation of 'sustainable places', fall within the descriptions of societal systems advanced by transition scholars (see, for example, De Haan and Rotmans 2011).

2.3.5 Sustainable urban transformation.

The goal of sustainable development that meets the needs of the current generation without compromising the ability of future generations to meet their needs will not be achieved through 'business as usual' policy-making and development paths which often go contrary to the principles of SD. Instead, SD calls for systemic sustainability transformations (Olsson et al. 2014). At the same time, Seto et al. (2010) have argued that ongoing rapid urbanisation has the potential to catalyse the transition towards sustainability due to the effects of urban agglomeration, innovation, and increasing urban wealth, which, Nevens et al. (2013) agree, will only be achieved with better governance structures in place. Put simply, the transition to urban sustainability requires structural transformations in order to realise the interlocking aims of sustainable urbanisation, human well-being, and sustainability. In the context of urban areas, McCormick et al. (2013, p.1) have defined sustainable urban transformations (SUT) as *"structural transformation processes (multi-dimensional and radical change) that can effectively direct urban development towards ambitious sustainability goals"*.

Notwithstanding the previously-acknowledged universal desire to transform urban areas into sustainable places, questions arise as to how the desired transition to urban sustainability can be achieved. Accordingly, Childers et al. (2014) have reviewed the challenges and opportunities of advancing urban sustainability theory and action, and they argue that setting a sustainable transition in motion requires an in-depth understanding of the transitions which urban systems may experience or have the potential to experience. Based on the review by Childers et al. (2014, p.324) of other scholarly work, they suggest that in order to model and exploit the transition to sustainability, the following processes and conditions must be considered:

(a) "The endogenous and exogenous triggers that have led to stresses and crises in contemporary cities must be documented and the causes behind them elucidated (Lucy and Phillips 2000; Graham and Marvin 2001).

(b) Cases in which contemporary cities are in crisis or are under threat as complex systems must be understood (Graham and Marvin 2001; Nassauer and Raskin 2014).

(c) We should explore the motivations that are deliberately moving contemporary cities to become more sustainable—that is, cities that are undergoing this transition because of desire, not crisis (Steiner 2014).

(d) We must expand our view of cities beyond the Global North model of the sanitary city to include non-sanitary cities that are more likely to occur in developing, or Global South nations, and to include regions that have not yet urbanised or are urbanizing in novel ways.

(e) The interaction of key resources that affect all cities provides a focus for understanding the opportunities and constraints that may characterise cities that are more sustainable (Brunner 2007; Kennedy et al. 2007)".

Especially for Africa, where data availability and access are dire, it is not exactly clear how this transition to sustainability will unfold. From the literature, it can be established that SSA countries are developing and urbanising rapidly (Darkwah and Cobbinah 2014). This suggests that they are already undergoing some form of socio-economic transformation. The central question here is: under the conditions of rapid urbanisation, pervasive poverty, and inequality, to what extent is it possible to induce and coordinate transformations that will result in a transition to urban sustainability? Understanding the continent's transition towards sustainability, therefore, demands an assessment of its current urban sustainable development situation which will help situate it within appropriate pathways that will facilitate a long-term and place-specific structural urban transformation process.

Transitioning from the contemporary SSA city to the sustainable SSA African city of the 21st century will require ongoing development and application of new knowledge in order to understand, support, and guide the dynamic processes of transformation. As O'Riordan (2016, p.1) wrote in an editorial issue, "*any transition to sustainability will be challenging*. *Not to confront it when the opportunity is rife is a letdown to those most deserving of the effort to safeguard their betterment*". The ongoing urbanisation and associated transformation in the cities of SSA, thus, make them a demonstrable exemplar within which processes related to sustainability principles determine whether they are making progress towards their goals? Progress must be measured. To this extent, the literature is replete with evidence on how sustainability has been evaluated from various perspectives, using different indicators, and through the development of different models. For example, in a study to delimit the concept of sustainable development, Bolis et al. (2014) employed an axiological perspective to develop a conceptual model that evaluated SD in the context of natural

resource, decision-making, and satisfaction of basic needs. Also, Holden et al. (2014) have evaluated the sustainable development of countries based on four equal measures of ecological sustainability, satisfaction of basic needs, and inter- / intra-generational equity. From the above, it appears there is no universal way of sustainable development. An apparent gap in the literature, therefore, is the lack of an evaluation framework which offers a holistic appreciation of sustainable development by facilitating the integration of its dimensions alongside the endogenous and exogenous factors such as poverty, globalisation, among others, which affect how the concept manifests. The rapid pace of urbanisation and the low level of development in SSA means that its situation is far more extensive and complex than discussed, and imminent action is required if the region's 21st century sustainability challenges are to be addressed to any extent. The next section will, therefore, review the literature on urban sustainability assessment by exploring the concept of sustainability indicators and modelling. This will be done with a view to providing the basis upon which sustainability status can be evaluated and understood, and also to help account for the necessary transformations needed in the transition to urban sustainability.

2.4 Urban sustainability assessment

2.4.1 Sustainability indicators (SI).

International, national, and private organisations can make decisions based on information available to them. Indicators and related data are widely recognised as key sources of such information for decision-making (Nader et al. 2008). Like many other concepts, there is no universally-accepted definition of an indicator; many definitions have been found in the extant literature (Joung et al. 2012; Singh et al. 2012). Joung et al. (2012, p.150) describe an indicator as "a measure or an aggregation of measures from which conclusions on the phenomenon of interest can be inferred". Indicators may be quantitative or qualitative, thus providing measurements in absolute or relative terms (Toth-Szabo and Varhelyi 2012). Indicators have the propensity to make problems more visible to stakeholders (Dahl 2012). This ensures that organisations and the public are sensitised on issues, and in effect, widens the basis of decision-making. The existence of an 'indicator industry' (Herzi and Hasan 2004) is a testament to the level at which indicators have gained importance.

As yardsticks of progress, indicators are believed to be vital in measuring and understanding any path towards sustainability (Lynch and Mosbah 2017). This is because they give indications as to whether a particular system is functioning well or not. Particularly relevant to the sustainability discourse is the high likelihood of well-chosen indicators to facilitate communication of information to a variety of audiences, as well as to minimise the amount of data required to describe and understand a situation (Keirstead 2007). As part of the global response to the need to develop sustainably, chapter 40 of Agenda 21 gives the impetus and reasons for the development of indicators in order to measure progress towards sustainability. It subsequently urges both national and international organisations to make efforts to identify and develop these indicators (Barrera-Roldan and Saldivar-Valdes 2002). Thus, the OECD (2005, para.1) defines sustainable development indicators as "statistical measures that give an indication of the sustainability of social, environmental and economic development". Some countries have made great strides in implementing effective sustainability indicator systems. For example, the United Kingdom and Switzerland have developed coherent and policy-relevant indicators which are linked to national strategies on sustainability (Dahl 2012). However, Weststrate et al. (2019) assert that some governments sometimes 'adjust' the definitions of indicators in order to enhance performance on the aspect being assessed by an indicator. At the city level, cities including Melbourne, Mexico City, Barcelona, Pune, and Taipei have all adopted indicators to measure and monitor the performance of their respective urban sustainable development plans (Shen et al. 2011).

2.4.1.1 Development of sustainability indicators.

The development of sustainability indicators (SI) in the 1990s was underpinned by the need to "provide a solid basis for decision-making at all levels and to contribute to a self-regulating sustainability of integrated environment and development systems" (United Nations 1992, p.346). This 'solid basis', Pires (2011) argues, failed to materialise as the overly technical and expert-driven nature of initial indicators developed (see Mitchell 1996; Bell and Morse 1999; Bossel 1999; Spangenberg 2002) rendered them irrelevant for use in urban management and governance. Therefore, rather than becoming tools for driving urban development, a lack of explicit links to local level policies meant that sustainability indicators became isolated technical databases (Bell and Morse 2001). As some authors argue, the exploitable functions of indicators mean that their development cannot be a purely technical issue but should rather involve a transparent policy process (Scipioni et al. 2009; Ramos 2009). Subsequent years have, however, seen the improvement of indicator development as scholars have argued and advanced the idea of linking SI development and urban policy and governance (Bell and Morse 2001; Herzi 2004; Herzi 2006; Rosenström 2006; Lehtonen,

2012). To this effect, Pupphachai and Zuidema (2017) have argued that designing SIs well means that, not only should they be linked to existing policies, but they should also be part of the process of developing new policies.

The development of indicators requires the consideration of key attributes which are expected to facilitate the compilation and analysis of data, as well as enable easy communication and uptake of indicator information. The desired criteria and characteristics of indicators have been well-elaborated by scholars (for example, Moss and Grunkmeyer, 2007; Joung et al. 2012). Consequently, this study adopts the integrated criteria (Table 2.3) established by the Sustainable Measures Group (2009) and Moss and Grunkmeyer (2007, as cited by Joung et al. 2012).
Characteristic	Description
Measurable	A measurable indicator is one that can be simply and easily measured
	by quantitative or qualitative means within a given time frame for data
	collection and evaluation.
Relevant	A relevant indicator is one that directly relates to a meaningful and
	purposeful aspect of sustainability.
Understandable	An understandable indicator is one that is easily interpreted by the
	community and lay people.
Reliable / Usable	A reliable / usable indicator contains trusted and accurate information
	from the organisation.
Data accessible	Similar to reliable / usable, an indicator must be based on data and
	information that can be easily accessed and acquired.
Timely manner	Data and information collection, calculation, and evaluation for an
	indicator must be done in a timely manner for informed decision-
	making.
Long-term	Indicators must ensure their future use, development, and adoption as
oriented	an organisational or product sustainability standard.

Table 2.3: Characteristics of indicators.

Source: Joung et al. (2012).

2.4.1.2 Selection of sustainability indicators.

Sustainability indicator selection is, in itself, a decision-making phenomenon for which various approaches exist. From the literature, methods for SI selection include, but are not limited to, 'top-down and bottom-up approaches', and Driver-Pressure-State-Impact-Response (DPSIR) models. It is worth noting that there is no consensus on the best approach to indicator selection. Magee and Scerri (2012) describe SI selection as a complicated process in which traditional top-down or bottom-up approaches may be adopted. Fundamentally, the top-down approaches rely on the inputs and assessments of experts with minimal public involvement, while the bottom-up approaches are citizen-driven (Lynch and Mosbah 2017).

Epistemologically-rooted in scientific positivism, top-down approaches are structured on traditional and formal hierarchies, while bottom-up approaches are based on a 'participatory' philosophy' (Turcu 2013). A succinct description of both approaches ranges from scenarios where indicators are chosen based on what experts see as relevant, to the identification of indicators by the ordinary citizen through a participatory approach (Fraser et al. 2006). Accordingly, both approaches to selection have been critically examined by researchers. For example, Wilsdon et al. (2015) have suggested that metrics used in standardised top-down approaches of indicator selection do not necessarily reduce the need to provide locally appropriate measurement tools. Conversely, the same author argues that bottom-up selection approaches are not well-tailored to support continuity and comparability of indicators. An aspect of indicator selection that has been given critical scholarly attention with respect to both approaches is local community participation, which is expected to influence the quality and relevance of the indicators selected (Pissourios 2013). Inclusive participation is crucial to the sustainability agenda as it enables communities to influence sustainability policies (Barry 1999). However, it is argued that inclusive participation will only be realised when the barriers to community participation have been overcome (DETR 2000; Wolfe et al. 2001). Ensuring an efficient selection of sustainability indicators will, therefore, require an optimal trade-off between both approaches. Scholars have proposed various measures aimed at providing a balance between the respective approaches in the selection of sustainability indicators. Astleithner and Hamedinger (2003) contend that collaboration between experts and scholars will enhance the effectiveness of sustainability indicator selection by neutralising the adverse effects of community participation (Lynch and Mosbah 2017). Other proposals include 'building communities of practice' whereby practitioners and scholars are grouped together during indicator selection (Gonzalez et al. 2011), and selection systems where indicators are co-produced by experts and citizens (Graymore 2014).

In the context of urban sustainability, Shen et al. (2011) assert that the indicator selection process should be based on the careful analysis of indicators which have fundamental significance and carry a maximum likelihood of producing accurate information, rather than just gathering data on all indicators. Others have also emphasised that indicator selection should be guided by the questions which they are expected to answer (Li et al. 2009; Joumard and Gudmundsson 2010). Joumard and Gudmundsson (2010) have, thus, introduced ten criteria for indicator selection which have subsequently been organised into three main categories, as shown in Table 2.4.

Category	Specific Criteria
Representation	Validity, reliability, sensitivity
Operation	Measurability, data availability, ethical concerns
Policy Application	Transparency, interpretability, target relevance, actionability

Table 2.4: Criteria for indicator selection.

Source: Journard and Gudmundsson (2010).

2.4.1.3 Challenges to the use of sustainability indicators.

Despite the benefits of an indicator-based approach to sustainability, it has been argued that achieving an optimal trade-off between the political relevance and scientific accuracy of indicator use remains 'a balancing act' due to certain methodological challenges (Hilden and Rosenstrom 2008; Mieg 2012). Mieg (2012) has outlined these challenges as problems of 'incompleteness', 'aggregation', and 'specificity'. Accordingly, questions that have led to the realisation of these methodological challenges include:

(a) What makes up a complete set of indicators? This question highlights a lack of a universal theory for (urban) sustainability where relevant and accepted indicators are well defined. For example, while ten general indicators were used in the European Common Indicators project to cover all dimensions of SD, the United Nations Commission on Sustainable Development (UNCSD), in a similar project, used nineteen indicators to represent each dimension of sustainability. Thus, as Mieg (2012, p.253) argues, "if a measurement focuses on relevant indicators it risks being incomplete; if it attempts to cover everything the measurement risks becoming irrelevant".

(b) How can different indicators be combined into a single valuation? A widespread problem of indicator use in (urban) sustainability study and practice is the aggregation of indicators into a single but standard valuation. In a different scenario, this methodological challenge has been closely linked to the concepts of weak and strong sustainability (Mieg 2012), which are discussed further in Section 2.6.3. Here, weak sustainability will undermine the general applicability of metrics to be used in an indicator system as it allows for a deficiency in one metric to be compensated for by a stronger metric. For example, where a metric representing environmental sustainability is lacking, it may be compensated for by a metric depicting

economic growth. In a similar vein, strong sustainability affects the aggregation of metrics into a single valuation as it fundamentally requires certain metrics to be present without substitution or compensation (Hopwood et al. 2005). In this case, strong sustainability raises questions on 'relevance' as it ensures the inclusion of some metrics as a basic requirement irrespective of whether or not they are relevant to what is being measured.

(c) Should general indicators be preferred to local ones? Specificity is a common problem area in the use of indicators. This leads to a dichotomy of choice between either general indicators or a place-specific local indicator. Mieg (2012) is of the view that while general indicators are highly useful in comparing cities, place-specific indicators may be more appropriate, especially in the context of (urban) sustainability. The author cites a typical example of the use of place-specific indicators as Seattle in the United States, where certain traits in the growth of salmon as they return to freshwater sites (for example, size at age, age at maturity, contaminant loads, among others) are examined for evidence of changing marine conditions (Sustainable Seattle 1998; Irvine and Riddell 2007). Thus, the argument for place-specific indicators is that they promote a strict local approach, but at the same time, they maintain the common criteria for (urban) sustainability management (Pastille Consortium 2002).

Furthermore, when considered within the context of important global commitments such as the New Urban Agenda³, Caprotti et al. (2017) suggest that the uptake and usefulness of sustainability indicators are potentially undermined in the arena of urban development practice due to factors which include: how sustainable cities are standardised, the role of experts, as well as the appropriateness of measurements and data. For example, Caprotti et al. (2017) argue that aside from the lack of data, the use of indicators to capture urban realities tends to be challenging due to the entanglement of measurements with the lives and priorities of urban citizens. The foregoing should, however, not be construed to imply that these challenges have limited the usefulness of sustainability indicators, rather they suggest that more thinking is needed to make indicators more 'fit for purpose'.

³ The New Urban Agenda (NUA) is a global commitment to promote urban sustainability that was agreed by national governments at the Habitat III summit held in Quito (Ecuador) in 2016 (Valencia et al. 2019).

Critically, the literature has shown that practical issues such as indicator selection, as well as methodological challenges of indicator use, influence the broader application of these indicators. These practical challenges mean that there is little gain in sustainability performance, as indicators tend to lead to mixed results on application in certain parts of the world (Selman 1999; Alshuwaikhat and Nkwenti 2002; Seabrooke et al. 2004). The challenge, therefore, for international governments, international organisations, and urban managers, especially concerning cities in developing countries, is to facilitate the universal application of sustainability indicators, by identifying best practices and making attempts to aggregate key aspects of the many different systems. This is expected to improve links between sustainability indicators and the policy process, thereby ensuring the effective and conceptual use of indicators.

Furthermore, apart from a few studies that have attempted to measure urban sustainability in the global South (for example, Turok and Borel-Saladin 2014), extant studies on (urban) sustainability generally relate to cities of developed nations (see Tanguay et al. 2010), a situation that is sometimes justified based on the lack of data in developing nations (Cloete 2015). The implication for Africa, as Cloete (2015) argues, is that effective assessment of sustainability is currently unrealistic. Many countries in SSA lack baseline sustainability data, both at the local and national levels. Hence decisions on sustainability tend to be taken based on potentially erroneous or outdated information. Although the development of a complete new set of indicators to measure progress towards sustainability in SSA cities is beyond the scope and focus of this study, it can be inferred from the literature that the potential experience that could be gained by applying carefully-selected indicators on a comparative basis will essentially contribute to the development of standard processes of urban sustainability performance assessment. The next section will, thus, discuss the various models within which indicators could be adopted to assess sustainability.

2.4.2 Modelling sustainability.

Sustainability decision-making is a daunting task that requires the consideration of economic, social, and environmental interactions in a mutually reinforcing manner. In specific sustainability terms, the literature shows that several theoretical models have been developed to explain sustainability as a concept (for example, see Spindler 2011; Bott and Grassl 2013). However, as Van den Bergh and Hofkes (1998) note in their survey of economic modelling

of sustainable development, there is no best approach to modelling sustainability, but rather a multitude of approaches exists which provide varied but complementary insights into sustainability. Also, according to Ali-Toudert and Ji (2017), the background of an observer examining sustainability determines how the concept is perceived and the particular aspects that will be highlighted. Thus, the different approaches to modelling sustainability are attributed to the distinct theoretical starting points, as well as the complexity of its different dimensions (Van Den Bergh and Hofkes 1998). Todorov and Marinova (2011, p.1-2) have identified a general typology of models used in representing and studying sustainability according to five major categories, as shown in Table 2.5.

Model Category	Description
Quantitative models	Mathematical, statistical, data-based, econometric and computer simulation.
Pictorial visualisation	Venn diagram, graphical representation.
Conceptual models	Representing particular concepts and theories.
Standardising models	Including indicators, benchmark values and targets.
Physical models	Physical version of object / system that enables further investigation.

Table 2.5: Categories of mode	els.
-------------------------------	------

Source: Todorov and Marinova (2011).

(a) *Physical models* have generally been used to describe the environmental components of sustainability. While physical models reduce uncertainty, enforce trans-disciplinarity, and are participatory, they have been criticised for having limited time spans and being overly specific (Todorov and Marinova 2011). As an example, Karlsson (2008) modelled the lead losses in Sweden through the use of lead-acid batteries.

(b) *Pictorial visualisation models* use flow charts, Venn diagrams, among others, to emphasise the importance of the inter-linkages among the dimensions of sustainability, and stress the need for an interdisciplinary and trans-disciplinary approach to sustainability. For example, the World Conservation Union (2006) has used overlapping circles of a Venn

diagram to portray the intersection of the pillars of sustainability, as shown in Figure 2.3. The merits of visualisation models include their emphasis on trans-disciplinarity, while a significant drawback is the limited amount of information they provide (Todorov and Marinova 2011).



Figure 2.3: Intersection of the pillars of sustainability.

Source: World Conservation Union (2006).

(c) *Conceptual models* are underpinned by particular concepts and theories, and are developed based on long-term and inter-generational considerations. A major weakness is their poor reaction to uncertainty. Nevertheless, they are known for triggering policy responses.

(d) *Standardising models* make use of indicators and other measurement tools in order to give a snapshot of a prevailing situation. They may be participatory and can be used for evaluation at both global and local levels. While most standardising models are designed with a long-term view, the temporal nature of the indicators means that the model cannot be entirely futuristic.

(e) *Quantitative models* employ statistical, econometric, and mathematical approaches to evaluate situations. They are developed from a long-term perspective and are equipped to

deal with uncertainty. However, their development can be less participatory, and they usually do not have a holistic appreciation of situations.

2.4.2.1 Urban sustainability modelling.

Converting the vision of a sustainable urban space into reality requires significant shifts in policy from the local to the global level. From the literature, there has been significant attention given to advanced methods through which urban sustainability is assessed and evaluated (Ali-Tourdert and Ji 2017). Kupiszewska (1997) has identified possible approaches to urban sustainability modelling to include 'cross-sector approach' and 'activity approach'. A cross-sector approach may be: resource-oriented where the flow of a resource is tracked and evaluated for current and future use; or population-oriented where factors affecting an environmental sector are tracked and evaluated. Activity approach involves the selection of a human activity and an assessment of all of its impacts on sustainability. The Organisation for Economic Cooperation and Development (OECD) has also developed the DPSIR system (DPSIR: driving force, pressure, state, impact, response) as a way of modelling urban sustainability based on the organisational process of urban development (Dizdaroglu and Yigitcanlar 2014). The core principle, however, is that due to the various interpretations and the complexity of urban sustainability, as well as the number of approaches available, the choice of approach or model is effectively determined by what the user wants to achieve with the model (Kupiszewska 1997; Ali-Tourdert and Ji 2017). For example, as one of the first attempts to tackle urban sustainability within a comprehensive framework, Spiekermann and Wegener (2003) have modelled urban sustainability in Europe as part of a European Union project called PROPOLIS (Planning and Research of Policies for Land Use and Transport for Increasing Urban Sustainability). In assessing urban strategies for their long-term effect on European cities, they developed a comprehensive evaluation system which consisted of integrated land use, environmental, and transport models together with indicator and presentation systems. Also, Li et al. (2016) have modelled transition to urban sustainability in the industrial city of Jianching in China. They argue that an integrated mixed method of urban metabolism modelling through material flow analyses complemented by a qualitative enquiry helps in understanding the complexity of urban systems.

2.4.2.2 Geospatial applications and urban sustainability.

The sustainability concept offers an opportunity to improve the quality of life through the careful integration of social, economic, and environmental concerns (Grundey 2008). As nature and society typically evolve in space and time, the spatio-temporal distribution of the pillars of sustainability should be taken into account during any decision-making. This is particularly important as spatial relationships are seen as some of the most vital factors that influence the various dimensions of sustainability (Sakalauskas 2010). Furthermore, an overall measure of sustainability cannot be achieved without geographic considerations of the locations where the sustainability parameters are to be measured. In essence, the complexity of sustainability as a concept makes it increasingly necessary to draw from different fields of knowledge. According to Ward et al. (2000), appropriate spatial information represents one of the most essential tools used in the analyses of questions of sustainability. To this end, (Ridd 1995) argues that remotely-sensed data can be a source of verifiable urban composition information.

Accordingly, remote sensing has gained recognition as an effective tool used in the analysis of spatio-temporal landscape change dynamics at different levels (Serra et al. 2008; Geri et al. 2010). The application of remote-sensing techniques in the study of land cover and land use can help improve our fundamental understanding of the dynamics of contemporary urbanisation and the complexities between urban systems and sustainability. Essentially, analysis of the spatio-temporal characteristics of land use / land cover (LULC) change is expected to provide a comprehensive understanding of the ecological consequence of urbanisation (Samal and Gedam 2015), and such analyses are an essential source of information for decision-making (Deng et al. 2009). The literature also highlights the importance of LULC-related data and maps for sustainable development (see Lambin et al. 2001; Turner et al. 2007). This is because some LULC data outputs are directly related to biophysical outputs such as ecosystem services and deforestation. Specifically, in terms of urban sustainability, LULC change outputs have enabled analysis of 'urban sprawl'⁴, which can be described as an implication of rapid urbanisation whereby there is uncontrolled expansion of urban areas and which sometimes results in urban areas moving beyond administrative boundaries and into adjoining districts (Zeng et al. 2005; Wang et al. 2020). As a result of urban sprawl, the urban landscape (form) is reshaped and urban areas attain

⁴ According to Wang et al. (2020, p.1), "urban sprawl refers to rapid, low-efficient, and disorderly growth of non-agricultural land towards peripheral areas".

attributes such as low density⁵ and scattered neighbourhoods, segregationist land use, and peri-urbanisation⁶ (Aurand 2007; Abudu et al. 2018). The sustainability implications of uncontrolled urban sprawl include, but are not limited to, air pollution, ecological habitat fragmentation, traffic congestion, and resulting air pollution (through increased GHG emissions), increase in urban temperature, socio-spatial segregation⁷, and restricted coverage and access to urban services (Polidoro et al. 2012). From a transition perspective, a time series analysis of land cover change is useful because according to Turnheim et al. (2015), practical efforts to examine transitions require the conceptualisation of the trajectory and connections between the past, present, and future. Therefore, remotely-sensed imagery offers the opportunity to analyse and perceive the trajectory towards the long-term vision of sustainability through continuous monitoring and appraisal over a particular period. Similarly, it has been argued that the complexities of transitions make it impossible to design and manage it from the outside (Lachman 2013). Pragmatic efforts to study transitions will, thus, require retrospective steering and stimulations in the context of its dimensions, speed, size, and time (Kemp and Loorbach 2003).

2.5 Sustainable development, urbanisation, and the urban citizen

Rapid urbanisation has consequences for sustainable development. Subsequently, the importance of citizens in the urbanisation and SD discourse cannot be overstated since they are the most important actors whose activities influence both the urbanisation process and the social, economic, and environmental dimensions of sustainable development. Furthermore, in the quest for solutions to the challenges that arise from the interaction between the processes of urbanisation and sustainable development, engaging solely with stakeholders who share the experts' view is akin to 'preaching to the choir', a situation that is inconsistent with the holistic characteristics of SD as a concept. In the urban context, a consensus within the sustainability literature is that community (citizen) support is a strong predictor of the extent

⁵ Low density areas result in single or small family neighbourhoods which are far from consolidated city centres where the core of urban services exist (Aurand 2007).

⁶ Peri-urbanisation is the extension of urban areas beyond administrative boundaries into adjoining rural districts such that there is a clash between urban and rural uses of land (Aurand 2007).

⁷ Socio-spatial segregation, as a result of urban sprawl, leads to a situation where the urban poor are far removed from the urban central district where economic opportunities are concentrated, thereby potentially missing out on jobs and urban services (Polidoro et al. 2012).

to which cities are committed to sustainability (see, for example, Laurian and Crawford 2016; Swann 2017). Thus, in an increasingly urbanised world, an underlying assumption is that the demographic and landscape change will lead to (positive and negative) alterations in urban systems with wider implications for overall sustainability experience. In this context, the literature already reviewed has shown that governments have formulated various policies and programmes to manage urbanisation in a manner that promotes sustainable development. A fundamental expectation is that these policies will shape the urbanisation process in a way that promotes economic prosperity including: adequate income and job creation; social wellbeing including access to basic services; environmental well-being including pollution reduction, and conservation of ecological resources; and the overall sustainability of urban areas. As development is dynamic, many of the expected outcomes of these development policies and programmes will only be observable in the medium to long-term if appropriately measured. Nevertheless, the question inevitably arises: what are the perceptions of citizens based on their lived-experience in the context of urbanisation and sustainability? Addressing this question requires an understanding of the perceptions of citizens in order to identify factors that shape their sustainable development experience, an exercise that is important for improving and moulding policy towards achieving sustainability.

Perception has been defined as "the organisation, identification and interpretation of sensory information to represent our environment" (Xiao et al. 2016, p.258). It is recognised that the perceptions of stakeholders provide useful constructs for the exploration of organisation behaviour (Delmas and Toffel 2004). Accordingly, the exigency to achieve sustainable development across all spheres, including in urban areas, has prompted the proliferation of research in diverse disciplines on perceptions and attitudes of citizens in the context of sustainability. From an urban sustainability perspective, Macke et al. (2018) have studied the subjective quality of life in a smart city by exploring the perception of citizens in a Brazilian case study. They found that the overall perception of the respondents about the main elements characterising Curitiba as a smart city implied their low level of satisfaction. They concluded that exploring the perceptions of citizens helped to deconstruct the interconnected facets of quality of life domains in the context of smart cities. Worth noting here is the fact that the majority of these studies have either focused on perceptions of sustainability in disciplines such as higher education, or when conducted in the context of urban sustainability, they have mainly focused on places outside SSA. Thus, the limited coverage of citizens' perceptions in the context of urban sustainability in SSA is a critical gap in the literature. Filling this gap is

important because the subjective experience of citizens through an exploration of their perceptions is a powerful tool through which they can provide information about the characteristics of their environment (Hsieh 2012). Furthermore, the acceptance and support of citizens are critical if a city is to successfully realise its sustainability ambitions (Laurian and Crawford 2016; Swann 2017).

According to Moganadas et al. (2013), people differ in their views, priorities, and motives with regards to sustainability. While prevailing conditions such as the political economy may shape the wider experience, the perception literature suggests that other factors may be at play. These factors have been highlighted to include demographic factors such as age and gender (Olsson and Gericke (2016), socio-economic status, and education attainment (Aina et al. 2019). For example, Narducci (2019) found significant differences based on age, education, political ideology, and gender in respondents' perceptions of ecosystem services in their study which evaluated the implications of urban growth and farmland loss for ecosystem services in the western United States. Also, Tuncer (2008) found significant differences in the perceptions of female and male respondents in a sustainable development survey of students. The general literature (for example, Voon 2012) also shows that diverse economic and psychological influences shape individual perceptions. The inference made from the above explication of perception literature is that, differences in demographic characteristics and socio-economic conditions among others, potentially shape the perceptions of citizens. In other words, two individuals could face the same wider economic conditions but have different perceptions about the impacts of the economic conditions due to differences in demographic and socio-economic characteristics.

The reverberation of SD as a concept has led to the promotion of practices such as recycling in urban and sustainability-related programmes and policies in order to influence stakeholders to embrace sustainability. However, the consensus within the SD literature is that sustainability cannot be achieved without a corresponding change in attitude across all levels, including at the national level and at the level of citizens (Kalsoom 2019). In this regard, Fairfield et al. (2011) in their study of the influences on the organisational implementation of sustainability argue that how organisations prioritise various sustainability issues will strongly determine the extent to which specific sustainability practices are implemented. This accentuates the need to gauge the interests of stakeholders like employees in the uptake of these sustainability practices (Velazquez et al. 2005). According to Pappas and Pappas (2014,

p.12), "individual behaviour creates the foundation for action in social, economic and environmental sustainability, and potentially guides human ability to work with one another to make life-affirming decisions". They conclude that when the day-to-day behaviour of individuals is aligned to well-stated values, the potential for greater sustainable community action is enhanced. Furthermore, Kalsoom (2019) has deconstructed the concept of 'attitude' from an SD perspective. He describes 'attitude' as a reflection of the degree to which people value sustainability and its several dimensions. He shows that attitude may be evaluated along the lines of the extent to which an individual favours a particular action, and that may be positive, neutral or negative. He further shows that attitude is not directly observable but may be inferred from what an individual says, intends or does. Finally, he expands on the link between attitude and behaviour; and argues that attitude is a proxy measure of people's behaviour, and inserts a caveat that positive attitudes do not always translate to behaviour. Extending the above to the context of this study, SD policies may promote the uptake of prosustainability behaviour or practices such as recycling by citizens; a question that emerges however is: how important are these practices to the targeted citizens, and to what extent are citizens willing to commit to in the implementation process? As the successful implementation of sustainability-related initiatives is shaped by the prevailing political economy (Davidson and Arman 2014), and as a result of the diversity that exists in people's views, motives, and priorities (Moganadas et al. 2013), it is critical to understand attitudes to sustainability in the context of the extent to which people are willing to partake in or promote pro-sustainability behaviour or practices. It is also recognised that the perceptions of stakeholders (citizens in this context) reflect their capability and willingness to promote sustainable practices (Gordon et al. 2012; Kerselaers et al. 2013). Therefore, as part of any inquiry on urbanisation and sustainability, it would seem appropriate to explore the nature of citizens' perceptions and their attitudes towards sustainability-related behaviour or actions in a way that also appreciates differences in views among people based on factors including demography and socio-economic conditions.

Critically, from what has been discussed so far, the importance of citizens' perceptions cannot be understated, if understanding the interactions between urbanisation and sustainability at the local urban level, and making recommendations for a transition to urban sustainability (which are central to this thesis), are considered. For example, considering the perceptions of citizens in this thesis would facilitate a social construction of knowledge about the realities that underpin the interactions between urbanisation and sustainable development

at the local urban level. Furthermore, in making recommendations for a transition to urban sustainability, exploring the perceptions of citizens would facilitate the provision of evidence-based guidance for sustainability transition that reflects the priorities, needs, concerns, and experiences of citizens. This is crucial if citizens are to embrace policies and programmes designed for sustainability transition (see Matschoss et al. 2019). Importantly, as is captured in the DPSIR (Driver-Pressure-State-Impact-Response) framework that will be proposed later on in Section 2.6.7.2, citizens and their perceptions thereof, constitute an important 'response' for managing the interactions between urbanisation and sustainable development into more sustainable outcomes.

2.6 Towards a framework for understanding urbanisation and sustainable development

The sub-sections that follow discuss theories and concepts which are relevant to the present discussion and provide a foundation upon which a conceptual research framework, that facilitates an understanding of the interactions between urbanisation and sustainable development in an integrated manner, is proposed.

2.6.1 Theories of urbanisation and development.

Apropos the debate on the causes and effects of urbanisation in developing countries, three theories which present different, but sometimes harmonious explanations of urbanisation, are considered useful for discussion in this thesis.

2.6.1.1 Modernisation and urbanisation.

Modernisation theory emerged in the middle of the 20th century as a model to explain the contrast and transition from an underdeveloped or traditional society to a modern one. Thus, the core concepts of the modernisation theory are 'tradition' and 'modernity' (Fangjun 2009). Theorists of the school of modernisation point out that there is a bi-directional relationship between urbanisation and industrialisation (Berliner 1977), further highlighting a release of surplus labour from agrarian regions to support industries in the urban areas (Dutt 2001). Furthermore, modernisationists consider urban areas as places of modernising institutions, such as educational facilities or factories (Bradshaw 1987). Within this line of thought, the underlying claim is that these modernising institutions serve as pull factors which draw people from rural areas to urban areas. Therefore, the more modernised a place is, the more urbanised it will become.

The modernisation school has been criticised for being a subjective conceptualisation that unalterably divided society into 'traditional and modern'; and also as a mechanism to spread American ideology (Yuan 2009). In the African context, criticisms of the modernisation theory abound. Matunhu (2011) has critically assessed the impact of the modernisation theory on Africa. He argues that ideas of modernisation contributed significantly to poverty on the continent. This is because the theory placed more value on external aid with inhibiting conditions at the expense of creativity and initiative of Africans themselves. Furthermore, he notes that the theory placed more emphasis on the supremacy of metropolitan regions in marshalling Africa's development, which undermined its already established structures of values and beliefs. The result was the discarding of Africa's own development path and the adoption of externally-controlled ones. This scenario can be implicitly used to explain the current trajectories in SSA, where Western interpretations basically underpin issues such as development and urbanisation. As Bull and Boas (2012) note, despite various criticisms, the modernisation theory still exists in contemporary theoretical frameworks, development strategies, and policy discourses. The foregoing, therefore, reinforces questions as to whether contemporary urbanisation in SSA should be pursued within the status quo or deliberate efforts should be made to steer urbanisation in a manner that fully considers prevailing conditions on the continent.

2.6.1.2 Dependency and urbanisation.

Dependency theory emerged in the 1950s as part of the debate on how unfair trade rules between poor and wealthy countries stagnated economic growth in poor countries, as compared to the wealthy industrialised countries (Ferraro 2008). Subsequently, dependency theory emerged as a medium for analysing issues of 'underdevelopment' within the global political economy (Namkoong 1999). From the literature on dependency, scholars have observed a dependent status by the poorest countries of the global South in the international economic and political systems (see, for example, Rodrick 2000; Knack 2001). Haq (1976) has viewed dependency from a historical perspective. He links the inequality in economic exchanges between the global North and South to the colonial era which strategically positioned the North at the centre of the world and the South on the periphery - as suppliers of raw materials. Essentially, colonial systems were designed to stifle any prospects of industrialisation in African cities by transforming them into European dependencies. By using communication as an example, Njoh (2005) describes how no direct telephone calls

were allowed between any two African nations, and how telephone services were exclusive to European residents in African cities. Therefore, a call from Abidjan in Ivory Coast to Accra in neighbouring Ghana had to pass through routers in France and Britain before reaching Accra. The fundamental argument in Haq's (1976) account, as summarised by Agbebi and Virtanen (2017), is that economic exchanges were more likely to be fair between equal partners rather than the unequal partnership promoted by colonialism.

An important dimension to dependency theory is the role of Foreign Direct Investment (FDI), usually through Multi-National Corporations (MNCs). Scholars have noted an increase in the level of international economic exchanges between advanced countries and poorer countries in the form of MNC investments (Bornschier and Chase-Dunn 1985; Ross and Trachte 1990). They argue that the conditions around these investments sometimes erode revenue, which could have been channelled into development programmes by poor countries. An example is when countries offer incentives in the form of tax and wage reductions to MNCs in order to attract more foreign investments (Ross and Trachte 1990; London and Ross 1995). MNCs are sometimes seen as impediments to development in poorer countries by hampering government policies for the people that are not in the interest of the MNCs (Bornschier and Chase-Dunn 1985; Ross and Trachte 1990). Furthermore, from the perspective of dependency theorists, the inflow of FDI is believed to impact negatively on economic growth and income distribution (Adams 2009). For example, according to Bornschier and Chase-Dunn (1985), an economy mainly fuelled by FDI will potentially result in the 'underutilisation of productive forces' due to a monopolised industrial structure. The effect is a stagnant economy, as the multiplier effect, through which demand in a particular sector creates further demand in another sector, is weakened (Adams 2009). This is especially true for SSA where most FDIs go into the extraction of natural resources (Pigato 2000).

Proponents of the dependency theory try to link it to urbanisation by explaining the interlinkages with issues including the availability of land and other resources, especially for people in rural areas. Some scholars are of the view that urbanisation (for example, as a result of MNC projects) potentially leads to landlessness in the rural areas and forces the rural population to move into cities in search of better opportunities (Njoh 2003). An example of this situation could be when farmers in a rural area are paid and dispossessed of their farming land in order for a mining project to take place. Without viable alternatives to employment, there is a tendency for these farmers to move to cities in search of better opportunities. Furthermore, others have argued that rapid urbanisation tends to distort urban-labour markets (Bradshaw and Noonan 1999). Also, dependency theorists have been instrumental in giving the developmental issues of the poorest countries the attention rightly deserved (Agbebi and Vitanen 2017). While not directly linked to the urbanisation process in general, this brief overview of the dependency theory is deemed relevant to the present discussion as it gives a theoretical underpinning on how the present developmental state in the area of interest has evolved.

2.6.1.3 Urban bias and urbanisation.

Coined by Lipton (1977) in his seminal work on urban growth and development in developing countries, 'urban bias' denotes an observation of a heavy systematic bias that skews resource allocation in favour of urban residents. In his classic account of Why Poor People Stay Poor, Lipton (1977) argues that the urban social elite exploit their powerful positions to skew development in favour of urban areas. He contends that reliance on the urban resource base by the urban elite means that they use their influence and lobbying powers to support macro- and micro-economic policies that promote urban development over rural development. In essence, the urban bias theory reflects the rent-capturing influence of urban elites in areas of national development. Nevertheless, the 'urban bias theory' has been criticised. For example, Byres (1979) highlighted Lipton's lack of recognition of the power wielded by rural people by citing the example of India, while Van Arkadie (1978) argued that Lipton's proposition failed to capture urban poverty. Over time, some scholars have also expounded arguments on the theory. For example, Majumdar et al. (2004) observe from a non-lobbying perspective, that the informational advantage held by urban residents, as well as the electoral imperatives of governments wishing to retain power, result in the uneven allocation of resources between the rural and urban residents of a country. They identify factors contributing to the informational advantage to include more educated and wealthier urban residents, as well as a stronger urban focus in media reportage. Ades and Glaeser (1995) also add an important insight by demonstrating the contingent effects of autocratic regimes on urban bias theory. They argue that such regimes offer disproportionate political power to urban residents, compared with rural residents, in order to hold on to power. A critical factor here is the ability of the urban class to mobilise politically. In response to criticisms of his theory, Lipton collaborated with Robert Eastwood to argue that structural adjustment and economic liberation programmes had failed to achieve their desired effects in

the poorest countries (Eastwood and Lipton 2000). Furthermore, in a re-statement of the theory, Lipton ([2005], as cited in Jones and Corbridge 2010) reiterated his view that urbanbiased government policies had led to 'substantially worse rural than urban outcomes' in the developing world.

For SSA, urban bias is seen as a by-product of colonialism. Colonial policies are believed to have led to a high degree of landlessness and land inequality, which in turn badly influenced productivity, distribution of public expenditure, and political and ethnic instability (Carter 2000; Sokoloff and Engerman 2000; Frankema 2005; Allcott et al. 2006). In addition, the colonial inheritance of dualistic economies fuelled an emergence of industrialisation policies in SSA, with the aims of steering economies away from export dependence to the path of industrialisation (Bezemer and Headey 2008). To this end, Bezemer and Headey (2008, p.1343) have singled out urban bias as "the largest institutional impediment to growth and poverty reduction in the world's poorest countries". Progressing the argument on the urban bias theory in SSA, Bates (1984) shows how governments used instruments of taxation to channel resources from rural areas to urban areas. The foundations of Bates' analysis rest on the notion that governments devised means of diverting revenues from agrarian sectors in order to keep the largely non-farm but powerful urban populace happy. This was achieved through measures such as manipulating exchange rates against export-bound farm products and setting administratively low domestic prices for exportable products and selling them later at prevailing world prices in order to make a profit.

In the wake of an increasing scale of urban poverty and an era of rapid urbanisation, one might be tempted to ask whether the theory still holds in the context of contemporary urbanisation, and to what extent? Answering such a question might be out of the scope of this study. However, from the perspective of this study, the relevance of this theory is expressed in order to shed light on the dynamics of contemporary urbanisation. For example, the analysis by Majumdar et al. (2004) remains a suggestive reminder of how the roots of contemporary urbanisation could be tied to the urban bias theory. Importantly, they demonstrate that especially in bad economic times, the effects of urban bias become more profound resulting in an increase in rates of urbanisation as cities expand to accommodate more people who want to enjoy the enhanced provision of public goods in urban areas. This assertion re-echoed the view of the World Bank Report (2000, p.130) which argued that "Africa's pattern of 'urbanisation without growth' is in part the result of distorted incentives

that encouraged migrants to exploit subsidies rather than in response to opportunities for more productive employment".

2.6.2 Utopianism and sustainable development.

Utopianism, just as sustainable development, is a contested concept with multiple meanings (Garforth 2009). Thomas More's (1516, cited in Sarakemsky 1993) novel Utopia advanced a generative paradox which has been influential in introducing the distinction between the good place (eu-topia) and the no-place (ou-topia) based on which the utopian concept has been expounded further. Subsequently, in her work on the History of Utopian Thought published in 1922, Hertzler (1922, p.268) defines utopianism as: "the role of the conscious human will in suggesting a trend of development for society or the unconscious alignment of society in conformity of some definite ideal". Perhaps one of the most renowned theorists, Bloch's (1986) seminal theorisation of the utopian concept, The Principle of Hope, frames the utopian inquiry around a philosophy of hope that conceptualises the utopian concept as a way of 'dreaming forward'. From an interdisciplinary perspective, Levitas (1990, p.191) defines utopia as "the expression of desire for a better way of living and being". Proponents of the utopian theory highlight its tendency to advocate for holistic and long-term thinking about the 'good society' within a democratic boundary (Levitas 2007). Critics, however, take issue with the utopian theory because its ideals are deemed to be paradoxical as they tend to advance incommensurable practices and values (Hong and Vicdan 2016).

Like sustainable development, the utopian concept could be characterised as a liminal discourse. In this case, liminality is expressed as *"the highly provisional in-between moment, implicated in that-which-has-been but gesturing towards that-which-is-to come"* (Mondal 2011, p.430). Thus, the plausibility of characterising sustainable development as *utopian* lies in its vision of an ideal developmental state that is definitionally difficult to instantiate. Importantly, this alignment of sustainable development with the utopian concept is an analytical one, rather than a descriptive one. Thus, based on the Brundtland Commission's definition of sustainable development, the potential connection between SD and utopianism is evident. For instance, in making the case for development without compromising the ability of future generation to also develop, SD, as a concept, assumes a utopian posture where the developmental realities of the current generation, whether critical or not, are not prioritised over that of the future generation, but are rather based on a perfected state of

egalitarianism. In other words, the prosperity of the future generation is a sequential outcome of the current generation. In his study on a new conceptual framework for sustainable development, Jabareen (2008) shows that sustainable development advances a vision for perfect human habitats which transcend ecological, social and political spheres. However, as a political agenda which is signed onto by national governments, the images of the utopian visions of sustainable development as a concept could potentially be obscured from the realm of public debate and interrogation (Levitas 2007) as they are complex and require multiple steps to be in motion simultaneously. Perhaps this should be a source of worry in the context of the sustainable development-utopia connection.

In sum, the paragraphs above have shown the importance of the links between sustainable development and the utopianism concept to the current study as it reveals, among others, the tension that the utopian features of SD as a concept sets up between future and present generations in the use of resources. For much of SSA, where poverty is pervasive and developmental levels are low, perhaps these tensions projected by the utopian vision of SD make its realisation a hollow dream. Essentially, in understanding the theoretical and practical implications of the sustainable development concept in SSA, one can, and must agree with Hertzler's axiomatic assertion that *"after all utopia is not a social state, it is a state of mind"* (1922, p.314). To some extent, all this explains why, just as with the utopian concept, SD, as a concept, remains a contested one. Nevertheless, despite its utopian tendencies, and consistent with Bloch's (1986) work, SD as a concept offers 'hope' through its clarion call for the redistribution of political power and equitable access to natural resources in order to guarantee humans a decent standard of living.

2.6.3 Natural capital and sustainable development.

First used by renowned economist, David W. Pearce, natural capital refers to the "set of all environmental assets" (Pearce 1988, p.599). While the importance of the natural capital concept is evident in its broad uptake in the economic literature, critics argue that it encourages a narrow perspective of the environment by reducing it to a mere asset with an economic value (Missemer 2018). In terms of its measurement, an indicator increasingly used to measure natural capital as a proxy for environmental quality is the ecological footprint (EF) (Solarin and Bello 2018). The ecological footprint reflects the extent to which people make demands on the environment by measuring the ecological assets that people require to

provide their needs (Mancini et al. 2017). The EF is measured in relation to *bio-capacity* (BC), which reflects the ecological budget in terms of biologically productive land and sea areas as used in EF calculations (Borucke et al. 2013). As such, there is ecological deficit when EF is higher than BC, an indication of unsustainability; while there is ecological surplus when EF is less than BC, an indication of sustainability (Ulucak and Bilgili 2018). Exploring the links between ecological quality and economic growth in the context of sustainable development, Holden et al. (2014) have argued that countries experiencing higher rates of socio-economic development (measured through HDI) are likely to have achieved that by expending significant amounts of ecological and renewable energy resources.

The natural capital concept is integral to two critical economic interpretations of sustainable development which are fundamental to the debate on how best to achieve it. The debate on weak and strong sustainability explores the relationship amongst the economic, environmental, and social dimensions of SD by the extent to which natural capital can be substituted with man-made capital (Ekins 2011; Wu 2013). The first approach, 'weak sustainability', takes its roots from the pioneering studies of renowned economists Robert Solow (a Nobel Laureate) and John Hartwick (a resource economist) in the 1970s (Neumayer 2013). Proponents of weak sustainability contend that the level of substitutability between natural capital (for example, biodiversity) and man-made capital (for example, socioeconomic infrastructure) should be unbridled and is sustainable as long as the quantity of capital stocks does not depreciate (Pillarisetti 2005; Fischer et al. 2007; Neumayer 2013; Greasley et al. 2014). More to the point, they argue that providing adequate goods, services, and infrastructure compensates for any depletion or pollution of environmental resources. In the context of this research, their argument would mean that rapid urbanisation and its impact on the ecological integrity of the earth is sustainable provided it results in economic growth. Conversely, some proponents of 'strong sustainability' argue that natural and man-made capital are meant to complement each other but are not substitutable (Daly 1995; Neumayer 2013). To put it succinctly, they argue that some natural capital cannot be substituted or replaced by any other form of capital due to their significance in maintaining ecological systems, as well as the irreversibility and uncertainty of such natural capital (Ekins et al. 2003). In the context of this research, their view implies that rapid urbanisation cannot be considered sustainable if it threatens the existence and functions of vital components of natural capital.

2.6.4 Equity and sustainable development.

The concept of equity is deeply embedded within the definition and praxis of sustainable development. Equity in the sustainable development context is premised on the concept of 'distributive justice' which can be defined as how opportunities and costs in life are distributed among members of a society (Armstrong 2012). Equity in the sustainable development discourse can be conceptualised in at least two ways: inter-generational equity and intra-generational equity (Kverndokk et al. 2014). From an inter-generational perspective, meeting the needs of the present generation should not compromise opportunities for future generations to realise the same. From an intra-generational perspective, there should be fairness in the distribution of resources within any contemporary generation. The intra-generational perspective, thus, assumes that for example, across social groups differentiated by income, gender, age, etc., there should be equitable access to opportunities for meeting basic needs that include, but are not limited to, employment, housing, education, and health (Trudeau 2018).

While equity remains a central tenet of sustainable development, the ability to realise it in practice is in question. Recent scholarship on sustainable development has elaborated on the equity concept and explored opportunities and barriers to achieving it practically. For example, Trudeau (2018) has argued that achieving equity in sustainable development practice hinges on the capacity of relevant institutions to champion this cause, demonstrate such commitment through the prioritisation of the concept at the onset of the development process, and the extent to which policies and resources are provided to support this cause. Accordingly, issues of social equity and social justice are widely seen as critical in any empirical analysis of sustainable development (Pearce et al. 1990). Of particular concern in the SSA context, is how endogenous and exogenous forces such as globalisation, market forces, and poverty may influence social equity. Thus, the interest here is for exploring how the interaction between urbanisation and sustainable development affects distributive justice in an intra- / inter- generational context. Equity in access to opportunities from the interaction of sustainable development and urbanisation as manifested in urban systems will, therefore, relate to how the engagement and participation of different actors are guaranteed.

2.6.5 Human well-being and sustainable development.

There is an explicit acknowledgement that human well-being (HWB) is of crucial importance in the analyses and measurement of sustainable development (McGillivray and Clark 2006). Nevertheless, like sustainable development, the human well-being concept is ambiguous as there are multiple definitions used to explain it. For example, HWB has been used to describe an objective measurement of people's living circumstances (Collard 2003). Others (for example, Mee-Udon 2009) have described HWB as a function of the meaningful and sustainable interactions between people and their physical and social environments. Thus, a parsimonious description consistent with the principles of sustainable development is used for the present discussion in which human well-being is described as "*a state of being with others, which arises where human needs are met, where one can act to meaningfully pursue one's goals and where one can enjoy a satisfactory quality of life*" (McGregor 2008, p.1).

In the context of sustainable development, the centrality of human well-being is embedded in multiple development agendas that range from social justice to environmental sustainability (Prescott-Allen 2001). In terms of its measurement, the ambiguity of the HWB concept means that the literature (see, for example, Diener et al. 2009; Scott 2012) acknowledges that its measurement could be both subjective and objective. The objective components of wellbeing which are usually measured quantitatively denote the material and social characteristics of a society as reflected in dimensions such as employment and income, education, housing, and health (King et al. 2014). However, the subjective components of well-being are captured qualitatively to reflect how one individual thinks and feels about life circumstances, usually in terms of satisfaction with specific dimensions (King et al. 2014). Importantly, Hagerty et al. (2001) have highlighted the importance of both subjective and objective components if the totality of life experiences is to be captured when analysing human well-being in the context of sustainable development. They argue that "a person may report a high level of subjective well-being, despite environmental conditions bad enough to significantly shorten life expectancy, hence affecting immediate future quality of life. Similarly, objective quality of life conditions (for example, health and material possessions) of a person may have very little to do with subjective well-being. For example, a person may be wealthy, yet feel very dissatisfied with life" (p. 8-9). The literature has reflected on the interconnectedness of human well-being and sustainable development in the context of socio-ecological systems. For example, Costanza et al. (2007) have developed a 'basic needs and capital inputs' model that conceptualises how natural capital together with built, human, and social capitals are required as inputs in the context of meeting human needs. The discussion has, therefore,

highlighted the critical influence of human well-being when analysing sustainable development as consistent with the 'meeting basic needs' requirement of the sustainable development definition advanced by the Brundtland Commission (WCED 1987).

2.6.6 Socio-ecological systems theory.

With the growing recognition of the bi-directional relationship between urbanisation and sustainable development (Lwasa 2014; Zhao and Wang 2015), a concomitant challenge is how this interaction will affect the overall stability of urban systems, and in the context of this study, how it affects human well-being and inter- / intra-generational equity. Furthermore, according to Feldstein and Glasgow (2008), the implementation of any sustainable development or sustainability policy programme is hindered when there is a lack of a comprehensive and practical conceptual framework within which the key features and predictors of sustainability are integrated. Thus, addressing the complexity of the interrelationship between urbanisation and sustainable development will require their organisation in a way that allows these complexities to be addressed based on sound evidence. Increasingly used to understand complex systems, socio-ecological systems (SES) theory emerged from the recognition of "the close interactions between society, in terms of socialeconomic systems and natural systems" (Petrosillo et al. 2015, p.1). As a precursor, the meaning of 'system' must be defined in order to examine SES adequately. Thus, a 'system' is defined here as an abstract appreciation of inter-related components which function as a whole (Schelbe et al. 2018). Subsequently, SES theory emphasises how the biological and biophysical processes of ecosystems interconnect with the cognitive, socio-cultural, and institutional processes of human systems (Berkes and Folke 1998; Adger 2006). In this context, the socio-ecological system is defined as an integrated human and natural system where feedbacks exist between the biophysical and human components (Ostrom 2009). Central to the SES perspective is holistic thinking that assumes that the socio-economic and ecological elements of sustainable development are interchangeable (Bastianoni et al. 2016). Reflecting on the interconnectedness of sustainability as a concept and socio-ecological systems, an extract from the United States Government Executive Order 13514 on federal leadership in environmental, energy and economic performance (2009, as cited in Stahl and Bridges 2013, p.551) argues that the general intent of sustainability "is to create and maintain conditions under which humans and nature can exist in productive harmony that permit fulfilling the social, economic and environmental requirements of present and future

generations". This creates a challenge of developing a means to help streamline sustainability across multiple scales (National Research Council [NRC] 2011). This challenge, according to the literature (for example, Summers and Smith 2014), is premised on the need for a unifying approach that connects scientific information and the normative values of sustainability.

From the foregoing, a variety of frameworks have emerged as viable options for diagnosing and understanding the sustainability of socio-ecological systems (Partelow 2018). Among the several frameworks that have been adopted to reflect the diverse research fields that are underpinned by socio-ecological systems is the DPSIR (Driver-Pressure-State-Impact-Response) framework (Petrosillo et al. 2015). The DPSIR framework holds promise to help organise complex socio-environmental problems and enable conceptualisations across socioecological systems (Gregory et al. 2013). It was originally developed by the OECD (1994) and subsequently enhanced by the European Environment Agency ([EEA] 1995) to help assess progress towards sustainable development by facilitating the identification of causeeffect relationships between human and natural systems (Smeets and Weterings 1999; de Stefano 2010). For proponents of the DPSIR framework, its strengths include potential for the integration of knowledge across different disciplines that allows for the gap between scientific disciplines to be bridged, and for the links between policy and management to be enhanced (Svarstad et al. 2008; Tscherning et al. 2012; Lewison et al. 2016). Critics, however, point to its limitations including the possibility of a mechanistic oversimplification of complex systems and the challenge of dealing with variables that may act both as a driver and pressure (Klijn 2004). The literature (for example, Bradley and Yee 2015, para. 2) recommends the application of a DPSIR framework to "capture, organise and visualise the environmental, social and economic outcomes of human decisions". A gap in the literature relates to the application of the DPSIR in a holistic spatial context. The literature on conceptualisation and assessment of socio-ecological systems using the DPSIR framework has focused on its application in human and natural systems. For example, Gisladottir and Stocking (2005) developed a conceptual framework model based on DPSIR in order to study the effectiveness of land degradation control measures. Nilsson et al. (2009) have also used DPSIR as a framework for strategic environmental assessment. Furthermore, Agyemang et al. (2007) have used DPSIR to organise information on environmental degradation in Ghana. From these examples, it appears that the application of DPSIR framework is mainly dominated by research on environmental impacts. Critically, due to its potential for

integrative and multi-dimensional assessment, the literature (for example, Pintér et al. 2005; Svarstad et al. 2008; Tscherning et al. 2012; Dizdaroglu and Yigitcanlar 2014) suggests that future research should include an application of the DPSIR framework that focuses on the conceptualisation and assessment of urban systems in the context of sustainability. Thus, considering the lack of a comprehensive theoretical framework based on which the complexities of sustainable development could be holistically understood (Jabareen 2008), a socio-ecological systems approach (DPSIR) is useful as a framework for advancing a holistic appreciation of urbanisation and sustainable development in a way that facilitates the inclusion of all components relevant for decision-making. In the context of this study, the inherent strengths of the DPSIR framework (including: its potential to organise and integrate knowledge across diverse disciplines; its potential to facilitate scientific communication by structuring and simplifying complex problems; its potential to establish cause-effect relationships; and its potential to facilitate the clarification and comprehension of the interrelationships between disparate concepts [Lewison et al. 2016]), mean that it can contribute to empirical research on urbanisation and sustainable development by organising their interrelationships in an integrated manner.

2.6.7 Towards a conceptual framework for understanding the inter-relationships between urbanisation and sustainable development in Ghana.

2.6.7.1 Theoretical assumptions.

Following Yang (2010), establishing assumptions are essential to designing a conceptual framework that will help understand the inter-linkages between the two concepts being studied. Thus, from the strands of literature reviewed on the distinct processes of urbanisation and sustainable development so far in this thesis, the following assumptions are made:

1. There is a bi-directional relationship between urbanisation and sustainable development which manifests at the spatial level of urban areas but transcend physico-administrative boundaries (see for example, Simon 2010; Gong et al. 2012; Peter and Swilling 2012; Turok and McGranahan 2013; Wu 2014; Zhou and Wang 2015; Zhou et al. 2015).

2. While potentially utopian, sustainable development as a concept assumes a limit on the extent of social, economic, and environmental development within which human well-being can be guaranteed to an acceptable standard and inter- / intra- generational equity can be

maintained (see for example, Prescott and Allen 2001; Levitas 2007; Holden et al. 2008; Jabareen 2008; Sassen 2011; Kverndokk et al. 2014; Trudeau 2018).

3. The urbanisation process results in radical changes to the landscape and demography of urban systems which have implications for achieving sustainable development objectives (see for example, Parnell et al. 2007; Duh 2008; Seto et al. 2010; Wu 2014; Samal and Gedam 2015; Cobbinah et al. 2015a; Pauleit et al. 2018).

4. The nature (i.e. pace) of the urbanisation process affects its interaction with the sustainable development process and has broader implications for human well-being and intra- / inter-generational equity (see for example, Biswas 2006; Duh 2008; Bryceson et al. 2009; Darkwah and Cobbinah 2014; Cloete 2015).

5. Endogenous and exogenous forces (including poverty, inequality, globalisation, resource use) underpinning urban systems determine how the interactions between urbanisation and sustainable development will affect human well-being and intra- / inter-generational equity (see for example, Prescott and Allen 2001; Lee and Zhu 2006; Peter and Swilling 2012; Afranie and Ahiable 2011; UN-Habitat 2014; Le Gales 2016).

6. The interaction between urbanisation and sustainable development can be understood from within a socio-ecological framework by selecting appropriate indicators (see for example, Dizdaroglu and Yigitcanlar 2014; Turok and Borel-Saladin 2014; Petrosillo et al. 2015; Bastianoni 2016; Lynch and Mosbah 2017; Partelow 2018).

2.6.7.2 DPSIR conceptual framework for understanding urbanisation and sustainable development in Ghana.

The processes of urbanisation and sustainable development, as well as the socio-ecological systems theory explicated in the previous sections, offer the points of departure based on which a conceptual framework for understanding urbanisation and sustainable development in Ghana is premised. As a result, a conceptual framework based on the DPSIR concept (shown in Figure 2.4) is proposed to organise and understand the interactions between urbanisation and sustainable development from a socio-ecological perspective. The conceptual framework is premised on a dynamic interplay between the forces of urbanisation,

and the economic, social, and environmental dimensions of sustainable development, herein referred to as 'drivers'. The next component 'pressure', reflects the factors that reinforce the manifestation of the drivers identified. The next component, 'state', captures the system-wide influences of the pressures identified. The next component, 'impacts', reflect the measurable impacts of the first three components, based on relevant indicators, as well as subjective experiences. The final component, 'responses', capture the societal, corporate, and institutional responses to the identified drivers, the pressures that reinforce their manifestation, and the mitigation of the state and impacts. Essentially, the DPSIR-based conceptual framework proposed is related mainly to the interaction between urbanisation and sustainable development. It assumes that there are endogenous and exogenous 'pressures' (poverty, globalisation, resource use) which influence the interactions between urbanisation and sustainable development, altering the 'state' of system characteristics (landscape, demography, human well-being) in the process. These then create measurable 'impacts' (land cover, urban form, population, living standards) of which the framework assumes a set of mitigating 'responses' (government, corporate, citizen) to manage the interactions into more sustainable outcomes. By proposing a conceptual framework, this research attempts to present a more holistic understanding of urban sustainability in the African context, offering a framework within which knowledge can be organised, and recommendations can be made on strategies for urban sustainability transformation.



Figure 2.4: DPSIR-based conceptual framework applied to this study.

Source: Author, as modified from OECD (1994), EEA (1995) and IEHIAS (2019).

Chapter Three

3. Data and Methods

3.1 Introduction

In this chapter, the primary research methods engaged in collecting and analysing data for this study are presented. Considering the already highlighted trans-disciplinarity of sustainable development, as well as its complex inter-relationships with urbanisation, a pragmatic philosophical viewpoint that offers freedom from "*the forced-choice dichotomy between positivism and constructivism*" (Creswell and Plano Clark 2007, p.44) is assumed for this study. Therefore, a positivist approach (supports quantitative research) which assumes that the external world is made of a singular objective reality that is not aligned to human perception (Poortman and Schildkamp 2012), and a constructivist approach (supports qualitative research) that seeks a social construction of realities (Florczak 2014), are both adopted to analyse the interactions between sustainable development and urbanisation in Ghana.

In general, given the focus of this research, and consistent with the philosophical viewpoint assumed, a mixed-method approach and case study design are adopted in this study. The methods used in collecting data include quantitative questionnaires, field observations, satellite data analysis, and documentary analysis. Data collection was guided by the research questions set out in Chapter One and took place throughout the study period. The field trips conducted as part of this study took place in stages. The first field trip was a pilot study which took place from October to November 2016. The pilot study enabled an analysis of the wider research project, and also provided the opportunity to test the validity and reliability of some of the research tools before designing an in-depth study. The primary data collection stage took place from January to March 2018. The amount and types of data used in this study include: (i) 624 questionnaires; (ii) four sets of satellite imagery; (iii) two urban policy documents; and (iv) numerous published data and grey literature from websites. The main locales for field data collection were Kumasi and Obuasi (which are located in central Ghana) as shown in Figure 3.1.



Figure 3.1: Map of Ghana showing Kumasi, Obuasi, and Appolonia City. Source: Adapted from Anarfi et al. (2020).

3.2 Methodological choices

In alignment with the pragmatic philosophical viewpoint adopted, a mixed-method approach, defined as "an approach to research in the social, behavioural, and health sciences in which the investigator gathers both quantitative (closed-ended) and qualitative (open-ended) data, integrates the two, and then draws interpretations based on the combined strengths of both sets of data to understand research problems" Creswell (2015, p.2) is used in this study. The usefulness of a mixed-method approach in this study is characterised by the ability to synergistically combine available methods in order to provide complete knowledge that is necessary to inform theory and practice (Burke-Johnson and Onwuegbuzie 2004). Critically, the integration of qualitative and quantitative data in the mixed-method design adopted is intended to achieve a fuller understanding of the research questions.

This study is premised on an overarching collective case study of urbanisation and sustainable development in Ghana, as well as two instrumental sub-case-studies which are designed to provide a comparative analysis of the interactions between urbanisation and sustainable development at the local urban level (Kumasi and Obuasi), and to provide an overview of sustainable city development (Appolonia City) in Ghana. In this thesis, a case study is defined as "an empirical enquiry that investigates a contemporary phenomenon within its real-life context, especially when the boundaries between phenomenon and context are not clearly evident", and it "relies on multiple sources of evidence" (Yin 1994, p. 13). In this context, an instrumental case study examines a particular case in order to gain broader insights, while a collective case study builds on an instrumental case study by examining multiple cases concurrently or sequentially to deepen understanding of a particular phenomenon (Stake 1995). Despite the difficulties associated with case studies, which include: complexities in scoping and designing case studies such that they adequately answer the research questions; accumulation of copious amounts of data; difficulty in making generalisations; and time-consuming tendencies (Yin 1994; Cavaye 1996; Darke et al. 1998); the case study approach in this study is expected prove advantageous by using diverse data collection and analysis methods to study the urbanisation and sustainable development processes in Ghana.

Within the case study setting, archival research was used to: (i) frame analysis of urbanisation and sustainable development in Ghana; (ii) analyse data on local level urbanisation and socio-

economic development; and (iii) analyse urban-related policy documents in Ghana. In this context, archival research involves the analysis of data that have been generated and stored primarily for purposes other than academic research (Heng et al. 2018). Unlike traditional research which uses prospective data by following processes which include generation of research question, research design, and data collection; archival research uses pre-existing data which means that data are collected before other research steps (Heng et al. 2018). Archival research uses administrative records and documents as the primary source of data. A key strength of archival research revolves around research replication, especially when publicly-accessible data are used. In this case, archival research promotes transparency and replicability of research as other researchers could attempt to replicate data analysis by using the same dataset and procedures. However, when archival data is privately-held, there is a potential lack of transparency as it becomes more difficult for other researchers to verify the correctness and reproducibility of the data analysis. Consequently, the archival data used in this research were publicly-accessible. Furthermore, remote sensing software was used to analyse landscape urbanisation through satellite data analysis, while a face-to-face survey method was adopted to explore the perceptions of citizens. A survey, as used in this study, is defined by Bryman (1989, p.104) as "the collection of data on a number of units with a view to collecting systematically a body of quantifiable data in respect of a number of variables which are then examined to discern patterns of association".

3.3 Data collection

According to Johnson and Turner (2003, p.299), "data collection methods should be combined so that they have different weaknesses and so that the combination used by the researcher may provide convergent and divergent evidence about the phenomenon being studied". Therefore, the next sub-sections will introduce the main data collection instruments used in this study.

3.3.1 Questionnaire administration.

According to Allery (2016, p.234), a questionnaire is "a series of pre-set questions, designed to address specific research questions, allowing a researcher to collect the same information, in the same way, from every respondent". The literature reviewed in Chapter Two alludes to the importance of sociological interpretations of citizens' perceptions and preferences, and how they shape their actions and way of life (see, for example, Macke et al. 2018). Also, it suffices to say that the rapid nature of contemporary urbanisation means that managers of urban areas are not sufficiently in sync with the changing dynamics of human values, and in the context of this study, the meanings and values that citizens attribute to the various dimensions of sustainability. Understanding these preferences and perceptions, therefore, represents an ideal foundation on which citizen involvement in urban decision-making could be advanced (Delmas and Toffel 2004). As such, the survey instrument upon which this research is based was designed in such a way that useful information required to gauge the perceptions of citizens concerning the main themes could be extracted from the data.

3.3.2 Observation.

The two sites for the comparative sub-case-study were observed multiple times throughout the data collection phase in order to capture the reality on the ground regarding the prevailing urban conditions and everyday-life routines of the urban populace in the study sites. Participants were observed during surveys based on ethical considerations.

3.3.3 Documentary analysis.

Document analysis is the process of eliciting, understanding, and developing empirical knowledge through the systematic review and evaluation of printed and electronic material (Corbin and Strauss 2008; Bowen 2009). Merriam (1988, p.118) argues that "documents of all types can help the researcher uncover meaning, develop understanding, and discover insights relevant to the research problem". Accordingly, the importance of document analyses in social science research, especially where it involves data from the national census, national policies, and public surveys, has been highlighted by scholars (see, for example, Matthews and Ross 2010). Documentary analysis was carried out in this research as a primary source of empirical data collection to complement other research methods (e.g., as ancillary data for land cover change analysis), and also for data triangulation purposes (see Bowen 2009).

3.3.4 Critical and systematic review of literature.

A critical review of literature was carried out to help guide this research and to achieve the different objectives of this study. The sources of data scoped included peer-reviewed journal articles and grey literature. Furthermore, consistent with suggestions about the difficulty of developing indicators for sustainability research that was highlighted in Chapter Two, a review of literature was conducted in order to help establish the indicators and related proxies that are appropriate for analysing urbanisation and sustainable development in Ghana. To this end, the Google Scholar database was searched for peer-reviewed articles published in scientific journals. The main search terms used in combination were 'urban sustainability assessment' and 'Africa'. The following filters were applied in order to streamline the search and identify the most relevant articles: (i) the article should be in English and it should have been published between 2010 (chosen to reflect the year in which Ghana became more urban than rural) and 2017 (chosen to precede the main field data collection period for this study in 2018); and (ii) the article should explicitly address urban or national sustainability assessment or indicators. Considering the scope of this study, the first 200 articles produced in the search were reviewed and the details of seven key articles based on which the indicators for sustainability assessment in this study were established are attached in Appendix 1. However, due to the already highlighted paucity of data on urban systems in SSA, the indicators or related proxies selected for sustainability assessment in this study have been based on factors such as data availability, measurability, relevance, as well as how they adequately facilitate answering of the research questions (see Journard and Gudmundsson [2010]).

3.4 Ethics and integrity

Ethical considerations are essential to any research process irrespective of the nature of data collection. Flinders (1992) has highlighted the relationships among the utilitarian, deontological, relational and ecological ethical approaches; and aspects of research including recruitment, fieldwork and reporting. In general, ethical issues in this research were considered based on a utilitarian approach (see Flinders 1992) as participants were recruited based on informed consent, fieldwork was conducted in a manner that avoided or minimised negative externalities on participants, and the confidentiality of reports and participants were guarded.

3.4.1 Ethical issues in the questionnaire survey.

The issue of confidentiality in conducting interviews and face-to-face surveys is widely reported in the literature (for example, see Ensign 2003; Kidd and Finlayson 2006; Clark and Sharf 2007). A prominent source where confidentiality could be compromised is during the write up of interview or survey findings, especially where quotes have been used (Sandelowski 1994; Rowan 2000). Also, while anonymising participants could make them unidentifiable by the public, they could still be identified by colleagues who have taken the same survey based on the information given out (Allmark et al. 2009). In keeping with the duty of protecting participant confidentiality, consideration was given to the idea of using initials or pseudonyms to anonymise participants and also the possibility of changing easily identifiable details in reporting the findings, as recommended by Richards and Schwartz (2002). Participants were constantly reminded of their right to opt out of the survey at any stage or time.

The issue of informed consent in research has been raised by several commentators (see, for example, Ensign 2003; Walker 2007). Informed consent in the survey is particularly important if the earlier-mentioned issue of confidentiality is to be protected (Cowburn 2005). Thus, at the centre of scholarly recommendations on consent in the survey process is the need to adequately furnish participants with full details on the nature of the research, and more importantly, to secure a form of written consent (Cowburn 2005). Ensign (2003) also argues for the use of oral assent in exceptional cases, for example, when dealing with vulnerable groups. Thus, an ethical imperative of this research was ensuring that participants were well-informed about the key objectives and given reasonable time to review and reflect on them before consenting. In order to achieve this, participants were given research information and participant consent sheets in hard copies, which were to be read, understood, and signed. In a less-developed country such as Ghana, where literacy is not universal, the researcher explained the research objectives to less-literate participants in the local language before seeking consent. The informed consent of participants was secured before every face-to-face survey.

'Harm' in qualitative research has been widely discussed (see, for example, Saunders et al. 2015). Harm in qualitative research could range from 'blows to self-esteem' to threatening participants' interests, position or brand (Saunders et al. 2015). The main implications of
'harm' on a research process are its impact on access and quality of data (Bird and Spier 2008). Also, Bird and Spier (2008) go on to suggest that potential conflict of interest from the publication of a study could result in pressure from various stakeholders to revise or omit sensitive conclusions. At all stages in this research process, the issue of 'harm' was guided by the often-posed question by McCall and Simmons (1969, p. 276): *"to whom shall harm be done in this study, and in what magnitude?"*

From the onset, the researcher set out to identify health and safety risks and other risks of this research to both the researcher and participants. The main risk posed by the researcher was based on the reality that the researcher, as an 'outsider', was unknown to some participants / organisations, was likely to 'waste people's time', and could potentially make conclusions based on weak evidence. This was partly mitigated by the initial research trip, which helped the researcher to familiarise with the locations and network with some stakeholders. Risks to the researcher included the fact that findings could be unpopular amongst stakeholders when communicated. This was partly mitigated by being transparent in communications with them, helping to establish trust. No other significant risks were identified. It is important to note that the researcher did not set out to harm or expose any participant or organisation during the survey process or in the subsequent presentation of findings. During the survey, there was ongoing reflectivity (see Cohn and Lyons 2003; Shaw 2003) as the researcher always circled back in reflection at various points in order to connect and make sense of issues arising, and at the same time respond sensitively to the needs of participants. Also, participants were assured continuously of the researcher's commitment to high ethical standards by presenting a fair and honest account of each interaction. There were no complex ethical dilemmas as a result of this research. In general, all protocols in this research were conducted within the ethical framework approved by Bournemouth University. This involved undergoing training on research ethics, scrutinising the research plan to identify ethical hotspots and ways of mitigating any possible factors that might compromise the ethical integrity of the research, and also establishing a list of assumptions and expectations as an ethical boundary that showed the limits within which the research should be conducted.

3.5 Methodological choices specific to research objectives

The main aim of this study is to explore the relationship between urbanisation and sustainable development in the context of Ghana in order to make recommendations for urban

sustainability and to develop a conceptual model that could facilitate decision-making for the transition to sustainability. Thus, this section discusses the specific methodological choices made and organises them in relation to the specific objectives outlined in Chapter One.

3.5.1 Methods for Objective One.

Objective One⁸ aims to apply a conceptual research framework that organises the interactions between urbanisation and sustainable development processes in an integrative and holistic manner to reflect the African situation based on the Ghanaian case. In the purview of the present research, the question arises as to the nature of the interactions between urbanisation and sustainable development in the African context. Therefore, in Africa where urbanisation is rapid and multiple developmental challenges abound (Cloete 2015; Cobbinah and Aboagye 2017), it is vital to present the real-life situation by devising a means of incorporating the main domains of urbanisation and sustainable development in a way that helps to identify the potential synergies and conflicts that must be managed and governed if progress is to be made towards achieving urban sustainability. Nevertheless, as established in Chapter Two, SD as a concept is complex and fuzzy, and there is a paucity of comprehensive frameworks based on which the complexities of SD could be holistically understood (Jabareen 2008). Furthermore, the trans-disciplinary nature of sustainable development means that the lack of a comprehensive framework of evaluation potentially undermines an understanding of its complex inter-relationships with the urbanisation process. Consequently, in building an integrated vision of urban sustainability in this research, the application of a conceptual framework is proposed. A framework has been described as an entity between a 'model' and 'method', that contains "a (not completely detailed) structure or system for the realisation of a defined goal" (Verbrugge 2019, para.4). Critically, depending on its use, a framework may consist of one or more models (Verbrugge 2019). A model represents a "simplified representation of a system at some particular point in time or space intended to promote understanding of the real system" (Bellinger 2004, para. 1). Norman (1983, apud Greca and Moreira 2000, p.5) defines conceptual models as "precise and complete representations that are coherent with the scientifically accepted knowledge". Fundamentally, conceptual models are critical for integrating and representing existing and generated knowledge (Struss 2004).

⁸ Note: Objective One has already been achieved and presented in Figure 2.4. Therefore, it is presented here in order to ensure overall coherency and completeness of the methods used in achieving the objectives in this study.

The choice of a conceptual framework over other types of models discussed in Chapter Two was based on its strengths in presenting particular concepts and also in triggering policy responses (Todorov and Marinova 2011). This choice was also influenced by the scope of the research, as well as the limited financial and time resources available to carry out this research. According to Pearce (2008), developing a model enables a researcher to transcend a descriptive boundary into the explanatory. The conceptual framework applied in this research is intended to help achieve a coherent and more holistic view of urbanisation and sustainable development in the African context. As shown in the literature, different geographic regions have different local environmental, socio-cultural, and economic conditions. This implies that models developed for a particular geographic area cannot achieve comparable results when applied in another jurisdiction, as respective regions have different ways of perceiving and achieving their developmental objectives. Nevertheless, some commonalities exist among regions. Therefore, the integrated conceptual framework expounded here is not intended to represent a rigid code of the urbanisation and sustainable development processes, but rather to integrate knowledge from different sources in order to help operationalise the urban sustainability concept and make policy recommendations irrespective of the socio-economic differences between urban areas.

3.5.2 Methods for Objective Two.

Objective Two aims to evaluate the overall sustainable development situation and urbanisation patterns in Ghana. This was achieved in three parts. The first two parts synthesised extant literature in order to provide a review of urban challenges and an overview of sustainable development in Ghana. In order to assess the overall sustainable development situation in Ghana, this study followed the methods used by Holden et al. (2014). Four primary dimensions of sustainable development were identified from the Brundtland Commission's definition of sustainable development (WCED 1987; Holden et al. 2014). These are (i) safeguarding long-term ecological sustainability; (ii) satisfying basic human needs (iii) promoting inter-generational equity, and (iv) promoting intra-generational equity. Table 3.1 presents the indicators used to measure the primary dimensions and the threshold values suggested for sustainable development based on the target year of 2030. The sustainable development situation was, therefore, analysed based on threshold levels for indicators for which development was characterised as sustainable based on a four-dimensional space called 'sustainable development space' (Holden et al. 2014).

Dimension	Indicator	2030 Threshold	Source of Data
(i) Safeguarding long-term	Yearly per capita	Maximum 1.7	Global Footprint
ecological sustainability.	ecological	Global hectares	Network (2019);
	footprint.	(gha) per capita.	UNDP 2019a.
(ii) Satisfying basic	Human	Minimum 0.630.	UNDP (2018; 2019;
human needs.	Development		2019b); World
	Index.		Population Review
			(2019).
(iii) Promoting inter-	The proportion of	Minimum 27%.	Energy Commission
generational equity.	renewable energy		(2016; 2018; 2019).
	to total energy in		
	primary energy		
	production.		
(iv) Promoting intra-	Gini coefficient.	Maximum 40.	World Bank (2019a).
generational equity.			

Table 3.1: Dimensions, indicators, and suggested 2030 threshold values for sustainable development.

Source: adapted from Holden et al. (2014).

The following explanations are offered for the thresholds used in examining the sustainable development space. According to Holden et al. (2014), ecological footprint is strongly correlated with energy consumption and they used 2.3 gha per capita as an adjusted value based on the call for a 15% reduction in energy consumption by 2030 (WCED 1987; WWF 2008; WWF 2016). However, a threshold of 1.7 gha was used in this study to reflect worldwide average bio-capacity as of the end of 2018 (UNDP 2019a). The minimum HDI of 0.630 was chosen to reflect the minimum requirement a country must meet in order to safeguard the basic needs of its citizens and this was chosen as an average based on the threshold HDI for countries with 'medium human development'⁹ (UNDP 2019a). The maximum Gini coefficient of 40 was chosen to reflect the target level set by UN-Habitat (United Nations 2010). Finally, the minimum 27% renewable energy reflects the proportion

⁹ According to World Population Review (2019), there are four tiers of the HDI: very high human development (0.8 -1.0), high human development (0.7 – 0.79), medium human development (0.55 – 0.69) and low human development (below 0.55).

needed by 2035 in order to be consistent with a 450 ppm CO_2eq stabilisation level recommended by the IPCC (2011). The sources of data consulted to evaluate sustainable development in Ghana are listed in Table 3.1.

For the third part of Objective Two, literature was reviewed in order to provide an overview and to assess the urbanisation situation in Ghana. The sources of data used are provided in Table 3.2. From these data, relevant information was extracted and presented in the form of tables and figures in order to guide the discussion of the overall urbanisation situation in Ghana.

 Table 3.2: Source of data for dimensions of urbanisation in Ghana.

Source of data
Cobinnah et al. (2015a) as adapted from
UNDESA/PD (2012) and UNDESA/PD (2013)
Cobinnah and Niminga-Beka (2017) as
adapted from GSS Records (2012; 2013).
Ghana Statistical Service (2014a).
World Bank (2019b).

3.5.3 Methods for Objective Three.

Objective Three aims to provide a deeper understanding of the Ghanaian situation through a comparative urban sub-case-study that specifically: (i) examines the patterns of landscape and demographic urbanisation and the implications for sustainable development; (ii) evaluates the sustainable development situation based on recognised and relevant indicators from the literature; and (iii) explores urbanisation and sustainable development based on the subjective experience (perceptions) of citizens.

3.5.3.1 Study site selection.

Two urban areas in Ghana (see Figure 3.1) have been selected as study sites in a comparative sub-case-study that investigates the urbanisation and SD process in the African context. The first case study site is Obuasi, located in the Ashanti region of Ghana. From a mineral resource economy perspective, the choice of Obuasi is of particular importance. This is because it is the location of the oldest gold-mining shaft in Ghana (operated by AngloGold Ashanti), and until the first decade of the 21st century it was the principal mining hub in Ghana (Wan 2014). In 2014, AngloGold Ashanti halted its operations in Obuasi due to the reduced performance of the mine and operations resumed in the first quarter of 2019 (Ghana Chamber of Mines 2019). The closure of the mine had development implications for Obuasi and its environs as AngloGold's operations in Obuasi are a key source of employment generation (Ghana Chamber of Mines 2019). The lack of research on urbanisation and mining has also been highlighted in the literature (Gough and Yankson 2012). As a result, Obuasi is a good case study for understanding the nexus between sustainability and urbanisation in the context of a resource-based city. This is particularly important within SSA, a region where most economies are based on natural resource extraction (UNECA 2012a; Turok and McGranahan 2013). The second case study site is Kumasi, the capital of the Ashanti region of Ghana. Kumasi was selected as a case study for four main reasons; it is the second biggest city in Ghana (after Accra, the national capital); it is centrally located with extensive transport links to all parts of Ghana; it is the fastest growing city in Ghana (Oduro et al. 2014); and Kumasi's primacy, by virtue of being the capital of the Ashanti region, presents an opportunity to examine the opportunities for and challenges to sustainable urbanisation in a rapidly expanding urban area. Overall, it is expected that comparatively assessing the interactions between urbanisation and sustainable development in Kumasi and Obuasi will prove useful as both urban areas exist within Ghana, and will, therefore, have broadly consistent context, history, and policy settings. This potentially helps to separate the influence of national factors such as policy, and makes it possible to focus on the more specific characteristics of Kumasi and Obuasi.

3.5.3.2 Demographic and landscape urbanisation.

The first part of Objective Three sought to examine the patterns of landscape and demographic urbanisation and the implications for sustainable development in two urban areas in Ghana. In terms of demographic urbanisation, literature was reviewed in a desk study to extract population and migration data on the two urban areas (See Ghana Statistical

Service 2014a; 2014b; 2014c; 2019). In terms of landscape urbanisation, a spatio-temporal analysis of land cover change was conducted. The sustainability of landscape urbanisation was discussed in terms of whether urban expansion in Kumasi and Obuasi resulted in sustainable urban forms (based on the definitions and assumed benefits of urban form characteristics provided in Table 2.1).

3.5.3.2.1 Spatio-temporal analysis of land cover change (LCC).

Four sets of satellite imagery were used to analyse the extent of landscape transformation in the two urban areas over a period of 32 years (1986-2018). Critically, Ghana's equatorial climate and proximity to the Gulf of Guinea result in the presence of significant cloud cover (50-80%) on satellite imagery for most parts of the year (see Coulter et al. 2016). Therefore, the imagery for this study were carefully selected to ensure that there is less cloud cover such that any further processing required would be within the expertise of the researcher and the scope of this study. The extent of landscape transformation in Kumasi meant that an arbitrary area of interest was demarcated as a buffer of 8 km from the official metropolitan boundary. This was to ensure that the full urban extent, both within and outside the official boundaries, was accounted for. Specifically, the 8-km boundary was chosen to reflect areas within what is now referred to as Greater Kumasi. A buffer below 8 km would cut off areas that fall under Greater Kumasi, while a buffer beyond 8 km would result in large swathes of land well outside Greater Kumasi being captured. The nature of landscape transformation in Obuasi, however, meant that the area of interest was limited to the official municipal boundary. The characteristics of the satellite imagery used are presented in Table 3.3. In terms of characteristics of the sensors used to acquire the images, the radiometric resolution¹⁰ is as follows: Landsat 5 TM and Landsat 7 ETM+ have radiometric resolutions of 8 bits (0 - 255 brightness values), and Landsat 8 OLI TIRS has a radiometric resolution of 12 bits (up to 4096 brightness values) which is transformed into 16 bits (up to 65536 brightness values) when processed into Level-1 data (Landscape Toolbox 2019a, 2019b, 2019c; University of Rhode Island 2019). In terms of spectral bands, the Landsat 5 TM has six 30m bands which cover the visible, near infrared and short-wave infrared areas of the electromagnetic spectrum, and one 60m thermal infrared band; Landsat 7 ETM+ has eight bands, including

¹⁰ Radiometric resolution describes the sensitivity of a sensor to electromagnetic energy. This means that the finer the radiometric resolution the higher its sensitivity to differences in reflected or emitted energy (Huete 2004).

six 30m bands which cover the visible, near infrared and short-wave infrared areas of the electromagnetic spectrum, one 60m thermal infrared band, and one 15m panchromatic band; and the Landsat 8 OLI_TIRS has eleven bands, including eight 30m bands which cover the visible, near infrared, short wave infrared, coastal / aerosol and cirrus areas of the electromagnetic spectrum, one 15m panchromatic band, and two 100m thermal infrared bands (Landscape Toolbox 2019a, 2019b, 2019c). The specific bands selected for the classification of the Landsat 8 OLI images (2018) were Band 2 (Blue [visible]), Band 3 (Green [visible]), Band 4 (Red [visible]), Band 5 (Near Infrared), Band 6 (Shortwave Infrared 1) and Band 7 (Shortwave Infrared 2). The specific bands selected for the classification of the Landsat 7 ETM+ (2000; 2010) images were Band 1 (Blue [visible]), Band 2 (Green [visible], Band 3 (Red [visible]), Band 4 (Near Infrared), Band 5 (Shortwave Infrared 1) and Band 7 (Shortwave Infrared 2). The specific bands selected for the classification of the Landsat 5 TM (1986) and Landsat 7 ETM+ (2000; 2010) images were Band 1 (Blue [visible]), Band 2 (Green [visible], Band 3 (Red [visible]), Band 4 (Near Infrared), Band 5 (Shortwave Infrared 1) and Band 7 (Shortwave Infrared 2). The specific bands selected for the classification of the Landsat 5 TM (1986) and Landsat 7 ETM+ (2000; 2010) images were Band 1 (Blue [visible]), Band 2 (Green [visible], Band 3 (Red [visible]), Band 4 (Near Infrared), Band 5 (Shortwave Infrared 1) and Band 7 (Shortwave Infrared 2). The spatial resolutions of all the image bands used in this study were 30 m. The Path / Row for the images were 194 / 55 (Kumasi) and 194 / 56 (Obuasi).

Location	Date	Satellite (sensor)	Specific bands used
Kuması	29/12/1986	Landsat 5 TM	
			1 2 2 4 5 5
Obuasi	29/12/1986	Landsat 5 TM	1, 2, 3, 4, 5, 7
Kumasi	04/12/2000	Landsat 7 ETM+	
Obuasi	04/12/2000	Landsat 7 ETM+	
Kumasi	02/06/2010	Landsat 7 ETM+	1, 2, 3, 4, 5, 7
			-
Obuasi	02/06/2010	Landsat 7 ETM+	
Kumasi	01/19/2018	Landsat 8 OLI	
Obuasi	01/19/2018	Landsat 8 OLI	2, 3, 4, 5, 6, 7

Table 3.3: Characteristics of satellite sensors and images.

(a) Image pre-processing.

The Landsat imagery acquired was processed according to the United States Geological Survey (USGS) Level-1 Product Generation System, which involves precision and terrain correction where radiometric and geodetic accuracy are provided by *"incorporating ground control points while employing a Digital Elevation Model (DEM) for topographic displacement"* (USGS 2019, para.3).

(b) Image classification.

Critical to the success of the complex image classification process is the selection of a suitable classification system (Lu and Weng 2007). Accordingly, seven distinct LULC classes of interest to the researcher in the context of urbanisation analysis for the study areas were identified (based on a modification of the Anderson Scheme Level 1 method). These include water, urban / built-up, mixed woody vegetation, bare land, forest, mine sites, and agricultural land (Anderson et al. 1976), as described in Table 3.4. However, as Kumasi and Obuasi have different land cover characteristics, five classes were identified as the final land cover class types for analysis in both sites, which include water, urban / built-up, mixed woody vegetation, agricultural land, and forest; whilst additional two classes consisting of mine sites and bare land were identified in Obuasi.

LULC Class	Description
Water	Permanent open water, lakes, reservoirs, rivers, bays.
Urban / Built-up	Residential, commercial, industrial, transportation and utilities, mixed
	urban and built-up land, paved-over areas.
Mixed Woody	Lower stature secondary growth vegetation and forest; gallery forests
Vegetation	along rivers, woody savannahs with high vegetation cover, tree
	plantations.
Bare Land	Bare soil, quarry and disturbed lands, dirt roads, exposed rocks.
Agricultural Land	Farmlands, fallowing shrublands.
Primary Forest	Dense primary forest, high stature forest.
Mine Sites	Mine sites and mine tailing ponds.

Table 3.4:	Land	cover	classes	analysed.

Source: Adapted from Anderson et al. (1976).

The satellite images were classified by supervised classification, through the widely-used maximum-likelihood classifier (Eum et al. 2011). Using the ERDAS IMAGINE (version 15.0) software, spectral signatures were created from training sites on the satellite images by digitising polygons representing all the identified land cover types shown in Table 3.4. As shown in Table 3.5, a total of sixty ground reference points were collected using a Garmin Global Position System (GPS) device for the classification of the 2018 image in Kumasi and consisted of the following: water (2); urban / built-up (18); mixed woody vegetation (20); agricultural land (10); and forest (10). For Obuasi, a total of sixty ground reference points were also collected for the 2018 image and consisted of the following: water (4); urban / built-up (18); mixed woody vegetation (10); agricultural land (10); mine sites (8) and forest (10), as shown in Table 3.6. By using the area of interest (AOI) tools in ERDAS IMAGINE (version 15.0), half of the ground reference points were used to create training polygons and the remaining used for accuracy assessment. The total number of ground reference points used in all four image dates are summarised in Table 3.5 (Kumasi) and Table 3.6 (Obuasi). For the historical images (1986, 2000 and 2010), the training sites were digitised based on a careful study of the imagery, Google Earth historical aerial imagery (where available), topographical maps, local knowledge and previous studies. The selection of reference points was done for unchanged land cover areas only and juxtaposed with the ground reference points collected during the 2018 field work. In doing so, it was assumed that there was a historical permanence in land cover types such as built-up, some mine sites, and some agricultural land; therefore, they had not changed over the study period. For each image (1986; 2000; 2010), half of the reference points were selected randomly as training samples, and the remaining half was retained for accuracy assessment. The underlying principle in the digitisation of training polygons was that the Euclidean distance between the respective spectral signatures represents the level of separability of the signatures created for the various land cover classes. Overall, scholars have acknowledged the importance of sufficient and representative training samples in image classification (see Hubert-Moy et al. 2001; Chen and Stow 2002; Mather 2004), and Mather (2004) recommends a training sample size of 30 times the number of bands per class. Critically, the number of reference points used in classifying the 2018 images was lower than the ones used for the 1986, 2000, and 2010 images (see Tables 3.5–3.6). This is because the number of GPS ground reference points the researcher could collect from the field for 2018 was limited by logistical and access constraints. However, for the other years (1986; 2000; 2010) the researcher used what was learnt from the ground reference points for 2018, in terms of spectral recognition, to train more areas. This

was because the researcher could not go back in time to collect field data. Following the collection of training sites and signatures, the maximum likelihood classifier was run in ERDAS IMAGINE (version 15.0) software based on a feature-space non-parametric decision rule. The statistics on the general change in the proportion of the different land cover types (especially in the urban / built-up class) have been used to show the extent of land cover change due to the process of urbanisation.

(c) Accuracy assessment.

For any map generated through remote sensing data, it is vital that its accuracy is assessed (Lunetta and Lyon 2004). As mentioned earlier, half of the ground reference points collected during the field work (for 2018 imagery) and half of the randomly sampled points (for 1986; 2000; 2010) were used for accuracy assessment (see Tables 3.5–3.6). A confusion matrix was constructed for each land cover map for accuracy assessment purposes.

Year		Water	Urban /	Mixed	Forest	Agricultural	Total
			Built-up	Woody		land	
				Vegetation			
1986	No. of	20	42	40	22	14	138
	RPs						
2000	No. of	10	50	18	10	36	124
	RPs						
2010	No. of	10	44	14	18	20	106
	RPs						
2018	No. of	2	18	20	10	10	60
	RPs						
			1				

Table 3.5: Number reference points (RPs) in the land cover class types used for training and accuracy assessment for Kumasi, 1986 - 2018.

Table 3.6: Number of reference points (RPs) in the land cover class types used for training and accuracy assessment for Obuasi, 1986 - 2018.

Year		Water	Urban	Mixed	Forest	Agricultural	Bare	Mine	Total
			/ Built-	Woody		land	ground	site	
			up	Vegetation					
1986	No. of	-	20	26	20	20	-	10	96
	RPs								
2000	No. of	2	30	30	10	10	-	10	92
	RPs								
2010	No. of	2	30	30	10	20	4	10	106
	RPs								
2018	No. of	4	18	10	10	10	-	8	60
	RPs								

Note: '-' is placed where a land cover is not identified on the image for a particular year.

3.5.3.3 Assessment of sustainable development in study sites.

The second part of Objective Three aims to evaluate the sustainable development situation in two urban areas in Ghana based on recognised indicators from the literature. Indicators were selected to assess sustainable development in the study sites based on the earlier-mentioned criteria for indicator selection (see Section 2.4). Therefore, the choice of indicators selected was informed by their measurability, comprehensiveness, clarity, reliability, relevance, and most importantly, data availability (Joung et al. 2012). This selection was made using a combined expert and citizen approach where choices were made based on the researcher's expert judgement through the literature reviewed, together with the opinions on indicators expressed by citizens in a pilot study during the earlier stages of this research (see, for example, Pissourios 2013). Based on the centrality of human well-being in urbanisation and sustainable development discourses (as discussed in the literature review), the indicators were selected to capture the part of the sustainable development definition advanced by WCED (1987) which sought to safeguard the basic needs of society in a sustainable manner. The indicators selected were, thus, inspired by the Sustainable Development Goals and assessed on the extent to which they achieved the Millennium Development Goals. Sustainable development of the two urban areas was assessed primarily on the dimensions of the Human Development Index (HDI), for which urban level data were available, or where there was a lack of data, proxies could be used. The selected dimensions of the HDI based on which sustainable development was assessed in the two study sites are presented in Table 3.7. Due to the lack of urban-level GDP, and also due to the contestations of GDP use as a measure of standard of living (see, for example, Costanza et al. 2009; Stiglitz et al. 2010), the levels of access to basic infrastructure and services in Kumasi and Obuasi were selected as proxy measures of standard of living (specifically, access to sanitation and water). Available and relevant data which sufficiently measured education included literacy level, pupil-teacher ratio (PTR), and gender parity index (GPI). Finally, available and relevant data which sufficiently captured life expectancy included Crude Death Rate (CDR) and Maternal Mortality Rate (MMR).

HDI Dimension	Indicators	Source of data
Life expectancy	Crude Death Rate, Maternal Mortality	Ghana Statistical Service
	Rate.	(2014a)
Education	Literacy level, Pupil-teacher ratio, Gender	National Development
	Parity Index.	Planning Commission (2019)
Standard of living	Access to improved sanitation, Access to a	National Development
	sustainable water source.	Planning Commission (2019)

Table 3.7: HDI dimensions, indicators used and source of data.

3.5.3.4 Face-to-face survey of citizens perceptions in study sites.

The third part of Objective Three explores urbanisation and sustainable development in the two urban areas in Ghana based on the subjective experience (perceptions) of citizens. The study adopted a quantitative survey to explore the perceptions and attitudes of respondents in the study sites concerning selected dimensions of sustainable development and urbanisation. Critically, the face-to-face survey used in this study helped to mitigate the lack of disaggregated urban data for the study areas by facilitating the soliciting of information on important indicators of sustainability based on the experiential knowledge of citizens in Kumasi and Obuasi.

3.5.3.4.1 Questionnaire Administration.

The study used a questionnaire designed by the researcher. Questionnaires consisting of a combination of close-ended and 5-point Likert-scale questions and statements were administered to residents of Kumasi and Obuasi in a face-to-face survey from January to March 2018. The choice of orally-administered face-to-face surveys was to mitigate any bias due to low-level literacy and the technical nature of the concepts being examined. A previous pilot study conducted in November 2016 provided real-world responses, and the feedback was critical in designing the main survey questionnaire as it served as an initial pre-test through which the main aspects of the survey were considered. Before the main survey period, the draft questionnaire was discussed with one academic from the local university (KNUST) and a professional planner. The draft version was then administered to 5 residents in Ayeduase, a suburb of Kumasi in order to determine how the respondents would react to the questions and statements. The final pre-testing process led to the reconstruction of sentences where understanding was not very clear, and the reduction of survey questions from 35 to 28. Overall, the construction of the questionnaire took into account relevant literature on urban sustainability and was designed in such a way that citizens' understanding and experience of sustainable development and urbanisation was captured. The questionnaire consisted of five main parts. The first part collected demographic information of respondents including age, gender, level of education, length of residency, housing type, and household size. It was essential to gauge citizens' perspective of the various dimensions of sustainability based on their own life experiences. After all, as the Thomas theorem opines: "if men [sic] define situations as real, they are real in their consequences" (Thomas and Thomas 1928, p.571-572). Thus, the second part explored citizens' views regarding the extent of agreement (on a scale of 1-5) with statements on the various dimensions of sustainability. The third part sought the level of agreement of citizens on the subject of urbanisation (based on a scale of 1 to 5) by exploring their views on issues such as the extent of demographic and landscape change, and the extent to which the level of development was commensurate with rates of urbanisation. The fourth part of the questionnaire explored attitudes towards urban sustainability, which required responses of 'yes', 'no' and 'maybe'. The final part featured an open-ended 'any comments' section where general views were sought from participants. As the questionnaires were administered to general citizens of diverse backgrounds, the questions were drafted in such a way that they conveyed the urban sustainability concept into a language that non-academics and experts could understand. Consistent with its quantitative foundations, the questionnaires were generally designed to ensure reliability, objectivity, and generalisability (see Mathers et al. 2007). As such, the results are intended to be replicable

irrespective of the person conducting the research. In effect, as Groves (1989, p.358) argues, the "goal is nothing less than the elimination of the interviewer as a source of measurement error". As face-to-face surveys require special skills in order to achieve consistency and robust insights into complex issues, the researcher took a course in 'research methods: surveys and questionnaires' as part of the researcher development programme in Bournemouth University. This ensured that the researcher honed his skills in effectively collecting, preparing, and analysing survey data. This was complemented by engagement with literature on face-to-face survey strategies and mock role-play of questionnaire administrations. Overall, the questionnaires administered were intended to provide a level of validation to the information gathered from literature and other methods employed. A sample of the questionnaire is attached in Appendix 2.

3.5.3.4.2 Participant selection.

A stratified random sampling method was adopted to select survey participants which involved drawing a random sample based on particular characteristics (Teddlie and Yu 2007). Survey participants were selected in a way such that the broad range of densities / classes were represented. In selecting participants in Kumasi, participants were approached from at least one suburb in each of the nine sub-metropolitan assemblies (at the time of the study) which made certain that low, middle, and high-income areas were covered. Unlike in Kumasi, Obuasi's population size means that it has no sub-municipals, but the municipality is rather organised around suburbs / towns. Based on local knowledge and ground observation, questionnaires were administered in mining quarters and estates to represent high-income areas, while every other location in the municipality was marked as low- / middle-income. Responses were elicited from both long-term residents and new residents who were aged 18 or older, although a minimum threshold of two years residence was used in order to ensure respondents were adequately familiar with a respective urban area. In all, questionnaires were administered in the closest six suburbs or towns from the centre. In administering the questionnaires, the researcher and an assistant (with no technical input), on reaching a survey community, explained the research and sought permission from a local representative to conduct the survey. Generally, the researcher sought to target every other household located in the main streets of suburbs. A total number of 624 questionnaires were administered, and the breakdown is presented in Table 3.8. The response rate was 100% because the survey only proceeded with the consent of the respondents. However, four of the questionnaires were invalid as they had missing values. Out of the 624 participants to whom the

questionnaires were administered, 15 of them (9 in Kumasi and 6 in Obuasi) provided comments which were considered relevant for discussion on the general perception of citizens based on the subject matter being surveyed. These comments are attached in Appendix 3.

Location	Total number administered	Invalid Questionnaires
Kumasi	321	3
Obuasi	303	1

 Table 3.8: Breakdown of questionnaires administered.

3.5.3.4.3 Analysis of survey data.

As the survey data were taken in paper form, the results were subsequently transferred manually and organised in Microsoft Excel software. This was followed by processing in IBM SPSS (version 20.0) software where statistical analysis was performed. In addition to frequency tables and bar charts used to present the survey data, the SPSS software was used to analyse group difference in perceptions of citizens. These were tested using non-parametric statistical tests. The differences in perceptions of respondents based on variables including age, education level, gender, and economic status were tested using the Mann-Whitney U test for two independent samples and the Kruskal-Wallis H test for more than two independent samples respectively. Only statements whose responses fulfilled key assumptions for the respective tests were selected for analysis. In terms of assumptions, for both tests, the groups being examined must be independent and the dependent variable should be continuous (Statistics Solutions 2019; University of Sheffield 2019). Consequently, the group differences in the survey responses were determined based on the following hypotheses.

(a) Gender differences in citizens perceptions of sustainability statements.

The decisions on statistical tests for gender differences were based on the following hypothesis.

(i) H_o: There is no statistical evidence of interaction between gender and the responses to a selected statement.

(ii) H₁: There is statistical evidence of interaction between gender and the responses to a selected statement.

(b) Differences in perception due to economic (employment) status.

The decisions on statistical tests for economic status differences were based on the following hypothesis.

(i) H_0 : There is no statistical evidence of interaction between economic status and the responses to a selected statement.

(ii) H₁: There is statistical evidence of interaction between economic status and the responses to a selected statement.

(c) Differences in perceptions due to age.

The decisions on statistical tests for differences in perceptions due to age were based on the following hypothesis.

(i) H_o : There is no statistical evidence of interaction between age and the responses to a selected statement.

(ii) H₁: There is statistical evidence of interaction between age and the responses to a selected statement.

(d) Differences in perceptions due to level of education.

The decisions on statistical tests for differences in perceptions due to level of education were based on the following hypothesis.

(i) H_o: There is no statistical evidence of interaction between level of education and the responses to a selected statement.

(ii) H₁: There is statistical evidence of interaction between level of education and the responses to a selected statement.

3.5.4 Methods for Objective Four.

Objective Four aims to examine the sustainability construct and operationalisation in Ghana's urban policies in order to determine their consistency with the principles of sustainable development. This research adopted Davidson and Arman's (2014) analytical approach for assessing municipal strategies and urban policies in Australia, in order to capture the dominant understanding of urban sustainability in selected sustainability-related public

policies and clarify their current positions in terms of sustainability constructs and operationalisation. Unlike David and Arman (2014) who assessed the sustainability constructs using an evaluation matrix that was based on UN-Habitat's eight emerging trends of urban sustainability, this study, as inspired by Goal 11 of the Sustainable Development Goals (SDGs) to "make cities and human settlements inclusive, safe, resilient and sustainable" (United Nations 2015, para. 1), used an evaluation matrix based on subdimensions of the City Prosperity Index (CPI) launched by the UN-Habitat in 2012 to measure sustainability and prosperity in urban areas (UN-Habitat 2016b). The choice of CPI over the UN-Habitat's emerging trends is based on the reality that the CPI is more comprehensive in terms of the pointers to sustainability, and it is intended to be an updated package for measuring urban sustainability. Thus, an evaluation matrix based on the subdimensions of the CPI was used to assess urban-related policies and the number of ticks assigned is explained as follows. Zero ticks imply an absolute lack of coverage of the subdimension reviewed. One tick was assigned when a sub-dimension has been highlighted but there is a lack of detailed discussion or associated implementation process. Two ticks were assigned when there is a detailed or associated implementation process captured in the document reviewed. The selected documents were the National Urban Policy (NUP 2012) which has been formulated by the Government of Ghana in order to steer Ghana's urbanisation to sustainable outcomes; and the Action Plan (MLGRD 2013) which is ancillary to the NUP and contains specific initiatives to be carried out in order to achieve the goal of the NUP.

Furthermore, unlike David and Arman (2014) who critiqued their sustainability constructs through the lens of Urban Political Ecology, the centrality of urban sustainability in the present discussion of the interactions between urbanisation and sustainable development means that the research drew on the World Bank's (2018) Urban Sustainability Framework (USF) as the lens of analysis for discussions of how the sustainability construct is deployed in the policies reviewed. Centrally embedded within the USF is its measuring framework, which is the benchmark used in ascertaining the deployment of sustainability constructs in this PhD thesis. The primary aim advanced in the measuring framework of the USF is to improve how urban sustainability is understood, and to promote evidence-based integrated planning through six dimensions of sustainability. The six dimensions prescribed by the USF are categorised into enabling dimensions and outcome dimensions, as shown in Table 3.9 and Table 3.10.

Enabling Dimensions		Goals				
1.	Governance and integrated planning	Achieve integrated well-planned urb	an			
		development.				
2	Fiscal sustainability	Ensure accountable governance and fisc	cal			
		sustainability.				

Table 3.9: Enabling dimensions and associated goals.

Source: World Bank (2018).

Ou	tcome Dimensions	Goals
1	Urban economies	Attain sustainable economic growth,
		prosperity, and competitiveness across all parts
		of the city.
2	Natural environment and resources	Protect and conserve ecosystems and natural
		resources into perpetuity
3	Climate action and resilience	Work toward mitigating greenhouse gas
		emissions while fostering the overall resilience
		in cities.
4	Inclusivity and quality of life	Work toward creating inclusive cities and
		improving cities' liveability, focusing on
		reducing poverty levels and inequality
		throughout cities.

Table 3.10: Outcome dimensions and associated goals.

Source: World Bank (2018).

The following questions were also used as a framework for corollary discussions on the review of selected policy documents:

- To what degree is sustainability / sustainable development accentuated within the policy documents?
- Which factors affect the sustainability operationalisation of the policy documents?

3.5.5 Methods for Objective Five.

Objective Five aims to examine sustainable city development in Ghana through a sub-casestudy. In order to analyse how a 'sustainable city' is being conceived and implemented in an SSA context, the 'Appolonia City' project (near Accra, Ghana's capital city), which was captured in the inventory of current 'sustainable city' developments in SSA presented in Section 2.3.3, was selected as a demonstrable exemplar. The location of Appolonia City, relative to the other case study sites, is shown in Figure 3.1. Literature was reviewed in order to give an overview of the Appolonia City project. An evaluation matrix was applied based on the sustainable urban form matrix for Jabareen's (2006) definition of an eco-city (shown in Table 3.11). Numerical values of 3, 2, and 1 were assigned for 'high', 'moderate' and 'low' rankings respectively. The definitions and assumed benefits of the factors that contribute to a sustainable urban form have already been presented in Table 2.1. Data for this sub-case-study, including project statistics and images, were acquired from the website of the developers of Appolonia City.

Factor	Jabareen's definition
Mixed land use	Moderate (2)
Diversity	Moderate (2)
Sustainable transport	High (3)
Sustainable energy	High (3)
Compactness	Low (1)
Greening	High (3)
Density	Moderate (2)

Table 3.11: Sustainable urban form matrix for an eco-city.

Source: Jabareen (2006).

3.5.6 Methods for Objective Six.

Objective Six aims to propose a conceptual model that could facilitate decision-making for the transition to urban sustainability in Ghana, and by extension, SSA. To achieve this, the findings and discussions were consolidated and the components of the DPSIR conceptual framework (applied to organise knowledge on the interactions between sustainable development and urbanisation) was modified. In so doing, the modified model produced an adaptive 'response' component based on the study findings which could facilitate decision-making for urban sustainability transition in Ghana.

Chapter Four

4. Urbanisation and sustainable development in Ghana: a critical analysis

4.1 Introduction

This chapter is an overarching case study that provides a critical analysis of sustainable development and urbanisation in Ghana. Without prescribing and providing appropriate solutions to the challenges of contemporary urbanisation, the sustainable development of Ghana's urban areas will remain a challenge. Thus, the broader objective of undertaking a critical analysis of urbanisation and sustainable development in Ghana is to frame a national perspective that will provide the building blocks upon which every nuance of the interaction between the two phenomena at the local urban level (presented in Chapter Five) could be captured and expressed. In terms of past, present, and future sustainable development and urbanisation in Ghana, key questions emerge which will guide the present discussion. In the context of this thesis: What are the existing urban challenges in Ghana? What are the key factors that influence urban development in Ghana? What is the overall development situation in Ghana in the context of the principles of sustainable development? What are the characteristics and outcomes of Ghana's urbanisation process? What are the independent or mediating factors that shape the urbanisation process? What are the wider implications of urbanisation for sustainability in Ghana?

Following this introductory section, the remainder of this chapter proceeds in three sections. The first section provides an overview of urban challenges, as well as selected key factors that influence urban development in Ghana. The second section provides an assessment of the sustainable development situation in Ghana. The third section provides an analysis of urbanisation in Ghana and discusses the implications of the country's urbanisation for sustainable development. This chapter does not attempt an empirical analysis of the extent to which the various dimensions of sustainability are affected by the urbanisation as the complexity of both processes makes it challenging to assess individual strands in isolation. Rather, for this study, this chapter presents an overall perspective on the sustainability implications of urbanisation in Ghana. This chapter, therefore, does not exhaust all the details of urbanisation and sustainable development in Ghana. Nevertheless, it captures relevant

information on the two phenomena based on available data and situates them within the wider context of this thesis.

4.2 Review of urban challenges and key factors that mediate urban development in Ghana

4.2.1 Urban challenges.

Duranton (2014, p.3) maintains that a key challenge to poor and developing nations "*is to ensure that their urban systems act as drivers of economic growth*" rather than act as a break. Like many other developing countries, the dualistic nature of urbanisation in Ghana presents opportunities as well as challenges. Using health as an example, the concentration of health facilities in urban areas could be an advantage of urbanisation (Vlahov et al. 2005), while rapid urbanisation could also create challenges of inequality in health access in urban areas (Matthews et al. 2010). While challenges may typically differ from one urban centre to another due to factors which include, but are not limited to, population size, geography, and human development, some common challenges can still be identified. At an annual growth rate of 3.4%, rapid urbanisation in Ghana has complicated the government's capacity to provide essential infrastructural and social services (Farvacque-Vitkovic et al. 2008; Okeke 2014). Essentially, Yeboah and Obeng-Odoom (2010) have argued that the urbanisation process has outpaced planning provisions in Ghana. The following sub-sections will, therefore, give an overview of some of the significant challenges confronting urban dwellers and managers in Ghana.

4.2.1.1 Environmental degradation.

The inevitable force of urbanisation has direct and indirect ecological implications for cities (Baabou et al. 2017). In Ghanaian urban areas, the impacts of urbanisation and urban growth on the environment are manifold. Like other cities in SSA, urbanisation has resulted in increased vehicular ownership, increased use of fuel for cooking, poor waste management practices, industrial emissions, and dust from unpaved roads (Amegah and Agyei-Mensah 2017). For example, scholars have observed a worsening level of air quality as cities in the country are known to have very high levels of particulate matter (Dionisio et al. 2010). Dionisio et al. (2010) have estimated annual particulate matter ($PM_{2.5}$) levels in Accra (Ghana's capital city) to range from 39 to 53 µg/m³ at roadside sites and 30 to 70 µg/m³ at

residential sites. This estimation means that particulate matter (PM_{2.5}) levels in Accra are higher than the levels recorded in the cities of Europe and America, which are less than 20 µg/m³ (Brauer et al. 2012). Air pollution (through increasing particulate matter) has farreaching implications for the overall well-being of the urban population in Ghana, as urban air pollution has been linked to increased incidence of cardiovascular, respiratory and lung diseases in SSA cities (Dalal et al. 2011; WHO 2013). The literature shows that urbanisation and associated urban growth in Ghana are taking place in an undirected manner, which have implications for the sustainable use of environmental resources in urban areas and towns (Yankson and Vough 1999; NUP 2012). Increasing affluence in cities tends to alter consumption patterns and affects resource use (Myers and Kent 2003). A study by Mantey and Sakyi (2019) has shown that high-income urban residents in the city of Accra produced high GHG emissions. They further noted that high-income residents who were aware of climate change produced higher GHG emissions compared with those who were not aware (Mantey and Sakyi 2019). Furthermore, in an era of climate change, there is an increasing vulnerability of Ghana's urban citizens to environmental hazards. For example, the country has experienced a frequent occurrence of natural disasters, including floods and droughts over the years (Okyere et al. 2012). In Accra, where there is annual flooding, the situation is further exacerbated by issues such as weak planning regimes and inadequate disaster response capacities, making the urban poor who live in densely populated slums vulnerable (Okyere et al. 2012). In a nutshell, the pace of Ghana's urbanisation exerts pressure on available environmental resources and hampers the ability to respond to environmental hazards in urban areas.

4.2.1.2 Unemployment and informality.

The population of Ghanaians living in 'urban areas' officially exceeded those in rural areas in 2010 (Ghana Statistical Service 2014a; Cobinnah et al. 2015a). The country's urban labour market is fundamental to the subsistence of urban dwellers as it transmits economic growth to workers and their dependents. A burgeoning urban population in a less-developed country such as Ghana means that the demand for jobs exceeds what is available, making urban unemployment widespread (Frank 1968). The negative impacts of urban unemployment are particularly strong for vulnerable groups such as children, youth, elderly, disabled, and the low-income population (OECD 2013). To put the above assertion in perspective, a study in Britain by Bell and Blanchflower (2010) estimated that the cost of being unemployed for six

months at the age of 22 translates to about 8% wage reduction at the age 23. Thus, for the youth aged between 15-24, who constitute about 20% of Ghana's population, and who are 'not in education, employment or training', unemployment presents a potentially harmful effect on their future well-being (OECD 2013).

As is observed in other developing countries, most migrants who move from rural to urban areas end up in the informal sector due to various reasons including a lack of social network (Ofori 2009; Osei-Boateng and Ampratwum 2011). The dominance of the informal sector in Ghana is attributed to reasons which include, but are not limited to, public sector employment freezes and the effects of globalisation, thereby resulting in local private sector firms folding up due to competition from foreign firms (Osei-Boateng and Ampratwum 2011). For urban areas, the precarious employment situation, together with an increasing urban population results in an inadequate supply of formal jobs (Ofori 2009; Osei-Boateng and Ampratwum 2011). The contribution of informality to the employment situation in urban Ghana is significant as the informal sector is estimated to constitute about 80% of Ghana's total labour force (Hormeku 1998; Osei-Boateng and Ampratwum 2011). Also, while Fajnzylber et al. (2006) have argued that small-scale self-employment is often an outcome of choice rather than the lack of formal employment opportunities, the same cannot be said of Ghana, where like other developing countries, unskilled and uneducated citizens abound (Falco et al. 2011). Gillespie (2016) has conducted empirical research related to the struggles over urban space in Ghana's capital city, Accra, and his observation of a large 'informal proletariat' kept out of formal waged employment and housing provision succinctly captures the challenge of informality in urban Ghana. Academic explanations given to urban informality in Ghana have mainly centred on the dispossession of the urban poor of their lands, which in most cases, were a source of livelihood and shelter (Obeng-Odoom 2013a). Informality has profound implications on wage determination and income equality as it is generally associated with low wages compared to formal employment (Bargain and Kwenda 2011; Angel-Urdinola and Tanabe 2012).

4.2.1.3 Poverty and inequality.

A lot has changed since the turn of the millennium as the country has experienced sustained economic growth (World Bank 2015). The overall incidence of poverty in Ghana has reduced substantially, and the country has achieved its MDG target of halving the proportion of the

population living in poverty. Statistical evidence shows that poverty levels had reduced from about 56% in the 1990s to about 24% in 2013. Nevertheless, rapid urbanisation and population growth have resulted in the proportion of urban-poor increasing over time. For example, although the poverty rate in Accra reduced from 23% in 1992 to about 4% in 1999, it increased again to 11% by 2006 (UNICEF 2012). Available evidence also suggests that inequality has increased at the national level as economic gains have not been evenly distributed (World Bank 2015; Cooke et al. 2016). The country's Gini coefficient rose from 37% in 1992 to 42.3% by 2013 (Cooke et al. 2016). Also, by 2006, the wealthiest 20% of the population earned over half of Ghana's income, up from about 44% of national income in the 1990s; the poorest fifth saw their share of national income reduce from 6.9% to 5.2% within the same period (Cooke et al. 2016). From a gender perspective, the literature (for example, Awumbila 2006; Danquah and Ohemeng 2017) suggests that urban poverty is sometimes determined by gender and may be shaped by cultural factors such as social status assigned to men and women, as the level of poverty experienced by women in Ghana's urban regions tends to be more severe than men. The widening inequality in Ghana is inimical to economic development and poverty reduction (Obeng-Odoom 2012). In terms of health, Wilkinson and Pickett (2009) have established a relationship between high inequality and high social and health problems in their study, and they suggest that the former causes the latter. Moreover, they suggest that a country will be better positioned to safeguard the basic needs of its citizens through a reduction in inequality, which will, in turn, reduce socio-economic (health and social) problems. Subsequently, as admitted by the Ghana Statistical Service (2007, p.17), "the decline in poverty [was] offset by increasing inequality". These observations have implications for the overall well-being of the country's citizens (for example see Easterly 2007), especially in urban areas where most of the population is now concentrated.

4.2.1.4 Housing.

A fundamental issue faced by urban areas, as the unabated process of urbanisation continues, is housing. As a basic necessity of life, housing is seen as crucial to human well-being across all spatial levels (Adoeye 2016), a provision Turok (2016) notes will shape the dynamics of society and help to determine whether successive generations are in better situations than their predecessors. Turok (2016) makes a case for access to better urban housing, which he believes has the potential to support two related objectives. He argues that access to durable and affordable urban housing will provide alternative means of sustainably managing urban

poverty as it creates conducive and liveable conditions through which citizens can acquire relevant skills. He further highlights how access to decent housing has the potential to raise urban productivity and output, through the expansion of economic and social opportunities in well-functioning cities.

As in many other countries in SSA, provision of housing, especially for the urban-poor, is a great challenge. The housing deficit in Ghana has been recorded at over 1.7 million (Daily Graphic Online 2018). Ghana's urban housing predicament is summed up by Yakubu et al. (2014), who have observed that the housing situation in most Ghanaian cities is characterised by supply difficulties, deteriorating conditions, and overcrowding. Their study highlighted the inter-linkages between poor housing conditions, health, and poverty in cities, further arguing that the plights of the urban-poor are exacerbated by a lack of access to public goods and services due to an exclusion created by the limited coverage of the urban-poor under State planning. In a study to determine the affordability of housing in two Ghanaian cities (Kumasi and Tamale), Boamah (2010) highlighted the generally worsening conditions of housing. Furthermore, a study by Konadu-Agyeman (2001) that examined the relationship between population growth and housing conditions in Accra between the 1950s and 1990s showed a general deterioration of housing conditions in the 1990s as compared with the 1950s. A situation he argued, was made worse by a poor maintenance regime, a struggling economy, and poor rent control.

4.2.1.5 Transport

The role of the transport sector in the sustainable functioning of cities has been well-recognised (see Dalkmann and Sakamoto 2011; Birago et al. 2017). Urban transport serves as a vehicle through which many economic and social activities are carried out (Hidalgo and Huizenga 2013). The transport sector is estimated to account for 13% of global GHG emissions, as well as 23% of energy-related GHG emissions (UN-Habitat 2011 as cited in Cevero 2013). According to Dalkmann and Sakamoto (2011), the motorised nature of global transport, which is fundamentally based on fossil fuels, is not sustainable as it has farreaching economic, social, and environmental effects. Subsequently, Hidalgo and Huizenga (2013, p.76) have argued that a paradigm shift is required in order to move the trajectory of urban transport onto a sustainable path by "adopting policies to avoid long and unnecessary motorised travel, to shift the growth trends of individual motor travel to favour non-

motorised and public transport, and to improve technology and operational management of transport activities". Godard (2013) has identified urban sprawl and fuel and energy costs as two critical factors that affect sustainable urban mobility in many African cities. He highlights the tension between urban sprawl and mobility as one that contradicts the criteria for sustainable mobility as it results in longer travel distances and increased energy consumption. He further points out that high energy costs can either coerce users and operators to make more energy-efficient choices or cause potential social tensions, as was the case in Nigeria in 2012 where there were strikes and riots due to fuel price increments.

The main forms of urban transport in Ghana are private vehicular transport, public transport, and other non-motorised transport (such as walking). The country's transport system and infrastructure have been characterised as poor, congested, and inadequate (Kwakye and Fouracre 1998; Amoh-Gyimah and Aidoo 2013). Increasing urbanisation and urban growth have, thus, exacerbated pressures on the already-stressed urban transport systems. For example, car ownership has increased tremendously, which is reflected by the number of newly registered cars (Obeng-Odoom 2009a), and this has profound effects on transport infrastructure and institutions. Although specific urban figures do not exist, about 94,846 cars were newly registered in 2009 compared to 52, 881 in the year 2000 (Ministry of Roads and Highways, Ministry of Transport, and Ghana Statistical Service [MRH, MoT and GSS] 2011). In a study which examined the mode of transport to work by government employees in Kumasi, Amoh-Gyimah and Aidoo (2013) observed a high level of traffic congestion in the metropolis especially during the extended peak periods when workers travel to work. Scholars have also noted the prevalence of road accidents, increase in travel times, and pollution as a result of urban growth and the resulting infrastructural bottlenecks in the urban transport system (Kwakye and Fouracre 1998; Amoh-Gyimah and Aidoo 2013). From the preceding observations, it can be argued that urban transport in Ghana is affected, and reflected by the extent of urbanisation. With urbanisation and urban growth expected to increase (Cobbinah and Niminga-Beka 2017), a key question is whether this fundamentallystressed urban transport system can cope with the demographic shift in a sustainable manner, especially in a less-developed country such as Ghana, which is less endowed with the financial and human resources to deal with the transport problems of urbanisation. Although the country's urban road network tripled from 4064 km in 1985 to 12400 km in 2010 (Environmental Protection Agency [EPA] 2006; Government of Ghana [GOG] 2010), a lack of alternate forms of transport, such as rail transport, coupled with weak transport management (Obeng-Odoom 2009a), means that current transport provision is unlikely to match the evolution of transport needs of Ghana's urban residents. The mutually interactive nature of transport with other parts of the urban system reflects the need to further explore the dynamics between urban development and urban transport, especially where respective policies strongly affect each other.

4.2.2 Selected key factors that influence urban development in Ghana.

4.2.2.1 Politics.

As much as urbanisation and urban sustainable development in Ghana are shaped by factors including migration, natural population growth, land use, market forces, globalisation etc., the urbanisation and sustainable development processes in the country are inherently political. As a result, the interactions between urbanisation and (sustainable) development in Ghana have evolved through key political shifts at the national level which have, in turn, shaped outcomes at the local urban level (Obeng-Odoom 2009a; Fuseini and Kemp 2015). For example, urbanisation and its attendant development in Ghana have been shaped by the agenda of the ruling regime since: (i) the pre-independence era where the interests of the British colonial government determined urban development and; (ii) the post-independence era (after 1957) where urban development has been shaped by mass industrialisation due to the socialist ideologies promoted by Kwame Nkrumah (Ghana's first president), as well as the neo-liberal structural adjustment programmes of the 1980s, etc. Post-independence, a relatively young independent Ghana pursued an extensive State-led industrialisation programme under the leadership of President Kwame Nkrumah. This kick-started a mass rural exodus to urban areas in order to take advantage of new economic opportunities. The adoption of 'growth pole' strategies by various regional authorities in the 1970s led to increased urbanisation in urban centres like Ho, Cape Coast, and Tamale as many people moved from all over the country to take advantage of the ventures established (Yaro et al. 2011). The programmes under the growth pole strategies were essentially economic ones and were not considered as interventions for urban development and urbanisation. Nonetheless, the impact of the programmes under the growth pole strategies on the above-mentioned urban areas cannot be understated as they helped to transform those areas into prominent urban centres with better facilities and infrastructure necessary for development. Growth in urbanisation rates in Ghana from 1980 was spurred by the structural adjustment programmes (SAPs) prescribed by the International Monetary Fund (IMF) to support the country's

economic recovery in the aftermath of the global oil crisis in the 1970s. This period began the mainstreaming of neo-liberal ethos in Ghanaian policy framework which has resulted in increased private sector participation, as well as an extension of the markets and minimised direct State provision (Obeng-Odoom 2012).

Furthermore, the Fourth Republic of Ghana (1992- present) has been dominated by politics between the two most dominant political parties; the National Democratic Congress (NDC) and the New Patriotic Party (NPP). Considering how the politics of the Fourth Republic affect urbanisation and development, a review of Ghana's 2008 Election manifestos showed the convergence, inadequacies, and tensions in the proposals proffered by the political parties towards the management of the country's urban transition (see Obeng-Odoom 2009b). For example, in terms of urban regeneration, there was a convergence in thinking about how the prevailing weak urban planning regime undermined the aesthetic quality of the built environment in Ghana's urban areas. To address the urban planning dilemma in the context of urban regeneration, the NPP proposed to reduce illegal building structures by strictly enforcing urban planning regulations, as well as by providing resources for the acquisition of mapping and photography equipment that would enable the enforcement of building codes (NPP 2008; Obeng-Odoom 2009b). The NDC proposed to tackle illegal building and builders, as well as improve sanitation, by preventing and improving slums (NDC 2008; Obeng-Odoom 2009b). The NDC further proposed to enhance urban regeneration by pursuing the expansion of water infrastructure. Despite the convergence in thinking between the NDC and NPP in the context of urban regeneration which has been demonstrated above, the potential inadequacies in the proposals proffered by both parties are evident. In this context, while the propositions by the NPP and NDC could potentially contribute to urban regeneration in Ghana, it is worth noting that urban regeneration is not limited only to improving the aesthetics of the built environment and providing water infrastructure, but rather encompasses addressing the economic and social well-being issues of urban communities by providing avenues for socio-economic growth and employment (Syms 2002; Gruneburg 2006). Furthermore, both the NPP and NDC proposed to solve urban transport problems in Ghana by promoting mass transit infrastructure. Tensions emerge, however, as the NDC appeared to favour public provision of transport services and at the same time proposed to support private transport operators to be more effective, efficient, and competitive (NDC 2008; Obeng-Odoom 2009b).

4.2.2.2 Land tenure.

Land plays a strategic, yet complex, role in development. In simple terms, land tenure describes the set of procedures that determine how access is granted for the use, control, and transfer of land in a jurisdiction. According to Owoo and Boakye-Yiadom (2015), the livelihoods of many people in less-developed countries fundamentally depend on access to land. This assertion especially holds in places such as Africa where many people engage in subsistence farming, thereby making land tenure security crucial for meeting the development needs of people (Lemmen 2010). For the poor, a lack of access to land resources potentially represents an existential threat as the capacity of poor people to participate in, and exploit social, economic, and civic opportunities tends to be undermined. In the context of urbanisation, land tenure and access are critical, as population growth due to urbanisation creates demand for more land to accommodate expanding urban functions. In some cases, demand for land tends to outpace supply within a specific urban boundary, thereby affecting surrounding areas.

In Ghana, where customary land tenure predominates, traditional leaders assume ownership of land for and on behalf of communities (Aubynn 2006; Kalt 2019). In specific terms, about 80% of land is held under customary ownership, while the rest is primarily owned by the State (Aubynn 2006; Tsikata and Yaro 2011). With the permission of the owner, land in Ghana may be rented or leased over a period of time (e.g., months, years) for various purposes including agriculture and housing (Nyame and Blocher 2010). This means that outright ownership of land in Ghana is limited. Therefore, the land and development needs of most Ghanaians are, sometimes, schematically influenced by the interests of land owners and other stakeholders such as the government, multi-national companies (MNCs), etc. In Ghana's mining regions, Nyame and Blocher (2010) suggest that with the decreasing value of agricultural land-use due to factors such as unpredictable income, climate stresses, etc., landowners appear to perceive mining leases as more economically-viable options. As a result, the livelihoods of some citizens (especially farmers) in the mining areas of Ghana are possibly threatened as land owners gravitate towards leases for small-scale mining activities. Even in areas where large-scale mining activities conducted by MNCs exist, the land-use interests of general citizenry tend to be subservient to those of MNCs whose interests are usually protected by the central government. Therefore, in most cases, citizens in communities are rendered landless, as land is annexed for large-scale mining operations through the application of legal instruments, because of the mutual economic benefits both

the MNCs (e.g., profits) and the government (e.g., taxes and royalties) are likely to accrue. For example, in the development of Newmont Ghana's Ahafo gold mine, over 9,000 people (mostly farmers) were displaced from their lands (Jnr et al. 2016). While compensation is sometimes paid to communities, and MNCs usually undertake corporate social responsibility (CSR) projects to maintain their social licence to operate, these might be insignificant when the livelihoods and long-term development needs of communities are considered.

4.2.2.3 Spatial and urban planning.

Historically, the extractive and agricultural sectors provided the basis for the uptake of spatial planning practice in Ghana as mining and cocoa investments created the need for Ghana's British colonial masters in the pre-independence era to make spatial planning provisions. As a result, Guggisberg's ten-year development plan (1920-1930) was launched in 1920 as the pioneering development plan that attempted to ensure orderly, safe, healthy, and socio-economically prosperous settlements in pre-independent Ghana (Leith 1974; Fuseini and Kemp 2015). Guggisberg's plan provided the foundation for the emergence of Ghana's first comprehensive spatial planning framework referred to as the Town and Country Planning Ordinance of 1945 (i.e CAP 84). The CAP 84 was Ghana's version of Britain's post-war restructuring of the country's planning provisions that had been extended to its colonies. Nevertheless, the geographical extent of planning in the pre-independence era was limited as planning provision mainly occurred within the boundaries of the 'Golden Triangle' (the vertices of the 'Golden Triangle' consisted of Kumasi, Accra, and Takoradi) where the bulk of exploitable resources that were of interest to the colonialists were confined (Adarkwah 2012).

The immediate post-independence planning regime aligned with the mass industrialisation strategy of that era and expanded planning beyond the 'Golden Triangle' to other parts of Ghana (The New York Times 1964). As a result, planning education was promoted, new planned townships were established, and institutional capacity was strengthened by establishing planning departments across the country (Inkoom 2009; Adarkwah 2012). However, it is worth noting that the three decades that followed Ghana's independence are known to have contributed to the urban challenges prevalent in Ghana today as planning capacity was overwhelmed by the urban population explosion (Fuseini and Kemp 2015). The assertion above is partly because the numerous coup d'états between 1966 (overthrow of first

post-independent government) and 1992 (when Ghana return to multi-party democracy) led to poor macro-economic performance and political instability. That situation undermined the stable politico-economic environment needed for planning to flourish (Fuseini and Kemp 2015). Following the liberalisation of Ghana's economy through the uptake of the IMF's SAPs which were designed to increase private sector investment and participation, as well as to promote decentralisation of governance (Obeng-Odoom, 2012; 2013b), the country's 1992 constitution provided the complementary legal and institutional framework for urban sustainable development in Ghana. Key legislations relevant to urban development were subsequently passed (see Fuseini and Kemp 2015) and these included: (i) the Local Government Act (Act 462 of 1993); (ii) the National Development Planning Commission (NDPC) Act (Act 480 of 1994) and; (iii) the National Development Planning System (NDPS) Act (Act 490 of 1994). In the years that followed, other key legislation and policies were introduced as Ghana became increasingly urban, including the National Urban Policy in 2012 (NUP 2012). In general, the legislations and policy highlighted above are among the key ones designed to enable planning institutions to respond to the opportunities and challenges of decentralisation, liberalisation, and rising urbanisation in the last three decades.

Critically, despite the various measures, including the legislations and policies highlighted above, that have been implemented to improve urban governance and management, urban planning in present day Ghana is dominated by a top-down decision-making process just as it was in the pre-independence era where planning in Ghana was top-down and Eurocentric (Adarkwa 2012), compared to the strong bottom-up that is common in developed nations. As a result, the most important decision-making on urban planning is controlled by central government agencies, with marginal inputs from authorities at the local urban level. The above scenario is explicitly captured by Obeng-Odoom (2009a, p.51) who wrote that: *"Tetteh's analysis (p. 22–30) of the 'role of Architecture and Planning' in the Future of Our Cities was very insightful. He argues that central government takes the responsibility of planning the cities while city authorities are tasked with controlling the development they did not plan for".*

4.2.2.4 How politics and land tenure affect urban planning and development.

It could be argued that politics contributes significantly to how planning processes and institutions have been undermined in Ghana. For example, as will be highlighted later in

Chapter Six with regards to the specific case of Ghana's National Urban Policy, a change in government could potentially undermine how the policy is implemented, and whether the contents of the policy align with the political ideologies of a new government. Furthermore, although legislations such as the Local Government Act (Act 462 of 1993) have been designed to enhance effective decentralisation by establishing Metropolitan, Municipal, and District Assemblies (MMDAs) which have oversight control of planning processes in respective jurisdictions, and whereas the NDPC ACT (ACT 480 of 1994) mandates the NDPC to ensure that local-level planning aligns with the broader national development plans and policies, local-level urban planning is effectively compromised through political patronage (Owusu 2004). The political patronage of local urban planning is attributable to the situation where the ruling government routinely appoints the leadership of both the MMDAs and the NDPC. The implications of political patronage at the local urban level in Ghana has been succinctly captured in published literature that surveyed the views of planners as follows: "the politician is only motivated by the desire to win or retain power which is a matter of vote. A nicely planned neighbourhood does not vote but people who live in unapproved developments do. There is no chance for any [planning] proposal succeeding if our political leaders think it has the potential to affect their electoral fortunes" (Yeboah and Obeng-Odoom 2010, p.87). Therefore, it is evident that eradicating undue political interference in urban planning processes in Ghana is critical for urban sustainable development.

Furthermore, published research alludes to how land tenure (discussed earlier), especially customary ownership, influences land use and urban planning decisions, and by extension, urbanisation and urban development in Ghana (Boamah et al. 2012; Ahmed and Dinye 2012; Siiba et al. 2018). Essentially, Siiba et al. (2018) argue that the land tenure system in Ghana has made it possible for traditional leaders, who are the key persons in customary ownership, to usurp formal planning institutions in the areas of land use determination and management. This is because while national institutions provide plans and policies to guide urban development, it is the traditional leaders who lease and allocate land within the respective jurisdictions. Where this dominance by traditional leaders (in terms of land allocation) is unregulated, as is mostly the case, they tend to assume the role of planning institutions, and, together with their surveyors, make land use decisions that may not align with the core values of urban planning. It is, however, interesting to note that the suggested interferences on land use and urban planning-related issues, mostly by customary land owners, are sometimes

embraced by a wider section of the Ghanaian society due to reported corruption and bureaucracies encountered when engaging formal planning institutions. This is not surprising as the literature (for example, Boamah et al. 2012; Ahmed and Dinye 2012), shows that defiance of Ghana's statutory land use plans instead of adherence by developers is common. This does not suggest that customary land ownership is the cause of the urbanisation and land use challenges in Ghana, but it demonstrates the key role it plays in (un)sustainable urban development in Ghana. In all, it is evident that land tenure and access, perceived or factual, are critical in the context of SD and urbanisation, especially in cases where people need to sustain their livelihoods or establish themselves as citizens in urban areas.

4.3 Sustainable development in Ghana

4.3.1 An overview of progress towards sustainable development in Ghana.

According to UNECA (2012a; 2012b), achieving SD requires a broad-based quality economic growth for which the prerequisite fundamentals include, but are not limited to, poverty and inequality reduction, as well as sound socio-economic infrastructure. Like most developing SSA countries, Ghana has taken important steps towards achieving SD. A plethora of developmental plans, policies, and initiatives have been formulated and implemented at national, regional, and local government levels in order to help guide the SD process and address its challenges (Alliance for Development 2014) which include:

- (a) Ghana Shared Growth and Development Agenda.
- (b) Sustainable Development Action Plan.
- (c) Growth and Poverty Reduction Strategy, I and II.
- (d) National Social Protection Strategy.
- (e) National Environmental Sanitation Strategy and Action Plan.
- (f) National Policy Framework on Climate Change.
- (g) Ghana Education Strategic Plan (2003 2015; 2010 2020).
- (h) Food and Agricultural Sector Development Plan, I and II.
- (i) Medium Term Agricultural Investment Plan.
- (j) Forestry Investment Plan.

For example, the government implemented a two-phased Ghana Poverty Reduction Strategy (GPRS I and II) aimed at reducing absolute poverty. A UNICEF-sponsored report in 2016

showed that the country's level of poverty fell from 56.5% to 24.2% between 1992 and 2013, helping it to achieve MDG target 1 of halving the proportion of the population living in absolute poverty (Cooke et al. 2016). However, by 2006, the annual average rate of poverty reduction had slowed by 1.1 percentage points, down from 1.8 percentage points in the 1990s. This shows that despite the successes in poverty reduction, the level of poverty is still significant; a situation Domfeh et al. (2012) argue challenges the country's SD progress as it affects nutrition, housing conditions, access to health, and education, etc. The National Environmental Sanitation Strategy and Action Plan (NESSAP) provided the policy framework for Ghana's environmental sanitation and helped provide access to improved water sources for the country's citizens. Nevertheless, an estimated 4,000 children still die annually from diarrhoea, and 23% of children in the country are affected by malnutrition linked to water and sanitation (UNICEF 2016). Another important area where Ghana has sought to promote SD has been through education. The role of education in promoting sustainable development is expressly captured by the various United Nations initiatives such as Education for Sustainable Development (ESD) and the Decade of Education for Sustainable Development (DESD) which sought to promote education and learning as fundamental for a more sustainable world (Pigozzi 2007). Some of the significant policies include the Free Compulsory Universal Basic Education (FCUBE) policy introduced in 1995 (Acheampong 2009) and the Free Senior High School Policy introduced in 2017 (Arthur-Mensah and Alagaraja 2018). These policies stress the relevance of education to the SD journey of Ghana and recognise how the unique challenge of poverty undermines the achievement of universal education as, for example, most parents were unable to afford the cost of education (Alderman and Brundy 2011). In demonstrating the importance of education to augment the production possibilities of citizens in driving social and economic development, Sen (1999, p.295) has argued that "a person may benefit from education through reading, communicating, arguing, in being able to choose in a more informed way, in being taken more seriously by others and so on. The benefits of education, thus exceed its role as human capital".

Despite the efforts of the national government in promoting SD, the private sector plays a key role in Ghana. Springett (2013) describes the private sector (in this case, Multi-National Corporations [MNCs]) as an influential player that employs a market-based approach to control and shape the meaning of SD, and Chapter 30 of Agenda 21 outlined the roles of the private sector in realising SD (Van Alstine and Barkemeyer 2014). This paved the way for
the private sector to be seen as key stakeholders in finding solutions for developmental problems in developing countries rather than as opponents of development (Fox 2004). It also marked a shift in perceptions held about MNCs in the 1970s and 1980s when their agendas were seen to be in direct conflict with the developmental goals of developing countries (Hilson 2012). Using the case of MNCs in Ghana's mining industry as an example, they have implemented CSR programmes (in addition to paying royalties and taxes) aimed at contributing to the sustainable development of the communities in which they operate. Accordingly, IIED (2002, cited in Kemp 2009, p.202) has suggested that "mining operations are sometimes the only viable option that remote communities have for social development". However, the role played by MNCs as partners in the quest for SD is one that could put the country's sustainable growth in the progeny of the free markets. This is because of the potential risks of MNCs to arrogate the environmental aspects of SD in a way that promotes economic growth and maximises profits. As Hilson (2012, p. 133) maintains, "few industrial activities have as large an environmental footprint and are capable of wielding as much influence on the well-being of a society as a large-scale mine or oil and gas project". While there is some evidence on Ghana's performance (for example, annual UN and national MDG reports) towards realising some of the SD-centred international agreements and goals (see, for example, UNDP 2015), empirical data on how well the country has performed are scarce. One could therefore question the extent to which both public and private efforts have been able to reconcile the country's economic, social and environmental ambitions in the context of SD.

4.3.2 An assessment of sustainable development in Ghana.

4.3.2.1 Human Development Index (HDI) for Ghana.

In Figure 4.1, Ghana's HDI score, which is a comprehensive measure of human well-being that incorporates key socio-economic dimensions of education, life expectancy, and income (Hicks 1997; Hou et al. 2015), is presented based on available data for the period 1990–2018. It is seen from Figure 4.1 that Ghana's HDI values increased from 0.454 in 1990 to 0.596 in 2018. This shows that overall, there has been an improvement in the level of development (as measured by the HDI). Table 4.1 also shows the disaggregated values for the components of Ghana's HDI for the period 1990–2018. From Table 4.1, it is seen that for an HDI value of 0.454 in 1990, life expectancy at birth was 56.8 years, expected years of schooling was 7.6 years, mean years of schooling was 4.9 years, and per capita gross national income was

U.S.\$. 1,897. By 2018, the respective components had improved significantly as for an HDI value of 0.592; life expectancy was 63.8 years, expected years of schooling was 11.5 years, mean years of schooling was 7.2 years, and GNI per capita was U.S.\$. 4,099.



Figure 4.1: Human Development Index for Ghana, 1990-2018.

Source: UNDP (2018; 2019b).

 Table 4.1: Component values of Ghana's HDI, 1990-2018.

Year	Life	Expected	Mean years	GNI per	HDI value
	expectancy	years of	of schooling	capita (2011	
	at birth	schooling		PP\$)	
1990	56.8	7.6	4.9	1,863	0.454
1995	57.5	7.7	5.7	1,992	0.472
2000	57.0	8.0	6.1	2,152	0.483
2005	58.7	8.7	6.4	2,475	0.508
2010	61.0	10.9	6.7	2,977	0.554
2015	62.8	11.7	6.9	3,735	0.585
2016	63.1	11.6	7.1	3,756	0.587
2017	63.5	11.5	7.1	3,943	0.591
2018	63.8	11.5	7.2	4,099	0.596

4.3.2.2 Inequality-Adjusted HDI (IHDI) for Ghana.

The inequality-adjusted HDI for Ghana for the year 2018 is presented in Table 4.2. The IDHI compensates for the inequality in the distribution of the components of the HDI by discounting the average value of each component based on its level of inequality. From Table 4.2, it is seen that when Ghana's HDI of 0.596 in 2018 (Table 4.1) is adjusted for inequality, it falls to 0.427. Specifically, inequality in the distribution of the HDI indices which includes inequality in life expectancy at birth (24.2%); inequality in education (34.9%); and inequality in income (25.3%) led to an overall loss of 28.3% in HDI value.

Table 4.2: Ghana's IDHI for 2018.

IDHI	Overall loss	Human	Inequality in	Inequality in	Inequality
value	(%)	Inequality	life expectancy	education	in income
		coefficient	at birth (%)	(%)	(%)
0.427	28.3	28.1	24.2	34.9	25.3

Source: UNDP (2019b).

4.3.2.3 Ecological Footprint for Ghana.

In Figure 4.2, Ghana's ecological footprint score, which is a measure of the quantity of ecologically productive areas needed for the maintenance of human natural resource consumption and the assimilation of human waste, is presented based on available data for the period 1965–2016. It can be seen that the country's ecological footprint increased from 1.1 gha per capita in 1965 to 2.0 gha per capita in 2016. At the same time, its bio-capacity decreased from 2.60 gha per capita in 1965 to 1.3 gha per capita in 2016. It can be seen that in 1995, Ghana's ecological footprint equalled its bio-capacity, and by 2016, Ghana had a bio-capacity deficit of -0.7. Therefore, if one adheres to the threshold set by WCED (1987), then the ecological environment in Ghana, as used for the production of natural resources and assimilation of waste, appears to be on an unsustainable trajectory.



Figure 4.2: Ecological Footprint and Bio-capacity statistics for Ghana, 1965-2016. Source: Global Footprint Network (2019).

4.3.2.4 Gini coefficient scores for Ghana.

In Figure 4.3, the Gini coefficient score, which measures the distribution of income, is presented for Ghana based on available data for the period 1987-2016. The values used to measure inequality based on the Gini coefficient are between 0 and 100, where 0 represents total equality. This means that the higher the value of the Gini Coefficient, the greater the level of inequality. Therefore, as Figure 4.3 shows, the inequality in Ghana has increased drastically from 35.3% in 1987 to 43.5% in 2016.



Figure 4.3: Ghana's Gini coefficient values, 1987-2016. Source: World Bank (2019a).

4.3.2.5 Proportion of renewable energy in total primary energy in Ghana.

In Figure 4.4, the proportion of renewable energy (R.E.) in total primary energy production in Ghana based on available data is presented for the period 2005-2017. From this, it is seen that the proportion of R.E. has decreased gradually from 41.7% in 2005 to 29.4% in 2017. Published data, however, shows that reduction in R.E is due to the exponential increase in the use of non-renewable energy such as oil (fossil fuel), rather than a drastic decrease in R.E. sources such as hydro-energy (Energy Commission 2016; 2018; 2019).



Figure 4.4: Proportion of renewable energy in total primary energy in Ghana, 2005-2017.

Source: Energy Commission (2016; 2018; 2019).

4.3.2.6 Sustainable Development Space (SDS): ecological sustainability and basic needs.

Figure 4.5 shows the graphical representations of the relationship between ecological footprint and HDI in Ghana. For parsimony^{11,} only fifteen data points for the years between 1990 and 2016 were plotted. The SDS for Figure 4.5 is the lower right quadrant where development would be deemed as sustainable at a maximum EF of 1.7 gha and a minimum HDI of 0.63. From Figure 4.5, the observed pattern shows that none of the data points plotted fell into the SDS space, as ecological footprint increased alongside an increasing HDI.



Figure 4.5: The sustainable development space (SDS) for ecological footprint and human development in Ghana.

¹¹ Note: Data on an aspect such as ecological footprint starts from 1965, while that of HDI starts from 1990. Therefore, this was done in order to ensure matching dates for the data on the variables used to plot the SDS graphs.

4.3.2.7 Sustainable Development Space: ecological sustainability and intra-generational equity.

Figure 4.6 shows the graphical representations of the relationship between ecological footprint and the Gini coefficient for Ghana. For parsimony, only six data points for the years 1988, 1992, 1998, 2005, 2012, and 2016 were plotted. The SDS for Figure 4.6 is the lower left quadrant where development would be deemed as sustainable at a maximum EF of 1.7 gha and a maximum Gini Coefficient of 40. From the SDS space shown in Figure 4.6, it was only in 1988 (36) and 1992 (38.4) when inequality was low that development, in the context of ecological sustainability and intra-generational equity, could be deemed as sustainable.



Figure 4.6: The sustainable development space (SDS) for ecological footprint and Gini coefficient.

4.3.2.8 Sustainable Development Space: ecological sustainability and inter-generational equity.

Figure 4.7 shows the graphical representations of the relationship between ecological footprint and renewable proportion of total primary energy. For parsimony, only twelve data points for the years between 2005 and 2016 were plotted. The SDS for Figure 4.7 is the lower right quadrant where development would be deemed as sustainable at a maximum EF of 1.7

gha and a minimum R.E proportion of 27%. From the SDS space shown in Figure 4.7, it was only in 2005 when E.F was 1.7 gha that development, in the context of ecological sustainability and inter-generational equity, could be deemed as sustainable.



Figure 4.7: The sustainable development space (SDS) for ecological footprint and renewable energy.

4.3.2.9 Sustainable Development Space: basic needs and intra-generational equity.

Figure 4.8 shows the graphical representations of the relationship between HDI and Gini Coefficient in Ghana. For parsimony, only five data points for years between 1990 and 2016 were plotted. The SDS for Figure 4.8 is the upper left quadrant where development would be deemed as sustainable at a minimum HDI of 0.63 and a maximum Gini Coefficient of 40. None of the years plotted fell into the SDS in the upper left quadrant.



Figure 4.8: The sustainable development space (SDS) for Human Development Index and Gini coefficient.

4.3.2.10 Sustainable Development Space: basic needs and inter-generational equity.

Figure 4.9 shows the graphical representations of the relationship between HDI and renewable proportion of total primary energy. For parsimony, only seven data points for years between 2005 to 2016 were plotted. The SDS for Figure 4.9 is lower right quadrant where development would be deemed as sustainable at a minimum HDI of 0.63 and a minimum R.E proportion of 27%. None of the years plotted fell within the SDS in the context of basic needs and inter-generational equity.



Figure 4.9: The sustainable development space (SDS) for Human Development Index and renewable energy.

4.3.2.11 Sustainable Development Space: inter-generational equity and intra-generational equity.

Figure 4.10 shows the graphical representations of the relationship between the Gini coefficient and renewable proportion of total primary energy. For parsimony, only three data points for years between 2005 and 2016 were plotted. The SDS for Figure 4.10 is the upper left quadrant where development would be deemed as sustainable at a maximum Gini Coefficient of 40 and a minimum R.E. proportion of 27%. None of the years plotted fell into the SDS in the context of inter- and intra-generational equity.



Figure 4.10: The sustainable development space (SDS) for Gini coefficient and renewable energy.

4.3.3 Discussion.

4.3.3.1 Sustainable Development Space: ecological sustainability and basic needs.

The consistent increase in Ghana's per capita ecological footprint (as shown in Figure 4.2) could be directly related to the relatively improved levels of socio-economic well-being in Ghana (as measured by the HDI in Figure 4.1). Here, it can be argued that the strive to provide the basic needs of an increasing population places greater demands for ecological resource exploitation, leading to an increased ecological footprint (Holden et al. 2014). A potential explanation for this observation is the scenario that, in an increasingly modernising world, the application of new technology has led to better ways of resource exploitation. In the Ghanaian case, this can be explained by the shift from subsistence farming to mechanised farming. Here, the application of technology potentially leads to diverse outcomes. For example, leveraging technology for resource exploitation could lead to increased efficiency and productivity such that wastes are reduced. Alternatively, maximising full technological benefits for resource exploitation and compensating for the capital-intensive nature of such technologies could lead to the exploitation of more resources and increased ecological footprints. For example, more land will be needed for commercial farming in this scenario.

Thus, with an increasing population and the need to improve socio-economic well-being, technology has potentially been leveraged to maximise resource exploitation.

The interface of ecological sustainability and basic needs is premised on the notion that there is no trade-off between environmental development and socio-economic development in terms of achieving basic needs such as income and education, but rather decisions on environmental development are fundamentally underpinned by the need to provide the basic needs of society within the biophysical limits of the environment. Humans, as an important part of socio-ecological systems, are the bridge between the ecological systems and the socioeconomic systems (see for example, Costanza et al. 2007). This is because it is humans who determine the resources they require from the ecological environment to maintain their subsistence. Thus, in this case, the quality, quantity, and exploitation levels of ecological resources will directly or indirectly affect the ability of humans to meet their socio-economic needs (basic needs in this case). The pattern observed in Figure 4.5 shows that Ghana's ecological footprint per capita increased alongside increases in HDI, resulting in none of the years plotted falling within the SDS. This is consistent with the findings of Holden et al. (2014), whose study on 167 countries showed increases in ecological footprint when HDI increased. Figure 4.2 has shown that Ghana is increasingly dependent on and potentially over-exploiting its ecological resources because while its ecological footprint per capita increased, its bio- capacity decreased within the period measured. A significant inference made from Figure 4.2 is that, theoretically, Ghana had a bio-capacity deficit of -0.7 ha per capita in 2016 compared to a bio-capacity reserve of 1.5 ha per capita in 1965. Thus, with rising concerns about the limits of the biophysical environment, there is an imminent risk that the viability of the ecological environment to help provide the basic needs of society will diminish, should the current regime and practices continue.

4.3.3.2 Sustainable Development Space: ecological sustainability and intra-generational equity.

The interface of ecological sustainability and intra-generational equity is premised on the notion that inequalities in the current generation will compromise ecological sustainability. From the literature (see Section 2.3.1), this has been succinctly captured by the Brundtland Commission (WCED 1987, p.72) as they stated that *"those who are poor and hungry will often destroy their immediate environment in order to survive: they will cut down forests;*

their livestock will overgraze grasslands; they will overuse marginal land; and in growing numbers they will crowd into congested cities. The cumulative effect of these changes is so far-reaching as to make poverty itself a major global scourge". Consistent with the widening inequality (Figure 4.3) and increasing ecological footprint in Ghana (Figure 4.2), a positive relationship between ecological footprint and Gini coefficient was observed for the data points plotted in Figure 4.6 and none of the data points fell within the SDS. The logical implication of this finding is that pervasive inequalities, which are exacerbated by high levels of poverty in Ghana, mean that the country (including people) is compromised in its ability to make decisions that could potentially safeguard ecological sustainability. Unlike Holden et al. (2014) who could not establish a relationship between ecological sustainability and intragenerational equity in their study, the arguments based on Figure 4.6 are consistent with the findings of Wilkinson and Pickett (2009) who argue that communities with low levels of inequality perform better when assessed on a wide range of positive features including life expectancy when compared to communities with high inequalities. In this case, widening inequality in Ghana means that Ghana's scores on environmental quality will be lower, as evidenced in increasing ecological footprint. The implications of this scenario can be demonstrated with the analogy of how poverty perpetrates wide inequalities and threatens ecological sustainability, as well as the very survival of citizens. In this instance, widening inequality in Ghana means that the poor will potentially attempt to survive by exploiting and overexploiting ecological resources at their disposal. For example, the poor could attempt to harvest ecological resources such as forest products to sell for income. This leads to deforestation, which has attendant consequences, including flooding, loss of species, carbon release, etc. Thus, the exploits of the poor for survival could create a range of pernicious consequences which threaten ecological sustainability, for which the poor are often the primary victims (see, for example, Okyere et al. 2012). From the findings, it can be argued, in the context of the relationship between ecological sustainability and intra-generational equity in Ghana, that inequality compromises ecological sustainability as it potentially reduces the urgency and public demand for the protection and exploitation of ecological resources within acceptable biophysical limits. This is because inequality potentially leads to increasing dependence by the poor on these resources for their livelihoods, while it reinforces the capacity and powers of the rich and market actors (for example, MNCs) to prevent such exploitation or protection.

4.3.3.3 Sustainable Development Space: ecological sustainability and inter-generational equity.

A central question that emerges from the findings is whether the present Ghanaian generation has an underlying duty to conserve and protect the environment for its future generation. Beyond the reality that current patterns of exploitation of ecological resources (as measured by the ecological footprint) are at the highest levels ever (Figure 4.2), by signing on to the various international charters on sustainable development (for example, the Rio Declaration on Environment and Development [1992]), Ghana establishes an indirect fiduciary relationship with the future generation with respect to ecological sustainability. In this context, Ghana has a passive obligation under these international instruments to pursue development in a way that does not compromise the ability of the future generation to do the same (WCED 1987). The evidence from Figure 4.7 implies that although the renewable energy remained above the 27% minimum threshold recommended, development was still deemed as unsustainable due to increasing E.F. This situation contrasts with the findings of Holden et al. (2014) who found that countries with a higher proportion of renewable energy tend to have lower ecological footprints. In a practical sense, it can be inferred from the evidence that although Ghanaian stakeholders (including government and private sector) might offer some statements in their development visions or in corporate social responsibility statements to underline their commitment to long-term and future-focused ecological sustainability, the development priorities of government and investment interests of private sector stakeholders mean that few of these statements are translated into tangible decisions and actions. It is true that in providing for the basic needs of its citizens, Ghana relies directly or indirectly on the extraction of its ecological resources and the country also relies on these same ecological resources to assimilate waste. It is also not a straight-forward process to determine the quantum savings on ecological resources that the present generation must make for the future generation. Nevertheless, a rising ecological footprint, made worse by a decreasing share of the renewable energy proportion of the country's total primary energy supply, may be an indication that the ecological sustainability ambitions of the present generation are incongruous with the ecological sustainability prospects of the future generation. In this case, one can argue that Ghana's sustainable development journey should be about more than just environmental development or protection, but rather, it should have a temporal focus or threshold. This is important in order to ensure that the current generation lives within the biophysical limits of the environment without compromising the ability of the future generation to do the same. Overall, the findings are an indication that the ecological

sustainability of Ghana's future generation is dependent on the altruism extended by the current generation, and for a less-developed country like Ghana, the ecological sustainability of the current generation depends on the extent to which they exploit ecological resources.

4.3.3.4 Sustainable Development Space: basic needs and intra-generational equity.

The interface between basic needs and intra-generational equity in this study is premised on the notion that a fair opportunity exists for all members of society to meet their basic needs and improve their levels of socio-economic well-being (as measured by the HDI). The consistent increase in Ghana's HDI (as shown in Figure 4.1), implies that there have been significant improvements in the socio-economic well-being of Ghanaians. At the same time, inequality has widened with a consistent increase in Ghana's Gini coefficient score (as shown in Figure 4.3). Subsequently, evidence from Figure 4.8 shows that while HDI levels have improved, Gini coefficient scores have risen and thus, Ghana has exceeded the maximum Gini coefficient level of 40 and has also not reached the minimum HDI level of 0.63 in order to fall into the SDS in the context of 'basic needs and intra-generational equity'. In other words, while general socio-economic well-being has improved, inequalities have increased in Ghana and compromised the ability to safeguard the basic needs of all citizens (Holden et al. 2014). This finding confirms those from previous studies (Holden et al. 2014; Wilkinson and Pickett 2009) which suggest that the possibility of safeguarding the basic needs of all citizens is higher in countries with low inequalities than it is in countries with higher inequalities. The findings (Figure 4.1; Figure 4.3; Table 4.2) imply that the well-being gains made by Ghana have not been transposed into greater and fairer levels of welfare for all members of society. This occurrence is elucidated by the inequality-adjusted HDI (IHDI) values for Ghana in 2017 (as shown in Table 4.2). In 2018, an HDI value of 0.596 was recorded for Ghana (see Table 4.1). However, as shown in Table 4.2, discounting the HDI value for inequality in the distribution of the components of the HDI results in a 28.3% loss in HDI value to an IDHI of 0.427. This finding is an indication of the state of well-being of a sizeable number of Ghanaians. This is because inequalities in the context of fulfilling the basic needs of the present generation mean that access to public and personal goods such as health, education, and income are affected. This has potentially damaging effects on the opportunities that exist for citizens to fulfil their basic needs, as the higher inequality levels could affect the social mobility of citizens. Thus, the findings correspond with that of Wilkinson and Pickett (2009) who highlighted how increasing inequality can cause health problems (see Section 4.2). The

findings also give essence to the picture portrayed in Section 4.2 (see, for example, Okyere et al. 2012; OECD 2013) where the urban poor in Ghana bear a disproportionate burden of the impact of socio-economic inequalities. The scenario extended here is that, when it comes to fulfilling basic needs, as shaped by the HDI, the poor in Ghana are potentially likely to attain low education, low incomes, and worse health outcomes, compared to the rich who are likely to be well educated and healthier. To sum up, it is suggested from the interpretation of findings on SDS in the context of 'basic needs and intra-generational equity' that access to basic needs in Ghana is socially defined (Wilkinson and Pickett 2009). While this assertion does not assume an institutionalised form of limitation on citizens, it suggests that the specification of the level of access of citizens to fulfil basic needs is influenced by the social organisation created by wide inequalities (as portrayed by Ghana's Gini coefficient scores over the years). Essentially, addressing the problem of disparities in intra-generational access to basic needs in Ghana will require efforts to mitigate the stratification created by wealth inequality. It is in cognisance of this reality that policies such as Free Compulsory Universal Basic Education, which was formulated and implemented in 1995 with the aim of achieving universal education by 2005 (Acheampong 2009), were introduced to promote sustainable development (see Section 4.3.1). While discussions on progress with respect to whether targets were reached, especially in relation to benefits for the poorest households, are beyond the scope of the current discussion, the exigency to address intra-generational inequalities in Ghana (in terms of access to basic needs like education), is reflected in the implementation of a Free Senior High School Policy by the government in power in 2017 (Arthur-Mensah and Alagaraja 2018).

4.3.3.5 Sustainable Development Space: basic needs and inter-generational equity.

The interface between basic needs and inter-generational equity in this study is premised on the notion that a fair opportunity exists for improved socio-economic well-being for the future generation if the current generation can adequately meet their basic needs (as measured by the HDI). The analysis from Figure 4.9 shows that although renewable energy proportions have remained above the minimum threshold of 27%, Ghana's HDI level, despite the improvements, has not reached the minimum threshold of 0.63 beyond which Ghana will fall into the SDS in the context of 'basic needs and inter-generational equity'. An important observation from the data (Figure 4.1; Figure 4.4; and Figure 4.9) was that as HDI levels improved, renewable energy proportion decreased. This is consistent with findings from Holden et al. (2014) who concluded from their research that countries creating a high HDI expend high amounts of energy, making it likely that their renewable energy sources will deplete at fast rates. In a practical sense, the links between providing basic needs and intergenerational equity in Ghana can be expressed in different ways. For example, as the current generation puts in place socio-economic structures to provide their basic needs, these same structures are expected to affect the future generation. In this case, better education of the current generation means that they can create the necessary foundations for the education of the future generation. Similarly, adequate healthcare for the current generation is an indication of beneficial health investments which could accrue benefits for the future generation.

4.3.3.6 Sustainable Development Space: inter-generational equity and intra-generational equity.

It is assumed that Ghana's sustainable development expectations involve providing opportunities for the desirable well-being of the present generation as well as the future ones. Accordingly, the analysis from Figure 4.9 shows that while renewable energy proportions (Figure 4.4) have remained above the minimum threshold of 27% for all the data points plotted, Gini coefficient (Figure 4.3) has increased beyond the maximum threshold to fall into the SDS in the context of 'intra-generational equity and inter-generational equity'. This is consistent with the findings from Holden et al. (2014) whose research suggested a weak but positive correlation between Gini coefficient and renewable energy proportion of total primary energy. From the considerations above, the complexity of the inter-linkages between inter-generational equity and intra-generational equity is clear, but of considerable importance. Exploring Ghana's sustainable development from an intra-generational vs intergenerational perspective has given a broader approach that accounts for inequalities in the distribution of socio-economic progress and ecological resources within generations and between generations. The findings subsequently indicate an inherent conflict between the distribution of resources for the current generation and opportunities for the future generations in Ghana to enjoy the same. The implication of the findings is that if Ghana continues as a country without an aversion for social, environmental, and economic inequalities, the depressive effect on the overall well-being of its citizens will persist within and among generations. Therefore, based on the evidence presented, in order for Ghana to

develop sustainably, it must offer the same opportunities for a better life to its future generations as it does for the present generation.

4.3.4 The state of sustainable development in Ghana.

It is a long journey to sustainable development. The graphs produced to capture the sustainable development space (Figure 4.5; Figure 4.6; Figure 4.7; Figure 4.8; Figure 4.9 and Figure 4.10) have helped to determine Ghana's sustainable development trajectory within the boundaries of the dimensions analysed. Based on the evidence gathered and subsequently discussed, it is clear that Ghana falls short with respect to meeting the threshold conditions for sustainable development. In Ghana's journey to sustainable development, the evidence presented in Table 4.1 and Figure 4.1 for HDI shows that Ghana is increasingly making socio-economic progress in a bid to safeguard the basic needs of its citizens. At the same time, from the evidence presented in Figure 4.2 (Ecological Footprint), Figure 4.3 (Gini Coefficient) and Figure 4.4 (renewable energy proportion), it is seen that Ghana's ecological footprint has increased and exceeded its bio-capacity, inequality has risen, and from an intergenerational perspective, its renewable energy proportion of primary energy has been decreasing. This has potential consequences of eroding the earlier progress by increasing environmental costs and weakening social connections, as well as undermining other poverty reduction efforts. Essentially it prompts questions about how far Ghana is willing to go in order to safeguard the basic needs of its citizens.

Sustainable development assumes an egalitarian principle which gives equal opportunities to all members of society (WCED 1987). Thus, in the context of intra-generational equity, it can be inferred from the evidence presented that Ghana has retrogressed in translating socioeconomic growth into approaches that deal with inequalities, as shown by the steady rise in Gini Coefficient over the years. This suggests that intra-generational equity is exacerbated by government policies and programmes, or shaped by systemic factors, of which poverty is prominent (see Section 4.2). In the context of inter-generational equity, the perspective of this study, based on the evidence presented, is that the uptake and enforceability of the fiduciary duty of ensuring inter-generational equity are questionable. Despite stakeholders' commitment to upholding the rights of future generations to equal opportunities in life by consenting and ratifying various international charters on SD, inter-generational equity is either mere rhetoric, or not the topmost priority of Ghana's successive governments due to the pressing need to provide basic needs for citizens; or the concept of inter-generational equity has not been adequately streamlined into Ghana's sustainable development policies, plans, and projects. Overall, a constant factor which permeates any attempt to rationalise what may influence the developmental choices and options available to Ghana in her quest for sustainable development is the currently low developmental levels (Holden et al. 2014) which can be described through poverty (i.e., a lack of economic resources both at the State level and also on the part of individuals). Based on the scenario described above, one can argue that both the current and future Ghanaian generations must have adequate economic resources in order to satisfy their basic needs and to protect the integrity of the ecological environment, which, in this case, includes utilising the ecological resources that are meant to be protected. In other words, the capacity of Ghanaian society to pursue sustainable development may seem to be dependent on economic performance. Nevertheless, going by the Brundtland Commission's definition of sustainable development, it is evident that if indeed a country such as Ghana is committed to SD principles, then its SD performance should not merely hinge on its socio-economic performance. While this does not discount the critical role of economic development, the argument advanced here is that economic performance in development is not the sole objective of sustainable development (Holden et al. 2014), a position succinctly captured in the first Human Development Report in the 1990s as: "the purpose of development is to offer people more options. One of their options is access to income—not as an end to itself but as a means of acquiring human well-being. But there are other options as well, including long life, knowledge, political freedom, personal security, community participation and guaranteed human rights. People cannot be reduced to a single dimension as economic creatures" (UNDP 1990, p.iii). Based on the evidence presented in the findings and taking cognisance of the 'poverty factor' in Ghana, the perspective of this study is that, as much as sustainable development in Ghana depends on economic, social, and environmental development progress, it also depends on the political salience attached to it. Therefore, Ghana's progress from a sustainable development perspective is also primarily a matter of political priority. This is because rather than following utopian interpretations of sustainable development or making decisions based on political expediency, Ghana, as a less-developed country, could still pursue political choices that will put her on a path to sustainable development. For example, irrespective of the economic situation, Ghana could still pursue SD if the country's leaders make choices that ensure equality of opportunity irrespective of gender, generation, among others, such that citizens are not left behind by economic progress, social mobility and environmental conditions.

4.4 Urbanisation in Ghana

4.4.1 An overview of urbanisation in Ghana.

In 2010, Ghana's urban areas were officially home to more than half of the country's population (Cobbinah et al. 2015a). Therefore, the management of the process of urbanisation is critically important, and will determine whether the country will reap or lose dividends from the demographic and structural shift. As already mentioned in Chapter Two, there is no common agreement on the definition of an 'urban' centre. How an urban area is constituted and defined remains an open question, usually subject to the interpretation of the particular country involved. Thus, the notion of what constitutes an urban area in developed countries is different from that of developing countries. However, Ghana uses a population (geographic) approach by which an urban area is defined as a settlement with a minimum population threshold of 5,000 or more people (NUP 2012). As in other parts of the developing world, increasing urbanisation in Ghana has been attributed to factors which include rural-urban migration, natural population growth, and reclassification of settlements (Ardayfio-Schandorf et al. 2012).

The urbanisation process in Ghana has led to changes in demography and landscape. From a demographic perspective, post-independent programmes including the industrialisation and structural adjustment programmes widely influenced urban population growth. A World Bank (2015) study showed that the urban population in Ghana increased from 4 million in 1984, to about 13 million by 2013. The demographic transition has, in turn, resulted in changes to the landscape due to the progressive conversion of land to meet urban needs. Numerous documented examples exist of pressure being brought to bear on land owners to convert fragile land cover types such as water bodies and wetlands to urban use due to the high demand for land (see, for example, Wiegleb 2016; Amo et al. 2017). For example, Amo et al. (2017) conducted an investigation into the acquisition and development of wetlands in Ghana and established that wetlands were increasingly being converted into residential and commercial uses in Kumasi due to rapid urbanisation. Accordingly, research has emerged to document the urbanisation process and patterns in Ghana. For example, Coulter et al. (2016)

have conducted a classification and assessment of land cover and land use change in southern Ghana using remotely-sensed satellite imagery. Their findings indicate that the increasing urban population in Ghana has resulted in urban expansion, and has led to the reduction of natural land cover in Ghana. In the southern part of Ghana where urbanisation is rapid, Coulter et al. (2016) concluded that urban expansion has reinforced environmental degradation patterns that started before the year 2000. Similarly, Attua and Fisher (2011) have analysed historical and future land cover change in the New Juaben Municipality in Ghana. Their study provided empirical evidence of rapid urbanisation in Ghana, and they concluded that urbanisation, as characterised by expanded urban physical infrastructure, had led to the decline in natural vegetation cover which was primarily characterised by rapid loss in hitherto arable lands. Nevertheless, the rapid nature of the urbanisation process in Ghana means that there is still limited understanding; and the existing literature has mainly focused on a single city or region as the unit of analysis (see, for example, Attua and Fisher 2011; Coulter et al. 2016). This potentially means that differences in urbanisation patterns of different city scales (for example, between small cities and large cities) are rarely discussed, although variations in factors such as socio-economic conditions or geophysical conditions may shape their respective urbanisation patterns (Abulibdeh et al. 2019). The argument above suggests that gaining a deeper understanding of the urbanisation process in Ghana will require an appreciation of the patterns in multiple cities. Accordingly, the limited and highlynuanced nature of scholarly work on urbanisation in Ghana means that in the context-specific study of the interactions between urbanisation and sustainable development, bespoke analysis of local urbanisation patterns and dynamics becomes a critical requirement.

The urbanisation process in Ghanaian urban areas shares similar characteristics with other urban areas in the sub-region, as it has occurred without significant industrialisation and transformation of rural areas. The country's recent wave of urbanisation since the turn of the millennium, coincided with significant socio-economic growth as, for example, industrial and service jobs grew from 38% in 1992 to 59% in 2010 (World Bank 2015). However, these employment opportunities have not happened in sectors considered critical to sustained economic growth (World Bank 2015). It is plausible to observe that increased urbanisation and population growth in Ghana have given rise to urban poverty, as has been discussed in Section 4.2.1.3. Thus, in the coming decades, a significant challenge for the country is how it will develop while maintaining the economic, social, and ecological integrities of urban systems.

4.4.1.1 Urbanisation and mining in Ghana.

Ghana has significant mineral resources and is one of the leading producers of gold in the world (ICMM 2007). Pre-independence, Ghana was known as Gold Coast, an indication of its mineral wealth. Mining provides significant taxes, revenues, and dividends for the country. For example, mining contributed to about 7% of the country's Gross Domestic Product (GDP) and about 40% of total merchandise export in 2011 (Akafia and Kuenyehia Sr. 2012). Scholars suggest that significant natural resource exports propel urbanisation, especially when the financial rewards accrued are used to provide urban goods and services to the populace (Gollin et al. 2013; Jedwab 2013; Cavalcanti et al. 2014). However, others (UNECA 2012a; Turok and McGranahan 2013) have also shown that natural resource-based economies are not naturally urban-oriented. The situation can be linked to the emerging concept of Resource-Based Cities (RBC), which have been defined as urban areas whose existence and economies are primarily based on the exploitation, extraction and processing of natural resources (Li et al. 2013), and include oil extraction, mining of mineral ores and forestry, etc. Given the importance of mineral exports to the national economy of Ghana, the influence of mining on urbanisation and its inter-linkages with the development of the urban areas which host the mining projects should not be overlooked. Concomitant with the start of mining operations is the migration of people into mining centres. As Hilson (2002a; 2002b) reports, the Ghanaian mining industry has been attracting migrants from within and outside Ghana for over one hundred years, fuelled by the desire of those migrating to seek better life opportunities. In a different vein, Hilson et al. (2014) note that as of the year 2005, there were over 50,000 Chinese migrants engaged in illegal mining in Ghana. Thus, together with natural population growth, migration has provided a central impetus to urbanisation in mining centres in Ghana. Given the country's history of mining (Hilson 2002a), several studies have been conducted to explore the relationship between the mining industry and the national economy, its impact on the environment and the role of corporate social responsibility (for example Aryee et al. 2003; Bloch and Owusu 2012, Hilson 2012; Standing 2014). However, there is little research on the interactions between mining and urbanisation. Gough and Yankson (2012) argue that although a lot has been written about mining in Ghana and also about urban settlements, these studies have been considered independently by the scholars in the respective fields. There is a poor understanding of the inter-linkages between mining and urbanisation due to the limited literature that consider both fields together. For instance, while the physical legacy of mining is understood in terms of the harmful externalities on the

environment, not much has gone into exploring the relationships between such externalities and the profound impact of mining on urbanisation and urban growth.

The mining settlement of Obuasi serves as a useful example. Obuasi is the location of the oldest gold mine in Ghana, which has been in operation for over one hundred years (Wan 2014). Mining has fuelled far-reaching processes of urbanisation in Obuasi, resulting in a rapid increase of its population from 22,800 in 1960, to about 203,554 in 2019, according to census data (Ghana Statistical Service 2002; Ghana Statistical Service 2014c; Ghana Statistical Service 2019), making it one of the largest urban areas in Ghana. The capitalintensive nature of large-scale mining operations means that only a few thousand people are directly employed at the mine, while the diverse nature of the local economy ensures that the town is not entirely dependent on the mine. Nevertheless, Gough and Yankson (2012) maintain that the principal reason for the existence and continuous growth of Obuasi is due to the mine. Aside from socio-economic benefits accrued from mining, it has been reported that processes associated with mining in Obuasi and its environs have led to widespread water pollution (ActionAid Report 2006; Foli et al. 2012; Azumah et al. 2019). Furthermore, it has been reported that mining-related activities have contributed to an increasing number of school dropouts in the Obuasi area as some youth tend to engage in small-scale mining at the expense of their education (Azumah et al. 2019). As the largest and most important town in the history of mining in Ghana, a lot of mining-related studies have been conducted on the Obuasi mine and its catchment area. Although not an exhaustive list, for example, Adu-Poko et al. (2012) have studied land cover changes in the Obuasi mine catchment, while Wan (2014) has also explored the environmental justices and injustices in three mining communities in Obuasi. In addition, Opoku-Mensah and Asare-Okyere (2014) have studied company-community conflicts over gold mining in Obuasi. Despite these studies, the extent to which the ills and the virtues of the large and complex urban agglomeration, of which Obuasi is at the core, is owed to mining, is poorly understood and rarely acknowledged. This is surprising because, like many other countries in SSA, mineral resource mining is fundamental to the Ghanaian economy and resource-based towns feature prominently in the country's economic and social geography. Particularly important for this research, there is limited knowledge on urbanisation and mining in the context of sustainable development, whereby the implications of urban growth through both population and landscape change are constructively considered.

4.4.2 An assessment of urbanisation in Ghana.

4.4.2.1 Urbanisation trends.

In order to highlight the trends of urbanisation in Ghana, the analysis that follows deploys past, current, and projected urbanisation in Ghana and refers to global statistics. In Table 4.3, past and projected urban population, urbanisation level and urbanisation rates over a tendecade period are presented. As seen from Table 4.3, Ghana had an urban population of 769,000 people in 1960, which represented an urbanisation level of 15.44%. By 2000, Ghana's urban population was 8,424,000, representing an urbanisation level of 43.95%. By 2010, over half of the Ghanaian population was living in urban areas as it recorded an urban population of 12,492,000, representing an urbanisation level of 51.22%. In terms of the future urban population, Ghana is projected to have an urban population of 35,520,000 people by 2050, representing an urbanisation level of 72.33%. The rate of change in urbanisation level was relatively higher, at 3.94% for the one-decade period (2000-2010) compared with 4.79% for the five-decade period (1950-2000). The table also shows that up until 2010, urbanisation in Ghana had been below or consistent with global levels. However, projected future urbanisation for 2010-2050 is expected to be higher in Ghana at 2.61% compared with the global level of 1.41%.

 Table 4.3: Urban population, urbanisation level and urbanisation rates in Ghana compared with World figures.

Location	ion Population ('000)			Urbanisation level (%)			Urbanisation rates (%)				
	1950	2000	2010	2050	1950	2000	2010	2050	1950- 2000	2000- 2010	2010- 2050
Ghana	769	8,424	12,492	35,520	15.44	43.95	51.22	72.33	4.79	3.94	2.61
World	745,495	2,858,632	3,558,578	6,352,175	29.44	46.69	51.60	67.18	2.69	2.19	1.41

Source: Cobinnah et al. (2015a) as adapted from UNDESA / PD (2012) and UNDESA / PD (2013).

4.4.2.2 Urbanisation patterns in the major urban areas in Ghana, 1970-2010.

Rapid urbanisation is also observed through the urbanisation patterns of the major cities in Ghana (as shown in Figure 4.11). From this, it can be seen that urbanisation is skewed

towards the two largest cities, Accra and Kumasi, as their combined population represented 63% and 32% of Ghana's urban population in 1970 and 2010 respectively. In 1970, only Accra had a population of over 500,000. By 2010, both Accra and Kumasi were estimated to have populations of over 2 million people each. It is also evident that smaller urban areas have also urbanised rapidly in both absolute and relative terms. For example, the urban population in Sekondi-Takoradi was estimated at 500,000 in 2010. Despite the spread of urbanisation across the country, the findings show that the two biggest cities, Accra and Kumasi, remain as the major containers of urban population and economic opportunities.



Figure 4.11: Urbanisation patterns in the major urban areas in Ghana, 1970 - 2010. Source: Cobinnah and Niminga-Beka (2017) as adapted from GSS Records (2012; 2013).

4.4.2.3 Rural-urban migration and natural population growth.

Table 4.4 shows the contribution of rural-urban migration and natural population growth to urbanisation in Ghana between 1984 and 2010. Overall, natural population growth was the greater contributor to urban population growth, contributing an average of 60% over the period. Their respective contributions also remained constant over the period. For example, urban growth due to migration from rural areas contributed 37.4% and 40.7% respectively, for the periods 1984-2000 and 2000-2010. At the same time, natural population growth contributed 62.6% and 59.3% to overall urban growth during the periods 1984-2000 and

2000-2010 respectively. When considered on a regional basis, the findings show that the contributions of rural-urban migration and natural population growth to urbanisation are mixed. For example, in the Western Region, urban growth due to migration contributed 59.2% between 1984-2010, a statistic that had reduced to 31% for the 2000-2010 period. However, in the Northern Region, urban growth due to migration contributed 11.3% between 1984-2010, a statistic that had increased to 44.6% for the 2000-2010 period. This suggests that how urbanisation manifests at the local level in Ghana is not homogenous.

	Urban Popu	ilation	% of growth due to		% of growth due to		
Region			migration		natural increase		
	1984-2000	2000-2010	1984-2000	2000-2010	1984-2000	2000-2010	
All	8,274,270	12,545,229	37.4	40.7	62.6	59.3	
regions							
Western	698,418	1,007,969	59.2	31.0	40.8	69.0	
Central	598,405	1,037,878	51.1	58.3	48.9	41.7	
Gt. Accra	2,547,684	3,630,955	4.1	28.1	95.9	71.9	
Volta	441,084	713,735	54.8	50.5	45.2	49.5	
Eastern	727,914	1,143,918	54.6	46.5	45.4	53.5	
Ashanti	1,853,065	2,897,290	56.3	45.7	43.7	54.3	
Brong	678,780	1,028,473	53.9	40.6	46.1	59.4	
Ahafo							
Northern	483,790	750,712	11.3	44.6	88.7	55.4	
Upper	144,282	219,646	57.5	41.4	42.5	58.6	
East							
Upper	100,848	114,653	71.6	-123.5	28.4	223.5	
West							

Table 4.4: Contribution of rural-url	ban migration	and natural j	population	growth to
urbanisation in Ghana.				

Source: Ghana Statistical Service (2014a).

Note: The regions in Ghana increased from 10 to 16 in 2019.

4.4.2.4 Urbanisation and economic growth.

In order to explore the relationship between urbanisation and economic growth in Ghana, the country's Gross National Income (GNI) figures were compared with the levels of urbanisation over time, as shown in Table 4.5. From the comparison, there is no clear link between the level of urbanisation and GNI. In 1970, GNI was U.S.\$ 250 at an urbanisation level of 29%. This increased to U.S.\$ 340 in 1984 at urbanisation levels of 32%. However, almost two decades later, at urbanisation levels of 44% in 2000, GNI in Ghana had decreased to U.S.\$ 270.

Table 4.5: GNI per capita vs urbanisation levels in Ghana, 1970-2010
--

Census year	GNI per capita (U.S. \$)	Urbanisation level (%)
1970	250	29
1984	340	32
2000	270	44
2010	1,230	51

Source: Cobinnah and Niminga-Beka (2017) as adapted from GSS Records (2012; 2013).

4.4.3 Discussion.

4.4.3.1 Trends in urbanisation and economic growth.

The findings presented in the previous sub-section provide supportive evidence for the pattern and nature of urbanisation in Ghana. Based on the numbers captured in Table 4.3, it is clear that urbanisation in Ghana has proceeded at a fast pace, a situation that has resulted in over half of Ghana's population living in urban areas (Cobbinah and Niminga-Beka 2017). Overall, between 1950 and 2010, Ghana's urban population had grown at an average annual exponential rate of 4%, increasing from 769,000 (15.44% urban) in 1950 to 12,492,000 (51.22% urban) in 2010. This resulted in the populations of its two biggest cities, Accra and Kumasi, hitting over 2 million each. While Ghana's urbanisation was either below or marginally on a par with global urbanisation rates between 1950 and 2010, critically, its future urbanisation level (72.33%) is projected to exceed the global level (67.18%) for the

period 2010 - 2050 (Cobinnah et al. 2015a). This highlights the enormity of the scale of urbanisation in Ghana.

As argued in Section 1.1.2, the relationship between urbanisation and economic growth in Africa remains an enigma, as it is not clear whether or not it always brings economic growth (Turok 2013). Published international and national statistics do not allow easy analysis of the contribution of individual urban areas to GNI per capita in Ghana. However, from the national statistics for Ghana, a comparison of the GNI per capita and urbanisation levels shown in Table 4.5 does not suggest any positive relationship between the two. This is because when GNI was U.S.\$ 250 in 1970, urbanisation level was 29%. When GNI increased to U.S.\$ 340 in 1984, urbanisation level was 32%. Subsequently when urbanisation level was 44% in 2000, GNI was U.S.\$ 270. As such, while economic growth tends to be associated with urbanisation, urbanisation does not necessarily lead to economic growth, especially in the African case (Turok 2013). The idea behind this assertion is that there are independent or mediating forces which are critical to determining whether urbanisation will lead to equitable economic growth. At this stage, an important fact to note is the resource-dependent nature of the general economy in Ghana, which, like other African countries, is suggested not to be urban-oriented, but helps situate the relationship between Ghana's urbanisation and economic development within the context of why urbanisation in Africa does not necessarily result in economic growth or development (UNECA 2012a; Turok and McGranahan 2013). While it is recognised from the foregoing that GNI per capita in Ghana may or may not increase alongside urbanisation rates, at the very least, any discussion on the effect of urbanisation on urban economic development in the country will have to also recognise the prevalence of poverty, informality, and their impacts on the livelihoods of the urban populace. As shown in Section 4.2.1.3, endemic poverty and inequality resulting from wide-spread unemployment and informality are among the structural ills plaguing urban societies in Ghana (UNICEF 2012; World Bank 2015; Cooke et al. 2016). As much as the products of informality, including slum formation, may be due to poor urban management; the informal sector is an indispensable part of Ghanaian urban centres as it is crucial for employment creation, especially for migrants (Gillespie 2016). From Section 4.2.1.3, it is true that poverty levels in Ghana have reduced from 56% in the 1990s to 23% in 2013. However, it is also true that as of 2006, the wealthiest 20% of Ghanaians earned over half of Ghana's income, up from 44% in the 1990s; while at the same time, the poorest one fifth saw their share of national income reduce from 6.9% to 5.2%. Therefore, inferring from the review of urban challenges in

Ghana presented in Section 4.2. and the discussion so far, it may be argued that even when urbanisation has spurred economic growth, it is not a panacea for all urban economic challenges. The trends in Ghana's urbanisation and economic growth have, thus, justified why settlements in Ghana are becoming increasingly 'urban' in their outlook. Even when urbanisation should result in an economic growth scenario, an interplay of factors could lead to the creation of urban areas where there is inequality and poverty.

4.4.3.2 Driving forces of urbanisation in Ghana.

Given the rapid nature of the urbanisation in Ghana, it is important to understand the key driving forces underpinning the process. Based on the evidence presented in this section and the literature (see, for example, McGranahan et al. 2009; Jedwab et al. 2017), it can be argued that urbanisation in Ghana is driven by internal forces and exacerbated by external factors. These forces are inter-related, and in an increasingly globalised world, it is their synergistic effect that shapes the urbanisation phenomena in Ghana. The discussion on the driving forces of urbanisation in Ghana will, therefore, consist of two parts: 'internal forces' which will touch on factors such as population growth and rural-urban migration, and 'external forces' which covers issues such as globalisation. For this discussion, population growth, rural-migration and reclassification of settlements will be considered under the umbrella of 'demographic shifts'.

(a) Demographic shifts.

Synthesis of the findings shows that consistent with the literature (for example, McGranahan et al. 2009; Jedwab et al. 2017), the main demographic movements underpinning urbanisation in Ghana are natural population growth and rural-urban migration. A third force, 'reclassification', exists which in the view of this study is a product of natural population growth (McGranahan et al. 2009; Jedwab et al. 2017). The findings show that a greater majority (60%) of urban growth in Ghana between 1984 and 2010, is attributed to natural population growth; and the contribution of rural-urban migration to urbanisation was at an average of 40% between 1984 and 2010. The contribution of rural-urban migration to urbanisation in Ghana is dual. On a primary level, the mere movement of migrants from rural to urban areas contributes to urban growth. On a secondary level, the reproductive functions of rural migrants who become urban citizens also contribute to urban growth (Cobinnah et al. 2015a). Overall, the agglomeration effects due to the primacy of cities such as Accra and

Kumasi have ensured that alongside natural population growth, they have also expanded due to the inflow of migrants seeking better opportunities in life. For example, at an average annual growth rate of 5% (Oduro et al. 2014), Kumasi is the fastest growing big city in Ghana. Similarly, the increase in the distribution of urbanised towns has been fuelled by urbanisation and 'reclassification' of some previously rural populations through natural population growth and migration (McGranahan et al. 2009; Jedwab et al. 2017). Nevertheless, the regional differences in urbanisation evidenced in Table 4.4 show that urbanisation does not automatically manifest in the same spatial pattern, as other development factors might be at play. Looking at the trends overall, it is clear that demographic drivers have not only led to an increase in the total urban population, but have also led to an increase in the sizes of Ghana's biggest cities.

(b) Market forces.

Consistent with the notion of cities as hotspots of economic growth in an increasingly globalised world (Zhang 2016), the role of market forces in driving the urbanisation process in Ghana is critical. The Ghanaian economy is a mixed one, characterised by a gradual shift from a State-dominant economy to a market-led and liberalised one (see Obeng-Odoom 2012). Thus, the increases in GNI over the past decades (shown in Table 4.5), as a measure of economic performance, have come about from a combination of minimal State control of economic activities and increased private sector freedoms. In order to analyse the causal links of market forces on urbanisation in Ghana, they are discussed in terms of the supply and demand for land and housing, and labour. Inevitably, an increasingly urban population which results from the process of urbanisation triggers the conversion of land to meet the construction demand for residential and commercial properties, and other public goods such as transportation infrastructure. An important factor to consider in the scenario above is the availability of land to meet the demands of different stakeholders. Assuming that various stakeholders can freely compete for limited urban land resources, then realistically, this means that only private investors and wealthy individuals could afford to invest. The urban land market, thus, influences the urbanisation process by determining the spatial concentration of economic opportunities and infrastructural provisions. With a housing stock deficit of at least 1.7 million (Daily Graphic Online 2018), the land and housing markets also influence the spatial structure of urbanisation in Ghana. Rural-urban migration and natural population growth create a demand to provide more urban housing. Considering the lack of welfare housing provision and the generally weak planning enforcement, the bid to supply

the housing demand further drives the urbanisation process. This is because, with the State unable to supply affordable public housing, market actors become the determinants of new property locations (Daily Graphic Online 2018). They will attempt to achieve this by pursuing options that will extract the highest value from the land and offer maximum investment returns. For example, market actors could decide to undertake projects near locations with relevant opportunities and services. This means that more people could decide to move to such areas to enjoy these opportunities and services. The effect of this could be the creation of property portfolios which do not necessarily conform to the logic of planning frameworks and could potentially distort urban form.

The more liberalised nature of Ghana's economy enhances the creation and expansion of a free labour market (see, for example, Grant and Nijman 2002; Obeng-Odoom 2012). When considered together with rural-urban migration discussed above, it can be argued that urbanisation in Ghana is shaped by rural 'push' factors and urban 'pull' factors (Jedwab et al. 2017), although the focus here will be on urban 'pull' factors. The unrestricted movement of people into urban areas is in itself a market behaviour as the migration of people to cities is usually in response to the 'pull' factors, such as better employment opportunities and higher wages. This means that provided the perceived economic incentives for migrating into urban areas remain strong, urbanisation will proceed with the flow of people from rural to urban areas. Thus, the urbanisation process in Ghana is shaped by market forces through its coexistence and interaction with State mechanisms. Essentially, the reality that urbanisation in Ghana has outpaced planning provisions (Yeboah and Obeng-Odoom 2010), is indicative of the weakened State-control of how these market forces are unleashed in a more liberalised setting.

(c) Globalisation.

As was elucidated in the literature (see Section 2.2.4), globalisation connects national economies by facilitating the movements of goods, services, and people freely across borders. Considering how globalisation shapes the urbanisation process in Ghana requires an examination of the everyday construction of globalisation. The narratives of globalisation generally assume an in-flow of foreign direct investments that facilitate the mobility of capital, goods, services, and people. This in-flow, in turn, provides the foundations for the overall development of a country by inducing economic growth. Inevitably, a significant proportion of Ghana's GDP is derived from foreign direct investments. For example, mineral

resource exports contributed to over 7% of the country's GDP in 2011 (Akafia and Kuenyehia Sr. 2012). Moreover, as can be seen from Table 4.5, Ghana's GNI per capita increased from U.S.\$ 340 in 1984 to U.S.\$ 1,230 in 2010.

Practically, urbanisation due to globalisation tends to be in response to decisions taken at higher spatial levels (for example, at the national, regional or global levels). The manner through which globalisation drives urbanisation in Ghana can, therefore, be demonstrated with the following illustrations. At the global level, an investment decision by a Multi-National Corporation (MNC) to establish operations in Ghana generally leads to the migration of capital, technology, and people from across the world and also from within the country (as can be seen in population growth of areas which host these MNC operations) as more people arrive to take advantage of increased opportunities. Here, decisions made at the global headquarters of any particular MNC could have implications at the local level, including changes in the spatial structure of urbanisation. At the regional level, Ghana's membership of political associations such as the Economic Community of West African States (ECOWAS) facilitates the free movement of people, goods, and services across the borders of member countries. This, for instance, has implications for labour markets, and impacts the spatial characteristics of urban areas in member countries, as there is a potential for the influx of people from poorer countries into the urban areas of relatively wealthier countries, in response to economic conditions.

4.4.4 Sustainability implications of urbanisation in Ghana.

Central to the current discussion on urbanisation in Ghana is the implication of the country's urbanisation process for the achievement of sustainability goals. As has been implied by the previous findings and discussions, urbanisation is not only premised on economic factors but also social and environmental factors (Cobbinah et al. 2015a). Therefore, the sustainability of urban areas is about pursuing economic growth while protecting the integrity of the environment and improving social equity. Interwoven with the perception of modern living, an increasingly urban population will stimulate a shift in lifestyle choices from rural to urban, thereby resulting in increased resource use, increased energy use, and increased car dependency, etc. Inferring from the urban challenges facing Ghanaian urban areas (as presented in Section 4.2), and taking into consideration the less-developed status of Ghana, it is clear that rapid urbanisation in Ghana would potentially exacerbate these challenges. For

example, increased car dependency could affect the health of the urban population and the environment due to air pollution. This makes it imperative to discuss whether the urbanisation process in Ghana has proceeded in a manner consistent with the principles of sustainable development, and in ways that assure a reasonable quality of life for Ghana's urban citizens. This is not an attempt to highlight only the 'negatives' of the urbanisation process in Ghana, but rather to reinforce the idea that a well-managed urbanisation process is crucial to the achievement of the country's broader SD goals.

From an economic perspective, globalisation and market liberalisation are increasingly shaping the fundamentals of urbanisation in Ghana. Critically, under the conditions of a liberalised economy, market forces could distort the urbanisation process in a way that threatens the economic foundations of urban areas. For example, there is an inevitable possibility of change in the actors who control and regulate the use of vital resources such as land from local communities into the hands of market actors. This potentially makes market forces determinants of the spatial concentration of vital opportunities and services. Such a situation will have a spill-over effect on the nature of urbanisation. The free labour market created by the liberalisation of State-control in Ghana means that rapid urbanisation due to rural-urban migration could lead to an over-supply of urban labour as the rate of migrantflow into urban areas could exceed the rate of urban job creation. This could exacerbate the already precarious urban unemployment situation (see Section 4.2.1.2) and deteriorate the quality of employment in Ghana. Furthermore, as elaborated earlier, globalisation could spur economic growth due to the inflow of foreign direct investments. This helps drive urbanisation as people move to areas of investment in search of better opportunities in life. For Ghana, FDI investments typically come in the form of MNC investment in the extraction of natural resources. The implications for economic sustainability, and by extension overall sustainability, is the risk of dependency on such investments (see Section 2.6.1.2). This is because there is a likelihood that MNCs will prioritise their profits and shareholders' interests. Therefore, in the event of decreasing profits, MNCs could respond by halting operations, withdrawing entirely, or moving to areas where more favourable conditions exist. This could have severe implications for urban areas whose local economies are heavily dependent on the operations of such MNCs (UNECA 2012a; Turok and McGranahan 2013). A typical example in Ghana is the case of Obuasi Municipality, which has hosted the mining operations of AngloGold Ashanti for over a century. The existence of the mining company means that Obuasi became highly urbanised, as it attracted a variety of people seeking

improved life opportunities. However, until its redevelopment and reopening in 2019, the operations of the company in Obuasi had been irregular since 2014, leading to the retrenchment of mine workers and reduced life opportunities for the municipality's citizens (Ghana Chamber of Mines 2019). Thus, for an urban area such as Obuasi, which is characteristically a resource-based city, rapid urbanisation in an increasingly globalised world is a double-edged sword in the context of sustainability.

From a social perspective, the urbanisation process could enhance sustainability in urban areas in Ghana. For instance, the lobbying power wielded by urban elites and the urban bias tendencies (Lipton 1977) could lead to the provision and improved access to relevant social infrastructure, including education and transportation. Again, for marginalised groups such as women living in rural areas, for whom access to certain opportunities may be limited or nonexistent, urbanisation represents a pathway for increased access to an array of opportunities including education, employment, and participation in politics. Nevertheless, the discussions in the previous sections also demonstrate that the combination of endogenous market forces and the exogenous force of globalisation which underpins the urbanisation process could polarise wealth and resource utility which could, in turn, threaten social sustainability. Such polarisation effects could lead to sharp divisions in urban societies, as deepened inequalities could threaten social harmony. Also, as mentioned earlier, the Ghanaian economy significantly depends on the extraction and export of natural resources, which are generally controlled by foreign corporations. Therefore, depending on the nature of FDIs and the behaviour of their actors, globalisation could drive urbanisation in ways that either contribute to poverty and inequality or alleviate them. For example, the arrival of a major global investor could lead to the forced relocation of a section of an urban population. For some, this might constitute a direct threat to their source of livelihoods. Viewed from another angle, the corporate social responsibility (CSR) initiatives of corporations which sometimes involve alternative livelihood programmes could help alleviate poverty and help protect vulnerable groups. However, although not in the scope of this thesis, it is important to mention that more often than not, these programmes have had minimal impact on alleviating the plight of the poor, as the poor are primarily denied access to natural resources which hitherto were available to them (see for example, Opoku-Mensah and Asare-Okyere (2014).

The quest for overall urban development, in the contexts of neo-liberal economies, could result in side-stepping environmental concerns as socio-economic development is pursued.

From an environmental perspective, population growth due to urbanisation in Ghana could generate substantial challenges to the ecological integrity of urban environments due to the tendency of neo-liberal systems to ignore environmental realities (see Le Gales 2016). Accordingly, in mainstream thinking, urbanisation is generally seen as a driver of environmental impacts (Grimmond 2007; Parnell et al. 2007). In Ghana, the urban challenges explicated in Section 4.2 typify the extent to which the environmental sustainability of urban areas could be compromised. For example, population growth due to urbanisation and its attendant change in consumption patterns could lead to more waste being generated, compounding the already significant waste management challenges facing urban areas in Ghana. Weak land use controls also increase the likelihood of the unscrupulous conversion of protected areas and agricultural land, on which many flora and fauna may depend, into residential and commercial areas (see Le Gales 2016). This is consistent with the position advanced by Redclift (2005) that less-developed nations could potentially prioritise socioeconomic development even at the expense of ecological integrity. On a positive note, however, a well-directed urbanisation process has the potential to mitigate environmental implications. For example, with the needed political will and requisite capacity, Ghana could positively manage the environmental implications of contemporary urbanisation by retrofitting existing urban areas and building new ones based on approaches that are consistent with sustainability principles. In such a scenario, approaches which promote compactness, for example, are likely to lead to shorter commuting distances which will, in turn, reduce air pollution (see Table 2.1).

In sum, the rapid pace of urbanisation in Ghana and the scale of urban challenges imply that the Government cannot adequately manage the process in a manner that is consistent with sustainability goals (relative to available resources). Accordingly, the liberal economic posture assumed by the State means that more and more urban needs are left to private sector provision (see Obeng-Odoom 2012; Leon 2015; Le Gales 2016). The effect of which could be a fragmented urbanisation process which undermines distributive equity, creates pernicious environmental impacts, and threatens wider sustainability goals. The discussions above have considered only some of the important implications of Ghana's urbanisation on the sustainability of its urban areas. The complex and bi-directional relationship between urbanisation and sustainability means that urbanisation underpins many factors that will determine the sustainability of urban areas in Ghana. While these have been explicated from social, economic, and environmental perspectives, no single perspective dominates as the

relative degree to which each dimension of sustainability is affected depends on dynamic local conditions. However, taken together, they set the context in which the sustainability of Ghana's urban areas may be affected by rapid urbanisation and stimulate a rethinking of how urban areas are managed.
Chapter Five

5. Urbanisation and sustainable development in Ghana: comparative case studies in Kumasi and Obuasi.

5.1 Introduction

This chapter presents the findings of the comparative sub-case-study on two urban areas in Ghana (Kumasi and Obuasi) selected to explore the interactions between urbanisation and sustainable development at the urban spatial level. This strategy was chosen because of the complexity of the inter-relations between the phenomena mentioned above at the local urban level which is, possibly, best understood through case-based investigation (Yin 2003). The rest of the chapter consists of four sections and proceeds as follows. The first three sections consider the urbanisation-sustainability link by focusing on i) demographic and landscape urbanisation in the two areas and their implication for sustainability; ii) the assessment of selected socio-economic indicators related to sustainability; and iii) the subjective experience of citizens in the two urban areas through a survey. The fourth section provides an integrated discussion on the findings in Sections 5.2–5.4.

5.2 The sustainability implications of landscape and demographic urbanisation in Kumasi and Obuasi

5.2.1 LULC Characterisation

The overall accuracies of the land cover maps were above 80%, and the Kappa coefficients ranged between 0.76 and 0.86 (see Appendix 4) with the exception of Obuasi 2010 (Table A4.7, Appendix 4), which imply that the greater majority of image classifications were of sufficient quality (Anderson et al. 1976). The lower overall accuracy and Kappa coefficient recorded for Obuasi 2010 (Table A4.7, Appendix 4) is attributable to classification error due to possible spectral mixing caused by overlap in land cover types such as 'mixed woody vegetation' and 'forest', potentially resulting in mixed-pixel problems which undermined separation of these areas in the imagery.

The maps of LULC classes for Kumasi in 1986, 2000, 2010, 2018 are shown in Figures 5.1 - 5.4, and the corresponding LULC statistics are shown in Table 5.1. Taken together, Figures 5.1–5.4 and Table 5.1 show that 'mixed wood vegetation' (77%) was the dominant land cover type in Kumasi in 1986. However, by 2018, urban expansion had led to the dominance of 'built-up' land cover (62.5%), at the expense of the other land cover classes. Overall, the land cover change in Kumasi within the study period could be characterised as one which is dominated by urban expansion, and the loss of agricultural land and forest. There was conversion of mixed woody vegetation to agricultural land, then onto urban or built-up. However, evidence of the conversion of some agricultural land to mixed woody vegetation or forest during 2010–2018 may be possible signs of sustainability or conservation strategies.



Figure 5.1: Land cover classification of Kumasi, 1986.



Figure 5.2: Land cover classification of Kumasi, 2000.



Figure 5.3: Land cover classification of Kumasi, 2010.



Figure 5.4: Land cover classification of Kumasi, 2018.

LULC Classes	Area Coverage (km ²)				Changes in area (%)			
	1986	2000	2010	2018	1986-	2000-	2010-	1986-
					2000	2010	2018	2018
Built-Up	43.23	232.92	390.23	509.20	438.79	67.53	30.50	1105.77
Agricultural Land	64.75	400.71	373.96	118.55	518.85	-6.68	-68.30	83.00
Mixed Woody Vegetation	628.76	176.34	46.57	113.35	-71.95	-73.59	143.39	-81.97
Water Body	0.50	0.96	0.96	0.67	92.00	0.00	-30.20	34.00
Forest	78.11	4.42	3.63	73.58	-94.34	-17.87	1924.51	-5.91
Total	815.35	815.35	815.35	815.35		•	·	·

Table 5.1: LULC characteristics in Kumasi (1986-2018)

For Obuasi, Figures 5.5–5.8 represent the thematic maps for its spatio-temporal LULC classes in 1986, 2000, 2010, and 2018 respectively, while Table 5.2 provides the corresponding LULC statistics. The land cover types observed in Obuasi varied slightly from Kumasi as it is a historic gold mining area. From Figures 5.5–5.8 and Table 5.2, it is evident that land cover in Obuasi was dominated by 'mixed woody vegetation' (42.5%) in 1986. However, by 2018, 'agriculture' land (33.3%) had moved ahead of 'mixed woody vegetation' (30.9%) as the dominant land cover type in Obuasi. Crucially, 'built-up' land cover increased from 6.2% in 1986 to 15.2% in 2018. Therefore, while not dominant, 'built-up' land cover increased significantly due to urbanisation. Overall, the land cover characteristics in Obuasi during the study period could also be characterised as one which is dominated by urban expansion, increase in mining land, and loss of natural vegetation (agricultural land, mixed woody vegetation and forest). However, aside from the transition from mixed woody to agricultural and onto urban / built- up, the LULC patterns in Obuasi were not so obvious as some other land classes fluctuated (for example, bare ground) between dates (possibly due to classification error).



Figure 5.5: Land cover classification of Obuasi, 1986.

Note: Mixed woody vegetation is represented in the legend as 'Mixed Vegetation'.



Figure 5.6: Land cover classification of Obuasi, 2000.



Figure 5.7: Land cover classification of Obuasi, 2010.



Figure 5.8: Land cover classification of Obuasi, 2018.

LULC	Area	Area Coverage (km ²)			Changes in area (%)				
Classes									
	1986	2000	2010	2018	1986-	2000-	2010-	1986-	
					2000	2010	2018	2018	
Built-Up	23.60	36.62	46.50	57.68	55.16	27	24	144.40	
Agricultural	126.64	80.13	110.03	126.83	-36.72	37.31	15.27	0.15	
Land									
Mixed	161.56	189.41	76.06	117.52	17.31	-59.84	54.51	-27.26	
Woody									
Vegetation									
Forest	58.16	70.08	119.04	71.55	20.49	69.86	-39.89	23.02	
Water Body	-	0.07	1.20	0.07	-	1614	94.17	-	
Bare ground	-	-	11.15	-	-	-	-	-	
Mining	10.55	4.20	16.53	6.86	-60.19	293.57	-58.5	-34.98	
Total	380.51	380.51	380.51	380.51		1	1	1	

Table 5.2:	LULC	characteristics	in (Obuasi ((1986-2018).
-------------------	------	-----------------	------	----------	--------------

Note: '-' is placed where a land cover is not identified on the image for a particular year.

5.2.2 Landscape urbanisation: Kumasi vs Obuasi.

For ease of comparison, especially in the context of urban expansion, a different set of land cover maps (See Appendix 6) was produced in which the land cover types were reduced to two classes (built-up and non-built-up). The statistics on the land cover maps in Appendix 6 are shown below in Table 5.3. The table shows that between 1986 and 2018, Kumasi's built-up land cover expanded by over 1,105%. Within the same period, the built-up share of Obuasi's land cover expanded by over 144%. In terms of exchanges or transfers between the different land cover types, the LULC change matrices (attached in Appendix 5) show that in both Kumasi and Obuasi, built-up or urban land cover was the most significant gainer, usually at the expense of natural land cover. Subsequently, the thematic maps in Appendix 6 were combined in order to produce maps that show the overall extent of landscape urbanisation in Kumasi and Obuasi over the study period as shown in Figure 5.9 and Figure 5.10 respectively. Overall, the spatial extent of land cover types over the thirty-two years

study period is a clear indication that urban development has replaced and eroded into surrounding natural vegetation and non-urban land cover types in Kumasi and Obuasi. Essentially, the application of remote sensing has proved to be an effective method for examining and understanding the LULC characteristics or landscape transformation in Kumasi and Obuasi.

Location	LULC	Area Coverage (km ²)			Chan	iges in are	ea (%)		
	Classes				•		1	-1	-
		1986	2000	2010	2018	1986-	2000-	2010-	1986-
						2000	2010	2018	2018
	Built-Up	43.23	232.92	390.23	509.20	438.79	67.53	30.50	1105.77
	Non-Built-	772.12	582.43	425.12	306.15	-24.57	-27.00	-28.00	-60.35
	up								
Kumasi	Total	815.35	815.35	815.35	815.35				
	Built-Up	23.60	36.62	46.50	57.68	55.16	27	24	144.40
	Non-Built-	356.91	343.89	334.01	322.83	-3.65	-2.87	-3.35	-9.55
	up								
Obuasi	Total	380.51	380.51	380.51	380.51				

Table 5.3: Built-up vs Non-Built-up land cover in Kumasi and Obuasi, 1986 - 2018.



Figure 5.9: Land cover change detection map for Kumasi, 1986 - 2018.



Figure 5.10: Land cover change detection map for Obuasi, 1986 - 2018.

5.2.3 Demographic urbanisation: Kumasi vs Obuasi.

The demographic data of Kumasi and Obuasi over the study period are presented in Table 5.4. It is worth noting that the years for which population statistics were available do not precisely correspond to the dates on which satellite data were captured due to gaps in published data, but for discussion purposes reference will be made to the closest dates possible. From Table 5.4, it is shown that the populations of both Kumasi and Obuasi expanded at least three times between 1984 and 2019. In terms of the rate of population growth, both Kumasi (14.81%) and Obuasi (11.94%) experienced the highest average annual rate of population change during the 1984-2000 intercensal period. The lowest average annual rate of change in Kumasi (2.41%) was recorded for the period 2010–2019, while Obuasi (2.43%) recorded its lowest annual rate of change between 2000 and 2010.

Location	Population					erage Ann es in Popu (%)	ual Ilation
	1984	2000	2010	2019	1984- 2000	2000- 2010	2010- 2019
Kumasi	496,628	1,170,270	1,730,249	2,105,382	14.81	4.79	2.41
Obuasi	60,617	115,564	143,644	203,554	11.94	2.43	4.63

Table 5.4: Population statistics for Kumasi and Obuasi.

Source: Ghana Statistical Service (2019).

Based on available data, the migration statistics measured by 'birthplace of dwellers' for Kumasi and Obuasi recorded from Ghana's 2010 Population and Housing Census are presented in Figure 5.11. From Figure 5.11, the migrant proportion of Kumasi (54%) is higher than recorded for Obuasi (37%). However, the proportion of migrants born outside the Ashanti region in each area is relatively higher in Obuasi (54%) when compared with Kumasi (38%).



Figure 5.11: Migration statistics for Kumasi and Obuasi, 2010. Source: Ghana Statistical Service (2014b; 2014c).

5.2.4 Population density in Kumasi and Obuasi

The population density figures for Kumasi and Obuasi are presented in Table 5.5. The table shows that population density in Kumasi decreased from 11,491 people / km^2 in 1984 / 1986 to 4,135 people / km^2 in 2018 / 2019. In Obuasi, the population density increased from 2,569 people / km^2 in 1984 / 1986 to 3,529 people / km^2 in 2018 / 2019. The sustainability implications of the changes in population densities in Kumasi and Obuasi are discussed in Section 5.2.5.4.

	Kumasi			Obuasi			
	Built-up area (km ²)	Population	Pop. Density	Built-up area (km ²)	Population	Pop. Density	
			(mean)			(mean)	
1984/1986	43.22	496,628	11,491	23.60	60,617	2,569	
2000	232.92	1,170,270	5,024	36.60	115,564	3,156	
2010	390.23	1,730,249	4,434	46.50	143,644	3,089	
2018/2019	509.20	2,105,382	4,135	57.68	203,554	3,529	

Table 5.5: Population density (number of people/ km²) in Kumasi and Obuasi.

5.2.5 Discussion.

5.2.5.1 Demographic urbanisation in Kumasi and Obuasi.

Contemporary demographic trends in Kumasi and Obuasi reflect the changes that have taken place in Ghana's demographic landscape over the last few decades resulting in over half of Ghana's population living in urban areas since 2010 (Cobinnah and Niminga-Beka 2017). As with other aspects of the demographic shift in Ghana, Table 5.4 has shown the remarkable changes in demographics in the study sites. From the evidence presented in Table 5.4, both Kumasi and Obuasi experienced increases, as Kumasi's population quadrupled and Obuasi's population tripled between 1984 and 2019. Consistent with the drivers of population growth identified in the literature, demographic urbanisation in the study areas is attributed mainly to natural population growth and migration (McGranahan et al. 2009; Jedwab et al. 2017). The migration data presented in Figure 5.11 confirms this, with the migrant proportion of Kumasi and Obuasi being 54% and 37% of their respective total population in 2010.

Among other possible reasons, a cogent explanation to describe the pattern of relative differences in demographic urbanisation in Kumasi and Obuasi is succinctly captured in the literature on urbanisation and economic growth in the African context (See Section 1.1.2). Accordingly, from the findings, two observations are made. First, while both areas experienced rapid population growth, that of Kumasi was higher in magnitude as its population quadrupled within the study period when compared with Obuasi whose population tripled. Second, the proportion of migrants in Kumasi's population was higher (54%) as compared with Obuasi (38%). Thus, using the proportion of migrants as a yardstick, it can be argued that Kumasi is potentially more economically-attractive to migrants than Obuasi. The simple explanation for this assertion is the potential for Kumasi's primacy in the region to derive the needed political and financial influence to attract people and services by offering relatively more opportunities for improved life, including improved access to urban services. In contrast, the local economy of Obuasi is primarily based on the mineral extraction activities of the gold mining giant AngloGold Ashanti. As a result, Obuasi's urbanisation process has been fundamentally shaped by the movement of people seeking to improve their livelihoods by exploiting the potential prosperity that the resource extraction operation offers in terms of job creation and enhanced urban infrastructure. However, with the down-turn in operations of the company (Ghana Chamber of Mines 2019), the capacity of the local economy to create and support jobs and wealth opportunities, a key attraction for migrants, has been potentially compromised. Therefore, on balance, Obuasi's population growth is currently more about natural population growth rather than through migration. Accordingly, from the findings in Table 5.4, Figure 5.11, and the discussion so far, it can be argued that Kumasi is more urban and more attractive to migrants than Obuasi. Consistent with the literature on urban bias discussed in Chapter Two, there is a heavy systematic bias that skews resource allocation in favour of urban residents (see, for example, Lipton 1977). The impact of urban-biased policies, thus, becomes more profound resulting in an increase in rates of urbanisation as cities expand to accommodate more people who want to enjoy the enhanced provision of public goods (Majumdar et al. 2004). Furthermore, the findings have demonstrated how the prevailing urban economic circumstances have shaped the contribution of natural population growth and migration to the demographic shifts observed in Kumasi over the study period. This is consistent with the arguments of Ferguson (1999) who suggests that population urbanisation should be conceptualised as two distinct and independent processes of natural population growth and migration for which the likelihood of either becoming the most important component of urbanisation is primarily determined by the

prevailing economic conditions. In sum, it is evident from the findings that the overall nature of demographic urbanisation in Kumasi and Obuasi were quite typical when considered in the context of observed patterns and identified leading causes of urbanisation discussed in Chapters One and Four; and consistent with the literature (see, for example, Ferguson 1999; David and Arman 2014), the urbanisation processes in the two areas have been fundamentally shaped by the prevailing economy.

5.2.5.2 Sustainability implications of demographic urbanisation in Kumasi and Obuasi.

With the results provided and discussed so far in relation to demographic urbanisation, a key question concerns the implications of the observations for sustainable development in Obuasi and Kumasi. As shown in Table 5.4, the urban growth in the areas under consideration provides local-level evidence of the ongoing rapid urbanisation in Ghana that has resulted in over half of the Ghanaian population living in urban areas since 2010 (Cobbinah et al. 2015a). While the combined population of Kumasi and Obuasi is roughly 7% of the overall national population (which is estimated at circa 30 million people), they are two of the ten biggest urban areas in Ghana (Ghana Statistical Service 2019). Thus, the scale of demographic shift presented in Table 5.4 for the two cities is both intriguing and also pinpoints to the extensive challenges faced by managers of Kumasi and Obuasi in relation to urban transition (see Section 4.2). For Kumasi, it is intriguing because of the uncertainty of having to deal with the fastest-growing urban area in Ghana, at 5% per annum (Oduro et al. 2014). For Obuasi, it is intriguing because of the uniqueness of its urbanisation process which has mainly been shaped by the resource extraction activities of AngloGold Ashanti and the recent down-turn of the company's operations (Ghana Chamber of Mines 2019). Overall, for both areas, it is intriguing because it presents unique opportunities to modernise, and in an increasingly globalised world, to transform into model cities that epitomise 21st century urban living in Ghana and by extension, SSA.

In the context of challenges, however, it has been argued that rapid urbanisation in Ghana has complicated the government's capacity to provide essential infrastructural and social services (Farvacque-Vitkovic et al. 2008; Okeke 2014) and the review of urban challenges in Ghana in Section 4.2 backs this argument. Therefore, from an economic perspective, rapid demographic urbanisation has implications for economic development in Kumasi and Obuasi (which are the two most populous centres in the Ashanti Region). While the high number of

migrants in both Kumasi and Obuasi is evidence of the attraction, power, and opportunities that both urban areas offer, an inevitable challenge will be how to engage all of these people in an economically productive manner that enhances their prospects for a better future. As already highlighted in Chapter One, a World Bank Report (2000, p.130) noted that "Africa's pattern of 'urbanisation without growth' is in part the result of distorted incentives that encouraged migrants to exploit subsidies rather than in response to opportunities for more productive employment". Furthermore, as argued by Potts (2016, p. 261), "migration rates go up when [urban] employment is being generated but are not sustained in the face of strong negative changes in urban economies, incomes and condition; they change in logical ways and according to the basic precepts of standard neo-classical migration theory" (see Section 1.1.2). Therefore, for a country where urban unemployment is pervasive, and poverty is enduring (see Section 4.2), the drastic change in urban population over the past decades in Kumasi and Obuasi is significant when considered in the context of the capacity of the cities to provide adequate job and wealth creation. From an environmental perspective, the sustainability implications of rapid urbanisation transcend beyond the urban areas themselves. This can be explained by the reality that an increasing population will expend more ecological resources to meet their basic needs (for example, food), as well as to assimilate their waste (see Chapter Four). Based on the evidence presented in Section 4.3.2.3 which showed that Ghana's ecological footprint has surpassed bio-capacity, the observed drastic upward changes in the population is expected to increase the ecological footprint in Obuasi and Kumasi, and potentially threaten ecological sustainability. From a social perspective, rapid urbanisation in Kumasi and Obuasi poses social challenges. While the inhabitants of Kumasi and Obuasi may be perceived as better off compared to their rural counterparts (in terms of standard of living), increasing populations mean that more housing, water, and sanitation infrastructure will have to be provided. In a country where a significant proportion of the population does not have access to basic infrastructure (see, for example, Amoh-Gyimah and Aidoo 2013; Yakubu et al. 2014), a rapid change in demographic urbanisation will likely mean that more people will compete for access to these infrastructures in order to meet their basic needs, which could widen already entrenched inequality. Also, the high proportion of migrants could potentially affect social cohesion and harmony. Under these scenarios, it can be argued and inferred from the literature (see, for example, Farvacque-Vitkovic et al. 2008; Smit and Parnell 2012; Okeke 2014) that if urbanisation (in the demographic context) in Kumasi and Obuasi proceeds without adequately-planned economic and social preconditions, then sustainability challenges become

inevitable. The implications are the potential growth of slums, worsening environmental conditions, deepened social tensions, and the overall deterioration of the quality of life for the cities' residents (see, for example, Ofori 2009; Osei-Boateng and Ampratwum 2011; Okyere et al. 2012). Therefore, the demographically-shaped urbanisation in Obuasi and Kumasi potentially affects the sustainable development requirement of meeting the basic needs of citizens as the trajectory of the urbanisation process may not be conducive for the realisation of sustainability ambitions.

5.2.5.3 Landscape urbanisation in Kumasi and Obuasi.

The extensive land use choices and demands stimulated by the urbanisation process have resulted in fundamental changes to the landscape in Kumasi and Obuasi, thereby posing significant risks to living conditions, the natural environment, and development. The analysis of the LULC maps shows that the urban areas under examination have gradually expanded due to the forces of urbanisation. In general, the findings suggest that landscape urbanisation in Kumasi and Obuasi is characteristically a two-step process as forest and mixed woody vegetation land cover types were lost to agricultural land cover, which, in turn, was lost to built-up or urban land cover (see Appendix 5). For Kumasi, the forces of urbanisation have set LULC patterns on a clear trend towards an increase in built-up cover and loss of natural vegetation. For example, between 1986 and 2018, significant portions of the forest land cover class in Kumasi and its immediate surroundings had been converted into a dominant urban landscape. Specifically concerning urban expansion, the findings (Table 5.3) indicate that the built-up share of Kumasi's land cover expanded by over 1,105% between 1986 and 2018. This has led to Kumasi's expansion beyond its 254 km² administrative boundary to merge with adjoining districts. In contrast, the landscape change in Obuasi has occurred within its administrative boundary and urban expansion has been scattered. As would be expected of an area with a history of mineral exploration and mining, urbanisation and mining activities resulted in built-up land cover and mining-related land cover classes being significant aspects of land cover transformation in Obuasi. Specifically concerning urban expansion, the findings (Table 5.3) indicate that the built-up share of Obuasi's land cover expanded by over 144% between 1986 and 2018. Thus, although not in the same magnitude as Kumasi, built-up land cover in Obuasi has expanded in relative proportions at the expense of non-built-up land cover over the study period. Essentially, however, there was densification of expansion in Obuasi as population density increased by 72% between 1984 and 2019, compared with Kumasi where population density decreased by 277% (see Table 5.5). In this context, unlike in Kumasi where urban expansion had spread out in all directions from the urban core, the urban land cover in Obuasi seemed to concentrate on the western side of the municipality. Based on local knowledge and personal observations, and consistent with observations in Google Earth imagery, the concentration of urban land cover in western Obuasi is explained by: (i) the location of AngloGold Ashanti's mining operations in the western side of the municipality; (ii) the location of the urban core in the western part of the municipality by mainly surrounding the mining operations; and (iii) the possibility that the prevailing land ownership regime, especially in the context of protected mining concessions, affects acquisition and development of new lands, thereby constraining urban expansion to the western side of the municipality. In the end, although no two urban areas are perfectly identical, the study sites exhibited some similarities that are relevant to the current debate as the impacts of land cover change, especially in the context of sustainability, potentially transcends physico-administrative boundaries. In all, the land cover classification and analysis have given a more fine-grained picture and diachronic analysis of landscape transformation in Kumasi and Obuasi (from a comparative perspective) over the last three decades. Consistent with the findings from the literature (for example, Attua and Fisher 2011; Coulter et al. 2016; see Section 4.4.1), the findings in this study demonstrate that rapid urbanisation in Ghana's urban areas has led to increased replacement of natural land cover by urban land cover. Furthermore, exploring the urbanisation process of landscape transformation in Kumasi and Obuasi highlights the importance of understanding the factors shaping LULC change in these cities. These are identified and explained in terms of (a) city size and administrative hierarchy; and (b) market and state forces.

(a) City size and administrative hierarchy.

From a landscape perspective, the size or hierarchy of an urban area can shape its landscape urbanisation process as it potentially determines available opportunities for improved life (Tselios 2014). Thus, as was noted in literature reviewed in Chapter Two, large cities (for example, capital cities and metropolitan regions) are shaped by urbanisation economies while small cities are shaped by localisation economies (McCann and Shefer, 2004). Furthermore, Scott and Storper (2003, p.581) suggested that *"large cities are locomotives of the national economies within which they are situated, in that they are the sites of dense masses of interrelated economic activities and offer a wider and better quality of the producer services that are essential to technological innovation"*. These observations are evident in the present

study and they are explained as follows. For instance, as the second biggest city in Ghana and the capital of the Ashanti Region, Kumasi is bigger than Obuasi, both in terms of population size, and also in terms of administrative hierarchy. The differences in population and administrative size mean that there is no horizontal equity in the availability of social, economic, and environmental opportunities between Kumasi and Obuasi. Thus, the potentially higher number of opportunities in Kumasi will lead to higher urbanisation levels as accentuated in the migration statistics in Figure 5.11. An increasing number of migrants, alongside natural population, will subsequently shape landscape urbanisation, as land will have to be converted to meet the basic needs of the increased number of people seeking better opportunities in life. In effect, the population size and hierarchy of Kumasi makes it a net gainer of resources, including institutional and human resources, as compared with Obuasi (see Fielding 1989; Kim and Rowe 2012), and determine the extent of landscape urbanisation in the process.

(b) The role of the market and administrative forces in the context of land.

A leading factor that shapes landscape urbanisation is the role of market forces (Brenner and Theodore 2002; Lee and Zhu 2006). Generally, rapid urbanisation in Ghana has created a significant demand for land. The rapid increase in built-up land cover in Kumasi and Obuasi (shown in Figure 5.9 and Figure 5.10 respectively) reflect this demand. Save for vested State lands, the majority of land in Ghana is held privately by customary landholders for communities or allocated long-term to powerful companies (Aubynn 2006; Tsikata and Yaro 2011). As argued in Section 2.2.4 and Section 4.2.2, land owners will seek to derive maximum economic benefits from their land assets (Nyame and Blocher 2010; Leon 2015). As a result, at any point in time, the market forces determine how land is used (Nyame and Blocher 2010; Leon 2015). The result of the population explosion in Kumasi and Obuasi discussed in the previous paragraphs is that it drives the market to make decisions on land conversion to accommodate the population's basic needs and services. Satisfying these needs could range from making land use decisions to construct houses to the intensification or commercialisation of agriculture in order to meet the food needs of the population. Marketinfluenced land conversion, in this case, can be favourable where it leads to the replacement of, for example, abandoned or unproductive land cover by urban development, while it could be deemed as unfavourable when it results from the encroachment or conversion of fragile land covers like water bodies and wetlands (see Section 4.4.1). In the end, the greater proportion of land use decisions that drives urbanisation are stimulated by the diverse ways in which landowners respond to market prices and policy incentives (Theodore and Brenner 2002; Lee and Zhu 2006; Gales 2016). What this means is that land use decisions by market force actors in Obuasi and Kumasi are important determinants of landscape urbanisation.

Separate from the role of market forces, the State (expressed as administrative and political forces) exerts critical influence on the relationship between urbanisation and LULC change. Although the State does not own the greater portions of land in Ghana, it is responsible for the regulation of land development through the various planning and enforcement instruments at its disposal. Also, it can trigger certain legal instruments such as compulsory purchase orders in order to acquire land from farmers for other uses. Even for the land that the State owns, land conversion from agricultural to urban uses can be a strategic way of attracting investments for much needed industrial growth. This could potentially lead to deagrarianisation in typically agrarian areas. Take the example of Obuasi where the investment decision for the operation of a gold mine resulted in the expropriation of land from farmers and the subsequent direct conversion of natural vegetation into mining land. The indirect rippling effect of the operation of the mine in Obuasi, potentially, is the conversion of natural land cover into urban land cover in order to provide urban infrastructure for the population. Thus, it could be argued that urbanisation-related LULC change is driven by economic and political instruments and incentives emanating from the marketisation of landscape development. In the end, there is a possibility for these instruments to, in turn, undermine the enforcement of land use plans and trigger unplanned and uncontrolled conversion of land.

The explication of factors shaping landscape transformation has shown that LULC change in Kumasi and Obuasi is a path-dependent process that is underpinned by political and administrative processes, socio-economic conditions, and historical decisions. The discussion so far shows that irrespective of planning provisions, the pace and nature of landscape change in Obuasi and Kumasi are shaped by other factors (as discussed above). As expected, differences between Kumasi and Obuasi, in terms of city size, have significant influence on landscape urbanisation. Also, the influence of the market in the context of land tenure and access in most of Ghana is a critical factor for LULC change. In the end, it can be argued that land use decisions which result in an urbanisation-related land cover change in the study sites are stimulated by the actions of landowners and managers in response to the market forces and policy. This implies that if urbanisation (and the resulting LULC change) in the two areas is to proceed sustainably, then externalities such as market forces will have to be effectively

managed. By adopting remote sensing techniques, the major land use and land cover changes in the study areas over three decades have been demonstrated. Essentially, it is evident from the findings that urban development is the single most important factor influencing land cover change in the study areas. Therefore, the pattern and manner of landscape urbanisation will be a reflection of how urban development drives changes in LULC.

5.2.5.4 Sustainability implications of landscape urbanisation.

Under increasing population and economic pressures, it is important to discuss the sustainability implications of the reported LULC changes in Kumasi and Obuasi. Table 2.1 has shown some of the attributes of a sustainable urban form, and the literature reviewed in Chapter Two (see Sections 2.3.3 and 2.4.2.2) has outlined some of the visions of sustainable cities and the sustainability implications of urban sprawl (as a result of landscape urbanisation). Furthermore, realising sustainable development requires that, at a minimum, the basic needs of society must be met (WCED 1987). The findings on sustainable development in Ghana in Chapter Four serve as a pointer for the developmental needs in Ghana as they suggested that while Ghana was making steady progress concerning meeting the basic needs of its citizens, its current development level means that it has not reached threshold levels beyond which it could safeguard the needs of its citizens. The links between urban sprawl and sustainable urban form, and meeting the basic needs of society, thus, become yardsticks for discussing the implications of the urbanisation-induced LULC change for urban sustainability in the study sites.

In general, consistent with the views of scholars (for example, Zeng et al. 2005; Wang et al. 2020), the findings (see Figures 5.9–5.10) suggest that landscape urbanisation (due to expansion of urban land cover) has resulted in urban sprawl in Kumasi and Obuasi. However, despite the reality that the populations of both urban areas expanded by at least three times within the study period, the sprawl was more significant in Kumasi as urban expansion in Kumasi moved beyond the city's 254 km² administrative boundary into adjoining districts, compared with Obuasi where urban expansion was within the administrative boundaries. The sustainability implications of the reported urban sprawl in Obuasi and Kumasi can be explained as follows. Consistent with the literature (for example, Aurand 2007; Abudu et al. 2018), the findings suggest that urban expansion (especially in Kumasi) has resulted in less-compact urban forms and low-density development. This means that urban sprawl (due to the

uncontrolled and poorly-managed urban expansion in Kumasi and Obuasi) has led to the consumption of more land resources than is probably necessary (see for example, Zhao et al. 2014). Also, for Kumasi, the prevailing urban expansion is characteristic of a low-rise sprawl as it can be inferred from the findings (Table 5.5) that the density of development has been low (see for example, Aurand 2007; Abudu et al. 2018). This means that, for example, citizens living in more spread out areas of Kumasi will potentially travel long distances in order to access economic opportunities and urban services in the city centre (see for example, Hamiduddin 2018). This, in turn, potentially leads to air pollution and climate change as higher quantities of pollutants such as GHGs are likely to be produced from urban transport (see for example, Polidoro et al. 2012; Baur et al. 2013). Low density developments also mean that social cohesion is potentially undermined as interactions among citizens are likely to be minimal in low density and scattered neighbourhoods of urban areas (see for example, Aurand 2007; Abudu et al. 2018; Bosehans and Walker 2020).

In terms of meeting basic needs, the transformational effect of urbanisation (as reflected in land cover change) has implications for food security. Using Kumasi as an example, it has already been suggested that there has been an outward expansion of its built-up area beyond administrative boundaries. At the same time from Table 5.1, natural vegetation cover consisting of agricultural land, forest, and mixed woody vegetation had decreased by 60% between 1986 and 2018. This makes it logical to argue that the dramatic expansion of urban land cover in the Kumasi area, for instance, has potentially led to the conversion of prime agricultural lands located within the city and in the previously peri-urban areas surrounding it. Thus, while agriculture might be a spatial concept, it is an important determinant in the urbanisation of previously rural areas in terms of land use change. Therefore, it is logical to assume that agricultural land will continue to be a repository for landscape urbanisation unless it can match the urban development pressure through market-based competition or well-enforced legal protection. This is because urbanisation-induced landscape changes affect agricultural production, thereby making it an important factor in any urbanisation-related conversation. Thus, although the contribution of urban agriculture to the sustenance of the over two million citizens in Kumasi cannot be overstated, its potential contribution to increasing access to food and serving niche markets (like the supply of fresh vegetables) cannot be understated. Taken together, the findings demonstrate that food security, among other relevant socio-economic indicators which have implications for urban sustainability, is intricately linked to LULC change.

The sustainability impacts of land use change due to urbanisation is a double-edged sword in relation to the source of livelihoods (employment) of urban citizens in Kumasi and Obuasi. LULC change due to urbanisation could imply that there is a more beneficial use of land which could improve the source of livelihoods of urban citizens. Take the case of Kumasi, where over two million citizens live, for which the pressure for a source of livelihood is potentially enormous. An increase in urban land use could result in the construction of industries or other forms of job creation, which would serve as sources of livelihoods for citizens. Nevertheless, the earlier-mentioned impact of market-forces means that agricultural land could be expropriated from farmers for other uses (see, for example, Leon 2015; Le Gales 2016). The review of urban challenges in Ghana in Chapter Four alludes to this occurrence based on the suggestion that urban informality in Ghana is fuelled by the dispossession of the rural-poor of their lands, which in most cases, were a source of livelihood and shelter (Obeng-Odoom 2013a). In Ghanaian cities, where there is widening inequality and poverty is pervasive (Obeng-Odoom 2012; Cooke et al. 2016), their plight is potentially worsened where alternative means of livelihoods are lacking (see Section 4.2).

The implications of land cover change for urban sustainability can also be discussed based on alignment with sustainability goals, as explicated through weak or strong sustainability (see, for example, Ekins 2011; Neumayer 2013; Wu 2013). The mere conversion of natural vegetation into urban cover in Kumasi and Obuasi can be interpreted as consistent with sustainability goals when viewed in the context of weak sustainability (see, for example, Ekins 2011; Neumayer 2013; Wu 2013), where natural capital is substituted for human capital. As shown in Tables 5.1 and 5.2, the dominant land cover change for both Kumasi and Obuasi over the study period was the conversion of natural vegetation to urban land cover. As urban land cover encompasses social, economic, and environmental infrastructure required by society, it can be argued that land cover change is sustainable as the multiple benefits of urban infrastructure are assumed to offset natural vegetation loss. For example, population increases recorded in both Kumasi and Obuasi will naturally create demand for additional housing (see, for example, Yakubu et al. 2014), which, more often than not, leads to the clearing of natural vegetation for housing construction. Land cover change, in this instance, can be viewed as consistent with sustainability principles as social benefits are accrued from housing construction. Conversely, if the principle of strong sustainability applies (see, for example, Ekins 2011; Neumayer 2013; Wu 2013), where natural capital cannot be substituted for human capital, the findings on land cover change in the two areas

might be interpreted as inconsistent with the goals of sustainability. As shown in the literature review, the functions and services provided by natural capital (for example, natural vegetation) cannot be replaced by human capital at any costs. Hence, using the same example of housing demand in response to a growing urban population in Kumasi and Obuasi, social and economic gains of land use change for the purposes of housing construction is inconsistent with the tenets of strong sustainability.

The reality, however, is that rapid demographic urbanisation in both Kumasi and Obuasi accentuates a dilemma of sustainable development where the basic needs of society must be met. This means that land will have to be developed to provide these needs, with the consequence being the transformation of vegetation cover. Accordingly, whether the sustainability implications of the land cover change recorded are judged through the lens of weak or strong sustainability, an inevitable reality is that land contains environmental assets that are necessary for the production of goods and services, and at the same time necessary for the maintenance of the integrity of the ecological environment. Therefore, if the principles of sustainable development, which include meeting basic needs of society, including housing, water and food will be achieved, then it is inevitable that a balance between strong and weak sustainability will have to be reached. Nevertheless, from the perspective of this research, the crux of the matter in relation to the sustainability judgment of land cover change in Kumasi and Obuasi based on the findings, hinges on the nature and magnitude of changes in land cover over the study period. From an overall system perspective, the magnitude of land cover changes in Kumasi and Obuasi (as mainly driven by urban development), bears the risk of producing undesirable alterations to the urban form, as well as the environmental and socioeconomic systems in the study sites. Taken together, the factors mentioned above make it safe to argue that land use change (due to the pressures of urbanisation) may be necessary if the minimum requirement of providing the basic needs of urban citizens are to be met. However, the discussion above also demonstrates that unmonitored and uncontrolled land use change potentially threatens sustainability in Kumasi and Obuasi by creating unsustainable urban forms.

5.2.6 Section summary.

In the last few decades, urbanisation and related-anthropogenic activities have led to extensive population growth and land cover changes that have implications for the sustainability of urban regions in Ghana. This section has provided a nuanced understanding of urbanisation in Kumasi and Obuasi and has shown that the process does not merely result in changes in landscape and population, but it is multi-dimensional and involves population, landscape, development, and society. The impact of urban growth and landscape transformation on the urbanisation process in two urban areas in Ghana (Kumasi and Obuasi) has been evaluated. The findings have shown that over a three-decade period, the population in Kumasi had quadrupled while that of Obuasi had tripled. Consistent with the identified drivers in literature, demographic urbanisation was primarily attributed to natural population growth and migration (Cobbinah et al. 2015a; Jedwab et al. 2017). In relative proportions, migration contributed more to urbanisation in Kumasi than it did for Obuasi. Factors that shaped this were identified to include city and administrative size; and market and administrative forces. From the point of view of this study, the relative differences in demographic shifts between the two areas based on the evidence presented are essentially judged as an economic one, as the local economies in the two areas determined their attractiveness for both emigration and immigration within any period. This observation is consistent with the literature (see for example, Ferguson 1998; Davidson and Arman 2014) which suggests that the urbanisation process is fundamentally shaped by the prevailing economy.

The study has produced land cover maps on Kumasi and Obuasi for 1986, 2000, 2010, and 2018 based on images remotely-sensed by sensors on board Landsat 5 (TM), Landsat 7 (ETM+) and Landsat 8 (OLI_TIRS). Accordingly, the dynamics of land cover change in the two areas have been quantified and presented. The following general trends in land cover change were recorded: urban expansion, agricultural land loss and deforestation. Specifically concerning urban expansion, the findings presented have shown that the built-up share of Kumasi's land cover had expanded from 5.3% in 1986 to 62.5% in 2018 (see Section 5.2.1). Spatially, land cover change due to urban expansion in Kumasi was outward from the centre, which meant that adjoining smaller districts and towns have been physically engulfed by Kumasi. In Obuasi, the findings have shown that the built-up share of land cover had expanded from 6.2% in 1986 to 15.2% in 2018 (see Section 5.2.1). Thus, although not in the same magnitude as Kumasi, built-up land cover over the study period.

Aside from the demographic and LULC information reported, the implications of the demographic shift and landscape transformation for sustainability in the study sites have also been considered. Taken together, population dynamics and LULC (in particular urban cover expansion) have varied implications for sustainability in Kumasi and Obuasi. These include the creation of unsustainable urban forms, threats to the capacity to meet basic needs (food insecurity), and population pressures on the capacity to provide urban services and economically engage citizens. The evidence presented in the form of demographic statistics and land cover maps for Kumasi and Obuasi has shown that urban challenges (see Section 4.2) would potentially be exacerbated if urbanisation continues in its present course. While the full implications of the quantum change in LULC and population growth are yet to filter down, what is clear from the findings is that there is a need for a paradigm shift; a complete change in perspective, in order to meaningfully address the impacts of past and contemporary demographic growth and landscape transformation on sustainability in Kumasi and Obuasi. In summary, the findings reported highlight the critical links between LULC, urbanisation, and urban sustainability in the two urban areas studied.

5.3 An assessment of the sustainability implications of urbanisation in Kumasi and Obuasi based on selected socio-economic indicators

A paucity of disaggregated and robust city-level indicator data (for example, on the MDGs and SDGs) for Kumasi and Obuasi means that the analysis on progress in human and sustainable development in this section proceeded with available and accessible data. Therefore, to better situate the data in context, reference will be made to Ghana's HDI and the respective components explored in Chapter Four. An important caveat in interpreting the city level statistics presented, therefore, is that they include very basic indicators which, one could argue, might be crude representations of development in Kumasi and Obuasi. Nevertheless, they are crucially the most widely available data that are useful to explore progress on sustainable development in the study areas.

5.3.1 Education and human development in Kumasi and Obuasi.

5.3.1.1 Literacy levels.

As a measure of a person's ability to read and write, Figure 5.12 presents the literacy levels in Kumasi and Obuasi based on the 2010 census report. From Figure 5.12, in Kumasi, 89.5% of

the total number of 1,303,168 persons who were 11 years and older were literate, with the remaining 10.5% illiterate; compared with Obuasi where 87.4% of the total 124,675 who were 11 years and older were literate, with the remainder illiterate. The findings show that although literacy levels are marginally higher in Kumasi than in Obuasi, on a general level, it is high in both areas.



Figure 5.12: Literacy levels of the population aged 11 years and older in Kumasi and Obuasi.

Source: Ghana Statistical Service (2014a; 2014b; 2014c).

5.3.1.2 Pupil-teacher ratio in Kumasi and Obuasi.

As a measure of the quality of education, the pupil-teacher ratio (PTR) in primary schools in Kumasi and Obuasi in 2012 are presented in Table 5.6. From the table, it is seen that while the PTR in Obuasi (1:22) was lower than the national level of 1:35, the PTR in Kumasi (1:38) exceeded the national level of 1:35. This means that, generally, classrooms were overcrowded in Kumasi compared with Obuasi.

Table 5.6: Pupi	il-teacher	ratio in	Kumasi	and	Obuasi	for	2012.
-----------------	------------	----------	--------	-----	--------	-----	-------

	PTR
National ratio	1:35
Kumasi	1:38
Obuasi	1:22

Source: National Development and Planning Commission (2019).

5.3.1.3 Gender Parity Index in Kumasi and Obuasi.

The Gender Parity Index (GPI) scores for Junior High Schools in Kumasi has been recorded as 0.95 for the 2011 / 2012 academic year, compared with 0.88 recorded in Obuasi (National Development and Planning Commission 2019). A GPI of 1 means that there is an equal number of boys and girls; a GPI of less than 1 means that there are more boys than girls, while a GPI greater than 1 means that there are more girls than boys. This means that the number of girls completing the full basic school cycle was higher in Kumasi than in Obuasi. Reasons attributed to the high GPI in Kumasi include increased sensitisation on female education, while the low GPI in Obuasi has been attributed to reasons that include girls dropping out of school to engage in illegal small-scale mining (see, for example, Azumah et al. 2019).

5.3.2 Life expectancy and human well-being in Kumasi and Obuasi.

The data on overall life expectancy in Ghana has already been presented in Table 4.1 (see Chapter Four). From the table, life expectancy in Ghana increased from 56.8 years in 1990 to 63.8 years in 2018. As a measure of life expectancy at the municipal level, data on aspects of mortality in Kumasi and Obuasi are presented to include Crude Death Rate and Maternal Mortality Rate.

5.3.2.1 Crude Death Rate (CDR).

Table 5.7 shows the Crude Death Rate, which measures the number of deaths in a year per 1000 mid-year population of a specific year, for the two areas. As shown in Table 5.7, CDR in Kumasi in 2010 was 4.8 compared with 4.4 recorded for Obuasi. This means that in the 12 months preceding the census in 2010, approximately five people died per 1,000 persons in Kumasi compared with an approximate number of 4 deaths per 1,000 persons recorded in Obuasi. This means that mortality (expressed herein as total crude death), as a measure of life expectancy, was relatively lower in Obuasi when compared with Kumasi. The CDR in both areas was, however, lower than the regional aggregate of approximately six deaths per 1,000 persons.

Table 5.7: Crude Death Rate (per 1,000 persons) in Obuasi and Kumasi based on total population and household death.

Location	Total Population	Death in Households	Crude Death Rate
Kumasi	1,730,249	8,110	4.7
Obuasi	168,641	778	4.4
Regional Aggregate	4,780,380	27,948	5.8

Source: 2010 Population and Housing Census, Ghana Statistical Service (2014a).

5.3.2.2 Maternal Mortality Rate (MMR) in Kumasi and Obuasi.

Table 5.8 presents the maternal mortality statistics, measured as the ratio of the number of deaths due to pregnancy and childbirth per 100,000 live births, in Kumasi and Obuasi for the 2015-2017 period. From the table, it is seen that MMR in Obuasi (600) was higher in 2015 and 2016 when compared with Kumasi (300). However, by 2017, the MMR in Obuasi had improved to 100 compared with Kumasi which remained constant at 300.

Table 5.8: Maternal mortality rate (MMR) in Kumasi and Obuasi, 2015-2017.

Location	MMR (per 100,000							
	live births)							
	2015 2016 2017							
Kumasi	300	300	300					
Obuasi	600	600	100					

Source: National Development and Planning Commission (2019).

5.3.3 Living standard in Kumasi and Obuasi.

5.3.3.1 Access to improved sanitation in Kumasi and Obuasi.

The reported proportion of urban population with access to improved sanitation between 2014 and 2017 is presented in Table 5.9. From the table, it is seen that access to improved sanitation is generally higher in Kumasi than in Obuasi. For example, in 2014, there was 84.5% access in Kumasi compared with 56% in Obuasi, although both were higher than the recorded average national value of 15% access. While access to sanitation in Obuasi improved to 68% in 2017, it was significantly below the 86% recorded in Kumasi but still higher than the national value of 15%.

Table 5.9: Access to improved	sanitation in	Kumasi	and	Obuasi.
-------------------------------	---------------	--------	-----	---------

	2014	2015	2016	2017
Kumasi	84.5%	84.7%	85%	86%
Obuasi	56%	60 %	62%	68%
National values	15%	15%	15%	15%

Source: National Development and Planning Commission (2019).

5.3.3.2 Sustainable access to safe water sources in Kumasi and Obuasi.

The reported proportion of the urban population with sustainable access to safe water sources in Kumasi and Obuasi between 2014 and 2017 is presented in Table 5.10. From the table, it is seen that in 2014, access to sustainable water sources was significantly higher in Kumasi (97%) when compared with Obuasi (75%), although both Kumasi and Obuasi scored significantly higher than the average national urban value of 55.9%. By 2017, access to safe water had improved significantly in Obuasi to 95% but marginally lower when compared with Kumasi (98%), although both urban areas recorded were significantly higher values than the national urban average of 75%.

		2014	2015	2016	2017
Kumasi		97%	98%	98%	98%
Obuasi		75%	80%	95%	95%
National	urban	55.9%	77.1%	76%	75%
values					

Table 5.10: Sustainable access to safe water sources in Kumasi and Obuasi.

Source: Source: National Development and Planning Commission (2019).

5.3.4 Discussion.

5.3.4.1 Education and human well-being in Kumasi and Obuasi: implications for sustainability.

A notable observation from the findings is the steady expansion of education access, as shown in the national level HDI improvement (see Chapter Four) and demonstrated at the urban-level by the high literacy levels in Kumasi and Obuasi. Table 4.1 (see Chapter Four) sets the context of education progress in Ghana as it shows that there has been a national improvement in the education component of human development (demonstrated by the increase in 'mean years of schooling' and 'expected years of schooling' from 7.6 years and 4.9 years respectively in 1990, to 11.5 years and 7.2 years in 2018). Accordingly, this reported improvement in education is corroborated in this study with evidence on high literacy levels at the national level (national average of 74.1% literacy). At the urban-level, literacy was marginally higher in Kumasi (89.5%) than in Obuasi (87.4%) based on the 2010 census report. Moving on to the quality of education, this study found contrasting evidence for Kumasi and Obuasi based on the pupil-teacher ratio (PTR). The PTR in Kumasi (1:38) was higher than the national average (1:35); compared with Obuasi which had a PTR of 1:22. The findings imply that, consistent with national improvements in education, access to education in Kumasi and Obuasi was high. However, based on the PTR, the findings imply that the quality of education in Kumasi is potentially compromised due to relatively overcrowded classrooms. This observation could be attributed to the impact of population pressure on education infrastructure. This is expected as the population in Kumasi has increased significantly in recent years compared with Obuasi (Ghana Statistical Service 2019). A normative interpretation from the findings includes the coverage of gendered needs in education. For example, as was shown in Section 5.3.1.3, sensitisation on female education

has led to a high proportion of girls who enrolled and completed the basic education cycle in Kumasi as its GPI was 0.95 in the 2011/2012 academic year. Thus, it could be argued that policy responses to promote female education have been effective. However, in the case of Obuasi the low GPI (0.88 in the 2011/2012 academic year) has been attributed to reasons including the engagement of pupils in small-scale mining (see, for example, Azumah et al. 2019). This is consistent with arguments made in Chapter Two concerning how the prevailing economy fundamentally shapes sustainability (see, for example, Davidson and Arman 2014). Nevertheless, these findings show that, consistent with national progress, Kumasi and Obuasi are making progress to meet SDG 4 through inclusive and equitable enrolment and retention of pupils at the basic school level.

In situating the findings on education (in relation to human well-being) within the overall context of this study, the implications of education as a component of sustainability in Kumasi and Obuasi are evident. In this case, one might question whether education functions as a driving force towards achieving sustainable development in the study areas. This is because while the increase in gender parity index and pupil-teacher ratio is in consonance with the improvements and expansion of access to education for the citizens shown in national level HDI, paradoxically, it will appear, based on the Ghanaian urban challenges identified in Section 4.2, that this has not translated into reduction in aspects of development such as unemployment or inequality. Nevertheless, as was shown in Chapter Four, education plays a prominent role in promoting sustainable development (see Pigozzi 2007). Subsequently, in Section 4.3.1, Sen (1999, p.295) has argued that "a person may benefit from education through reading, communicating, arguing, in being able to choose in a more informed way, in being taken more seriously by others and so on. The benefits of education, thus exceed its role as human capital". This precise notion of education being a key panacea to the sustainable development challenges in Ghana is also shaped by Ghana's development and education policies expressed in Section 4.3.1. For example, the Free Compulsory Universal Basic Education (FCUBE) policy introduced in 1995 (Acheampong 2009) and the Free Senior High School Policy introduced in 2017 (Arthur-Mensah and Alagaraja 2018), stress the relevance of education to the SD journey of Ghana and recognise how the challenge of poverty undermines the achievement of universal education as, for example, most parents were unable to afford the cost of education (Alderman and Brundy 2011). Thus, in congruence with the views already captured in the literature (for example, Sen 1999), the argument advanced by this study (based on the national and urban level findings) is that the

sustainability implications of human development through education in the study areas is not limited to labour market returns. To the extent that sustainable development in Ghana is fundamentally constrained by poverty and limited infrastructure (Domfeh et al. 2012), improvements in education in the study areas (as demonstrated by the improved pupil-teacher ratio) could potentially improve the sustainable development situation. Finally, as will be shown in Section 5.4, from the angle of citizens' attitudes and commitment to prosustainability behaviours, there is a clear role for development education in providing opportunities for citizens to be able to make informed decisions about how they can contribute to sustainable development through positive attitudes and practices. In sum, it is argued that while education in Kumasi and Obuasi (as represented by findings of high literacy levels and high pupil-teacher ratios shown in Table 5.6 and Figure 5.12) cannot exclusively achieve sustainable development in these areas, the findings demonstrate the fundamentally important enabling role that education plays (in coordination with other socioeconomic processes) to equip citizens with the skills necessary to play active roles in the economy, and in turn, stimulate socio-economic growth thereby supporting the quest for broader sustainable development in the study areas (Sen 1999).

5.3.4.2 Life expectancy and human development in Kumasi and Obuasi: implications for sustainability.

Healthy citizenry is central to achieving environmental, economic, and social development. The values of 'Crude Death Rate' in the study sites based on the 2010 census report (as shown in Table 5.7) suggest that mortality in Obuasi (4.4 deaths per 1,000 persons) was lower than in Kumasi (4.7 deaths per 1,000 persons), although both were below the regional average of 5.8 deaths per 1,000 persons. With respect to maternal mortality, it can be inferred from the findings in Table 5.8 that the MMR in Kumasi remained constant at 300 deaths per 100,000 live births from 2015 to 2017; compared with Obuasi where maternal mortality was 600 deaths per 100,000 live births from 2015 to 2015 to 2016, reducing to 100 deaths per 100,000 live births in 2017. These findings show that there is a clear stratification of mortality dimensions based on the size of the study sites and suggests that the general health in the population was better in Obuasi than in Kumasi. As MDG 5 considers maternal health, the very high maternal mortality rates in Kumasi and Obuasi in 2015, at 300 and 600 respectively, mean that both cities could not achieve the MDG target of 185 / 100,000 live births. With respect to achieving the SDG 3 goal of 'ensuring healthy lives and promoting

well-being for all at all ages', which pushes for an MMR target of less than 70 deaths per 100,000 live births by 2030, the latest figures in 2017 suggest that, all things being equal Obuasi (100 / 100,000), is well placed to meet the SDG target on maternal health compared with the unacceptably high MMR of 300 / 100,000 in Kumasi (as of 2017). Whatever the case, strong and sustained efforts will be needed to enhance maternal health in Kumasi and Obuasi.

While debating the reasons behind the state of health in Kumasi and Obuasi (as inferred from the statistics presented) is beyond the scope of this study, an undeniable reality is that rapid urbanisation and population growth will shape the contemporary and future health possibilities in these cities, and subsequently determine sustainable development progress in the two cities. Thus, consistent with the literature, urban advantage in health service provision (Vlabov et al. 2005) might enhance maternal health as the better facilities in Obuasi and Kumasi (compared with rural areas) could mean that mothers and new-borns have access to modern healthcare facilities, well-trained personnel, and reduced travel times to access these facilities, thereby potentially reducing maternal mortality. Alternatively, reported impacts of inequality in health access in urban areas (Matthews et al. 2010) might also imply that current poor performance in aspects such as maternal mortality in both Kumasi and Obuasi will be exacerbated by contemporary and current rapid urbanisation, thereby impeding sustainable development progress.

5.3.4.3 Living standards in Kumasi and Obuasi: implications of access to improved water and sanitation in Kumasi and Obuasi for sustainability.

Water and sanitation are critical to the sustainable development agenda (United Nations 2019). The findings showed that access to improved water and sanitation services in the urban areas studied were generally higher than the national average. Table 5.9 showed that between 2014 and 2017, Kumasi maintained relatively high levels of access to improved sanitation compared with Obuasi. However, within the same period (2014–2017), Kumasi and Obuasi scored higher than the national average score. Both Kumasi and Obuasi, therefore, exceeded the MDG 7 target of halving the population without access to improved sanitation facilities. Similarly, the findings presented in Table 5.10 showed that Kumasi recorded near-universal sustainable water access at 97% in 2014, rising marginally to 98% in 2017, compared with Obuasi, where an access level of 75% was recorded in 2014, rising
substantially to 95% in 2017. Again, these were above the national levels of 55.9% and 75% recorded in 2014 and 2017 respectively. This means that consistent with the national performance, Kumasi and Obuasi exceeded the MDG 7C target of halving the proportion of people without sustainable access to safe drinking water. An important caveat, however, is that these figures may have been adjusted to serve political ends (see, for example, Weststrate et al. 2019).

Considering SDG 6 'ensuring the availability and sustainable management of water and sanitation for all', a critical question in relation to the current discussion, however, is the extent to which the noteworthy progress reported for both cities with respect to the MDGs enhances their chances of meeting SDG 6. It could be argued that the challenge on the possibility of Kumasi and Obuasi to meet SDG 6 lies in the difference between the cities' performance and the national performance. Consistent with the idea that people move to cities in search of better opportunities for life (Njoh 2003); and taking cognisance of the rapid pace of population growth and the migrant proportion of the population in Kumasi and Obuasi (see Section 5.2), there is a likelihood that rapid urbanisation (due to natural population growth and migration) will lead to the concentration of populations in these areas and potentially overwhelm the capacity to provide urban services (Farvacque-Vitkovic et al. 2008; Okeke 2014). This, from the perspective of this study, creates an impression that rapid urbanisation in Kumasi and Obuasi creates intractable impacts that undermine the achievement of SDG 6. In sum, the findings and discussions so far highlight the critical dynamics between water and sanitation service provision and population growth, and stress the need for city authorities to constantly invest in new water and sanitation infrastructure while maintaining existing ones in order to keep to up with the rapid pace of urbanisation.

5.3.5 Section summary.

This section has demonstrated the variations in aspects of human and sustainable development in Kumasi and Obuasi as reflected through comparisons of indicators on education, health and living standards. Consistent with the overall national improvements in human development (see Chapter Four), the findings (based on data available) have shown that both Kumasi and Obuasi have experienced significant improvements in education (literacy rates, PTR, GPI), living standards (access to water and sanitation), and health (life expectancy); which, from the perspective of this study, have become dominant contributors

to overall improvements towards sustainability in the study areas. The findings are, therefore, consistent with suggestions in the literature (for example, Gong et al. 2012; Turok and McGranahan 2013) about how urbanisation potentially leads to improvements in living standards and increased access to utilities. Essentially, the findings and discussions generated a compelling narrative of how education is integral to the sustainable development of Kumasi and Obuasi, and how sustainable development in the two areas is potentially undermined by structural constraints, including poverty and inequality. In sum, the findings have shown that while the recent improvements in human development and well-being in Ghana, as represented by the HDI (See Chapter Four), has led to urban level (Kumasi and Obuasi) improvements in education, living standards, and health, the rapid pace of urbanisation in Kumasi and Obuasi could undermine the achievement of sustainable development targets in the two areas as conventional planning cannot keep pace with contemporary and future population growth. A key challenge will, therefore, be to recognise the bi-directional relationship between sustainability and urbanisation in order to provide adaptive solutions to the sustainability challenges in Obuasi and Kumasi due to rapid urbanisation.

5.4 An analysis of citizens' perceptions on dimensions of and attitudes towards urban sustainability in Kumasi and Obuasi

5.4.1 Demographic summary.

A valid total of 620 (out of 624) questionnaires administered in Kumasi (n = 318) and Obuasi (n= 302) were analysed. The summary of the demographic variables is shown in Table 5.11. The table shows that a small majority of respondents in Kumasi and Obuasi were female, while the rest were male. A greater number of respondents in Kumasi (39.0%) and Obuasi (42.7%) had been residents for more than ten years. The proportion of 'unemployed' respondents was approximately 40% for both Kumasi and Obuasi, with the remainder either 'employed with wage' or 'self-employed'. The proportion of respondents without formal education was 2.5% in Kumasi and 3.6% in Obuasi, with the remainder educated up to tertiary level. These numbers are comparable to the demographic distributions recorded in the 2010 Census for the study sites (Ghana Statistical Service 2014b; 2014c).

and Obuasi.	[
Category	Location			
Gender	Kumasi	%	Obuasi	%
Female	172	54.10	157	52.00
Male	146	45.90	145	48.00
Age (years)	Kumasi	%	Obuasi	%
18 - 30	139	43.70	126	41.70
31 - 45	113	35.50	97	32.10
46 - 60	51	16.00	57	18.90
> 60	15	4.70	22	7.30
Length of	Kumasi	%	Obuasi	%
Residence				
< 2	0	0	0	0
2 - 5	80	25.20	65	21.50
6 - 10	114	35.80	108	35.80
> 10	124	39.00	129	42.70
Level of	Kumasi	%	Obuasi	%
Education				
None	8	2.50	11	3.60
Primary	101	31.80	141	46.70
Secondary	128	40.30	95	31.50
Tertiary	81	25.50	55	18.20
Employment	Kumasi	%	Obuasi	%
Status				
Employed	47	14.80	51	16.90
with wage				
Self-	145	45.60	131	43.40
employed				
Unemployed	126	39.60	120	39.70
/ Retired				

Table 5.11: Demographic summary of questionnaires (valid) administered in Kumasi

5.4.2 Perceptions of the dimensions of sustainability and urbanisation.

5.4.2.1 Citizens' perceptions of economic sustainability statements.

The criteria for citizens' perceptions on economic sustainability included in the survey were income levels, cost of goods and services, and ease of employment. Table 5.12 shows the distribution of responses given to the statements on economic sustainability. The dominant observation made from Table 5.12 is that the perception of respondents on economic sustainability in both Kumasi and Obuasi is negative. For example, in terms of ease of finding a job (Q1), an overwhelming majority of respondents (Kumasi=84.5%; Obuasi=88.7%) disagreed or strongly disagreed with the statement.

 Table 5.2: Distribution of responses for perceptions on economic sustainability statements.

Statement	Ranking	Kumasi	Obuasi
		(%)	(%)
Q1. It is easy to find a job.	Strongly disagree	24.5	20.5
	Disagree	60.1	68.2
	Undecided	13.8	7.9
	Agree	1.6	3.3
	Strongly agree	0	0
Q2. The cost of goods and services are	Strongly disagree	20.1	15.6
acceptable (compared to elsewhere in	Disagree	52.2	32.1
Ashanti).	Undecided	21.4	50.3
	Agree	6.3	2.0
	Strongly agree	0	0
Q3. Income levels are adequate (compared to	Strongly disagree	6.6	9.6
elsewhere in Ashanti).	Disagree	25.5	51.3
	Undecided	21.4	25.5
	Agree	44.0	12.6
	Strongly Agree	2.0	1.0

5.4.2.2 Citizens' perceptions of social sustainability statements.

Respondents were asked about their perceptions on statements which centred on the adequacy of access to education, healthcare, transport, housing, as well as their views on the level of public safety. Table 5.14 shows the distribution of responses given to the statements on social sustainability. Responses were mixed; however, notable observations were made. For example, 73.2% of respondents in Obuasi agreed or strongly agreed that public safety levels were acceptable, compared with 7.2% in Kumasi. Also, compared to variables such as healthcare (Kumasi=13.8%, Obuasi=12%) and housing (Kumasi=9.7%, Obuasi=13.6%) where less than 15% of respondents in both locations agreed or strongly agreed, a significant proportion of respondents (Kumasi=71.1%, Obuasi=36.0%) agreed or strongly agreed with the statement on access to education (Q4).

Statement	Ranking	Kumasi	Obuasi
		(%)	(%)
Q4. Access to education is adequate.	Strongly disagree	0.6	3.0
	Disagree	14.2	40.1
	Undecided	14.2	21.5
	Agree	60.4	30.5
	Strongly agree	10.7	5.5
Q5. Access to healthcare is	Strongly disagree	13.2	13.6
adequate.	Disagree	49.7	38.7
	Undecided	23.3	35.8
	Agree	13.8	11.3
	Strongly agree	0	0.7
Q6. Access to housing is adequate.	Strongly disagree	15.1	7.0
	Disagree	56.0	48.3
	Undecided	18.2	31.1
	Agree	9.7	13.6
	Strongly agree	0.9	0
Q7. Public transport is widely	Strongly disagree	1.9	2.0
available.	Disagree	38.4	15.2
	Undecided	19.8	8.3
	Agree	39.6	60.6
	Strongly agree	0.3	13.9
Q8. Public safety levels are	Strongly disagree	20.1	7.6
acceptable.	Disagree	56.9	12.6
	Undecided	15.7	6.6
	Agree	6.9	31.1
	Strongly Agree	0.3	42.1

Table 5.3: Distribution of responses for perceptions on social sustainability statements.

5.4.2.3 Citizens' perceptions of environmental sustainability statements.

Respondents were asked about their perceptions on statements which centred on the adequacy of access to potable water, sanitation, and green spaces, as well as their views on

air pollution and recycling. Table 5.15 shows the distribution of responses given to the statements on environmental sustainability. Responses were mixed; however, notable observations were made. For example, a tiny proportion of respondents (Kumasi=7.2%, Obuasi=4.0%) agreed or strongly agreed on the acceptability of air pollution levels (Q9); however, as an indication of the degree of seriousness of the situation in either place, 9.1% of respondents in Kumasi strongly disagreed compared with Obuasi where a significantly higher number of respondents (32.1%) strongly disagreed. Similarly, there were disparities in responses from Kumasi and Obuasi with respect to access to potable water (Q12). For example, a combined 5.6% (Obuasi) and 33.6% (Kumasi) expressed agreement or strong agreement with the statement on access to potable water, compared with 79.8% (Obuasi) and 46.8% (Kumasi) who expressed disagreement or strong disagreement. The prominence of the disparity between the two cities is the level of strong disagreement, where 0.3% of respondents in Kumasi expressed strong disagreement compared with 22.5% in Obuasi.

Statement	Ranking	Kumasi	Obuasi
		(%)	(%)
Q9. The level of air quality or pollution is	Strongly disagree	9.1	32.1
acceptable.	Disagree	55.7	55.3
	Undecided	28.0	8.6
	Agree	7.2	4.0
	Strongly agree	0	0
Q10. The amount of accessible green and	Strongly disagree	8.2	26.5
public spaces is adequate.	Disagree	39.6	36.1
	Undecided	34.6	32.5
	Agree	17.6	5.0
	Strongly agree	0	0
Q11. The levels of sanitation service provision	Strongly disagree	19.6	18.9
and general cleanliness are adequate.	Disagree	59.6	46.4
	Undecided	13.6	32.5
	Agree	7.2	2.3
	Strongly agree	0	0
Q.12. There is good access to potable water.	Strongly disagree	0.3	22.5
	Disagree	46.5	57.3
	Undecided	19.2	14.6
	Agree	33.3	5.6
	Strongly agree	0.3	0
Q.13. Recycling and re-use of waste materials	Strongly disagree	0.3	0.3
are important to save the environment.	Disagree	0.9	1.3
	Undecided	12.9	10.6
	Agree	68.9	70.9
	Strongly agree	17.0	16.9

Table 5.4: Distributions of responses for perceptions on environmental sustainability statements.

5.4.2.4 Citizens' perceptions of urbanisation statements.

The respondents were asked about their perceptions on statements which centred on the urbanisation process, whether it is commensurate with the level of development, how it has affected their access to socio-economic opportunities, and whether they were content with being in their current city. Table 5.16 shows the distribution of responses given to the statements on urbanisation. Responses were mixed; however, notable observations were made. There was a general awareness among respondents about the urbanisation process as over 45% of respondents in both Kumasi and Obuasi agreed or strongly agreed with the statement on the rapidly changing urban landscape (Q14); while over 60% of respondents in both areas did not think that urbanisation rates were commensurate with the level of development (Q15). With respect to access to socio-economic opportunities (Q16), a combined 49.7% (Kumasi) and 54.0% (Obuasi) of respondents disagreed or strongly disagreed, and at the same time, a combined 24.4% (Kumasi) agreed to the statement compared with Obuasi where 9% of respondents agreed.

Statement	Ranking	Kumasi	Obuasi
		(%)	(%)
Q.14 The city's population and landscape are	Strongly disagree	0	2.0
changing rapidly.	Disagree	7.2	27.8
	Undecided	16.4	25.2
	Agree	52.2	31.5
	Strongly agree	24.2	13.6
Q15. The rate of urbanisation is commensurate	Strongly disagree	10.4	11.3
with the level of development.	Disagree	59.1	50.3
	Undecided	26.4	37.4
	Agree	3.8	1.0
	Strongly agree	0.3	0
Q16. There is great access to economic	Strongly disagree	9.4	9.6
opportunities, essential infrastructure and	Disagree	40.3	45.4
services from where I live in this city.	Undecided	29.9	35.1
	Agree	20.4	9.9
	Strongly agree	0	0
Q.17 I am happy to live in this city and I do	Strongly disagree	9.1	6.3
not seek to migrate to another place in this	Agree	28.6	35.8
country.	Undecided	27.4	23.2
	Agree	28.3	27.8
	Strongly agree	6.6	7.0

Table 5.5: Distribution of responses for perceptions on urbanisation statements.

5.4.3 Attitudes towards urban sustainability.

In order to explore the roles and opportunities for promoting urban sustainability in the context of urban citizens, the attitudes of respondents, in terms of willingness to commit to widely accepted practices that promote sustainability, such as waste recycling, patronising public transport, and resource (water, energy) conservation, were examined. The results presented in Figure 5.13 and Figure 5.14 show that there is a clear trend of positive attitudes towards pro-sustainability practices amongst respondents in both Kumasi and Obuasi. The results showed that aside from the 'willingness to recycle waste' category where respondents

in Obuasi showed greater relative willingness, the respondents in Kumasi were marginally more willing to commit to pro-sustainability behaviour in the other categories. For example, in Kumasi, 80.1% (n=255) of respondents indicated their willingness to recycle, with the remaining 19.9% (n=63) selecting 'maybe' as an answer; compared with Obuasi where 90% (n=272) answered yes, 9% (n=27) answered 'maybe' and the remaining 1% (n=3) answered 'no'. For the other categories, 81% (n=257) answered yes, 14% (n=45) answered maybe, and 5% (n=6) answered 'no' in Kumasi, to the question of whether they were willing to patronise public transport; compared with Obuasi where 81% (n=239) answered 'yes', 14% (n=51) answered 'maybe', and 5% (n=12) answered 'no'. Also, with respect to water conservation, 93.4% (n=297) answered 'yes', and the remaining 6.6% (n=21) answered 'maybe' in Kumasi, compared with Obuasi where 91.4% (n=276) answered 'yes', and the remaining 8.6% (n=26) answered 'maybe'. Finally, with respect to energy conservation, 92.1% (n=293) answered 'yes', and the remaining 7.9% (n=25) answered 'maybe' in Kumasi, compared with Obuasi where 86.8% (n=262) answered 'yes', 12.9% (n=39) answered 'maybe' and the remaining 0.3% (n=1) answered 'no'. Despite the differences, the result showed that overall, there is a positive attitude towards pro-sustainability behaviour in both areas.



Figure 5.13: Attitudes towards urban sustainability practices in Kumasi.



Figure 5.14: Attitudes towards urban sustainability practices in Obuasi.

5.4.4 Group differences in sustainability perception.

5.4.4.1 Gender differences in citizens' perceptions of sustainability statements.

The Mann-Whitney U test results for statements which fulfilled test assumptions (see Chapter Three) for gender differences in citizens' responses in Kumasi are summarised in Table 5.17. The table details the mean rank scores, U, Z and p-values produced by the Mann-Whitney U-test and the decision reached on whether to reject or accept the H_o as evidence of gender difference association. From Table 5.17, statistically-significant interactions in gender differences were found for Q5, Q11 and Q15. Specifically, the Mann-Whitney U-test showed that perceptions on Q5 (Access to healthcare is adequate) were greater for females (mean rank=169.79) than for males (mean rank=147.38), U=10786.00, z=-2.336, p=0.019. For Q11 (The levels of sanitation service provision and general cleanliness are adequate), the Mann-Whitney U-test showed that perceptions on Q11 were greater for females (mean rank=168.50) than for males (mean rank=147.73), U=10836.00, z=-2.279, p=0.023. For Q15 (The rate of urbanisation is commensurate with the level of development), the Mann-Whitney U-test showed that perceptions on Q15 were greater for females (mean ran=168.65) than for males (mean rank=148.73), U=10893.00, z=-2.189, p=0.029. Therefore, the null hypothesis (H_o) is rejected for Q5, Q11, and Q15. No statistically significant interactions were found for males (mean rank=148.73), U=10893.00, z=-2.189, p=0.029.

gender difference in the other perception statements in Table 5.17; therefore, H_0 was accepted for all of them.

Statement	Mear	n Rank	Test Statis	Test Statistic		
	Male	Female	U	Ζ	Р	
<i>Q1</i>	152.44	165.49	11525.00	-1.442	0.149	Accept
Q2	154.70	163.58	11855.00	-0.936	0.349	Accept
Q5	147.38	169.79	10786.00	-2.336	0.019	Reject
Q6	156.34	162.18	12095.00	-0.625	0.532	Accept
Q8	158.00	160.77	12337.00	-0.299	0.765	Accept
Q9	160.21	158.90	12452.00	-0.142	0.887	Accept
Q11	147.73	168.50	10836.00	-2.279	0.023	Reject
Q12	156.25	161.32	12071.00	-0.531	0.595	Accept
Q15	148.73	168.65	10893.00	-2.189	0.029	Reject
Q16	154.97	163.35	11894.50	-0.854	0.393	Accept

Table 5.6: Mann-Whitney U test results for responses in Kumasi

The Mann-Whitney U test results for gender differences in citizens' responses to statements which fulfilled the test's assumptions (see Chapter Three) in Obuasi are summarised in Table 5.18. From Table 5.18, although there were differences in the mean rank scores for males and females, no statistically significant interactions were recorded. Hence, H_0 was accepted for all the statements.

Statement	Mea	an Rank	Test Stat	Fest Statistic			
					Decision		
	Male	Female	U	Ζ	Р		
<i>Q1</i>	148.94	153.86	11011.55	-0.596	0.551	Accept	
Q5	144.58	157.89	10378.50	-1.402	0.161	Accept	
Q6	149.14	153.68	11040.50	-0.488	0.625	Accept	
Q9	142.56	153.68	10086.00	-1.1915	0.055	Accept	
Q12	142.98	159.37	10147.00	-1.825	0.068	Accept	
Q16	157.88	145.61	10458.00	-1.314	0.189	Accept	

Table 5.7: Mann-Whitney U test results for responses in Obuasi.

5.4.4.2 Sustainability perception by economic (employment) status.

The Kruskal-Wallis H tests for economic group differences in Kumasi are summarised in Table 5.19. Statistically-significant differences based on economic status were found in only two variables out of the total of nine which fulfilled the assumptions (see Chapter Three) of the Kruskal-Wallis H test for which H_o was rejected. These variables were Q1 (It is easy to find a job) and Q2 (The cost of goods and services are acceptable, compared to elsewhere in Ashanti). For Q1, a Kruskal Wallis H test found a statistically significant difference between economic groups, X₂ (2) = 40.694, p <0.001, with mean ranks of 188.07 for 'Employed with wage', 181.01 for 'Self-employed' and 124.09 for 'Unemployed'. For Q2, the Kruskal Wallis H test found significant difference between group responses, X₂ (2) =11.191, p=0.004, with mean ranks of 153.97 for 'Employed with wage', 176.33 for 'Self-employed' and 142.48 for 'Unemployed'.

Statement	Mean Ran	k		Test Statistic		Ho
						Decision
	Employed with wage	Self-employed	Unemployed	$H(X^2)$	P	
Q1	188.07	181.01	124.09	40.694	< 0.001	Reject
Q2	153.22	176.33	142.48	11.191	0.004	Reject
Q5	156.97	170.02	148.34	4.407	0.110	Accept
Q6	148.11	163.35	159.32	1.119	0.549	Accept
Q8	162.80	164.49	152.52	1.511	0.470	Accept
Q9	157.09	162.01	157.51	0.249	0.883	Accept
Q11	157.50	167.12	150.20	2.972	0.226	Accept
Q12	154.18	163.63	155.51	0.795	0.672	Accept
Q15	145.46	168.12	154.82	3.487	0.175	Accept

Table 5.8: Kruskal-Wallis H test results for group differences (economic) in responses in Kumasi.

Further comparisons were made to ascertain whether the differences in perception due to economic status were polarised towards certain economic groups. These are attached in Appendix 7. As the findings show, in Kumasi, no statistically-significant relationships were found for group differences in perceptions between respondents who were 'Employed with wage' and 'Self-employed' respondents in relation to Q1 (X_2 [1] = 0.209, p=0.648) and Q2 (X_2 [1] = 2.463, p=0.117), while statistically significant relationships were found for differences in perceptions between 'Self-employed' and 'Unemployed' in relation to Q1 (X_2 [1] = 32.688, p=<0.001) and Q2 (X_2 [1] = 10.918, p=0.001). For the difference between 'Employed with wage' and 'Unemployed', statistically significant relationships were found for Q1 (X_2 [1] = 22.393, p < 0.001), while no statistically significant relationships were found for Q2 (X_2 [1] = 0.565, p = 0.452). This shows that even within the groups, economic group effects were more influenced by the perceptions of the 'Unemployed' group in Kumasi.

Table 5.20 summarises the Kruskal-Wallis H test for economic group differences in Obuasi. The findings showed group differences regarding two variables only. These variables were Q1 (It is easy to find a job) and Q16 (There is great access to economic opportunities, essential infrastructure and services from where I live in this city). For Q1, a Kruskal Wallis H test found a statistically significant difference between economic groups, X^2 (2) = 38.487, p <0.001, with mean ranks of 202.38 for 'Employed with wage', 153.06 for 'Self-employed and 128.18 for 'Unemployed'. For Q16, the Kruskal Wallis H test found a significant difference between group responses, $X^2(2) = 10.599$, p= 0.005, with mean ranks of 162.89 for 'Employed with wage', 164.19 for 'Self-employed' and 132.81 for 'Unemployed'. Thus, H_o was rejected for Q1 and Q16 and accepted for the remaining variables.

Statement	Mean Rank			Test Sta	Ho	
						Decision
	Employed with	Self-	Unemployed	$H(X^2)$	Р	
	wage	employed				
<i>Q1</i>	202.38	153.38	128.18	38.487	<0.001	Reject
Q5	165.04	154.18	142.82	2.843	0.241	Accept
Q6	157.97	154.79	145.16	1.287	0.526	Accept
Q9	151.69	147.43	155.86	0.733	0.693	Accept
Q12	144.50	146.06	160.41	2.616	0.270	Accept
Q16	162.89	164.19	132.81	10.599	0.005	Reject

Table 5.9: Kruskal-Wallis H test results for group differences (economic) in responses in Obuasi.

Further comparisons were made to ascertain whether the differences in perception due to economic status were polarised towards specific economic groups. These are attached in Appendix 8. As the findings show, in Obuasi, statistically-significant relationships were found for group differences in perceptions between respondents who were 'Employed with wage' and 'Self-employed' in relation to Q1 (X_2 [1] = 19.400, p<0.001), while no statistically significant relationship was found for Q16 (X_2 [1] = 0.037, p=0.847). Statistically significant relationships were found for differences in perceptions between 'Self-employed' and 'Unemployed' respondents in relation to Q1 (X_2 [1] = 9.083, p=0.003) and Q2 (X_2 [1] = 8.948, p=0.003). For the difference between 'Employed with wage' and 'Unemployed', statistically significant differences were found for Q1 (X_2 [1] = 31.772, p<0.001) and Q16 (X_2 [1] = 5.504, p=0.019). Similar to Kumasi, results show that economic group effects were more influenced by the perceptions of the 'Unemployed' group in Obuasi.

5.4.4.3 Sustainability perception by age (years) of respondents.

The Kruskal-Wallis H tests for differences in responses based on age (years) group in Kumasi and Obuasi are summarised in Table 5.21. and Table 5.23 respectively. For Kumasi, it is seen (Table 5.21) that a statistically significant relationship was only found for Q17 (I am happy to live in this city and I do not intend to migrate to another place in this country) out of all the statements which fulfilled the assumptions of the Kruskal-Wallis H test. As a result, Ho was rejected for Q17 and accepted for the other statements in Table 5.21. In order to determine whether differences in perceptions on Q17 (based on age) were influenced by certain age groups in Kumasi, the groups were paired and the Kruskal-Wallis H test was repeated for paired age group interactions. The results (see Appendix 9) showed statistically significant interactions in all the age group pairings. In order to further ascertain which age groups contributed most to the differences in perceptions for Q17, the effect size (= H / n-1), which shows the magnitude of interactions, was calculated (Table 5.22). From Table 5.22, the largest effect size (0.182) for Q17 in Kumasi emerged from the interactions between '18 -30' and '> 60' age groups, while the smallest effect size (0.039) emerged from the interactions between '18-30' and '31-45' age groups. This indicates that the interactions between the youngest age group ('18 -30') and the eldest age group ('> 60') contributed more (18.2%) to the differences in perceptions due to age for Q17 in Kumasi. As shown in Table 5.23, H_o was accepted for all the statements tested based on age group in Obuasi because no statistically significant interactions were found.

Statement	Mean Ra	nk			Test Statistic		H _o Decision
	18-30	31-45	46-60	> 60	$H(X^2)$	P	
Q2	157.36	163.60	155.07	163.50	0.532	0.912	Accept
Q3	150.12	163.72	171.44	174.07	3.292	0.349	Accept
Q5	157.51	173.82	143.93	123.03	7.709	0.052	Accept
Q6	157.91	158.73	170.62	142.27	1.624	0.654	Accept
Q7	163.71	157.46	144.86	185.60	3.261	0.353	Accept
Q8	159.45	161.31	157.38	153.60	0.165	0.983	Accept
Q9	155.39	167.07	158.65	143.47	1.870	0.600	Accept
Q11	151.65	172.71	154.80	139.00	5.427	0.143	Accept
Q12	156.81	167.83	156.61	121.47	4.299	0.231	Accept
Q13	159.60	158.18	150.11	200.50	5.310	0.150	Accept
Q14	145.14	170.22	168.03	182.80	7.542	0.056	Accept
Q15	160.27	169.57	144.72	126.73	5.931	53.549	Accept
Q17	128.44	161.80	208.02	265.10	0.115	< 0.001	Reject

Table 5.10: Kruskal-Wallis H test results for differences in responses based on age (years) group in Kumasi.

Table 5.11: Effect size for age group differences in perceptions in Kumasi for Q17.

Group Interaction	Effect Size
'18 – 30' and '31 – 45'	0.039
'18 – 30' and '46 – 60'	0.155
'18 – 30' and '> 60'	0.182
'31 − 45' and '46 − 60'	0.065
'31 − 45' and '> 60'	0.175
'46 − 60' and '> 60'	0.119

Statement	Mean Ra	nk			Test Statistic		H _o Decision
	18-30	31 - 45	46-60	> 60	$H(X^2)$	P	
Q1	147.96	147.29	161.41	164.64	2.471	0.480	Accept
Q3	155.23	150.16	148.07	144.95	0.549	0.908	Accept
Q6	165.51	145.23	136.65	149.52	6.984	0.072	Accept
Q7	155.17	140.10	163.54	149.52	3.854	0.278	Accept
Q8	148.97	144.77	162.66	166.73	2.555	0.465	Accept
Q9	147.74	154.65	150.47	161.80	0.845	0.839	Accept
Q10	156.34	150.11	151.90	128.86	2.101	0.552	Accept
Q11	148.51	146.69	159.16	170.02	2.177	0.537	Accept
Q12	155.08	136.27	171.11	147.34	7.630	0.054	Accept
Q13	145.33	159.05	154.80	144.98	2.441	0.486	Accept
Q14	142.30	165.75	149.11	147.59	4.380	0.223	Accept
Q15	153.56	142.90	169.06	132.16	5.369	0.147	Accept
Q16	146.06	147.80	156.63	185.68	4.911	0.178	Accept

Table 5.12: Kruskal-Wallis H test results for differences in responses based on age group in Obuasi.

5.4.4.4 Sustainability perception by level of education of respondents.

The Kruskal-Wallis H tests for differences in responses based on level of education in Kumasi and Obuasi are summarised in Table 5.24 and Table 5.25 respectively. The tables show that there were no statistically-significant interactions in responses (based on level of education) for all the statements which fulfilled the assumptions of the Kruskal-Wallis H test in Kumasi and Obuasi. As a result, H_0 was accepted for all the statements in Tables 5.24 -25.

Statement	Mean Rank				Test Statistic		H _o Decision
	None	Primary	Secondary	Tertiary	$H(X^2)$	P	
Q1	123.56	168.04	153.27	162.24	3.596	0.309	Accept
Q3	192.19	149.68	160.98	166.18	2.955	0.399	Accept
Q5	150.50	167.36	156.07	156.01	1.291	0.731	Accept
Q6	152.88	171.54	149.62	160.75	4.012	0.260	Accept
Q7	160.75	169.68	152.81	157.25	2.251	0.522	Accept
Q9	174.88	158.88	158.88	160.17	0.305	0.959	Accept
Q10	174.50	146.33	167.78	161.35	3.769	0.288	Accept
Q12	148.63	163.70	155.54	159.58	0.647	0.886	Accept
Q13	172.06	151.47	166.27	157.57	2.476	0.480	Accept
Q14	156.94	170.41	157.48	149.35	2.953	0.399	Accept
Q16	126.56	152.05	174.78	147.90	7.247	0.064	Accept
Q17	148.75	168.99	158.83	149.79	2.246	0.523	Accept

Table 5.13: Kruskal-Wallis H test results for differences in responses based on level of education in Kumasi.

Table 5.14: Kruskal-Wallis H test results f	or differences in	responses based	on level of
education in Obuasi.			

Statement	Mean Rank				Test Statistic		H _o Decision
	None	Primary	Secondary	Tertiary	$H(X^2)$	Р	
Q5	134.64	149.12	151.01	161.82	1.441	0.696	Accept
Q6	166.09	142.71	154.59	165.78	3.896	0.273	Accept
Q7	164.09	155.39	154.22	134.31	3.545	0.315	Accept
Q9	141.77	150.26	150.47	158.39	0.653	0.884	Accept
Q10	138.27	149.64	157.59	148.40	0.942	0.815	Accept
Q11	133.09	155.37	145.93	154.89	1.436	0.697	Accept
Q12	163.77	152.00	150.65	149.24	0.336	0.953	Accept
Q13	134.18	143.59	157.76	164.44	5.150	0.161	Accept
Q14	137.68	142.57	157.24	167.23	4.246	0.236	Accept
Q16	173.95	152.58	149.33	147.98	1.041	0.791	Accept

5.4.5 General comments by citizens.

Details of the comments made by citizens which were recorded in the extra space provided at the end of the questionnaire are attached in Appendix 3. Overall, fifteen of the 'further comments' (six in Obuasi and nine in Kumasi) provided by the participants were deemed relevant to the current conversation. The statements highlighted how socio-economic considerations dominantly shaped the perceptions of citizens, sometimes, at the expense of environmental considerations. For example, a participant in Obuasi (ID: O304) said that "I used to engage in 'galamsey'¹², but there has been a clamp down by the government and the company [AngloGold Ashanti] due to environmental concerns and the fact that the land belongs to the company. But what should we do since there are no jobs? I know 'galamsey' destroys the land and water [environment] but what choice do have when 'galamsey' is the only means through which we can survive". Similarly, a participant in Kumasi (ID: K727) said that "I understand the fact that you are sensitising us on the things we could do to promote good development. However, if we want to recycle our waste where do we send it? We can only do it if adequate provisions are made. These days, in order to be able to go to work you must buy a car if you can afford it. The 'trotro' [public transport] is unreliable and you spend a lot time waiting it".

5.4.6 Discussion.

5.4.6.1 Perceptions of sustainability and urbanisation dimensions.

(a) Economic sustainability dimensions.

The state of the economy is always going to be an important consideration when exploring the overall well-being of urban citizens in the context of sustainability. The literature (see, for example, Bell and Blanchflower 2010; OECD 2013) emphasised how unemployment can impact negatively on the livelihoods of urban citizens, a situation which is particularly damaging for vulnerable groups such as the physically-challenged and the elderly. The findings presented in Table 5.13 attest to those realities as most responses in Kumasi and Obuasi on the economic statements expressed a general pessimism on the representations of the economy. The prevailing opinion was that difficulties exist in securing employment. For

¹² Galamsey is the local name in Ghana for small-scale mining activities (mostly illegal) where the top soil of ore-rich land is excavated and chemicals including mercury are used to extract the mineral ore (Azumah et al. 2019).

example, when asked for their perception of how easy it is to find a job (Q1), 84.5% of respondents in Kumasi disagreed or strongly disagreed, compared with 88.7% in Obuasi. Differences in perceptions among respondents emerged in the question about whether income levels were acceptable (Q3), for which 46% of respondents in Kumasi agreed or strongly agreed compared with 13.6% of respondents in Obuasi who agreed or strongly agreed. In order to explain these observations, it is argued that while urban unemployment is pervasive in Ghana (see Section 4.2), consistent with the literature (see, for example, Ferguson 1999; David and Arman 2014), it is the local economic context that has underpinned respondents' perceptions of the economy in both areas. For example, at the time of the study, Obuasi, whose urbanisation is mainly premised on its mineral extraction links, was struggling economically. This is because of the down-turn in operations of AngloGold Ashanti's mine, the mainstay of the local economy. The mine was closed in 2014 and re-opened in 2019 (Ghana Chamber of Mines 2019). The implications of the mine's closure on the local economy were the loss of jobs and livelihoods for the various citizens, as well as the impact on the various sectors of the local economy. Thus, the views of respondents in Obuasi reflect the economic vulnerability of resource-based cities like Obuasi which depend on the economic empowerment opportunities provided by companies such as AngloGold Ashanti. However, the relatively better perception of the economy by respondents in Kumasi may be attributed to Kumasi's primacy regionally and nationally. Kumasi's primacy means that the city is likely to attract more substantial resources and opportunities including jobs and investments. Furthermore, Kumasi's primacy means that it wields significant administrative and political powers which potentially influence economic decisions in its favour. Overall, the perceptions of citizen on the economy are the implications of the findings in Section 5.2 which demonstrated that the populations in Kumasi and Obuasi have expanded by at least three times since 1984, and as the literature suggests (see, for example, Frank 1968), a burgeoning urban population in a less-developed country such as Ghana means that the demand for jobs exceeds supply, making urban unemployment widespread.

(b) Social sustainability dimensions.

From a human well-being perspective, the raison d'etre underpinning all sustainabilityrelated policies and projects is to provide opportunities for citizens to meet their basic needs (see, for example, WCED 1987). Thus, the findings presented in Table 5.14 explored the views of citizens on selected statements on social sustainability. By considering the distribution of citizens' responses to statements on access to social goods such as education and healthcare (as shown in Table 5.14), the responses of respondents in Kumasi and Obuasi were remarkably similar for some statements and disparate in others. For example, on the statement concerning access to health (Q5), a combined 62.9% in Kumasi disagreed or strongly disagreed, compared with 52.3% in Obuasi. Significant disparities emerged in responses on public safety (Q8) where only 7.2% of respondents in Kumasi agreed or strongly agreed that public safety levels were adequate, compared with 73.2% in Obuasi. Possible explanations for these disparities can be given in the context of city size and population pressures. For example, except for the existence of the gold mine which has diversified the community make-up of Obuasi and shaped its urbanisation, it is typically an agrarian area where residents have lived in proximity with each other. In contrast, Kumasi's population is ten times that of Obuasi (Ghana Statistical Service 2019), and the evidence previously presented (See Section 5.2) has shown that the migrant population in Kumasi (53%) is higher than in Obuasi (38%). This has implications for public safety as the literature (see, for example, Opoku-Mensah and Asare-Okyere 2014) suggests that high populations potentially compromise social harmony. Therefore, it could be argued, based on this survey, as an exploratory test of social sustainability, that public safety is favourably evaluated in Obuasi compared with Kumasi. On the whole, despite the specific differences, an important pattern observed in Table 5.14 was that respondents in Kumasi and Obuasi were more alike than different with regard to perceptions concerning aspects of social sustainability.

(c) Environmental sustainability dimensions.

The findings in Table 5.15 showed unique differences in perceptions of the respective respondents with regard to environmental sustainability statements. For example, concerning air quality / pollution levels (Q9), the responses in Obuasi were unmistakably clear as 87.4% of respondents disagreed or strongly disagreed with the notion that the level of air quality was acceptable, compared with 64.8% in Kumasi. However, on paper, AngloGold Ashanti, for most parts of its operations, is reported to have developed a relatively healthy control and oversight mechanism to monitor and deal with the pollution effects of its activities. Another observation is that, the respondents in Obuasi were more dismissive about potable water (Q12), as over 79.8% disagreed or strongly disagreed with the notion that access to potable water was adequate; compared with Kumasi where 46.8% disagreed or strongly disagreed with the same notion. A possible explanation for this observation can be inferred from the well-documented pollution of water sources in Obuasi. For example, the literature (see ActionAid Report 2006; Foli et al. 2012) alludes to the arsenic contamination of water bodies

in the mining environs and how this affects the socio-economic development of the residents. While AngloGold Ashanti, through its corporate social responsibility activities, complements the Ghanaian government's political mandate to provide public goods such as potable water to the citizens in Obuasi, the findings indicate a serious lack of access to potable water. This again reflects the peculiarity of urban sustainability problems faced by resource-based urban communities. The findings also show a severe potable water access problem in Kumasi, which is not all that different in magnitude to that of Obuasi. For Kumasi, the reasons that can be inferred from Section 4.4.1 (see, for example, Amo et al. 2017) include the impacts of land use change, as well as population pressures which lead to the conversion of fragile land cover types such as water bodies and wetlands. Therefore, respondents' perceptions in the context of the environment, appear to corroborate evidence from Section 4.2, thereby suggesting that the manifestations of urban sustainability problems within the same regional geographic area can be drastic but also diverse.

(d) Citizens' awareness of urbanisation.

Across both study sites, citizens demonstrated their awareness of the demographic and landscape transformation brought about by the urbanisation process. As the literature suggests (see for example, Misilu et al. 2010; Page 2012; Miller 2014), urbanisation will either lead to the creation or reduction in socio-economic opportunities. The study also explored citizens' perceptions on their level of access to socio-economic opportunities. In all, the findings on perceptions of urbanisation presented in Table 5.16 generate three key observations. First, there is a general notion that urbanisation rates were not commensurate with the levels of development (Q15), as 69.5% of respondents in Kumasi and 61.6% of respondents in Obuasi disagreed or strongly disagreed with the statement. Second, the findings were an indication of the impact of urbanisation on the ability of citizens to meet their basic needs, as 49.7% of respondents in Kumasi and 55.0% of respondents in Obuasi disagreed or strongly disagreed that despite the rapidly changing urban landscape they still had good access to socio-economic opportunities and infrastructure (Q16). Differences, however, also existed as 20.4% of respondents in Kumasi agreed that they have access to socio-economic opportunities and infrastructure despite rapid urbanisation, compared with 9.9% of respondents in Obuasi. The final observation is that a relatively higher number of people in Obuasi (42.1%) were not happy to live in the city and were actively seeking to migrate to somewhere else in Ghana (Q17), compared with 37.7% in Kumasi. Essentially, the observations outlined above shed further light on the nature of the urbanisation process in the

study areas. They provide indirect evidence that despite the rapidly changing landscape and population boom, urbanisation has not happened in a way such that its inherent benefits are maximised to support the development of people.

5.4.6.2 Group differences in citizens' responses to sustainability statements.(a) Gender differences in citizens' perceptions of sustainability.

Section 4.2 established the persistence of gender inequality in urban areas in Ghana (see for example, Awumbila 2006; Obeng-Odoom 2012; Danquah and Ohemeng 2017). Generally, gender differences in responses in Obuasi were weaker than Kumasi as no statisticallysignificant gender differences were found for all the statements tested in Obuasi. Statistically significant differences were found in some aspects of social and environmental sustainability in Kumasi; and also in some aspects of urbanisation. For example, the Mann-Whitney U-test showed that perceptions on Q5 (access to adequate healthcare) were greater for females (mean rank = 147.38) than for males (mean rank=169.79). While the scope of the survey did not include finding causal relationships, the gender differences in responses could be explained by reasons established in the literature (see, for example, Awumbila 2006; Danquah and Ohemeng 2017) which include, but are not limited to, poverty, cultural beliefs and traditions linked to gender-role expectations. These findings are also consistent with other scholars (for example, Tuncer 2008; Olsson and Gericke 2016) whose research reports differences in the perceptions of female and male respondents in sustainable development surveys. The findings are a clear indication that gender differences are important when seeking to understand the urban sustainability experience. The findings imply that women might be uniquely positioned to contribute to and lead efforts for urban sustainability. In this regard, urban sustainability-related policies must take into consideration women's potential penchant for pro-sustainability behaviour in order to effectively promote sustainabilityrelated initiatives. Overall, the findings provide empirical support for the dominant idea in gendered sustainable development literature (Tuncer 2008; Olsson and Gericke 2016; Aina et al. 2019) that societal gender relations shape the extent to which people experience sustainable development and suggest that gender must be considered when making urban sustainability decisions.

(b) Differences in citizens' perceptions based on economic status.

The findings showed that a person's economic status could significantly affect their perception judgements. For example, in Kumasi (see Table 5.19), the Kruskal Wallis H test found significant difference between group responses for Q2 (Acceptability of cost of goods and services), with mean ranks of 153.97 for 'Employed with wage', 176.33 for 'Selfemployed' and 142.48 for 'Unemployed'. In Obuasi, the Kruskal Wallis H test found a significant difference between group responses for Q16 (Access to economic opportunity and infrastructure), with mean ranks of 162.89 for 'Employed with wage', 164.19 for 'Selfemployed' and 132.81 for 'Unemployed'. Further statistical tests were done to explore where the differences due to economic status were emanating from (see Appendices 6 and 7). Accordingly, in Obuasi, for example, statistically significant relationships were found for differences between 'Self-employed' and 'Unemployed' in relation to Q1 and Q2. For the difference between 'Employed with wage' and 'Unemployed', statistically significant differences were found for Q1 and Q16. Thus, the findings have shown that to some extent, a respondent's economic (employment) status influenced responses, especially when the contribution of economic opportunities to overall sustainable development is considered. This is consistent with the position in the literature (for example, Aina et al. 2019) that the socioeconomic characteristics of respondents are strong determinants of their perceptions of sustainable development. Critically, the findings give an indication about the outcome for the urban poor in terms of the impacts of urbanisation. The reality that the group differences were more as a result of the responses of the economically inactive (unemployed) shows the harsher realities of urbanisation challenges on the urban poor. Consequently, the practical challenge is to design and develop sustainability-related policies and projects that cater for the needs of all citizens, while also being conscious of economic status-based vulnerabilities, especially among the urban poor. This is because the literature has thrown light on the challenges of the urban poor, especially slum dwellers, and how increasing inequality worsens their plights (Ghana Statistical Service 2007; Obeng-Odoom 2012). The discussion above shows how poverty challenges the country's SD progress (see, for example, Domfeh et al. 2012), and how the prevailing economic conditions determine the sustainability outcomes of urbanisation (Ferguson 1999; Davidson and Arman 2014).

(c) Differences in citizens' perceptions based on age and level of education.

The findings suggest that the level of education had no influence on the perceptions of citizens both within and across the urban areas studied. In terms of age group differences, the findings suggest that age group had no influence on the perceptions of respondents in Obuasi

as no statistically significant interactions were found for all the statements tested in Table 5.23. However, in Kumasi, age group of respondents explained differences in perception only to some extent as statistical significance was found in just one statement, Q17 (I am happy to live in this city and I do not intend to migrate to another place in this country), out of the thirteen statements tested (see Table 5.23). Age appears to be related to perceptions of citizens in Kumasi in relation to Q17, as despite the urban challenges in Ghana, the older generation (>60) seems to be more content and are not seeking to migrate, compared with the younger generation (18-30) who seem to be discontent with the overall state of urban development in Kumasi (see Table 5.22). The observation above could be explained in terms of aspects of urban challenges in Ghana, such as unemployment (see Section 4.2). The younger generation (18-30) who are not in employment, education or training might seem discontent with current urban development in Kumasi potentially because of the scarcity of employment opportunities. Therefore, they may aspire to migrate to other places in Ghana (or beyond) where there may be adequate economic opportunities for a better quality of life.

5.4.6.3 Willingness to engage in pro-sustainability practices.

On the whole, over 90% of respondents gave a favourable response to questions posed about their willingness to engage in pro-sustainability behaviour in both Kumasi and Obuasi. While the survey did not examine factors that may affect citizens' ability to engage in prosustainability behaviour and actual actions, the results suggest a generally favourable attitude towards sustainability. Based on the picture portrayed about the state of urban development in Ghana in Section 4.2, where urban poverty and inequality are pervasive, slums exist, and there is a scarcity of basic socio-economic infrastructure, a situation that is further exacerbated by rapid urbanisation (see Section 4.4); one might question the reasons underpinning the reported willingness of the respondents to engage in sustainable behaviour. It could be argued that the respondents were willing to conserve water and energy (electricity) because the provision and availability of these social or environmental resources are limited by their scarcity. Also, willingness to use public transport is explained by the reality that it may be the only available option (if available at all). Therefore, while the study did not explore the reasons why they would choose to engage in such behaviours, and considering the well-documented urban development problems presented in Section 4.2, the results imply that the need to promote urban sustainability by engaging in pro-sustainability behaviour currently resonates well with respondents, and they may potentially engage if conditions permit. This is consistent with the assertion by Pappas and Pappas (2014) that there is the potential for sustainable action when the day-to-day behaviour of individuals are aligned to well-stated values. In sum, the results on respondents' attitudes towards prosustainability practices in Kumasi and Obuasi have shown that the selected practices were widely acceptable. This presents an opportunity to formulate policy, or if such is already available, to implement policies aimed at transforming these positive attitudes into normative sustainability behaviour in Ghanaian urban areas. Essentially, however, appropriate macroconditions such as basic infrastructure (for example, kerbside recycling bins; public transit infrastructure) must also be provided if the implementation of such policies is to be successful.

5.4.6.4. General inference from citizens survey.

The urban sustainability dimensions explored in this study focused on the different dimensions of sustainability and attempted to represent the needs, values and priorities of the Ghanaian society as perceived from the review of the country's urban challenges (see Section 4.2). Although common interpretations of sustainability are premised on the integration of the different environmental, social, and economic dimensions (see, for example, Jeronen 2013), the findings have shown that the respondents' perceptions about sustainability are primarily influenced by their socio-economic circumstances, rather than a desire to witness any defined description of sustainability where environmental, social and economic goals are balanced. While the findings presented in Section 5.4.2 have shown the differences and similarities in the perceptions of respondents in Obuasi and Kumasi, the comments recorded in Appendix 3 and sampled in Section 5.4.5 succinctly capture the context within which the interactions between urbanisation and sustainability at the local urban level manifest from the perspective of citizens.

The general inference made from the comments is that as much as citizens recognised the various dimensions of sustainability in the context of urbanisation, their overall perceptions were shaped by their socio-economic circumstances. While there is certainly no basis for generalisation, what resonates throughout the comments is the tendency for citizens to prioritise issues concerned with basic socio-economic provisions, sometimes, over environmental concerns, in the midst of poverty. The survey of the subjective experiences of citizens, thus, corroborates the initial premise in this thesis on how poverty manifests and

uniquely shapes the urbanisation and sustainable development experience in the urban areas of SSA. Consequently, the findings presented in this chapter embed the understanding that the circumstances in SSA (in the context of national and citizen poverty) mean that perceptions and interpretations of urban sustainability at the local level potentially vary from the utopian conditions for urban sustainability suggested in the literature (see Section 2.3.3), albeit equally legitimate.

Overall, although the responses for Kumasi and Obuasi in Tables 5.13–5.16 were not tested for statistical significance, and despite the different local conditions in Kumasi and Obuasi, the findings suggest that there is more commonality among responses in the context of the dimensions of sustainability than might be suggested based on their respective geographic, administrative and macro-economic structures. This is evident from the pattern of identical responses on the level of agreement with the selected sustainability statements among respondents from both urban areas. This is because while Obuasi's economy is fundamentally based on mining-related activities compared with Kumasi's economy which is underpinned by commerce and service provision, the findings seem to suggest that there is potentially more agreement than previously thought. Furthermore, the findings give an indication of the trade-offs and mediating effects of the different dimensions of sustainability and the role of governance in achieving such. For example, a basic tenet of sustainable development is the ability of citizens to meet their basic needs (WCED 1987), which includes education. This requires having the economic means to do so. Therefore, although the analysis on economic dimensions showed that the general perceptions of respondents were negative to a large extent, the findings indicate that the impacts of such negative perceptions are sometimes mitigated by some dimensions of social sustainability. Take the scenario that Ghana prioritised access to education for the past decades and successive governments, irrespective of political ideologies, have sought to widen universal access to education (see, for example, Acheampong 2009; Arthur-Mensah and Alagaraja 2018), with the result that a broader group of citizens now enjoy the fruits of these social policies. Therefore, the introduction of education policies such as Free Compulsory Universal Basic Education (FCUBE) in 1995 (Acheampong 2009) and the Free Senior High School Policy in 2017 (Arthur-Mensah and Alagaraja 2018) certainly in intent, and almost by definition, extends the safety net to cover the urban poor in Ghana. This is reflected by the relatively positive responses in perception of education access in Kumasi and Obuasi.

The findings also have important policy implications when viewed from the perspective of gender difference. If gender differences exist in the perceptions and attitudes of citizens towards urban sustainability, then policy needs to address this with appropriate gender-specific provisions. The importance of gender is supported by evidence in the literature (see, for example, Awumbila 2006; Danquah and Ohemeng 2017). The findings in Kumasi suggest differences in responses based on gender regarding access to healthcare, level of sanitation provision, and on whether the development level was commensurate with urbanisation levels. Thus, for example, if challenges to sanitation provision hamper the sustainability experience and progress of women, then policy provisions aimed at improving sanitation services for women in cities, may have critical impacts on their sustainability experience and in turn urban sustainability.

However, the perception judgment of citizens could potentially be biased in diverse ways, as the factors underpinning respondents' perception remain complex. It is, therefore, likely that subjective perceptions without the assistance of monitoring data may not be an entirely accurate description of objective sustainability conditions. Nevertheless, local respondents are able to grasp the true core of local problems as they form their opinions based on their intuitive judgements, as underpinned by their experiential knowledge (Hsieh 2012). In sum, the findings have captured the perceptions of residents on selected sustainability statements based on their lived experiences and agree with the scholarly view that exploring the perceptions of citizens helps to deconstruct the interconnected facets of urban sustainability (Macke et al. 2018).

5.4.7 Section summary.

It is important to consider the perspectives of citizens when evaluating urban sustainability. The present section had the objective of exploring the perceptions of respondents in Kumasi and Obuasi on aspects of urbanisation and dimensions of sustainable development in the context of urban sustainability. The findings do not provide a clear way forward. However, in the context of urban sustainability policy formulation and planning, the findings highlight the divergence and convergence of preferences and attitudes among citizens; and therefore, they highlight the importance of an approach to urban sustainability that achieves an appropriate balance across the sustainability dimensions and incorporates the needs of vulnerable groups. The findings suggest that once the right conditions are in place, citizens' ownership of

sustainability ideas, policies, and projects could be encouraged. The 'further comments' made by the participants helped to inform the understanding of the dominant perception of respondents as they gave an indication that citizens are likely to prioritise socio-economic interests over environmental considerations. Overall, the findings contribute to the literature on urban sustainability in Africa from a citizens' perspective. To the best of the researcher's knowledge, this is the first study to empirically assess the sustainability and urbanisation perceptions of citizens in Ghana, more so in a comparative framework.

5.5 Integrated Discussion

The analyses in Sections 5.2–5.4 have demonstrated the spatial and temporal dynamics, as well as perceptions of citizens, on the interactions between urbanisation and sustainable development at the local urban level in Ghana. In order for development to be classified as sustainable, it must proceed in a way that ensures a balance among economic, social and environmental aspects. For example, from an economic perspective, the findings in Section 5.2 have provided evidence on how rapid urbanisation has influenced a demographic shift in Kumasi and Obuasi. However, consistent with the national urban employment situation (see Section 4.2), the demographic shift in both urban areas does not seem to have been accompanied by adequate economic opportunities for citizens as evidenced by citizens' perceptions of economic sustainability statements (Table 5.12) in Section 5.4. From a social perspective, the findings in Section 5.3 suggest that urbanisation in Kumasi and Obuasi has resulted in a higher concentration of urban social services and infrastructure. This has led to improvements in aspects of social development such as education, maternal health, and access to sanitation and water in Kumasi and Obuasi compared with average national performance. This observation was corroborated by the somewhat positive perceptions of citizens to the social sustainability statement concerning access to education in Section 5.4. It is, however, argued that adequate investment must be provided in order to ensure that rapid urbanisation does not undermine the capacity to provide urban services and infrastructure for citizens in Obuasi and Kumasi. From an environmental perspective, the findings in Section 5.2 suggest that urban land cover has expanded at the expense of natural vegetation in Kumasi and Obuasi, a situation that potentially undermines the integrity of urban environments in the two areas. Overall, the interactions between urbanisation and sustainable development in Kumasi and Obuasi have created urban forms which are possibly characterised by attributes such as less-compact and low-density development. Considering the general urban poverty and unemployment situation in Ghana (see Section 4.2) and the perceptions of citizens on economic sustainability statements in Section 5.4, it could be argued that the role of market forces, uncontrolled landscape urbanisation, and the resulting urban forms could undermine sustainability by potentially creating impacts such as socio-spatial segregation. Consequently, socio-spatial segregation could threaten the livelihoods of the urban poor in Obuasi and Kumasi, as they will potentially become far removed from the concentration of economic opportunities in that exist in city centres (see Polidoro et al. 2012). In sum, the analyses in Sections 5.2–5.4 suggest that the interactions between urbanisation and sustainable development at the local urban level in Kumasi and Obuasi are currently not proceeding in a sustainable manner.

Chapter Six

6. National Urban Policy and sustainability in Ghana: a critical analysis

6.1 Introduction

Cities and their urban forms, in contemporary times, provide visual expressions of policies framed at the national level and implemented at the local level (Uitermark 2005). This means that as much as urban areas are shaped by the forces of urbanisation (including globalisation) (Zhang 2016), in no small extent, they tend to reflect the national urban planning and development philosophies. Alongside the popularisation of the sustainability agenda, the rapid pace of urbanisation has resulted in the increasing adoption of National Urban Policies (NUP) as a way of sustainably managing the urbanisation process (UN-Habitat 2016a). In this context, one could easily assume that the existence of such policies could help remedy the sustainability challenges of urbanisation and improve the resultant urban forms. However, despite the implications of urban policies for local sustainable development, there is limited understanding of how sustainability principles are incorporated (Davidson and Arman 2014). Accordingly, the salience of SD and urbanisation-related issues on both global and local scales makes it imperative to discern the extent to which urban sustainability principles are embedded in urban and development policies. In other words, if the gap between rhetoric and practise is to be bridged, then the sustainability focus of these policies should be examined.

The chapter provides an evaluation of the sustainability context of selected urban-related policy documents in Ghana and the remainder proceeds in four parts. The first part provides an overview of the selected policy documents and highlights the main objectives and foundations of the documents. The selected documents are the National Urban Policy (NUP 2012), which has been formulated by the Government of Ghana in order to steer Ghana's urbanisation to sustainable outcomes; and the Action Plan ([AP], MLGRD 2013), which is ancillary to the NUP and contains specific initiatives to be carried out in order to achieve the goal of the NUP. Due to their interdependence, they were analysed as one document. The second part presents findings on the evaluation matrix for the NUP and AP based on the sub-dimensions of UN-Habitat's (2016b) City Prosperity Index (CPI). The third part presents discussions on the sustainability constructs of the NUP and AP under the lens of the World Bank's (2018) analytical urban sustainability framework (USF). The fourth part presents

discussions on the factors that potentially affect the operationalisation of sustainability in the NUP and AP. Overall, it is important to note that the focus in this thesis is not to superficially criticise or praise the policy documents under consideration. However, an adage associated with issues of public policy and administration: 'seeing is believing', applies in this context. Therefore, an obvious caveat is the reality that these policy documents have only been launched in the last decade, and it will take time to implement and judge their effectiveness. Thus, the following sections will provide a review of selected urban-related policy documents and their overarching influence on urban sustainability in Ghana, rather than an assessment of policy outcomes.

6.2 Overview of National Urban Policy (NUP)

An overview of the National Urban Policy (NUP) is presented as follows. Ghana's urban vision, as articulated in the main goal of the NUP, is "to promote a sustainable, spatially integrated and orderly development of urban settlements with adequate housing, infrastructure and services, efficient institutions, and a sound living and working environment for all people to support the rapid socio-economic development of Ghana" (NUP 2012, p.21). The policy document outlines the following twelve objectives to be pursued in order to realise the goal of the NUP through the AP:

- (a) to facilitate balanced re-distribution of urban population.
- (b) to promote a spatially integrated hierarchy of urban centres.
- (c) to promote urban economic development
- (d) to improve environmental quality of urban life.
- *(e) to ensure effective planning and management of urban growth and sprawl, especially of the primate cities and other large urban centres.*
- (f) to ensure efficient urban infrastructure and service delivery.
- (g) to improve access to adequate and affordable low-income housing.
- (h) to promote urban safety and security
- *(i) to strengthen urban governance.*
- (j) to promote climate change adaptation and mitigation mechanisms.
- (k) to strengthen applied research in urban and regional development.
- *(l)* to expand sources of funding for urban development and strengthen urban financial management.

In general, the NUP sets out the Ghanaian government's response to the reality that half of Ghana's population lives in urban areas. The NUP is the outcome of a set of activities that began in March 2009 and included: situation analysis and framework for urban development and management in Ghana; stakeholder consultations; technical expert reviews; strategic environmental assessment; stakeholder (public) consultation workshops; and central government consideration and approval (Awortwi 2015). The NUP recognises the need to provide for the increasing urban population *vis a vis* the prevailing urban challenges in Ghana (see Section 4.2). Overall, the NUP specifically provides a framework and direction for remedying the challenges in Ghanaian urban areas by addressing urban security issues; urban economic development issues, with emphasis on the informal sector; urban infrastructure provision; urban environmental improvement issues; and urban planning issues (NUP 2012; MLGRD 2015).

6.3 Assessment of urban sustainability in development and urban policy

An evaluation matrix showing how the NUP and AP align with the dimensions of CPI is presented in Table 6.1. The number of ticks assigned is explained as follows. Zero ticks imply an absolute lack of coverage of the sub-dimension in the documents reviewed. One tick is assigned when a sub-dimension has been highlighted, but there is a lack of detailed discussion or associated implementation process. Two ticks are assigned when there is a detailed or associated implementation process captured in the document reviewed. For example, this means that for an indicator such as income: (i) zero ticks have been assigned if there is absolutely no mention or provision for income in the NUP; (ii) one tick has been assigned if income is explicitly mentioned in the NUP but not directly linked to a specific programme or objective in the AP; and (iii) two ticks have been assigned if income has been explicitly captured in the NUP, and has also been linked to a specific programme or objective in the AP. Therefore, from Table 6.1 (which assesses the content of the NUP and AP based on 18 selected sub-dimensions of the CPI), it is seen that a greater majority (13 out of 18) of the CPI sub-dimensions considered were assigned two ticks. This means that there was a policy initiative and / or associated key activity for the following: economic strength, employment, economic agglomeration, housing infrastructure, urban mobility, safety and security, land use, social inclusion, waste management, energy, municipal finance and accountability, participatory and institutional capacity, and governance of urbanisation. Four

sub-dimensions (economic equity, ICT, public spaces, gender inclusion) were mentioned but with no detailed policy initiative or key activity. Only one sub-dimension, air quality, received zero ticks as it is neither explicitly mentioned nor associated with a specified detailed action. Overall, the NUP and AP provide adequate scope for the sustainability of urban areas in the context of the CPI dimensions.

No	City Prosperity Index	Urban Sustainability Framework	NUP / AP
	(CPI) Sub-dimension	(USF) Dimension	score
1	Economic Strength	Urban economies	2
2	Employment	Urban economies	2
3	Economic Agglomeration	Urban economies	2
4	Housing Infrastructure	Urban economies	2
5	ICT	Urban economies	1
6	Urban Mobility	Urban economies	2
7	Public Space	Natural environment and resources	1
8	Safety and Security	Inclusivity and quality of life	2
9	Land Use	Natural environment and resources	2
10	Economic Equity	Urban economies	1
11	Social Inclusion	Inclusivity and quality of life	2
12	Gender Inclusion	Inclusivity and quality of life	1
13	Air Quality	Natural environment and resources	0
14	Waste Management	Natural environment and resources	2
15	Energy	Climate action and resilience	2
16	Municipal Finance and	Fiscal sustainability	2
	Accountability		
17	Participatory and	Governance and integrated planning	2
	Institutional Capacity		
18	Governance of	Governance and integrated planning	2
	Urbanisation		

Table 6.1: Evaluation matrix for NUP and AP based on CPI sub-dimensions.
6.4 Evaluating the policy approach to sustainability

In the context of the documents under review, it can be argued that the NUP and AP are primarily concerned with the establishment of a long-term fundamental urban strategy for Ghana. Thus, Ghana's NUP represents a strategic policy response that recognises the opportunities and threats of rapid urbanisation and attempts to provide guidelines aimed at steering the path of development in cities to a sustainable one (UN-Habitat 2016a). In light of the findings in the previous section, a review of the sustainability context of the NUP is discussed based on the outcome and enabling dimensions of the Urban Sustainability Framework (as introduced in Chapter Three).

6.4.1 Urban economies.

The urban economy dimension is stipulated in diverse aspects in the NUP and they are all aimed at stimulating urban economic growth. A wide range of urban economic subdimensions are captured in the NUP covering the entire urban economic continuum, from high-end formal activities to low-end informal activities. For example, in highlighting the potential role of urban areas as engines of economic growth, the NUP makes a case for the designation of some urban areas as 'growth centres' for economic activities and infrastructural development by enhancing their investment portfolios and making provisions to mobilise resources for their investment and infrastructural needs. The NUP further recognises the impact of globalisation and seeks to make Ghanaian urban areas globally competitive. At the lower end of the urban economic continuum, the NUP widely captures the urban informal sector.

While it is impossible to elaborate on all the mechanisms that the NUP proposes to help steer urban economic development, a key theme that emerges is the role of the informal economy. From the information on informality and unemployment presented in Section 4.2.1.2, it can be argued that the intricacies between informality and unemployment in Ghana have an overarching effect on urban employment and overall urban economic development because the informal sector is estimated to constitute about 80% of Ghana's total labour force (Hormeku 1998; Osei-Boateng and Ampratwum 2011). Rather than merely making efforts to reduce informality, the NUP and AP recognise how critical the informal economy is to the overall urban economy, as four of the eight policy initiatives under urban economic

development are explicitly based on the informal economy. Specifically, these policy initiatives are intended to: (i) change official attitudes towards informal enterprises- from neglect to recognition and policy support; (ii) ensure that urban planning provides for the activities of the informal economy; (iii) build up and upgrade the operational capacities of the informal enterprises; and (iv) improve funding support for the informal economy. The attention given to the informal economy in the NUP gives cognisance to the idea that the rate of population growth due to urbanisation will render it difficult for urban areas in Ghana to create adequate productive employment opportunities for an increasingly urban population (Osei-Boateng and Ampratwum 2011). It can also be argued that the wide focus on informality in the NUP shows that informality is not ephemeral, but rather a mainstay in the urban economic space in Ghana (Hormeku 1998; Osei-Boateng and Ampratwum 2011), hence it becomes prudent that necessary policy actions (as evident in the NUP) are taken to support the informal sector in order to ensure that its positive contributions to the overall urban economy are maximised. The attention given to informality in NUP marks a shift away from elite policy-making (see Lipton 1977) and ensures that the interests of the most vulnerable are captured at the national level. However, when considered vis a vis the need to develop rural areas, such heavy focus on urban informality could be seen as a potential 'urban bias' as it potentially changes the narrative or conditions in favour of urban residents, and could give further impetus to rural-urban migration (Lipton 1977; Majumdar et al. 2004). Nevertheless, an obvious argument that can be advanced based on evidence presented so far in this thesis is that, the rapid nature of urbanisation vis a vis a fundamentally weak national economy, where poverty and unemployment are pervasive (Domfeh et al. 2012; Cooke et al. 2016), will more often than not, produce a large informal sector. Furthermore, if any national policy could be tailored to better deal with this urban challenge then it is the NUP. In general, the NUP captures elements of the urban economy and proposes policy initiatives aimed at promoting economic growth and making Ghanaian urban areas competitive across the continuum of urban economic activities.

6.4.2 Fiscal sustainability.

Meeting the infrastructural demands for urban areas is capital-intensive, a situation made worse in fundamentally weak economies (Farvacque-Vitkovic et al. 2008; Okeke 2014). In the traditional Ghanaian setting, development activities in urban areas are primarily funded by the central government. Rapid urbanisation will create demand for larger urban

investments in order to provide the necessary infrastructure and services for a growing urban population (Farvacque-Vitkovic et al. 2008; Okeke 2014). This, in effect, makes it unsustainable for the government to fund urban development solely (see, for example, OECD 2007; Lauermann 2017). Both policy documents recognise the issue of urban development finance and fiscal sustainability. For example, the NUP advocates for financially innovative ways of funding urban development and managing the economic sustainability of urban investments. This, it does, by promoting project financing approaches, such as public-private partnership (PPP) as an alternative option that will help alleviate the financial pressures on government (see, for example, OECD 2007; Lauermann 2017); and by promoting the implementation of fiscal decentralisation proposals through which the central government will devolve its fiscal tools to local metropolitan and municipal authorities, usually in terms of responsibility for revenue mobilisation and spending decisions. Such proposals will have benefits as well as challenges. For example, the fiscal decentralisation promoted by the NUP will allow revenue to be mobilised and spending decisions made based on locally-determined needs and priorities. Challenges could, however, arise where the devolution of fiscal tools threaten macro-economic stability; for example, where the local development priorities of a municipal authority (which are dynamic) do not align with the overall national development goals.

6.4.3 Inclusivity and quality of life.

From a housing perspective, the NUP gives explicit policy support for the provision of housing infrastructure for urban citizens. With a national housing stock deficit of at least 1.7 million units (Daily Graphic Online 2018), an increasingly urban population, together with the significantly low levels of social housing provision, mean that satisfying the housing needs of the urban population tends to rest on private provision. Accordingly, the NUP advocates for an enabling environment in order to propel private sector delivery of affordable housing in urban areas. At the same time, it makes a case for the upgrading of slums in urban areas and promotes private sector involvement in slum upgrade and redevelopment. An apparent threat to inclusivity here is the neo-liberal outlook of the policy. For slum upgrade, there is a possibility that the pursuit of often capital-intensive redevelopment projects could marginalise the urban poor who are usually the intended beneficiaries of such efforts and make them landless in the process (Obeng-Odoom 2013a). The discussion above corroborates the evidence by Davidson and Arman (2014) on how the dominance of neo-

liberal urbanism can impede the potential of urban policies to deliver transformational change in the context of sustainability.

A central theme in Ghana's main development vision is the reduction of poverty. Similarly, earlier sections of this thesis have demonstrated the pervasiveness of poverty in Ghana (see Section 4.2.1.3), especially in urban areas, and how it makes the urban-poor vulnerable, thereby undermining urban sustainability (Easterly 2007; Okyere et al. 2012). However, the analysis of the NUP shows an inadequate focus on poverty reduction. It is true that some of the objectives of the NUP, including for example, 'promoting urban economic development' may lead to poverty reduction. It is also true that some programmes, such as slum upgrade and policy initiatives on informality, may have the urban poor as intended beneficiaries. However, it is not clear in the NUP how these will lead to overall poverty reduction as there is no explicit link to any poverty reduction strategy in the document. Critically, there was no detailed breakdown or specification of actions to be carried out to mitigate or reduce urban poverty. The factors adduced above, therefore, reflect the need to formulate urban policy in a way that ensures sustainability through the incorporation of explicit inclusivity measures in policy design.

6.4.4 Natural environment and resources.

As has been shown already, urbanisation represents one of the most significant manifestations of population and landscape dynamics (Samal and Gedam 2015; World Bank 2015), both of which potentially exert varying degrees of pressure on the natural environment. This scenario has made it imperative to pursue urban sustainability actions in order to find ways of managing the delicate trade-offs between the anthropogenic impacts of urbanisation and the threats to the integrity of the natural environment. From an urban environment perspective, the NUP seeks to 'improve the environmental quality of urban life'. Four of the policy initiatives captured in the Action Plan focus on sanitation and waste management. Policy initiatives for creating environmental awareness through public education, as well as environmental management in mining towns are also captured. The remaining initiatives focus on the protection of environmentally sensitive sites such as wetlands or coastal areas. Curiously missing, perhaps, is a focus on air quality. Also, while 12 policy initiatives were outlined under the Action Plan to cater for the urban environment, there were no environmental targets set in the NUP and AP. For example, while four of the

policy initiatives focused on waste management, no target is captured; for example, to achieve 10% urban plastic waste recycling by a particular year. Without such targets, it is difficult to decipher how the policy initiatives will contribute to urban sustainability. In sum, the NUP and AP provide a broad guiding framework rather than the means and tools for achieving specific environmental targets. Nevertheless, they represent a strong starting point for mainstreaming environmental concerns into urban policy and decision-making.

6.4.5 Climate action and resilience.

The NUP tackles climate change with a policy objective as follows: 'to promote climate change adaptation and mitigation mechanisms'. Ghana's NUP and Action Plan show that its urban climate action emphases are on public education for climate and energy conservation, cleaner technologies, emission control, institutional governance and coastal or wetland management. As a country that is expected to reach an urbanisation rate of 70% by 2050 (Cobbinah et al. 2015a), one of Ghana's most significant challenges is managing the trade-off between urban growth and climate change processes (Okyere et al. 2012). It is important to note that mitigating climate change requires a multi-sectoral approach where sectors such as the power generating industry and manufacturing industry may have already implemented national plans towards GHG reduction and climate change mitigation. Nevertheless, the fact that cities contribute to 70% of global energy consumption and greenhouse gas emissions (Hoornweg et al. 2011) makes urban level climate change action strategic in managing the impacts of overall climate change.

Despite the climate actions highlighted in the NUP and the Action Plan, a question remains: are Ghanaian urban areas ready for climate change? A striking feature of the climate-related policy objective, as captured within the NUP and AP, is the lack of specific targets for climate change action (for example, emission reduction targets). For example, none of the policy initiatives under climate makes mention of greenhouse gas (the main proponent in the climate change conversation) or provides any targets for their reduction. Also, consistent with Wheeler (2008, p.488), whose research concluded that the majority of plans *"lack the strong actions and political and institutional commitment needed to mitigate emissions"*; rather than provide concrete evidence-based measures, the initiatives promoted in the NUP to manage climate change were somewhat vague. For example, one of the key activities to encourage the progressive reduction of hazardous substances by industry is to 'apply appropriate and

deterrent sanctions to defaulting firms / companies'. However, relevant pollutant or climate emission targets based on which the proposed punitive measures will be applied are hardly mentioned. This points to an apparent lack of urban pollutant or climate emissions data to inform urban policy at the national level. Accordingly, based on what is captured in the NUP and AP, and the apparent lack of data for urban-related climate action or decision-making, it cannot be confidently argued whether the NUP and AP's climate initiatives encompass the necessary elements that are needed for successful climate change adaptation in Ghana's cities in ways which could eventually contribute to urban sustainability. This is because while the NUP recognises the importance of climate change management in achieving overall sustainable development in cities, and the AP outlines the initiatives Ghanaian urban authorities should pursue to cope with their climate vulnerabilities, the lack of targets makes it difficult to determine how ambitious these initiatives are, and the national capacity thereof, to achieve them. Essentially, it remains to be seen what the capacity of local authorities for climate action in cities will be if national-level urban policy documents are limited in scope on progress and targets for climate change.

6.4.6 Governance and integrated planning.

The complexity of urbanisation and its sustainability dynamics demand comprehensive policy integration such that relevant policies are synthesised to achieve common goals (Pelletier 2010; Vojnovic 2014). Accordingly, governments tend to face criticism at all levels for the perceived inability of national programmes, policies, and funding to deliver multiple benefits, as they tend to fail at cutting across all departmental levels (see, for example, Niyonkuru 2016). Thus, from the perspective of this study, realising the sustainability imperatives of Ghana's NUP will require integrating national and urban development through the balancing of national and local interests. Ghana's NUP takes into cognisance the role of effective governance and planning in dealing with the challenge of population growth and development pressure due to rapid urbanisation. One of the key features of the NUP is the introduction of two cross-cutting initiatives which are intended to achieve multiple objectives namely: The Land Use and Spatial Planning (LUSP) Bill; and Development Guidelines. Passed as the LUSP ACT, 2016, it essentially provides revised legislative foundations based on which sustainable development of land and human settlements will be pursued. The Development Guidelines serve as instruments for the steering of spatial development and

comprise planning models, planning standards and land use zoning regulations. The planning models captured in the NUP are presented in Table 6.2.

Planning model	Coverage	
National Spatial Development Framework	National.	
Regional Spatial Development Framework	Administrative regions, sub-region, multi- region.	
District Spatial Development Framework	District, multi-district.	
Structure Plan	District, multi-district, sub-district.	
Local Plan	Sub-districts (specific localities).	

Table 6.2: Planning models captured in Ghana's National Urban Policy.

Source: NUP (2012).

How the sustainability construct in urban policy is framed and understood at the national level will eventually determine the extent of adherence or resistance at the local level (see, for example, Uitermark 2005). The aforementioned initiatives are, therefore, intended to cut across six out of the twelve objectives outlined in the NUP, giving coverage to issues including legislation, urban settlements, urban economy, integrated planning, local plans, land-use management, and planning standards.

6.5 Challenges to sustainability operationalisation in Ghana's NUP

From the discussion so far, one could argue that development in urban areas is underpinned by the policy regimes in force and manifest as such (see, for example, Portney and Berry 2010). In this context, then, any policy intervention in urban areas will have to take into consideration related sustainability factors. The review of the NUP and AP shows that they provide a strong stimulus for urban areas in Ghana to develop in a manner consistent with sustainability principles. In a provocative stance, it could be said that the issue of haphazard implementation of urban policy in SSA (Pillay 2008) raises the question as to the extent to which sustainability principles can be operationalised in urban policies. Answering such a question would require an understanding of the challenges in the implementation of the NUP, especially in the context of sustainability operationalisation. These are now discussed under the following themes: distributional focus and integration of sustainability dimensions; sustainability rhetoric; policy coordination; and political barriers.

6.5.1 Distributional focus and integration of sustainability dimensions.

For all of its developmental intentions, the NUP shows some inherent tensions concerning its aspirations for economic, social, and environmental progress that might not be consistent with general sustainability principles (see, for example, Pelletier 2010; Vojnovic 2014). As has been shown in the literature (see Section 2.3.1) and reiterated throughout this thesis, in the context of the various dimensions of urban systems, the sustainability agenda requires holism (see, for example, Jeronen 2013). Thus, any policy initiative that can be attributed to one dimension would have implications for other dimensions (Pelletier 2010; Vojnovic 2014). For example, 'promoting urban economic development' cannot be disconnected from the fact that underlying factors such as 'market forces' which shape urban economies could have exclusionary tendencies that potentially threaten inclusivity for vulnerable groups, thereby undermining social sustainability (Young et al. 2006; OECD 2007). In a similar vein, the market forces could have an overarching influence on the objective of 'improving the environmental quality of urban life' by producing environmental costs due to market failure, or by producing environmental benefits through solutions to environmental problems. The urban sustainability challenge evident in this context is that, while the objectives and policy initiatives outlined in the NUP are critical to achieving progress in all dimensions of sustainable development in Ghana's urban areas, the policy fails to outline their interrelationships such that there is a better integration of the balance of economic, social, and environmental dimensions in the context of sustainability (Pelletier 2010; Vojnovic 2014).

Of equal importance is the distributive focus of sustainability principles in the NUP. As inferred from the research that studied the sustainability constructs in urban policies in the literature reviewed in Chapter Two (for example, Lubell et al. 2009; Portney and Cutler 2009), the extent to which sustainability principles are incorporated in urban policies often depends on the respective local socio-economic and demographic contexts. Despite the recognition of the importance of embedding sustainability principles in urban policies, it does not extinguish the desire of countries such as Ghana (which is less-developed) to develop at all costs (Zhang et al. 2018). In fact, it is evident from perusing NUP and AP that there is fundamental prioritisation of economic development. This means that while the government

subscribes to various SD-focused international agreements such as the SDGs, it will not voluntarily relinquish its desire to move Ghana to a higher level of development. Hence, the political intentions of the Ghanaian government will determine whether an objective is pursued irrespective of whether or not it is consistent with sustainability principles. This aligns with the views of Davidson and Arman (2014) who have argued that the potential of urban policies to deliver transformative change in the sustainability context is limited by the prevailing political situation. Thus, overall, the NUP illustrates a stronger focus on economic outcomes as compared with social and environmental ones. Although the NUP does not discount the significance of achieving social and environmental progress, however, it can be inferred from its objectives that it values stronger economic progress as a springboard to support social and environmental development. From a sustainability perspective, this is consistent with the notion highlighted in Section 2.3.1 about how developed countries tend to stress the 'sustainable' aspect of the concept, while less-developed countries tend to stress the 'development' aspect (Zhang et al. 2018).

Furthermore, the wide-ranging objectives set out in the NUP could potentially derive benefits and costs that compete or conflict with each other. Consider the illustrative scenario of Ghana where unemployment is prevalent and inequality is deeply entrenched (see Section 4.2.1.2), the seemingly inexorable conflict of the NUP's economic objective that seeks to 'promote urban economic development', against the environmental objective of promoting 'climate change adaptation and mitigation mechanisms' which potentially includes reduction in greenhouse gas emissions, is set in play. However, as succinctly captured in the Action Plan: "over time, all policy initiatives with related activities will need to be fully carried out, but in the situation of limited resources, it is crucial that priorities are clearly identified and actively pursued by all implementing bodies" (MLGRD 2013, p.11). Thus, when it comes to setting priorities and making strategic choices, sustainability is potentially undermined as its competing costs and benefits may be difficult to reconcile. This demonstrates that the desirability of the sustainability agenda in public urban policy potentially loses traction when faced with competing macro-scale growth imperatives and it is ultimately shaped by the prevailing political economy (Davidson and Arman 2014).

It can be argued that despite the enthusiasm surrounding SD, the review of the NUP showed a limited incorporation into urban policy, as reflected by the overall sustainability outlook. This is consistent with the findings by Portney and Cutler (2010) who concluded, based on their

assessment of the sustainability policies of thirteen American cities, that few were on the path to sustainability. Critically, the NUP fails to articulate sustainability objectives and targets. A look at the objectives of Ghana's NUP shows that it advocates for some form of development. Nevertheless, these objectives are in no way operationalised into concrete sustainability goals. Perhaps while 'political will' may exist for incorporation, the fuzziness of sustainability makes it difficult to capture how its principles are met in the document. Nevertheless, within the purview of the discussion so far in this thesis, is the idea that the pervasiveness of the sustainable development agenda warrants an explicit incorporation of sustainability imperatives in urban policy discourse.

6.5.2 Sustainability rhetoric.

A fundamental challenge for a country such as Ghana, that has recognised the need to steer contemporary urbanisation to a sustainable future, and has rightly formulated a national urban policy, is to translate its objectives into reality (see Pillay 2008). In general, the NUP is moderately-ambitious, and its Action Plan provides an expansion on the initiatives designed to achieve the policy objectives. The plan sets specific policy actions and measures, but with only rudimentary development. That is, it cites specific measures for application to particular local points with varying degrees of definition, but without actually justifying them and developing them in depth. This means that translating the objectives of the NUP into practice on the ground, in terms of how it will achieve sustainability, is difficult. As has been noted throughout this thesis, the Ghanaian government has embraced SD, and although not explicitly mentioned, it is reflected in policy goals and government programmes (Alliance for Development 2014). Based on the notion that the extent to which sustainability principles are incorporated into urban policies typically depends on the respective local socio-economic and demographic contexts (Davidson and Arman 2014), the fuzziness of SD and Ghana's status as a poor less-developed country, mean that reconciling and operationalising sustainability principles through ambitious targets or objectives, as contained in the NUP, are imponderable and tend to be more of a rhetoric. Such a situation gives a clear indication of how the disjuncture between the policy's rhetoric and praxis potentially undermine the sustainability outcomes of the NUP and can be explained by the following. First, while the government has embraced SD, Ghana's urban areas and systems are heavily reliant on private and external input to thrive (see, for example, Springett 2013; Van Alstine and Barkemeyer 2014; Lauermann 2017). This means that its urban systems blatantly favour the interests of, and are

influenced by, market forces and external forces such as globalisation (see Sections 2.2.4 and 2.6.1.3). Such a situation, as was shown in Section 4.2.1.2, has for example, led to local firms folding up due to competition by foreign firms (Osei-Boateng and Ampratwum 2011) and reinforced wide-ranging inequalities in wealth and freedoms that hinder the development of urban areas (Young et al. 2006). Thus, despite the emergence of strong policy focus on SD, as captured in documents including the NUP, adherence to sustainability principles potentially remains rhetoric as strategies of contestation by the aforementioned forces will antagonise its operationalisation in order to protect their interests. This scenario is consistent with Pacione's (2007, p.248) assertion that *"the goal of sustainability is not an integral component of market capitalism and will inevitably encounter opposition from entrenched interests"*. The argument above is consistent with the views of other scholars (for example, Wheeler 2008) on how the absence of strong commitment to translating the sustainability features of urban policies into concrete actions renders their sustainability intentions into mere rhetoric.

6.5.3 Policy coordination.

The commitment of the Ghanaian Government to the SD agenda is captured in the plethora of programmes and policies aimed at promoting overall development (Alliance for Development 2014). Essentially, the formulation of the NUP recognised the importance of contemporary urbanisation in Ghana; including how it threatens human well-being or paradoxically serves as a platform for future prosperity, thereby reaffirming the country's commitment to SD (NUP 2012). The potential of urban policies to deliver sustainable outcomes is a function of the level of coordination among actors (McGuirk 2012). Accordingly, Peters (2018) has highlighted the advantages and challenges of coordination for the realisation of policy objectives. On the one hand, the author shows that the benefits of policy coordination include: dealing with cross-cutting problems; capturing changing demands; reducing duplication and contradictions among departments; and mitigating 'displacement' tendencies where an actor takes solo decisions without consultation that creates problems for another actor. On the other hand, the author highlights reasons for failure in policy coordination to include: power play in government that leads to inadequate information sharing; and where actors focus on their individual targets at the expense of collective goals. Yet, the NUP and AP do not document a holistic plan or, at the very least, a conceptual integrative map within which its objectives will be operationalised. Some of the

initiatives in the AP that are designed to achieve particular objectives are aligned to specific modal policies, but very little is shown in how they specifically integrate into the NUP. This makes it difficult to see precisely how the multiple objectives of the NUP will be achieved across all levels when, despite a clear goal, the steps to achieve its objectives remain rather nebulous. In essence, the initiatives outlined in the NUP appear to be standalone and the document, as viewed on its merit, does not illustrate the synergistic links between the various policy initiatives, neither does it translate into an integrated sustainability vision. This lack of convergence in sustainability thinking and action in the NUP potentially threatens cumulative sustainability impact. The above discussion provokes thought on whether the formulation of urban policies alone is enough evidence to show that cities are taking sustainability seriously (Portney 2003).

A review of the initiatives of the NUP reveals its reliance on existing policies and outlines a multiplicity of agencies through which it seeks to achieve its objectives. Thus, it can be argued that the NUP has been designed in such a way that the successes of its programmes are dependent on the successes of the other national policies and programmes that are already existing (McGuirk 2012). For example, the NUP makes reference to the National Housing Policy with regard to creating solutions to deal with urban housing problems. This means that any failure in that national policy will potentially affect urban housing. Thus, while the need to coordinate and integrate policies may depend on the context; complex issues such as sustainability and its bi-directional relationship with the urbanisation process which cut across different policy areas and actors, might warrant synergistically coordinating programmes and policies. Nevertheless, while the NUP and its ancillary AP outline how different agencies might contribute to the overall policy goal, they fail to show how these will be harmonised in an integrative manner such that benefits are maximised and conflicts reduced (see Peters 2018). Admittedly, any coordination might happen during the implementation phase, but taking into consideration the complexity of urban systems and the enormity of urban problems in Ghana (see Section 4.2), it is argued that any form of integrative coordination should be prospectively outlined at all stages from policy formulation through to implementation, and more importantly, captured in the main policy document. For example, the actors responsible for economic growth in urban areas may want to promote coal-powered industries while the actors responsible for the environment may want to introduce a strict climate change control regime. Therefore, it is important from the outset to outline in an integrative manner how actors, policies and programmes will be

coordinated if sustainability benefits are to be realised, as one actor could take decisions or implement programmes that may contradict or create problems for another actor. Again, concerning the coordination of different programmes, actors and policies, the NUP does not outline or set any national targets. Here, consistent with some of the reasons attributed to policy failure in the literature by Peters (2018), which include the potential of actors to sideline collective goals in order to meet individual targets, one might argue that setting national targets in the NUP will risk a scenario where no single agency takes responsibility for meeting the targets.

Again, consistent with the view of McGuirk (2012) on how the extent of coordination among actors determines how urban policies could translate into sustainable outcomes, it can be argued from this discussion that while the NUP promises a coordinated development of urban areas in Ghana, the scope of complementary policies and responsibilities of actor agencies seems complex. Effectively, there is some potential for the NUP to combine with other existing policies (for example, National Housing Policy) to reinforce each other, while at the same time the NUP's objectives may not be compatible with others. Although the aforementioned scenario may contribute to greater flexibility in the planning and governance of urban areas in Ghana, another consequence of such complexity may be the dispersion and thinning of resources such that their effect is reduced. As such, maximising the NUP's sustainability attributes would require improving coordination and cooperation both horizontally among national policies and vertically among the different government agencies in a way such that their cumulative effect creates a balance among environmental, social and economic goals.

6.5.4 Political barriers and the role of metropolitan and municipal authorities.

The Ghanaian government sets the context within which urban areas should develop through policies such as the NUP. However, it is within the remit of local governments (metropolitan and municipal assemblies) to design and implement specific local plans in order to achieve national objectives (Uitermark 2005). This provides an excellent case to illustrate the need to have a differential understanding of power when examining the implementation of national policies on the ground. Accordingly, a problematic aspect of the operationalisation of urban policies in Ghana is the top-down nature of formulation and implementation. Unlike in developed countries where municipal authorities exercise a higher degree of relative

autonomy in policy formulation and implementation, as well as possess considerable fundraising powers, city authorities in Ghana tend to be mere political transmitters of central government policies with little powers of their own to significantly tailor such policies or raise requisite funds needed for transformational urban development.

What is apparent here is the importance of taking a more-balanced view of the power relationships between major actors in the policy formulation and implementation arena (Davidson and Arman 2014). As mentioned earlier, the Government, through the relevant ministries and agencies, sets the national policies and is responsible for providing the legislative and regulatory frameworks within which urban areas operate, as well as providing the requisite funding. The role of urban authorities, thus, becomes one of coordinators and implementers at the local level. The challenge of municipal authorities here is not of political support as the central government automatically holds control of local governments. Instead, the challenge is in the effective decentralisation of governing powers due to a lack of a constitutional guarantee of local autonomy. This impacts effective local decision-making and the managerial capacity of local urban governance regimes as the powers of local governments to determine how urban development is carried out is limited. The NUP shows that there are multiple actors assigned to achieve various objectives which could be seen as a positive sign as it shows the depth of institutional capacity, among others. Nevertheless, more often than not, such agencies are likely to compete with or supersede local governments as managers and protagonists of their various responsibilities (see Peters 2018). Such a scenario could indirectly affect the potential of achieving sustainability objectives at the local level as the local government's ability to integrate and coordinate all forms of urban activities is limited.

Another principal area where power relations remain complex is in the area of finance. Broadly speaking, the main source of finance for local governments in Ghana is through central government funding, and in the larger cities, internally generated funds from local taxation. In the bigger cities, private involvement is almost inevitable due to market-based forces (see, for example, OECD 2007; Lauermann 2017). This means that where central government funding is not adequate or forthcoming, market forces could potentially gain considerable influence on decision-making that affects urban sustainability (Wu 2001).

The political barriers to operationalising the sustainability aspects Ghana's NUP are also considered from the perspective of a change in government. As may well reflect the scenario in other less-developed countries, there is a tendency in Ghana, albeit well-within their political mandates, for different governments to change the policy implementation and / or course taken by past governments based on their ideologies or for the purposes of political expediency. A particular government might commit to prioritising economic growth policies, while another might focus more on social development policies or programmes. Also, a government may choose to prioritise long-term development plans while another may decide to stick with short- or medium-term development plans. For example, the National Democratic Congress (NDC) under whose government the NUP and AP were formulated in 2013 also launched a 40-year National Development Plan for Ghana (2018-2057) in the year 2015 aimed at fostering socio-economic prosperity, which would have had an overarching influence on the NUP's implementation and its sustainability operationalisation. However, upon the change of government in 2017, the New Patriotic Party (NPP) government, based on their vision and ideology, launched a 7-year Coordinated Programme of Economic and Social Policies (2018-2024), with a basic aim of job creation, prosperity, and equal opportunity for all. As a central policy, this is also expected to have an overarching influence on the NUP. Essentially, while the NPP government did not abandon or change the NUP itself, the medium-term development route taken will either help to deliver a better implementation of the NUP and its sustainability operationalisation or make it more complex and difficult to achieve; a situation which at this point can only be judged by posterity. What is clear here is that changes in political power potentially influence and change how the wider policy frameworks and development plans will support the NUP over time. This, again, is consistent with the views of Davidson and Arman (2014) who have argued that the prevailing political situation limits the potential of urban policies to deliver transformative change in the sustainability context.

Chapter Seven

7. Sustainable city development and the transition to urban sustainability in Ghana

7.1 Introduction

The sustainability implications of rapid contemporary urbanisation in Ghana, as demonstrated in the previous sections by the comparative sub-case-study on Obuasi and Kumasi, provides a valuable testing ground for assessing the usefulness and practicalities of different concepts of sustainable city development (as explored in Chapter Two). The key question here is how the initial categorisation of sustainable city concepts can be interpreted in the context of sustainable city development in Ghana, and by extension, Sub-Saharan Africa. One might assume that the plethora of sustainable city options highlighted in the literature (see, for example, de Jong et al. 2015), and also evident in physical projects across the globe (see, for example, Joss et al. 2011), may represent an avenue for ready-made solutions that can be tapped into in order to positively shift the sustainability trajectory of urban areas in SSA. For urban areas in SSA, modernism and traditionalism represent two different inherent values (Yuan 2009; Matunhu 2011). An intimate contact with Western urban concepts (as a remnant of colonialism) exists, where most urban plans and processes are still rooted in colonial interpretations of what should be (see Allcott et al. 2006; Boon 2009). Ideas of modernist and ambitious city development concepts proffered by advocates as key to a sustainable urban future (Noorloos and Kloosterboer 2018) also exist. However, the combination of the factors mentioned above and the fragile economic conditions in most SSA cities means that, in practice, there is a perpetual turbulent struggle to discover an identity for a 'sustainable' African city.

Following this introductory section, the remainder of this chapter proceeds in three sections. The first section assesses sustainable city development in Ghana through a case study that analyses the extent to which sustainability principles are embedded in Ghana's pioneer satellite city (Appolonia City). The second section provides an overview of the barriers to urban sustainability in Ghana based on the evidence available from this study. The third section discusses recommendations for urban sustainability transition and proposes a conceptual model of adaptive responses that could facilitate urban sustainability transition based on evidence available from this study.

7.2 Sustainable city development in Ghana: a case study of Appolonia City project

7.2.1 An overview.

Owned by Rendeavour (Africa's largest city developer), the Appolonia City project proposes to deliver a 2,325-acre master-planned city about 20 km away from Ghana's capital city, Accra (Table 7.1). It is widely recognised as the first satellite city in Ghana to be built from scratch. It is expected to accommodate 100,000 residents and host up to 20,000 visitors daily. The developer expects to spend an initial U.S.\$. 250 million to provide the necessary urban infrastructure, out of which U.S.\$. 100 million has already been spent. As a mixed-use development, the developers propose to construct a complete city that provides urban infrastructure and services, including residential, retail, commercial, educational, healthcare, and recreational facilities. They propose to provide up to 300 acres of public parks and gardens, a 200-acre light industrial area and a 120-acre central business district (CBD). Rendeavour has engages other corporate actors in delivering its satellite cities. In Appolonia City, there are partnerships with financial institutions such as Ghana Home Loans Limited (recently acquired by First National Bank) to provide mortgages, and also with other estate companies such as Bijou Homes.

Falt (2019) has provided a historical account of the Appolonia City project which is summarised as follows. The Appolonia City project was officially launched in 2012 by Ghana's President at that time, John Dramani Mahama. The initial construction started after Rendeavour acquired the land from the traditional authorities in 2012. This reflects the dominance of customary land tenure in Ghana, as was highlighted earlier on in this thesis. The land was vacant and parts of it had been previously used by a handful of farmers for small-scale farming. The acquisition of the land was preceded by a two-year reconnaissance and negotiation process. The structure plan and first local plan were approved by the Tema Metropolitan Assembly under whose jurisdiction Appolonia City fell in 2012. By 2017, the first fifty houses and housing foundations had been constructed. The expectation is that the full four-phase Applonia City project will be delivered by 20 years from the start of construction.

Despite being a privately-funded satellite city project, Appolonia City seems to have considerable political and State support. This situation aligns with the already highlighted dominance of neoliberal urban governance in Ghana which has resulted in increased reliance

on private sector provision for urban development. In the specific case of Appolonia City, published research (Falt 2019) shows that the project is widely embraced and supported by both central and local government institutions due to a strong preference for privatised urbanism in Ghana. Such a preference is borne out of the idea (perceived or factual) that the private sector is better positioned to deliver urban development projects and services compared to the central government which lacks the capacity to meet the urban development needs of a rising population (Falt 2019). Consequently, at the launch of the first phase of a 2,000-unit 'affordable housing' project in Appolonia City in 2017, the speech given by Ghana's President, Nana Akufo-Addo, reiterated political support for the project as he acknowledged that "Appolonia City is a laudable initiative and I want to encourage others to emulate. I'm happy to note that the Appolonia Community is a shareholder in the project. The Government I lead is a national cheer leader of the private sector and we should do all we can to provide the enabling environment" (Appolonia City 2019c, para. 4).

Furthermore, Appolonia City project enjoys wide political and State support due to the rationality that associates urban development with economic progress (Falt 2020). This is promoted by how the vision of Appolonia partially aligns with the policy objectives outlined in the NUP in the previous chapter (see Section 6.2). On the one hand, the developer of Appolonia City claims to seek to go beyond the challenges of stifling urban congestion and a dearth of quality housing by creating infrastructure– *"living and working spaces, communities, schools and hospitals- that will help sustain and accelerate Africa's economic growth, meet the aspirations of Africa's burgeoning middle classes, and serve as a catalyst for further urban development"* (Appolonia City 2019b, para. 2). On the other hand, enhancing socio-economic development by promoting sustainable, spatially integrated, and orderly urban development is the kernel of Ghana's NUP. This convergence of objectives underpins the significant political support given to the Appolonia City project.

Finally, the communities that surround Appolonia City are fundamental to its existence (Isakson 2019). This is because the construction of the city would not have been possible if the traditional authorities had not sold the land, which they held in trust for the community, to Rendeavour. Although verifiable details are scant, the local communities are believed to own 10% of Rendeavour's subsidiary in Ghana, which is known as Appolonia City Development Company (Falt 2019). As a result, a community trust fund has been set up into which the yearly dividends would be paid and utilised for developmental activities in the local

communities. Therefore, the local communities have been designated as official partners of the Appolonia City project and the traditional authorities reinforce this partnership by publicly supporting the project and performing ceremonial functions (Falt 2019). For some inhabitants of the local communities, the construction of the Appolonia City project presents an opportunity to participate in a new labour market (Isakson 2019). In this context, Isakson (2019) has reported that some local residents had already been employed as security guards, construction workers, etc. Although no large-scale evictions occurred (Falt 2019), the construction of Appolonia City may represent a loss of livelihood for the few small-scale farmers who were evicted to pave way for the project. However, the reality for all residents in the local communities is the imminent transformation of the previously rural landscape into an urban one dominated by Western-world architecture. Overall, the local communities seem to support the Appolonia City project as published research by Falt (2019) has reported that there is no evidence of bold resistance to the city's construction.

Characteristic	Comment	
City type	Satellite, mixed use development	
Total size	2,325 acres	
Ownership	Private	
Initial investment	U.S.\$.250 million	
Population	Up to 100, 000 residents	
Expected daily visitors	Up to 20,000	
Housing units	25,000	
Central Business District	120 acres	
Light industrial park	200 acres	
Public parks and gardens	300 acres	
Planning	Master-planned	

 Table 7.1: Characteristics of the Appolonia City project.

Source Appolonia City (2019a)

7.2.2 A sustainability evaluation of Appolonia City.

An evaluation matrix showing how Appolonia City ranks based on available data, and in comparison to the rankings designated to an eco-city (sustainable city) as defined by Jabareen (2006) is presented in Table 7.2. The evaluation is explained as follows. Numerical values of

3, 2, and 1 were assigned for 'high', 'moderate' and 'low' rankings respectively. Therefore, based on Jabareen's (2006) definition, an eco-city would be characterised by the following: (i) moderate mixed land use; (ii) moderate diversity in the representation of different income groups, household size, etc.; (iii) high level use of sustainable transport means of transport such as cycling; (iv) high level use of sustainable energy forms such as passive solar energy; (v) less compact urban forms; (vi) high amount of green spaces; and (vi) moderate density of development. The definitions and assumed benefits of the factors contributing to a sustainable urban form have been presented in Table 2.1.

From Table 7.2, Appolonia scored 13 out of a possible total of 21, compared to the eco-city defined by Jabareen (2006) which scored 16 out of 21. Specifically, Appolonia City was ranked lower than Jabareen's eco-city in three categories (diversity, sustainable transport, and sustainable energy) out of the seven categories considered; the two cities were ranked equally in three categories (compactness, greening, and density), and Appolonia City was ranked higher than Jabareen's eco-city in one category (mixed land use). The general inference from the matrix presented, therefore, is that, to a considerable degree, Appolonia City does not conform to the visions of a sustainable city based on Jabareen's (2006) sustainable urban form matrix. What follows is a narrative that attempts to justify the evaluation of the extent to which the Appolonia City project meets the visions of a sustainable city.

Factor	Jabareen's evaluation	Appolonia City (author's
		evaluation)
Mixed land use	Moderate (2)	High (3)
Diversity	Moderate (2)	Low (1)
Sustainable transport	High (3)	Low (1)
Sustainable energy	High (3)	Moderate (2)
Compactness	Low (1)	Low (1)
Greening	High (3)	High (3)
Density	Moderate (2)	Moderate (2)
Total Score	16	13

 Table 7.2: Evaluation matrix for Appolonia City based on Jabareen's (2006) sustainable urban form matrix.

Source: Author as adapted from Jabareen (2006).

(a) Mixed land use.

Appolonia City was ranked high (3) in the mixed land use category as its master plan shows that the development is a strategically planned mixed land-use zone (see Figure 7.1 below). With approximately 2,325 acres of land area, its master plan proposes a land use mix that includes a 120-acre CBD to host commercial activities; a 200-acre light industrial area, over 300 acres of public parks, and residential housing areas. This means that Appolonia City could potentially maximise the sustainability benefits of agglomeration of mixed land use, which includes shorter commute distances for people who will live and work inside or near the city (see Table 2.1).



Figure 7.1: Map of Appolonia city showing land-use mix. Source: Appolonia City (2019f).

(b) Diversity.

Appolonia City was ranked low (1) in the diversity category due to an apparent lack of inclusivity. Despite the political support that the project enjoys, Appolonia City is an inherently commercial project whose top-most priority will be to generate profit in order to secure the interests of its investors. Consequently, the characteristics of the project shows that it has not been designed to suit the needs of low-income groups. This is succinctly captured in the vision of its developer which has already been expressed in Section 7.2.1. While the mixed-use design and the presence of commercial and industrial areas could attract prospective employees from low-income backgrounds who will diversify the social mix in the city, access to residential facilities, the most significant aspect of the city project, is likely to exclude people from low-income backgrounds. For example, the cost of a plot of land measuring '40ft x 70ft' in the Nova Ridge Estate in the city starts from GHC 112,750 (equivalent to U.S.\$ 20,000 [Appolonia City 2019d, para. 4]); while a 1-3-bedroom Oxfordtype house starts from GHC 478,750 (equivalent to U.S.\$ 80,000 (Appolonia City 2019g, para. 1). Taking cognisance of the level of poverty in Ghana (see Section 4.2.1.3), it is potentially unlikely that any part of Appolonia City is within the financial means of the average Ghanaian.

Consequently, the Appolonia City project has the potential to be an elite enclave and a practical manifestation of neo-liberal mainstreaming in urban development (see Section 2.2.4), as evidence of its initial phase suggests the construction of a modern but expensive city. In order to get the quantity of land needed to establish this city, the developer had to acquire land from land owners in communities which are typically rural. Researchers from the University of Bergen (Norway) have captured the contrast in the development between Appolonia City and the surrounding communities as shown in Figure 7.2 and Figure 7.3. On the one hand, Figure 7.2 shows that the houses in Appolonia City are well-planned and have been constructed with expensive materials. On the other hand, Figure 7.3 shows an unstructured cityscape with houses mainly constructed from inexpensive materials in the nearby village. Accordingly, the developer attempted to improve conditions in the nearby communities through its corporate social responsibility projects. For example, it has renovated the basic school building in the Appolonia village community in order to provide a conducive environment for learning and to enhance education in the community. This can be seen in Figure 7.4 which shows the state of the basic school in Appolonia village before and after renovation. While this is laudable, it appears tokenistic and has probably been done in

order to maintain the developer's social licence to operate. From the foregoing discussion, it is evident that inclusivity in urban development is low in the Appolonia City project.



Figure 7.2: Image showing 'The Oxford' types of houses under construction in Appolonia City. Source: University of Bergen (2019).



Figure 7.3: Image showing the housing conditions in a nearby Appolonia village. Source: University of Bergen (2019).



Figure 7.4: Image showing basic school in Appolonia village before and after renovation. Source: Appolonia City (2019e).

(c) Sustainable transport.

Appolonia City was ranked low (1) in the sustainable transport category due to a lack of an explicit mention of sustainable transport systems in any of its publicly available information. Appolonia City proposes to provide permanent paved roads (see Figure 7.5); dedicated pedestrian walkways, and jogging and biking trails. Nevertheless, for a satellite city that is expected to accommodate approximately 100,000 residents (in addition to about 20,000 daily visitors), a lack of a dedicated public transport system is of considerable significance. Recalling the general urban transport challenges in Ghana discussed in Section 4.2.1.5; including the increasing rate of vehicle ownership (Obeng-Odoom 2009a), it can be speculated that private vehicle use is expected to be the main source of transport in Appolonia City. This lack of alternative sustainable transport has potential consequences, including air pollution and further stress on the already-poor public road infrastructure. Surrounding communities are likely to be affected as the impacts of activities in urban areas transcend physico-administrative boundaries (see, for example, Simon 2010). It may be fair to argue that since the project is proceeding in phases there is a likelihood that a sustainable mobility option such as Bus Rapid Transit (BRT) may be provided later, and the capital requirements of an initiative like BRT mean that its financial viability might only be feasible in the later stages of the city's development. Nevertheless, the relevance of sustainable

mobility to sustainable city building cannot be understated, thereby warranting its explicit capture in project-related plans, statements, visions, and websites.



Figure 7.5: Image showing transport infrastructure in Appolonia City. Source: Appolonia City (2019).

(d) Sustainable energy.

Appolonia City was ranked moderate (2) in the sustainable energy category. This is because it proposes to provide a constant supply of electricity (a rarity in Ghana) to the prospective residents and businesses through dedicated bulk supply points and transformers. Furthermore, as shown in Figure 7.6, Appolonia City promotes sustainable energy use as some of its houses are fitted with solar panels. This ensures that the city, to a considerable extent, maximises the use of passive solar energy which is available in abundance due to the tropical climate in Ghana. The potential storage and use of solar energy has added sustainability benefits such as improving the local micro-climate in Appolonia City (see Table 2.1). However, the potential dominance of private vehicular use as the main source of transport in the city is likely to erode the possible sustainable energy gains due to increased vehicular emissions.



Figure 7.6: Image showing an Appolonia house fitted with solar panels. Source: Appolonia City (2019a)

(e) Compactness.

Appolonia City was ranked low (1) in the compactness category. While the mixed land use design adopted by Appolonia City could lead to a compact and more sustainable urban form, the cityscape is dominated by residential housing and, as shown in Figure 7.5, the majority are semi-detached, detached or terraced. These types of houses are typically designed for single families. This means that more geographical space is being used for house construction than would have been used if multi-storey buildings were to be constructed.

(f) Greening.

Appolonia City was ranked high (3) in the green category as up to 300 acres (over 10% of the total area of 2325 acres) has been designated for use as public parks and gardens. In addition, the residential housing constructed so far in Appolonia City (see, for example, Figure 7.2 and Figure 7.5) show the presence of manicured lawns. The significant greening component in Appolonia City could accrue a myriad of sustainability-related benefits (see Table 2.1). First, it will ensure that biodiversity in the city is protected, as well as create opportunities for people to interact with biodiversity. Second, the parks could enhance social cohesion within the city as they would serve as recreational avenues for residents to meet and mix. Third, potential woodlands in Appolonia City could help mitigate climate change by serving as carbon sinks in order to absorb the expected GHG emissions concentration due to the

potential dominance of private vehicle use discussed earlier. Overall, the urban parks and gardens will also make the city aesthetically pleasing which could accrue health benefits for people who patronise Appolonia City (see Table 2.1).

(g) Density.

Appolonia City was ranked as moderate (2) in the density category. While the expected population of about 100,000 people and the concentration of economic activities are typical of high-density areas, the city's plan (Figure 7.1) and the residential housing constructed so far (Figure 7.2; Figure 7.5) suggest otherwise. Essentially, the sustainability benefits of population concentration in Appolonia city is potentially eroded by increased GHG emissions as low-density residential housing influences private vehicle use.

7.2.3 Appolonia City: A sustainable city?

Reconciling the normative visions of a sustainable city expressed earlier in this thesis (see Section 2.3.3) and the characteristics of the Appolonia City project, the evidence and discussions so far do not suggest a total convergence of characteristics. While this study recognises the on-going construction of the Appolonia City project, the evidence and discussions have provided a snapshot of the mainstream conceptions of sustainable city development in Ghana. Accordingly, as evidenced by Appolonia City, it can be implicitly assumed that the practical implementation of concepts of a 'sustainable city' in Ghana leads to the potential construction of an urban space which is innovative and more compact; allows mixed land use; and is aesthetically pleasing. However, rather than promote inclusive urban development that enhances the achievement of greater and wider sustainability, the characteristics of the project portray a potential creation of an elite enclave that is beyond the reach of the ordinary Ghanaian and is potentially dominated by unsustainable practices such as private car dependency. Furthermore, when the images of Appolonia City are juxtaposed with those of the surrounding Appolonia villages, they demonstrate the dominance of Western ideas and a lack of cultural sensitivity to urban development in the Ghanaian context. While not in the same dimension, the findings are consistent with those expressed in the literature (see for example, Caprotti et al. 2015) where new city developments are potentially beneficial to its inhabitants but limited in its contribution to wider sustainability. Overall, from the perspective of this study, the general inference that can be made, based on evidence available on Appolonia City, is that there is a significant gap between the

conceptions of sustainable city development expressed in Chapter Two and its implementation in the Ghanaian context.

7.3 Barriers to sustainable city development in Ghana

From the review of the Appolonia City project in the previous section, it can be argued that the mere existence of sustainable city concepts which have been designed based on sustainability principles does not guarantee the creation of sustainable urban areas on implementation. Intrinsic barriers and external influences which may include, but are not limited to, geopolitics, land tenure, governance, economic, and environmental factors, tend to make it more challenging to address the urban sustainability problems of cities in less-developed countries, and specifically in this context, SSA. The discussions in the following paragraphs focus on how politics and finance critically moderate the interplay between urbanisation / city development and the broader sustainability agenda. While a myriad of barriers exists, from the perspective of this thesis, the politically-charged nature of urbanisation (Pieterse 2008; McGranahan et al. 2009; Jones and Corbridge 2010; Parnell and Simon 2010) and the pervasiveness of poverty in Ghana (see Section 4.2.1.3) make the factors of politics and finance significant in explaining why sustainable city concepts have been slow to gain a foothold in SSA.

(a) Politics.

Political power is inextricably linked to and influences any social process, and the successful implementation of sustainability-related initiatives is shaped, in no small extent, by the prevailing political economy (Davidson and Arman 2014). For Ghana, implementing any concept for urban sustainability, just as any other development process, is fundamentally a political exercise. Like most of SSA, where economies are dependent on natural resource extraction, there is a potential risk that Ghana's path to growth will fail to build connections between urbanisation and sustainable development as it is believed that resource-based growth is not naturally urban-oriented (UNECA 2012a; Turok and McGranahan 2013). This potentially puts governments under pressure to create alternative economic opportunities that support the livelihoods of citizens. Also, governments are keen to associate and endorse projects related to the concept of sustainable development for political expediency. Consequently, governments tend to be receptive to the idea of sustainable city projects as a way of diversifying their economies and transitioning from natural resource dependence. This

means that governments encourage private investors to develop projects such as Appolonia City as normative or prescriptive options for city building where economic growth, be it GDP increment or investment profits, leads to the achievement of public goals including urban regeneration (OECD 2007; Wang and Tomaney 2019). This makes it safe to assume that major city projects that are being promoted for their alignment with the principles of sustainability (see Section 2.3.3) are mainly pursued for their economic competitiveness. However, as legitimate as the goal of creating economic growth is, how it ensures the economic, social, and environmental well-being of the wider urban system and society is not always clear. In effect, governments could be focussed on how GDP is increased through such projects while the private-sector also focuses on their profits. Thus, given the geopolitical and physico-administrative realities of urban systems in SSA (Simon 2010), there is a risk of corporatisation and neo-liberalisation of urban systems. Subsequently, if priorities are not well-managed in the planning and distribution of potential benefits during the implementation of the sustainable city concept, sustainability outcomes could be undermined (Wu 2001; Young et al. 2006; OECD 2007). For example, as noted earlier, scholars, including Mishra (1999), have suggested that the neo-liberal realignment in urban areas promoted by globalisation has never guaranteed social integration, cohesion or the survival of the social groups which its market mechanisms tend to marginalise. The implications could be harsher in SSA as most of these concepts have been developed based on realities in the global North, where the pre-conditions for sustainable city are well entrenched in planning processes compared to SSA. Crucially, political dispensation could impede the achievement of sustainable outcomes in the implementation of sustainable city concepts as citizens' interests (especially of vulnerable groups) are projected as subservient to corporate and government interests (Wu 2001; OECD 2007; Lauermann 2017). This could be linked to the themes of urban entrepreneurialism, urban exclusion, and citizen participation, discussed earlier in Chapter Two.

(b) Finance.

In general, SSA cities lack the financial capacity to cope with urbanisation (Smit and Parnell 2012). Accordingly, the substantial capital requirements for establishing new cities like Appolonia City, or retrofitting existing ones in a way consistent with the principles of sustainability, mean that they are mostly financed by private investors or through public-private ventures (OECD 2007; Lauermann 2017). Furthermore, the Appolonia City sub-case-study has shown how the capital-intensive nature of implementing sustainable city concepts

means that more often than not, they become elite enclaves for people who can afford to live there (See, for example, Mishra 1999; Anthopolous 2010; Obeng-Odoom 2013a). Critically, the argument above shows that greater and well-targeted investment will be needed from governments and the private sector if the uptake of new city concepts is to shape the creation of fair and sustainable urban societies.

7.4 Towards a transition to urban sustainability in Ghanaian urban areas

The evidence and discussions put forward in this thesis (i.e. review of urban challenges in Ghana in Chapter Four; the general sustainable development situation in Ghana in Chapter Four; landscape and demographic urbanisation in Chapter Five; the subjective experience of urban citizens in Chapter Five; the sustainability appraisal of the Appolonia City project in this chapter) suggest that urban areas in Ghana, as currently constructed, are on a trajectory of unsustainability. These trends, if allowed to persist in an era of rapid urbanisation, will expand the catalogue of unsustainable urban practices and reinforce the unsustainability of Ghana's urban areas. The critical challenge for urban sustainability in Ghana, therefore, is to steer urban areas onto a trajectory of sustainability through an adaptive and systemic transformation of their urban systems in response to rapid urbanisation. Building on the findings and discussions from the previous chapters, this section introduces discussions on potential responses and strategies through which the transition to urban sustainability could be facilitated in the Ghanaian case. It is important to note, however, that the responses for a transition to urban sustainability recommended from this study are not exhaustive or mutually exclusive. Rather, within the scope of this study, they are intended to capture the distinct ways in which a transition from the status quo to sustainable pathways could proceed based on the findings and discussions produced. The remainder of this section proceeds in two parts. The first part presents recommendations as potential strategies that could facilitate the transition to sustainability in Ghana's urban areas. Using the DPSIR framework, the second part suggests an adaptive conceptual model to facilitate urban sustainability transition decision-making in the Ghanaian context, based exclusively on the scope and findings of this study.

7.4.1 Towards a transition to a trajectory of sustainability in Ghana's urban areas.

The sustainability solutions embedded in the various approaches to urban sustainability identified in Chapter Two recognise the nexus between urban sustainability and future global

sustainability. It will, therefore, be a logical corollary to assume that proponents of the respective approaches would have the belief that any particular approach offers the requisite tools and principles to support broader sustainability discourse and implementation. Nevertheless, it is evident from the findings on the comparative sub-case-study explored in Chapter Five that no single approach could adequately address fundamental sustainability processes *vis a vis* the multiple challenges to urban systems in Ghana described in Section 4.2.

While not discounting the sustainability solutions proffered by the different schools of thoughts on how best to induce a transition to a trajectory of urban sustainability through sustainable city development (see Section 2.3.3), the dynamics of urban systems in SSA, coupled with rapid rates of urbanisation, generally demand an eclectic strategic approach in order to address the complex needs of urban areas in SSA in a manner consistent with sustainability principles or the ideals of a sustainable city. From the perspective of this study, and consistent with the views of other scholars (see, for example, Campbell 2012), an eclectic approach, which in this context can be described as a consolidation and strategic application of a range of principles and policy prescriptions from different sustainable city concepts and transition strategies will help shape the urban sustainability trajectory in Ghanaian cities by providing opportunities to critically analyse what is being done (prescriptive actions), vis a vis what ought to be done (normative actions), rather than build cities based on particular concepts. The practical implication is that relevant decision-makers would have to carefully select the best available solutions from the various approaches to sustainable city development and tailor them to the challenges that need to be solved at the local urban level. This means that while the transition to future urban sustainability might be potentially too complex to manage due to rapid contemporary and future urbanisation, adopting an eclectic approach for urban transition offers an opportunity to identify and synthesise the value-added that consolidation of the sustainable city concept ideas might provide to urban sustainability processes. The following paragraphs will, thus, discuss recommendations that will potentially facilitate a transition to a trajectory of sustainability in Ghanaian urban areas. The recommendations are presented in ways that allow for the consideration of multiple pathways for facilitating a transition to a trajectory of sustainability, rather than considering a single most-likely pathway. Essentially, this discussion is not intended to be exhaustive, rather it is demonstrating the wide-ranging ways in which the transition to a trajectory of sustainability may be realised based on the findings of this study.

7.4.1.1 Policy and organisational context.

Transitioning to urban sustainability requires changes to the institutional foundations which shape dominant and prevailing practices. Accordingly, sustainable urban transformation, which is the basis for any transition to urban sustainability, relies to a large extent on urban governance. This is so because the prevailing governance mechanisms embody the structural rules that determine functioning of urban systems and the trajectory of sustainability. The findings in Chapters Four and Five have highlighted the rapid pace of urbanisation in Ghana and the intractable challenges it poses to sustainable development at the local urban level. While the prevailing conditions in Ghana's urban areas mean that urban design and planning provisions have been potentially outpaced, their transition to sustainability can be guided. This requires a fundamental overhaul of policies and governance practices that guide urbanisation and sustainable development. On a general level, the government must pursue an agenda for policy and institutional reform in the context of sustainability. This means that, if the government is truly committed to SD, it must fully align its policies, institutions and practices to the principles of SD.

(a) Urban policy.

Ghana's National Urban Policy was formulated and implemented in 2013 in order to facilitate the transition of its urban areas into sustainability (see Chapter Six). Centrallyembedded within the NUP is the sustainability concept as demonstrated in its urban development strategies contained in the Action Plan. Nevertheless, the analysis of the sustainability focus of the NUP showed that it is skewed towards economic dimensions at the expense of social and environmental dimensions. A general inference from the findings on Ghana's urban policy in Chapter Six is that the government's policy-making ideology is fundamentally shaped by an affinity to establish sustained economic growth. The transition to a trajectory of urban sustainability would, however, require an ideological shift in policy-making from the dominant focus on economic development to a more balanced focus among the environmental, social and economic dimensions of sustainability.

With over half of Ghanaians living in urban areas, the findings on the assessment of sustainable development and the citizens survey in Chapter Five demonstrate the importance of meeting the basic needs of people for achieving greater sustainability in urban areas. This requires formulating and implementing policy measures that will improve the well-being and quality of life of urban citizens in Ghana. Taking cognisance of the poverty situation in

Ghana (see Section 4.2.1.3), which is exacerbated by neo-liberal dominance in Ghana's urban systems, a transition to a trajectory of sustainability specifically requires formulating policies to address the socio-economic needs of urban citizens with an explicit focus on vulnerable groups and social justice. Furthermore, the perceptions and contributions of citizens towards enhancing urban sustainability in places they live tend to depend on the extent to which they can meet their current needs. The findings in Chapter Five have captured how citizens expressed their lived experiences in the context of sustainability (based on their perspectives of selected urban sustainability statements). In the Ghanaian case, evidence from Obuasi and Kumasi suggests that such perceptions are dominated by economic concerns. This creates a need to incorporate the preferences and opinions of the local population in sustainability policy-making and planning in order to achieve greater sustainability in Ghana's urban areas.

In order to steer Ghana's urban areas onto paths of sustainability, much more fundamental research needs to be undertaken. As was evident in the analysis of Ghana's NUP, the absence of relevant emissions targets shows that there is an apparent lack of data on urban systems, including climate, which can be used to inform policy. This makes it relevant for the authorities to conduct research that helps to develop Ghana-specific indicators and establishes criteria, based on which progress to sustainability could be evaluated for policy-making purposes. While evidence from the literature reviewed in Chapter Two highlights the various indicator sets that could be used to measure urban sustainability, this thesis has already stressed how there is no universally accepted set of indicators, as well as the difficulty in acquiring data on indicators. Furthermore, as evidenced by the Appolonia City project presented earlier, the societal and political conceptions of a sustainable urban area tend to contradict inclusive sustainable development. Therefore, carefully targeted research to produce a dynamic model of the quintessential Ghanaian urban area as a sustainable city would serve as a reference for policy-making based on which the sustainability transition trajectories of Ghanaian urban areas could be evaluated and facilitated.

From a policy coordination perspective, the urban policy analysis in NUP showed that the formulation and development of sustainability-related policies and plans can be challenging due to the fuzziness of sustainability as a concept (see, for example Jabareen 2008); a situation made complex by the need to ensure that such policies complement and are compatible with existing policies and plans. Therefore, stimulating a transition to a trajectory

of urban sustainability would require an enhanced coordination of sustainability-related policies and plans in order to ensure that their synergistic effects are maximised.

Last but not least, the political nature of sustainability as a concept means that the components of Ghana's NUP have potentially been selected based on what is politically feasible, which, in turn, is potentially imposed externally. For example, as the uptake of national urban policies has been recommended by an international organisation like the United Nations (UN-Habitat 2016a), there is the potential that components of the NUP that are considered deliverable have been framed by the perspectives of such international organisations. As demonstrated in Chapter Six, a resulting implication is the tendency of urban policies to be considered as rhetoric. Thus, from an urban policy perspective, a transition to urban sustainability will be successful if realistic and politically deliverable targets are chosen as policy solutions.

(b) Institutional.

Generally, the transition to a trajectory of urban sustainability will demand an adjustment of organisational infrastructure across all relevant institutions. From an organisational perspective, the narratives from the analysis of Ghana's NUP in Chapter Six highlight the need for greater decentralisation in order to enhance local government engagement in the context of sustainability. An ethos of centralisation in Ghana means that while municipal and metropolitan government authorities are the implementers of the NUP, it is not clear how they can effectively contribute to achieving the NUP's objectives independently in the absence of a strong local mandate. Crucially, in order for Ghana's urban areas to transition to a trajectory of sustainability, there is the need to create a synergy between the implicitly top-down nature of policy formulation and effective local government through greater decentralisation. Decentralisation, in this case, is expected to promote urban sustainability by reinforcing and empowering local level decision-making. In essence, the transition to a trajectory of urban sustainability will be shaped by the extent to which a shift in the hegemonic power structures governing urban areas in Ghana is achieved.

The expansion of urban land cover at the expense of natural vegetation (Chapter Five) and the ecological deficit caused by the ecological footprint of Ghana's population exceeding biocapacity (Chapter Four) demonstrate the unsustainable manner in which development is proceeding from a resource consumption perspective. Mitigating these trends of natural resource consumption and facilitating a transition to greater sustainability will, thus, require a shift from reactionary governance to precautionary governance. This includes the need to reform the country's tax systems in order to introduce instruments such as ecological tax. Furthermore, the analysis of urban policy in Chapter Six showed how the lack of mechanisms for assessing environmental impacts potentially undermine sustainability efforts. This demonstrates a need to enhance the capacity of relevant institutions to internalise the environmental cost of urbanisation and related socio-economic development. This requires concerted efforts for data collection and the development of relevant accountability mechanisms for effective environmental impact assessment.

The transition to a trajectory of sustainability in Ghanaian urban areas will benefit from deepened coordination between the central government and other relevant organisational stakeholders. From a resource-based city perspective, the Obuasi case study has demonstrated how Obuasi's local economy is fundamentally dependent on the activities of AngloGold Ashanti and how their activities have shaped the quality of the urbanisation process, including the provision of jobs and services for the increasing population in Obuasi. The strong emphasis placed by respondents on economic sustainability in the survey demonstrates the inter-linkages between the economic prosperity of citizens and the growth of businesses such as AngloGold Ashanti. Consequently, the capacity of organisational stakeholders, including AngloGold Ashanti, to shape the transition to sustainability from an economic perspective will be enhanced through stable macro-economic policies that potentially safeguard the growth of businesses, as well as through incentives like tax breaks. Furthermore, the deleterious environmental effects of the mining operations of AngloGold Ashanti have been widely acknowledged in the literature and from the responses of participants in the citizens survey (see, for example, ActionAid Report 2006; Foli et al. 2012). As such, environmental sustainability in a place such as Obuasi could be enhanced through coordinated partnership for environmental stewardship between AngloGold Ashanti and the relevant government authority. A coordinated partnership, in this case, would be expected to provide the financial and logistical resources, as well as an effective monitoring and compliance regime that may be necessary to achieve greater environmental stewardship and sustainability.

From a planning perspective, rapid urbanisation and the dominance of market forces pose potential challenges to the transition to sustainability due to challenges including weak planning regimes and capacity (see, for example, Bryceson et al. 2009; Turok and Parnell 2009; Bekker and Therborn 2012; Fox 2014). The time series analysis of land cover change and the growth of urban population captured in Chapter Five demonstrate the spatial and temporal ways in which sustainability as a concept challenges urban planning practice in Ghana. As the (un)sustainability of Ghanaian urban areas is primarily a matter of cumulative impacts, the adequacy of the prevailing planning regime is critical as it is the arena where strategies to achieve greater sustainability over the long term will be conceived, mobilised, accepted or rejected. The extent of urban expansion at the expense of natural vegetation cover in the two case study areas in Chapter Five demonstrates how market forces and the land tenure system (see Sections 2.2.4 and 4.4.1) have weakened the capacity of relevant authorities to take regulative measures to limit the spatial consumption of land. Furthermore, from a market-force perspective, the Appolonia City project assessed in the second section of this chapter demonstrates how the central government's objective of promoting local growth and supporting the private sector potentially weakens the regulatory powers of planning authorities. As the central government exploits the financial liquidity of the proponents of Appolonia City in order to alleviate the acute housing shortage in Ghana (estimated at 1.7 million units; see Daily Graphic Online 2018) the potential creation of an elite enclave and the car-dependent land use are not consistent with the sustainability of the wider region around Appolonia City. Achieving greater sustainability for the wider Appolonia City area, thus, requires enhancing the capacity and independence of planning authorities to ensure that such projects are well anchored within the wider regional plans in ways that lead to just and sustainable urban development. This includes protecting the role of experts like planners in decision-making processes for such projects.

The fuzziness and trans-disciplinary nature of sustainability as a concept has been stressed throughout this thesis (see, for example, Jabareen 2008). This means that not only urban instruments such as plans, finance mechanisms, and policies, but crucially the strategic competence of the human resources who deal with such instruments, are critical for implementing them in ways that promote a transition to sustainability. Therefore, from a human resource perspective, a transition to a trajectory of sustainability will require that decision-makers are abreast with relevant knowledge on the synergies among the various dimensions of sustainability. This includes sensitising and training decision-makers on new technologies and new sustainability ideas, as achieving greater sustainability in Ghana's urban areas in the midst of rapid urbanisation will potentially create uncertainties which
require innovative thinking. Crucially, opportunities to sensitise and train human resources are expected to enhance their capacity and competency to respond to complex social, environmental and economic issues that affect urban systems and citizens in Ghana. Overall, the quest for a transition to a trajectory of sustainability in Ghanaian urban areas requires a reshaping of the institutional capacity around sustainability in order to enhance the management of the spatial implications of rapid urbanisation and population growth. This requires pursuing an agenda for integrated sustainability planning, decision-making, and investment; as well as enhanced planning regulation enforcement mechanisms, and good urban governance.

(c) Private Sector.

From a strategic niche management perspective, 'niche' developments can emerge to promote novel visions and proffer alternative solutions for sustainability transition (see Section 2.3.4). Consequently, the sub-case-study on Appolonia City demonstrates how actors from the private sector are capitalising on the salient urban challenges in Ghana by promoting new satellite cities as 'niches' meant to challenge the status quo on urban development in Ghana. Critically, these new city projects have been purposefully designed to disrupt the current 'regime' of urban development in Ghana through articulated visions of practice, which proponents believe are superior alternatives which will enhance urban sustainability in Ghana. Using the case of sustainable transport, the review of urban challenges in Ghana in Section 4.2 highlighted the poor-quality state of transport infrastructure in Ghana's urban areas and how this has contributed to increased private vehicle use in Ghana. In the context of the Appolonia City project, a paradox emerges as its proponents pledge transport infrastructure such as pedestrian walkways and biking trails while the absence of public transport suggests that the overall design promotes private car dependency. Critically, achieving greater sustainability for the Appolonia City project and the surrounding region requires its proponents to reorient its masterplan to introduce sustainable forms of transport, including BRT, in order to mitigate long-term environmental costs due to potential private car dependency. Furthermore, while private sector developments such as Appolonia City represent avenues of experimentation through which sustainability principles could be infused into urban development, an effective transition to sustainability should involve the uptake of urban development practices which are suited to the individual conditions prevalent in most urban areas of Ghana. This includes acknowledging the legacy of the existing local areas and infrastructure. Ultimately, the discussions above have shown that the effectiveness

of 'niche' developments like Appolonia City in disrupting the urban development status quo into a trajectory of sustainability is not guaranteed unless careful considerations are made to tackle issues such as wider inclusivity and the role of sustainable transport in promoting sustainable lifestyles. Overall, the private sector stakeholders must maintain good corporate governance mechanisms which simultaneously ensure investment returns and mitigate externalities (social, environmental, economic) on other stakeholders and the wider urban system.

7.4.1.2 Individual and Community contexts.

While the Government of Ghana, through the various departments, is responsible for providing an efficient urban structure, the behaviour and practices of prospective users of the urban systems are critical as they are the ones who appropriate urban spaces in ways that suit their requirements. Generally, in order to realise sustainable development in Ghana's urban areas, there must be behavioural change at the local level.

(a) Individual.

The acceptance, support, and participation of citizens are critical if a city is to successfully realise its sustainability ambitions (Laurian and Crawford 2016; Swann 2017). With residents being the target groups and primary beneficiaries of urban sustainability initiatives, there is no better opportunity to develop more effective sustainability plans, policies, and projects than by actively involving urban citizens. In general, the findings on the perceptions of citizens in Chapter Five provided useful insights that could potentially help to improve sustainability-related strategies, as well as help in the planning and implementation of policies in ways that address the nuances in opinions and attitudes of citizens towards urban sustainability. For example, as it was found that citizens expressed a high willingness to engage in sustainability-enhancing practices such as recycling, efforts should be made to influence the behaviour and attitudes of individual citizens towards sustainability action. Such actions could be realised through education programmes that are geared towards increasing the level of awareness of the sustainability-related policies, projects and plans amongst local citizens.

(b) Community.

In addition to individual actions, participatory community action is required to induce a transition to sustainability in Ghana's urban areas. This arises out of the need to mitigate the weakened capacity of the government to promote urban sustainability due to issues such as weak planning and regulatory frameworks, inadequate finance, and politics, as has been stressed throughout this thesis (see, for example, Bryceson et al. 2009; Turok and Parnell 2009; Bekker and Therborn 2012; Fox 2014). Generally, participatory community action could promote voluntary and locally-led sustainability solutions. These include, but are not limited to, recycling initiatives, sustainable energy initiatives and sustainable transport initiatives. For example, with increasing private car dependency and its attendant environmental impacts in Ghana's urban areas (see Section 4.2), participatory community action could promote a shift toward sustainable lifestyles through initiatives such as car sharing and walking clubs. Local urban communities could also stimulate a transition to sustainability by generating knowledge about aspects of sustainability through participatory community research. Furthermore, as has been maintained throughout this thesis, sustainable development is a dynamic process. Thus, from a sustainable community development perspective, transitioning to sustainability requires resilience such that communities can adapt to changes accompanying the transition. Overall, it is expected that community-led action could potentially facilitate a bottom-up transition to a trajectory of urban sustainability by mitigating government inaction on sustainability-related issues or where existing, augment the government's sustainability efforts.

7.4.2 Conceptual model for adaptive response towards urban sustainability transition in Ghana.

An ideal model for an adaptive response towards a transition to urban sustainability in the Ghanaian context is shown in Figure 7.7. Rather than a grand conceptual model, Figure 7.7 captures the complexity of the interactions between urbanisation and sustainable development in Ghana through a reflection on the findings of this thesis. By modifying the response component of the DPSIR-framework (see Figure 2.4), Figure 7.7 tentatively emphasises how various actors and policies could potentially influence urban sustainability transition processes and reposition Ghana's urban areas onto a trajectory of sustainability. Ultimately, the conceptual model presented in Figure 7.7 demonstrates how the adaptive responses of various actors are important for creating the necessary conditions that will enhance the capacity of Ghana's urban areas to transition to sustainability. Rather than

propose responses top-down or bottom-up, it presents balanced and multiple practical starting points (for example, citizens, private sector, government) for inducing a transition to sustainability. Consequently, the relevant findings were sorted into the respective components of Figure 7.7 and are summarised as follows:

(a) Drivers.

The main drivers which shape the state and trajectory of urban sustainability in the study have been identified as urbanisation and sustainable development (environmental, social, economic) processes.

(b) Pressures.

The pressures identified encompass those that result from anthropogenic activities in the local urban areas, as well as at the regional level that could potentially exacerbate conditions in local urban systems, and these include: globalisation / neo-liberalism; poverty; and resource use / depletion.

(c) State.

As an indication of the system-wide influence of the identified pressures on urban areas, the following 'State' variables were identified: landscape; demography; and human well-being.

(d) Impacts.

The effects of the pressures on the urban system states have created localised impacts on aspects of the urban systems, and these have been identified to include: land cover; urban form; population; and living standards.

(e) Responses.

As indicators of responses to mitigate the impacts of urbanisation and sustainable developments, and to manage these processes in ways which facilitate a transition to a trajectory of sustainability in Ghanaian urban areas, the following relevant 'responses' were identified: government (institutional); citizens and communities; and corporate or private sector. Specifically, in order to enhance urban sustainability, the government must: pursue policy and institutional reform for sustainability; make policy prescriptions for poverty reduction and sustainable resource use; pursue integrated sustainability planning, decision-making and investment; enhance planning enforcement regimes; and pursue overall good

urban governance. The corporate / private sector must: pursue good corporate governance which ensures that their activities promote environment stewardship and pursue innovative approaches to urban development which deliver sustainable urban forms. Citizens and communities must exude positive behaviour and attitude towards urban sustainability. Overall, the synergistic effects of the responses will be maximised through coordinated partnerships among the various relevant stakeholders.

In terms of the inter-linkages between the components of the conceptual model, Figure 7.7 shows how urbanisation and the pursuit of sustainability generate 'pressures' (for example, poverty and resource depletion) which alter the 'state' of aspects of urban systems (such as landscape and demography). The 'pressures' and 'state' components create 'impacts' that affect aspects of urban systems (including land cover, living standards and population). The inter-linkages among the 'drivers', 'pressures', 'states' and 'impacts' trigger 'responses' that aim to mitigate cumulative effects of these inter-linkages. Adaptive 'responses' from the Government (for example, through integrated sustainability planning), Private Sector (for example, through sustainable urban form designs) and, Communities and Citizens (for example, through change in attitudes) are expected to facilitate a transition to sustainability by mitigating the 'impacts' of the interactions between urbanisation and sustainable development. Adaptive 'responses' from the Government (for example, through integrated sustainability planning, and policy prescriptions for poverty reduction and resource use) are expected to improve the 'state' of aspects of urban systems (for example, demography, landscape and human well-being) and to minimise the effects of 'pressures' on urban systems. Adaptive 'responses' from the Government (for example, through policy and institutional reform for sustainability) and Private Sector (for example, through good corporate governance) are expected to promote sustainable development and sustainable urbanisation. The adaptive 'responses' are expected to regulate the interactions between urbanisation and sustainable development, and thereby contribute to a transition to urban sustainability.

Critically, Figure 7.7 potentially shows that the DPSIR framework could be useful for promoting the transition to sustainability. For example, within this thesis, the framework has facilitated problem identification by helping to highlight and frame problems that emanate from the interactions between urbanisation and sustainable development. This is evidenced by how 'drivers' (for example, urbanisation and economic development) and 'pressures' (for

example, poverty and globalisation) were identified to affect 'state' (for example, landscape) and 'impact' (for example, land cover and living standards) aspects of urban systems. Therefore, the framework could prove useful in the practical realm of urban development where issues that affect the complex interactions between urbanisation and sustainable development have to be identified and structured for decision-making. The DPSIR framework has also facilitated an appreciation of the interactions across the different parts of the urban socio-ecological system. For example, Figure 7.7 shows how 'pressure' factors alter 'state' factors, and how these factors are, in turn, minimised or mitigated by 'response' factors. In the arena of urban development practice, acknowledging and understanding interactions between the multiple aspects of an urban system could be essential for achieving positive sustainable outcomes. Furthermore, the DPSIR framework has helped to generate possible solutions that could ameliorate the problems and impacts emerging from the interactions between urbanisation and sustainable development. The 'response' factors (for example, citizen and government) in Figure 7.7 are evidence of the multiple pathways through which urban sustainability transition could be achieved. Stimulating a transition to urban sustainability in real life possibly requires designing a range of strategies that provide pathways and solutions for meeting urban sustainability goals. Overall, Figure 7.7 shows that the DPSIR framework could be operationalised to promote the transition to urban sustainability if it is appropriately modified or improved.



Figure 7.7: Adaptive DPSIR-based conceptual model for urban sustainability transition in Ghana.

Source: Author's analysis (as modified from OECD 1994; EEA 1995; and IEHIAS 2019).

Chapter Eight

8. Conclusions

8.1 Introduction

The purpose of this study was to explore the relationship between urbanisation and sustainable development in the context of Ghana in order to make recommendations for urban sustainability, and to develop a conceptual model that could facilitate and enhance evidence-based decision-making for the transition to sustainability. Thus, in the next section, a summary of the salient issues identified in the literature review on urbanisation and sustainable development is provided, and conclusions are made in the context of the six specific objectives of this study. These were:

(1) To apply a conceptual framework that organises the interactions between urbanisation and sustainable development in an integrative and holistic manner.

(2) To evaluate the urbanisation trends and the sustainable development situation in Ghana.

(3) To provide a deeper understanding of the Ghanaian situation through a comparative urban sub-case-study (Kumasi and Obuasi) that specifically: (i) examines the patterns of landscape and demographic urbanisation and the implications for sustainable development; (ii) evaluates the sustainable development situation based on recognised indicators from the literature; and (iii) explores urbanisation and sustainable development based on the subjective experience (perceptions) of citizens.

(4) To examine the sustainability construct and operationalisation in Ghana's urban policies in order to determine their consistency with the principles of sustainable development.

(5) To examine sustainable city development in Ghana through a sub-case-study (Appolonia satellite city) in order to establish the extent to which it aligns with sustainability as a concept.

(6) To suggest recommendations for urban sustainability transition and to develop a conceptual model that could facilitate and enhance evidence-based decision-making for the transition to urban sustainability in Ghana, and by extension, SSA.

The third section of this chapter presents the policy implications and recommendations. The fourth sector reconsiders the study's findings in the context of the concepts of sustainability transition and sustainable urban transformation. The fifth section is an evaluation of the validity of the study. The sixth section of this chapter presents the limitations and pointers for further research. The seventh section is a summary that reinforces the key contribution to knowledge in the context of the overall study.

8.2 Summary of critical issues and main conclusions

8.2.1 The extent to which knowledge of urbanisation and sustainable development can be organised within a DPSIR conceptual research framework model.

Applying the DPSIR framework (Figure 2.4) to organise knowledge in this study addressed the first objective of this research and showed that: the concepts of urbanisation and sustainable development constitute driving forces which shape the functioning of urban areas; the interactions between urbanisation and sustainable development generate 'pressures' including poverty, globalisation, and resource depletion which alter the 'state' of urban systems and affect variables including demography, landscape, and human well-being (see Chapter Two); the 'impact' of the alteration of urban systems include changes to land cover, urban form, population, and living standards (see Chapters Four and Five); and 'responses' from relevant stakeholders are required in order to mitigate the 'pressures', improve the state of urban systems, and mitigate impacts of changes to land cover, urban form, population, and living standards (see Chapters Four, Seven and Eight). An important caveat for evaluating the usefulness of the DPSIR framework applied, however, is that the variables identified are not exhaustive representations of all possible 'pressures', 'impacts', 'states' and 'responses' on the urbanisation and sustainable development processes in Ghana, but are exclusively based on the literature reviewed and data available for this study. Notwithstanding, the lesson of the application of the DPSIR framework in this study is that examining the interactions of disparate concepts such as sustainable development and urbanisation stimulates a rethinking of the assumptions and methods that guide the valid construction of knowledge, as evidenced

by the pragmatic philosophy adopted for this research. Overall, in light of the complexity of the interactions between urbanisation and sustainable development, as well as a lack of a comprehensive framework for evaluation (Jabareen 2008), the application of a DPSIR conceptual framework addressed the first objective of the research as it enabled a holistic and integrative analysis of the concepts that were examined.

8.2.2 An overview of urbanisation and sustainable development in the Ghanaian context.

Chapter Four addressed the second objective of this study by presenting an assessment of urbanisation patterns and the overall sustainable development in Ghana based on four primary dimensions identified from the Brundtland Commission's definition of sustainable development (WCED 1987; Holden et al. 2014) which include: (i) the extent to which the basic needs of citizens are being met; (ii) how ecological sustainability is being safeguarded; and (iii-iv) how both inter- and intra- generational equity are being maintained. The findings highlighted the unsustainable nature of development in Ghana as evidenced by an ecological footprint that has exceeded bio-capacity; increasing Gini co-efficient scores (as a measure of widening inequality); increasing HDI scores (as a composite measure of the extent of progress on attaining the basic needs of education, health, and income); and a decreasing renewable energy component of total primary energy. As sustainable development advocates for a form of development that ensures a balance in environment, economic, and social considerations, the sustainable development space graphs produced highlighted how Ghana was failing to meet threshold conditions beyond or within which development could proceed in a sustainable manner. The findings demonstrate that focusing only on achieving socioeconomic development will not be conducive to protecting ecological integrity and ensuring inter- and intra-generational justice. The fundamental argument that emanated from these findings, thus, is that although Ghana has made significant strides towards promoting overall development, it is imperative that it pursues development in a sustainable, equitable, and inclusive manner; and more importantly, within the planetary boundary. Consequently, the political nature of SD as a concept means that the extent to which SD is pursued in the Ghanaian case will be determined by the importance the Government attaches to its achievement. This means that despite Ghana's developmental challenges, political choices could still be made that affect overall sustainable development. These choices include, but are not limited to, ensuring equality of opportunity irrespective of gender, generation (current or present), socio-economic status, etc., such that citizens are not left behind by economic

progress, social mobility, and environmental conditions. Overall, this part of the research represents a first attempt to frame a holistic perspective of SD in Ghana based on the primary dimensions of sustainable development, as advanced within the Brundtland Commission's definition of sustainable development.

This study also provided an overview of the overall urbanisation patterns in Ghana. Broadly, the findings showed that Ghana's urbanisation has reached unprecedented levels, with the proportion of the country's population living in urban areas increasing from 15% in 1950, to 51% in 2010 (Cobinnah and Niminga-Beka 2017). The key driving factors shaping the urbanisation process in Ghana were identified to include internal ones such as demographic shifts and market forces, and external ones such as globalisation. By juxtaposing the findings on Ghana's sustainable development and urbanisation in a discussion, Chapter Four highlighted the sustainability implications of urbanisation in Ghana and demonstrated the complexity of the interactions between urbanisation and sustainable development at the national level. For instance, market forces have increasingly shaped Ghana's urbanisation process since the implementation of the structural adjustment programmes (SAPs) recommended for Ghana's economic transformation in the 1980s, by facilitating the provision of urban infrastructure and services. Nevertheless, the characteristics of market forces tend to reinforce inequality and unemployment, and exacerbate environmental impacts, thereby undermining the positive sustainability impacts of urbanisation in Ghana. Essentially, the findings in Chapter Four support arguments about how rapid urbanisation inflicts a multitude of diverse challenges on the achievement of sustainable development (Allen 2009; Zhou et al. 2015). Consequently, it is argued that rapid urbanisation in Ghana complicates the achievement of sustainable development due to a constellation of factors, including existing low levels of development, the effects of market forces, and the desire to safeguard the basic needs of an increasingly urban population.

8.2.3 The extent and depth of interactions between urbanisation and sustainable development at the local urban level in Ghana.

Chapter Four framed a national perspective on urbanisation and sustainable development in Ghana. As the impacts of the interactions of sustainable development and urbanisation manifest at the local urban level, it was necessary therefore, to deepen this national assessment and explore the interactions between urbanisation and sustainable development in the local urban areas in Ghana. Chapter Five narrowed the focus to the local urban level and addressed the third objective of this research by developing comparative a sub-case-study on two urban areas in Ghana, namely Kumasi and Obuasi. The first investigation focused on the sustainability implications of demographic and landscape urbanisation in the two study sites. By classifying satellite imagery, landscape urbanisation in the two study sites was assessed through a spatio-temporal classification of land cover over the period 1986-2018. The findings of the study highlighted how the urbanisation process in Kumasi and Obuasi has stimulated the expansion of built-up (urban) land cover mainly at the expense of natural land cover. Specifically concerning urban expansion, the findings presented have shown that between 1986 and 2018, the built-up share of Kumasi's land cover expanded by over 1,105%, compared with Obuasi where built-up land cover expanded by over 144%. While the different rates of urban expansion between Kumasi and Obuasi could be attributed to differences in city size and administrative hierarchy, the findings reinforced the national level observation about how market forces shaped changes in land cover and the subsequent urban form. In a country such as Ghana where planning enforcement is weak, the expansion of built-up land cover due to the forces of urbanisation is noticeable and significant. For example, the thematic maps produced showed built-up land cover in Kumasi had expanded and transcended the city's administrative boundaries into adjoining districts, changing its urban form in the process. Critically, the findings of the study, thus, demonstrate how humaninduced development in Obuasi and Kumasi can aggravate changes in land cover and result in urban forms that undermine sustainability. In the broader context, the study demonstrated how visualising the extent of urban expansion at the expense of natural land cover can be a relevant input for analysing the sustainability implications of landscape urbanisation as they were identified to include aspects of development such as food security and urban form.

In terms of demographic urbanisation, the study provided an overview of changes in the populations of Obuasi and Kumasi over the period of 1986-2019. Consistent with national level demographic urbanisation, the findings highlighted a significant shift in the population in both study areas between 1986 and 2019, as Kumasi's population quadrupled while Obuasi's population tripled. These changes were fundamentally shaped by factors identified in Chapters Two and Five including migration, market-forces, and natural population growth; and consistent with the literature (see, for example, Ferguson 1999; David and Arman 2014), the urbanisation process in the two areas has also been shaped by the prevailing economy. In sum, the findings provided a nuanced understanding of urbanisation in Kumasi and Obuasi as

they demonstrated that the process does not merely result in changes in landscape and population, but it is multi-dimensional and involves changes in population, landscape, development, and society.

The second investigation in Chapter Five was an assessment of the sustainability implications of urbanisation in Kumasi and Obuasi based on selected socio-economic indicators. The study assessed the performance of the study areas based on indicators related to education, life expectancy, and living standards. In general, the findings of the study lent credence to the argument about how the agglomeration effects of urban areas enhance their levels of socio-economic development (Gong et al. 2012; Turok and McGranahan 2013; Earle 2016). The study showed how Kumasi and Obuasi, by virtue of their 'urban state', offered better opportunities for improved living standards compared with overall national performance (based on the indices assessed). Furthermore, considering the challenges of poverty and the generally low level of development in Ghana, the study argued that the achievement of sustainable development in Obuasi and Kumasi could be undermined as the rapid population growth in these areas could erode the quality of the development opportunities they offer. This observation is significant, especially when urban population growth is deemed to have outpaced conventional planning in Ghana.

A notable contribution of this study is that it provides insights into the interactions between urbanisation and sustainable development based on the subjective experience of citizens. Consistent with the case study approach adopted, and acknowledging the paucity of data on urban areas (Smit and Parnell 2012), it became necessary, therefore, to corroborate evidence from multiple sources. Thus, the third investigation in Chapter Five examined the central themes of this study by exploring the perceptions of citizens in Kumasi and Obuasi. The findings from the survey suggest that how citizens perceive statements about urban sustainability in a particular urban area depends, to a considerable extent, on how that urban area has enabled them to maximise their potential in terms of access to socio-economic opportunities and services such as jobs, housing, education, income, as well as access to an environment with minimal negative impacts. For example, 46% of citizens in Kumasi agreed or strongly agreed to the statement on income levels, compared with 13.6% in Obuasi. In contrast, however, 84.5% of respondents in Kumasi disagreed or strongly disagreed on the statement suggesting that it is easy to find a job, compared with 88.7% in Obuasi.

balanced level of environmental, social, and economic considerations. This makes it easier to assume that the perceptions of citizens on the sustainability statements will be ambiguous. However, the findings showed a fundamental dominance of socio-economic considerations, sometimes at the expense of environmental considerations. This was evidently captured in some of the statements provided by respondents (see Appendix 3) in the 'further comments' section of the questionnaire. These observations align with earlier discussions in the first investigation about the influence of the prevailing economy, city size and administrative forces in shaping the quality of landscape and demographic urbanisation in Kumasi and Obuasi. The findings have, thus, provided valuable insight into understanding the experiential perspectives of respondents on urban sustainability-related themes in Kumasi and Obuasi. Furthermore, the findings showed how males and females, age groups and socio-economic groups responded differently to some urban sustainability statements. For example, in Kumasi, there was a statistically significant difference between how females responded to the statement on access to healthcare, as well as statistically significant differences among the responses by those who are 'Employed with wage', 'Self-employed' and 'Unemployed'. Therefore, the findings support arguments in the literature about how the sustainable development experience of citizens is shaped by issues such as gender roles, age, and socioeconomic characteristics (see, for example, Tuncer 2008; Olsson and Gericke 2016; Aina et al. 2019). Therefore, to align with the egalitarian nature of sustainable development as a concept, those factors must be taken into account during sustainability decision-making.

Citizens' acceptance and participation in sustainability-related activities are important for achieving wider sustainability in urban areas. According to the findings, respondents in Kumasi and Obuasi strongly support pro-sustainability behaviour as over 90% of them were willing to participate in recycling, patronise public transport, and conserve water and energy. Thus, the findings of the study are an indication of the potential for citizens to adopt pro-sustainability values, attitudes, and practices which can enhance the pursuance of wider urban sustainability. Furthermore, the findings of the survey reinforce the role of citizens as a key component of the sustainability agenda in Ghanaian urban areas. Essentially, the findings showed that the interactions between urbanisation and sustainable development in Kumasi and Obuasi were not sustainable when the trade-offs between positive impacts (for example, improved living standards) and negative impacts (for example, uncontrolled urban expansion) were considered. Overall, by developing the comparative sub-case-study, Chapter Five provided deeper insights and enhanced the limited knowledge on the interactions between

urbanisation and sustainable development at the local urban level in Ghana, and by extension, SSA.

<u>8.2.4 How does policy interpret and apply the principles of sustainability in order to manage</u> contemporary and future urbanisation towards sustainability?

Chapter Six addressed the fourth objective of this research by examining the sustainability construct and operationalisation in Ghana's National Urban Policy and Action Plan. By assessing the content of the NUP and AP using an evaluation matrix based on selected dimensions of the UN-Habitat's City Prosperity Index (CPI), the study found that the NUP and AP provide adequate scope for the sustainability of urban areas in the context of the CPI dimensions. By using the World Bank's urban sustainability framework as a lens of analysis, the study discussed the sustainability context of the NUP and AP under themes that included urban economies, fiscal sustainability, inclusivity and quality of life, natural environment and resources, climate action and resilience, and governance and integrated planning. Demonstrating the gaps in the sustainability construct of Ghana's NUP provided a basis for identifying challenges to its sustainability operationalisation which include, but are not limited to, rhetoric tendencies, political barriers, and an unbalanced consideration of the various dimensions of sustainability. Essentially, the findings suggest that in its current form, the alignment of the NUP to sustainability principles is limited. Overall, by assessing the NUP and AP, this study framed a perspective on the extent to which sustainability principles are embedded in policy-making for urban sustainability in Ghana. It was argued that realising the multiple objectives of the NUP would require the urgent integration of the various dimensions of sustainability in a non-discretionary manner, such that power relations are well-managed and the policy objectives are coordinated with other independent government policies (for example, housing policy) to achieve wider sustainability.

8.2.5 To what extent can innovative approaches to sustainable city development be adapted to shape sustainable urbanisation within the African context?

An argument that emanated from the review of literature in Chapter Two was how the utopian ideals of the innovative approaches to sustainable city development (for example, eco-cities and smart cities) potentially undermine the development of sustainable cities in SSA. This is because they have been mainly implemented in wealthier countries, and their

designs and characteristics tend to overlook the exceptional circumstances that undermine sustainable development in SSA including, but not limited to, low development levels, lack of financial resources to support infrastructural development, weak institutions, and weak enforcement of planning regulations (Bryceson et al. 2009; Turok and Parnell 2009; Bekker and Therborn 2012; Fox 2014). Therefore, the first part of Chapter Seven addressed the fifth objective of this research by examining the development of sustainable cities in Ghana through an evaluation of the Appolonia City project. The sub-case-study on the Appolonia City project provided a synthesis of sustainable city development in Ghana as Appolonia City's (un)sustainability was assessed by juxtaposing it in an evaluation matrix with the sustainable urban form features of an eco-city proposed by Jabareen (2006). The findings showed that Appolonia City's characteristics contained features that could enhance wider urban sustainability, compared with 'business as usual' urban development in Ghana. For example, the project promoted the use of sustainable energy forms including solar energy, promoted urban greenery, and promoted mixed land use. These desirable features of the Appolonia City project provide diverse sustainability benefits which are explained in Table 2.1. Conversely, certain features of the Appolonia City project were deemed to potentially undermine its sustainability benefits. For example, despite proposing to host up to 100,000 residents, the project characteristics did not highlight any plans for a sustainable form of transport. This suggests that the project inherently promotes private car dependency. The dominance of private car use is expected to contribute to increased GHGs which erodes the sustainability benefits provided by a desirable feature such as urban greenery (see Table 2.1). Critically, the study argued that the Appolonia City project did not fully align with the normative visions of a sustainable city promoted in Chapter Two. Following that, the study discussed the main barriers of sustainable city development in Ghana to include finance and politics, as the implementation of sustainable city concepts requires substantial financial resources and political will. Overall, the findings on the Appolonia City project facilitated a deeper understanding of how the essential characteristics of contemporary sustainable city development in Ghana contributes to or undermines wider urban sustainability.

<u>8.2.6 Towards a trajectory of sustainability in Ghanaian urban areas: recommendations and a conceptual model to facilitate evidence-based decision-making.</u>

By juxtaposing and consolidating the findings that exposed urban challenges in Ghana in Chapter Four; the findings on overall sustainable development and urbanisation in Ghana in Chapter Four; and the findings on urbanisation and sustainable development at the local urban level through the comparative sub-case-study in Chapter Five, this study demonstrated that, in its current form, development in Ghanaian urban areas is not proceeding in a sustainable manner. Subsequently, while not dismissing the transformative strengths of innovative approaches to sustainable city development such as eco-cities and smart cities, the findings on the (un)sustainability of the Appolonia City project highlighted the difficulties in applying these concepts within the SSA context. Against this backdrop, this study argues that a transition to a trajectory of sustainability is required in order to enhance sustainability in Ghana's urban areas.

In making recommendations for a transition to a trajectory of urban sustainability, the study argues that there is no 'one-size fits all' formula for stimulating a transition. The different economic, political, and geographic situations in Ghana's urban areas imply that rather than prescribe a particular set of characteristics for developing sustainable cities, the relevant stakeholders should adopt an eclectic mix of solutions from the various approaches to sustainable city development and adapt them to the needs and characteristics of particular urban areas. Notwithstanding, based exclusively of the findings and observations in this research, the study discussed recommendations which could facilitate the transition to a trajectory of urban sustainability in Ghana's urban areas. The recommendations generally highlighted the need for: institutional and policy overhaul for sustainability outcomes; citizen and community action and behaviour for sustainability; and corporate or private governance practices that promote a healthy balance between investment returns and sustainable development.

A key highlight from this study's findings is that the transition to sustainability in Ghana's urban areas is in the nascent stage, if it exists at all. Therefore, Chapter Seven further addressed the sixth objective of this research by adapting the study findings and modifying the DPSIR framework (Figure 2.4) in order to develop a conceptual model (Figure 7.7) that promotes evidence-based decision-making in the context of facilitating the transition to a trajectory of sustainability in Ghana's urban areas. Essentially, the conceptual model developed in Figure 7.7 provided a better understanding of the interplay between urbanisation and sustainable development in Ghana in a holistic manner, and its adaptive responses serve as a predictor of the potential ability of the dynamic constellation of diverse stakeholders to shape the urban development process in ways that could stimulate a transition to a trajectory

of sustainability in Ghanaian urban areas. The model is consistent with the principles of SD and contributes to the important dialogue about the future of urban areas in SSA. Overall, as a first step, the study has provided a more reliable integration of disparate data on urbanisation and sustainable development in Ghana, and shows promise as a foundation based on which evidence-based decision-making on urban sustainability in Ghana, and by extension, SSA, could proceed.

8.3 Policy implications and recommendations

The findings of this research present many implications. The evidence of this study highlights the importance of the interactions between urbanisation and sustainable development for the overall sustainability of urban areas in Ghana and, by extension, SSA.

(1) The evidence of the study in Chapter Four shows that the overall development situation in Ghana is not sustainable, especially in terms of balancing resource use with the need to safeguard the basic needs of citizens, as well as ensuring inter- and intra- generational justice. These findings have implications for overall development policy as they suggest that sustainable development principles are undermined. Therefore, if development in Ghana is to be consistent with the principle of sustainable development, policy-makers must ensure that meeting one goal does not compromise the possibility of meeting another goal. In this context, legal and resource provisions should be made to ensure that a policy set such as the 7-year Coordinated Programme of Economic and Social Policies (2018–2024) equally safeguards the environment while delivering its objective of creating jobs, prosperity, and equal opportunities for Ghanaian citizens.

(2) Considering the evidence on landscape and demographic urbanisation at the local urban level in Chapter Five, the findings are helpful for policy-makers to better understand the complexity of the relationships among land use and land cover change, demographic shifts, and sustainable development. The findings on land cover patterns are potentially useful for providing guidance on better management of future land use as they help to map the extent to which different land cover classes are changing at temporal and spatial scales. As the extent of land cover change has implications for sustainability in Ghana's urban areas, it is recommended that emerging landscape management approaches are adopted to guide future land use planning. Furthermore, to remedy the poorly managed expansion of Ghana's urban

areas, relevant urban development institutions such as the National Development Planning Commission (national level) and the Town and Country Planning Department (local level) should be well resourced and granted the appropriate powers to enforce development controls and regulations without interferences.

(3) Considering the evidence on the perspectives of urban citizens in Chapter Five, the findings have policy implications in the context of citizen engagement for sustainable development. The evidence of the study is an indication of the subjective experiences of citizens, and the findings suggest that the perspectives of citizens on urban sustainability are dominated by socio-economic considerations, sometimes, at the expense of environmental ones. The findings also showed differences in the perspectives of citizens based on gender, age and economic factors. This makes it imperative to sufficiently engage urban citizens and incorporate their values and preferences in sustainability policy-making and planning. The evidence of the study relating to respondents' positive attitudes towards pro-sustainability practices in Kumasi and Obuasi also have implications for the role of citizens in steering Ghana's urban areas onto trajectories of sustainability. This requires setting the necessary preconditions for citizens to actively engage in pro-sustainability behaviour. The necessary preconditions include, but are not limited to, providing basic infrastructure such as kerbside recycling systems and public transport systems (such as BRT).

(4) The findings have policy implications for sustainable city development and the transition to sustainability in Ghana's urban areas and beyond. The poverty and institutional capacity situation in Ghana suggest that it might be unreasonable to expect the implementation of a particular approach to sustainable development to explicitly and successfully address the various challenges (see Chapters Four, Five, and Seven) that threaten the sustainability of Ghanaian urban areas. This will require a more coordinated approach to development where there is full alignment of the national urban policy to the principles of sustainable development, and where partnerships among government institutions, citizens, and the private sector in creating and implementing visions for a transition to sustainability are reinforced. For example, by taking Ghana's housing deficit (estimated at over 1.7 million units) and the focus on slum upgrade in the NUP into consideration, it is evident in this thesis that the interactions between market forces and the complex land tenure system affect the provision of socially inclusive urban housing. Therefore, policy provisions must target partnerships and collaborations between stakeholders such as the Ministry of Local

Government and Rural Development (MGLRD), the Ministry of Lands and Natural Resources (MLNR), Lands Commission, municipal authorities, utility service providers (e.g. Electricity Company of Ghana), traditional leaders, and local communities, in order to accelerate urban housing delivery by identifying and developing legally acquired land. Such partnerships should ensure that the housing needs of low-income citizens, as well as the socially, economically, and politically hegemonic citizens, are equally and adequately catered for.

(5) The findings of this study have policy implications for evidence-based urban sustainability decision-making in Ghana. The literature reviewed and the breadth of the DPSIR framework applied to organise knowledge in this study support an integrative analysis of the different dimensions of sustainability, and highlighted the sheer volume of indicators required to measure components of urban systems for decision-making. Considering the paucity of urban data in SSA (Smit and Parnell 2012), systematic efforts are required to enhance data availability and access in order to promote evidence-based decision-making for the urban sustainability transition in Ghana.

8.4 Re-assessment of sustainability transitions and sustainable urban transformation

The literature reviewed in Chapter Two highlighted the need for radical and structural shifts within urban systems in order to achieve sustainable urban societies (Rotmans and Loorbach 2009; Markard et al. 2012). However, it was argued that the sustainability transition concepts and theories (for example, transition management, strategic niche management) that have emerged to guide sustainable urban transformation have focused mainly on specific sectors of urban systems such as energy and health, rather than holistically capture transitions at all levels of urban systems (Coenen et al. 2012). A key point of departure from Chapter Two was the need to consider sustainability transition and transformation from an overall urban spatial context. On the basis of the findings in this study, it can be argued that the transition concepts and theories are helpful for understanding the particularities of the adaptive responses for a transition towards a trajectory of urban sustainability in Ghana. For example, from a transition management perspective, Figure 7.7 highlighted the need for a more coordinated partnership and interaction among various actors (for example, government, citizens, and private sector) in order to facilitate a transition to urban sustainability. From a multi-level perspective, the NUP, for example, represents a redirection of the national

strategy towards sustainable urbanisation at the 'landscape' level, while a change in attitude and behaviour at the local level by citizens represents a 'niche' which could potentially challenge and stimulate change in existing 'regimes' (for example, positive behaviour towards pro-sustainability initiatives). Furthermore, from a strategic niche management perspective, the Appolonia City project, for example, represents a 'niche' that has been deliberately created and supported (based on visions of a sustainable city) in order to induce change in 'business as usual' urban development 'regimes' in Ghana towards more sustainable outcomes. However, as the findings in this study have shown, development in Ghana's urban areas is proceeding in an unsustainable manner. The sustainability of development in Ghana's urban areas is further undermined by structural and institutional challenges such as poverty, weak institutions and out-dated regulations (Bryceson et al. 2009; Turok and Parnell 2009; Bekker and Therborn 2012; Fox 2014). As a result, while a transition to urban sustainability may be observed over a long period of time, it can be argued that urban sustainability transition in the SSA context is non-existent or, at best, nascent. Furthermore, the transition concepts and theories have been mainly generated and applied in Europe where the preconditions for a sustainability transition may generally exist (see, for example, Loorbach and Rotmans 2010). This raises a question over how far these theories and concepts are transferable to the SSA context where the preconditions for sustainability transitions do not exist. Therefore, it is argued that the extent to which the concepts and theories of sustainability transition 'hold' in the SSA context is limited.

8.5 An assessment of the validity of this research

Considering the lack of a theoretical framework to study the interactions between the concepts of urbanisation and sustainable development due to their disparate nature, the pragmatic philosophy adopted in this study allowed the adoption of a DPSIR framework to guide this research. Considering the paucity of data on urban areas in SSA (Smit and Parnell 2012), a strength of this study is its deployment of mixed methods and a multiple case study approach in order to collect and organise data on the interactions between urbanisation and sustainable development from multiple sources. The initial stage of this research involved an in-depth review of the extant literature on urbanisation and sustainable development in order to identify gaps and fortify the theoretical foundations of this research. The initial stage of the research was essential as it resulted in a pilot study which laid the foundations for the main stage of the research, which represents a detailed attempt to examine the interactions between

urbanisation and sustainable development in SSA based primarily on evidence available and accessible from Ghana. In doing so, the research has framed a perspective on the sustainable development and urbanisation situation in Ghana. The sub-case-studies developed provided a deeper understanding and subtle insights into the interactions between urbanisation and sustainable development at the local urban level. Critically, the case study approach made it possible to provide evidence on processes that were not directly observable, such as the perspectives of citizens. In essence, the mixed methods and multiple case study approach adopted enabled the study to fully explore the predefined categories of analysis based on the research objectives.

The comparative approach adopted for the sub-case-study at the local urban level is a key attribute of this research as the complexity of the relationship between urbanisation and sustainable development means that the majority of research on those processes has been based on the analysis of single study sites. Concerning study site selection, Kumasi could be classified as a large-sized urban area as it has a population of over 2 million, while Obuasi, with a population of just over 200,000, could be classified as a medium-sized urban area. The selection of Kumasi and Obuasi as study sites was important as it helped broaden the research on urbanisation and sustainable development whose focus has mainly been skewed towards large-sized urban areas. The case study sites were characteristic of the broader SSA region where the economies of countries are dependent on natural resource extraction, as they represent a mix of a resource-based city and a big city. Essentially, the comparative approach adopted facilitated an understanding of the complexity, diversity, and differences in the manifestation of the interactions between urbanisation and sustainable development at the local level which otherwise would not have been possible.

8.6 Limitations of the study and suggestions for further research

Despite the policy implications of this study, a number of limitations are identified. Like many less-developed regions, research in SSA is undermined by financial constraints which result in a paucity of published data. Thus, reinforcing the notion on which this study was premised, which suggests that there is a paucity of data on urban areas in SSA (Smit and Parnell 2012), significant gaps in available data on urbanisation and sustainable development were encountered, especially at the urban level where the tangible impacts of interactions between urbanisation and SD could be most practically captured. In general, while the

findings based on Ghana are in the context of SSA, structural and economic conditions potentially differ from country to country, thus limiting the extent to which the findings could be generalised in the SSA context. Further research could, therefore, offer deeper insights on the complex interactions between urbanisation and sustainable development by developing comparative studies on multiple countries and urban areas within SSA. Consequently, the identified limitations and suggestions for further research, as specifically related to the study objectives, include the following:

(1) The first set of limitations identified concern the conceptual model proposed (Figure 7.7). As shown in the literature reviewed, conceptual models are shaped by particular concepts and theories and are developed based on long-term and inter-generational considerations (Todorov and Marinova 2011). This was evident in the DPSIR conceptual framework (Figure 2.4) applied in this study as its structure assumed a seamless transition between 'pressures', 'states' and 'impacts' on the one hand, and responses on the other hand. This provided no consideration for uncertainty. Despite the limitations posed by the choice of conceptual framework proposed, it was argued in Chapter Two that there is no best approach to modelling sustainability, but rather, a variety of approaches exist which provide varied but complementary insights into sustainability. Therefore, the core principle in sustainability modelling is that due to the various interpretations and the complexity of urban sustainability, as well as the number of approaches available, the choice of approach or model is effectively determined by what the user wants to achieve with the model (Kupiszewska 1997; Ali-Tourdert and Ji 2017). Critically, the limitations posed by the DPSIR-based conceptual model (Figure 7.7) proposed were mitigated by the manner in which the DPSIR framework (Figure 2.4) was applied in this research. The DPSIR conceptual framework (Figure 2.4) was applied in order to provide an organisational structure based on which the complex interactions between urbanisation and sustainable development in SSA could be holistically captured, and in so doing help to bridge the knowledge gap created by the paucity of data on urban areas in SSA (see Smit and Parnell 2012). The conclusions in the preceding paragraphs have shown that, to a considerable extent, the conceptual framework has been successful in facilitating a reliable integration of variable data on urbanisation and sustainable development in Ghana. Overall, while based exclusively on the findings in this study, the conceptual model proposed in Figure 7.7 has also provided a foundation based on which evidence-based decision-making for a transition to urban sustainability could proceed. Further research could, therefore, offer improvements by developing other forms of models and incorporating other sources of data in order to deepen understanding of the interactions between urbanisation and sustainable development.

(2) In assessing the sustainable development situation in Ghana in Chapter Four, this study followed the methods proposed by Holden et al. (2014). Among the key issues which they highlighted, concerning the indicators used, was measuring ecological sustainability using the ecological footprint. They observed that the ecological footprint captured consumptions and emissions based on land areas. This means that other non-land-based issues such as emissions related to radioactivity and heavy metals are not captured, which represents a methodological weakness. Furthermore, they observed that the ecological footprint relied on GHG emissions and total energy use which could make them alternative indicators. However, they argue that a strong point for adopting the ecological footprint is because it covers more environmental issues than is possible with GHG emissions and total energy use. Consequently, considering the paucity of data on sustainable development in Ghana, the indicators and thresholds used are based on high quality sources which are robust and relevant enough to mitigate these potential methodological weaknesses. It will, therefore, be informative if future research could consider applying other relevant indicators to assess overall sustainable development in Ghana.

(3) The next set of limitations relate to the comparative sub-case-study developed in Chapter Five. First, concerning the assessment of landscape urbanisation, the extent of land cover detail was limited as a maximum of seven classes were mapped. For instance, apart from primary forest, all other forest types were classified as mixed woody vegetation. Nevertheless, the primary agenda in this research was to highlight the extent of built-up (urban) land cover in relation to other land cover types. This was achieved. Future studies could provide a more detailed and comprehensive analysis to include the other land cover types that were not captured in detail in this study. Second, a limitation of the study relates to the indicators used for sustainability assessment in the second comparative investigation on Kumasi and Obuasi. Since human well-being is central to this thesis, the indicators selected were underpinned by the Human Development Index (HDI), as the development of indicators was beyond the scope of this study. Indicators such as Maternal Mortality Rate and Crude Death Rate were the proxies used to represent life expectancy. However, as discussed in the literature reviewed in Chapter Two, the indicator selection process should be based on the careful analysis of indicators which have fundamental significance and carry a maximum

likelihood of producing accurate information, rather than just gathering data on all indicators (Shen et al. 2011). Therefore, despite issues with indicator selection, the indicators chosen were relevant and robust as they were related to the HDI, MDGs and SDGs. Critically, they were selected because the paucity of data meant that they were the indicators which had reliable, measurable, policy-relevant, and accessible data available (Joumard and Gudmundsson 2010). Future research could expand the indicator set for sustainability assessment in order to achieve a deeper understanding of the urban-level impacts in the context of sustainability.

Finally, in relation to the citizen survey, the following observations are made. In exploring the perceptions of citizens, the questionnaire used in the survey was close-ended. A shortcoming of such close-ended questionnaires is that they potentially inhibit respondents from expressing extended views. However, to mitigate this, a 'further comments' space was provided at the end of the questionnaire to record any other views which the respondents wanted to express. The crux of the matter is that considering the sample size, a more in-depth method such as a semi-structured interview was not pragmatically achievable within the financial and time constraints of this research project. Nevertheless, considering the technical nature of the urbanisation and sustainability discourse, and the reality that the sample was being drawn from the general population, the close-ended questions were appropriate for ensuring uniformity in questions asked and answers recorded. In the future, studies could infuse semi-structured questions into questionnaires in order to mitigate the depth of information challenges that emerge from using only close-ended questionnaires. Also, as stressed throughout this thesis, sustainability as a concept is multi-faceted, and a significant number of variables representing the economic, social, and environmental dimensions of sustainability could be included in a questionnaire that studies it. Nevertheless, there was a deliberate focus on the most important characteristics determined during the initial pilot study. This was necessary due to sample size and data collection restrictions. Future research could expand on these and include other sustainability variables.

(4) The study adopted the UN-Habitat's City Prosperity Index and World Bank's Urban Sustainability Framework to analyse data and frame discussions on the sustainability construct and operationalisation of Ghana's urban-related policy documents. This means that the findings and discussions were restricted to the selected dimensions and indicators used in the aforementioned index and framework. Nevertheless, as already highlighted, it was shown in the literature review that the indicator selection process should be based on the careful analysis of indicators which have fundamental significance, as well as carry a maximum likelihood of producing accurate information, rather than just gathering data on all indicators (Shen et al. 2011). Therefore, despite the potential bias, the indices used are relevant, reliable, and internationally accepted.

8.7 What is the key contribution to knowledge of this research study?

A contribution to knowledge is provided in that the processes of urbanisation and sustainable development have been aligned and explored in depth. How these processes interact has been explored not just theoretically but empirically in the context of Ghana. In taking the study forward, a mixed methods approach has generated data which shows that urbanisation in Ghana is proceeding at a rapid pace. By following the methods of Holden et al. (2014), this research gives empirical evidence on the sustainable development situation in Ghana. The findings demonstrate that while Ghana is making progress towards sustainable development, the country did not meet the empirical thresholds adopted for which its development trajectory could be classified as sustainable. In doing so, this study makes an original contribution to scholarly literature on sustainable development in Ghana, as to the best of the researcher's knowledge, it is the first study to provide empirical evidence on sustainable development in Ghana based on the four primary dimensions extracted from the Brundtland Commission's definition of sustainable development, which include safeguarding basic needs, maintaining ecological integrity, and inter- and intra- generational equity. Crucially, the discussions on the sustainability implications of Ghana's urbanisation provide evidence of the inter-relationships between the two processes.

By developing a comparative sub-case-study on two urban areas in Ghana, this study contributes to the literature on the complexity and diversity of local impacts due to the interactions between urbanisation and sustainable development. The sub-case-study provided empirical evidence on: the intricacies of landscape and demographic urbanisation at the local urban level; the sustainability situation based on selected socio-economic indicators; and the experiential sustainability perspectives of citizens. In doing so, this study makes an original contribution to scholarly literature, as to the best of the researcher's knowledge, it is the first study to provide comparative empirical evidence on the subjective experience of the interactions between urbanisation and sustainable development based on two urban areas in

Ghana. Overall, the comparative sub-case-study demonstrates that the interactions between urbanisation and sustainable development cannot be captured holistically without sensitivity to local complexities, diversity and differences.

By examining the sustainability construct and operationalisation in Ghana's National Urban Policy and Action Plan, this study contributes to the literature on policy responses to the interactions between urbanisation and sustainable development in Ghana. Considering the reality that Ghana's NUP and AP were launched in 2012, there is limited evidence of studies that have been dedicated to assessing their performance. Therefore, this study makes an original contribution to scholarly literature, as to the best of the researcher's knowledge, it is the first study dedicated to the empirical evaluation of the sustainability construct and operationalisation in Ghana's National Urban Policy.

This study contributes to the literature on innovative approaches to sustainable city development which have been promoted to maximise the benefits and mitigate the pernicious impacts of the interactions between urbanisation and sustainable development in order to transform urban areas for sustainability. The study highlights barriers that may undermine the uptake of innovative approaches to sustainable city development in SSA. By examining the Appolonia City project, this study makes an original contribution to scholarly literature on sustainable city development in Ghana, as to the best of the researcher's knowledge, it is the first study that has empirically assessed the (un)sustainability of the Appolonia City project. Overall, this study helps to bridge the knowledge gap on barriers that undermine the uptake of innovative approaches to sustainable development in SSA.

By consolidating the findings of this study and modifying the DPSIR framework (Figure 2.4) to develop Figure 7.7, this study makes an original contribution to scholarly literature on urban sustainability transitions in Ghana, as to the best of the researcher's knowledge, it is the first study to develop a conceptual model with adaptive responses that could facilitate and enhance evidence-based decision-making for the transition to urban sustainability in Ghana and, by extension, SSA. The thesis highlights the importance of the socio-ecological DPSIR framework as a tool to simplify, understand and organise knowledge on complex and disparate concepts like urbanisation and sustainable development.

To conclude, this thesis critically adds substantially to the geographical understanding of the interactions between urbanisation and sustainable development in the SSA context, especially at the local urban level where the tangible impacts of these interactions manifest, and also from the perspective of a researcher who, as a Ghanaian national, has experienced the interactions between urbanisation and sustainable development in Ghana to a considerable extent.

References

Abdullahi, S., Pradhan, B., Mansor, S. and Shariff, A.R.M., 2015. GIS-based modeling for the spatial measurement and evaluation of mixed land use development for a compact city. *GIScience & Remote Sensing*, 52 (1), 18-39, DOI: 10.1080/15481603.2014.993854

Abudu, D., Echina, R.Z. and Andogah, G., 2018. Spatial assessment of urban sprawl in Arua Municipality, Uganda. *The Egyptian Journal of Remote Sensing and Space Science*, article in press. https://doi.org/10.1016/j.ejrs.2018.01.008

Abulibdeh, A., Al-Awadhi, T. and Al-Barwani, M., 2019. Comparative analysis of the driving forces and spatiotemporal patterns of urbanisation in Muscat, Doha, and Dubai. *Development in Practice*, 29 (5), 606-618. DOI: 10.1080/09614524.2019.1598335

Acheampong, K., 2009. Revisiting Free Compulsory Universal Basic Education (FCUBE) in Ghana. *Comparative Education*, 45(2), 175–195. doi:10.1080/03050060902920534

ActionAid Report, 2006). Gold Rush: The Impact of Gold Mining on Poor People in Obuasi in Ghana (London: ActionAid).

https://www.actionaid.org.uk/sites/default/files/doc_lib/gold_rush.pdf Accessed 24/04/2019.

Adams, S., 2009. Can foreign direct investment (FDI) help to promote growth in Africa? *African Journal of Business Management*, 3(5), 178-183.

Adarkwa, K.K., 2012. The changing face of Ghanaian towns. *Afr. Rev. Econ. Finance*, 4(1), 1-29.

Ades, A. and Glaeser, E., 1995. Trade and Circuses: Explaining Urban Giants. *Quarterly Journal of Economics*. 110, 195-227.

Adger, W. N., 2006. Vulnerability. Global Environmental Change, 16(3), 268–281.

Adoeye, D.O., 2016. Challenges of Urban Housing Quality: Insights and Experiences of

Akure, Nigeria. *Procedia - Social and Behavioral Sciences*, 216 (2016), 260 – 268. doi: 10.1016/j.sbspro.2015.12.036

Adu-Poko, I., Drummond, J., and Li, Z., 2012. Land-cover change monitoring in Obuasi, Ghana: an integration of earth observation, geoinformation systems and stochastic modelling. *Journal of Earth Science and Engineering*, 2 (5). 1-14.

Afrane, S. and Ahiable, G., 2011. The informal economy and microfinance in Kumasi. In K.K. Adarkwa (Ed.), *Future of the tree towards growth and development of Kumasi*, University Printing Press, KNUST, Kumasi, Ghana.

Agbebi, M. and Virtanen, P., 2017. Dependency Theory – A Conceptual Lens to Understand China's Presence in Africa? *Forum for Development Studies*, 44 (3), 429-451. https://doi.org/10.1080/08039410.2017.1281161

Agyemang, I., McDonald, A. and Carver, S., 2007. Application of the DPSIR framework to environmental degradation assessment in northern Ghana. *Natural Resources Forum*, 31, 212-225. doi:10.1111/j.1477-8947.2007.00152.x

Ahmed, A. and Dinye, R.D., 2011. The impact of land use activities on Subin and Aboabo Rivers in Kumasi Metropolis. Int. J. Water Res. Env. Eng., 4, 241–251.

Aina Y.A., Amosa, M.K., Orewole, M.O., 2019. Students' Perception on Sustainability. In: Leal Filho W. (eds) Encyclopedia of Sustainability in Higher Education. Springer, Cham.

Ait-Kadi, M., 2016. Water for Development and Development for Water: Realizing the Sustainable Development Goals (SDGs) Vision. *Aquatic Procedia*, 6 (2016), 106–110.

Akafia, M.E. and Kuenyehia Sr, K., 2012. 'Getting the Deal Through, Mining 2012 Ghana', 89 (2012). www.kimathilegal.com/publications/Mining%20in%20Ghana%202012.pdf Accessed: 31/03/2016. Albala-Bertrand, J.M., 2003. Urban disasters and globalization. In: Kreimer A, Arnold M, Carlin A (eds) *Building safer cities: the future of disaster risk. Disaster Risk Management Series N. 3*. The World Bank, Washington DC.

Albrechts, L., 2010. More of the Same is Not Enough! How Could Strategic Spatial Planning Be Instrumental in Dealing with the Challenges Ahead? *Environment and Planning B: Planning and Design*, 37(6), 1115–1127. doi: 10.1068/b36068.

Alderman, H. and Bundy, D., 2011. School Feeding Programs and Development: Are We Framing the Question Correctly? *World Bank Research Observer*, 27(2), 204–221. http://wbro.oxfordjournals.org/content/early/2011/07/26/wbro.lkr005.full.pdf Accessed: 25/09/2017.

Ali-Tourdet, Fazia and Ji, L., 2017. Modeling and measuring urban sustainability in multicriteria-based systems — A challenging issue. *Ecological Indicators*, 73 (2017), 597–611. https://doi.org/10.1016/j.ecolind.2016.09.046

Alkemade, F., Hekkert, M.P. and Negro, S.O., 2011. Transition policy and innovation policy: friends or foes? *Environmental Innovation and Societal Transitions*, 1(1), 125-129. http://dx.doi.org/10.1016/j.eist.2011.04.009

Allcott, H., Lederman, D. and Lopez, R., 2006. Political institutions, inequality, and agricultural growth: The public expenditure connection. World Bank Research Working Paper 3902. Washington DC: The World Bank.

Allen, A., 2009. Sustainable cities or Sustainable urbanisation? perspective. UCL's journal of sustainable cities (summer edition), 1-3. Available from: https://www.ucl.ac.uk/sustainable-cities/results/gcsc-reports/allen.pdf Accessed: 20/3/2016.

Allery, L.A., 2016. Design and use questionnaires for research in medical education. *Education for Primary Care*, 27 (3), 234-238. https://doi.org/10.1080/14739879.2016.1175914 Alliance for Development, 2014. Ghana and Sustainable Development: Restoring the Present, Securing the Future. http://www.myjoyonline.com/opinion/2014/february-24th/ghana-and-sustainable-development-restoring-the-present-securing-the-future.php Accessed: 20/5/2016.

Allmark, P., Boote J., Chambers, E., Clarke, A., McDonnell, A., Thompson, A. and Tod, A.M., 2009. Ethical issues in the use of in-depth interviews: literature review and discussion. *Research Ethics Review*, 5 (2), 48–54.

Alshuwaikhat, H.M. and Nkwenti, D.I., 2002. Visualizing decision-making: perspectives on collaborative and participative approach to sustainable urban planning and management. *Environment and Planning B*, 29 (2002), 513-531.

Amegah, K.A. and Agyei-Mensah, S., 2017. Urban air pollution in Sub-Saharan Africa: Time for action. *Environmental Pollution*, 220 (Part A), 738–743. http://dx.doi.org/10.1016/j.envpol.2016.09.042

Amo, M., Bih, F.K., Agyeman, A., Gyamfi, T. A. and Mensah, M., 2017. Investigation into the acquisition and development of wetlands in built environment industry: a case study in Kumasi metropolis. *International Journal of Civil Engineering, Construction and Estate Management*, 5(4), 1-20.

Amoh-Gyimah, R. and Aidoo, N.E., 2013. Mode of transport to work by government employees in the Kumasi metropolis, Ghana. *Journal of Transport Geography*, 31, 35-43.

Anarfi, K., Hill, R.A. and Shiel, C., 2020. Highlighting the sustainability implications of urbanisation: a comparative analysis of two urban areas in Ghana. *Land*, 9 (9), 300. <u>https://doi.org/10.3390/land9090300</u>

Anderson, J.R., Hardy, E.E., Roach, J.T. and Witmer, W.E., 1976. A land use and land cover classification system for use with Earth Observation data, USGS professional paper 964, Reston, Virginia.

Anderson, B. N. and Michal, C. J., 1978. Passive solar design. *Annual Review of Energy*, 3(1), 57-100.

Angel-Urdinola, D.F. and Tanabe, K., 2012. Micro-determinants of informal employment in the Middle East and North Africa region. The World Bank, Washington DC.

Anthopoulos, L. and Fitsilis, P., 2010. From digital to ubiquitous cities: Defining a common architecture for urban development. 2010 Sixth International Conference on Intelligent Environments, (IE), IEEE (2010), 301–306.

Antrop, M., 2000. Changing patterns in the urbanized *countryside of Western Europe*. *Landscape Ecology*, 15, 257–270.

Appolonia City, 2019a. Welcoming webpage with project details. https://www.appolonia.com.gh/ Accessed: 09/9/2019.

Appolonia City, 2019b. Why choose Appolonia? https://www.appolonia.com.gh/why-choose-appolonia/ Accessed: 09/9/2019.

Appolonia City, 2019c. Nana Akufo-Addo launches affordable housing project at Appolonia City. https://www.appolonia.com.gh/news/akufo-addo-launches-ghana-home-loans-new-housing-project-at-appolonia-city/ Accessed: 10/9/2019.

Appolonia City, 2019d. Nova Ridge. https://www.appolonia.com.gh/now-selling/nova-ridge/ Accessed 10/9/2019.

Appolonia City, 2019e. Corporate Social Responsibility. https://www.appolonia.com.gh/csr/ Accessed: 10/09/2019.

Appolonia City, 2019f. Appolonia City Brochure. https://www.appolonia.com.gh/wp-content/uploads/New-Appolonia-brochure-spreads.pdf Accessed 9/9/2019.

Appolonia City, 2019g. Oxford Type Homes. https://www.appolonia.com.gh/now-selling/the-oxford/your-home/ Accessed: 10/9/2019.

Ardayfio-Schandorf, E., Yankson, P.W.K. and Bertrand, M., 2012. The mobile city of Accra: Urban Families, Housing and Residential Practices. Dakar, CODESRIA, 2012.

Armstrong, C., 2012. *Global Distributive Justice: An Introduction*, Cambridge University Press, Cambridge, GB.

Arthur-Mensah, N. and Alagaraja, M., 2018. Examining training and skills development of youth and young adults in the Ghanaian context: an HRD perspective. *Human Resource Development International*, 21(5), 493-508. DOI: 10.1080/13678868.2018.1468587

Aryee, B.N.A., Ntibery, B.K. and Atorkui, E., 2003. Trends in the small-scale mining of precious minerals in Ghana: a perspective on its environmental impact. *Journal of Cleaner Production*, 11 (2003), 131-140.

Astleithner, F. and Hamedinger, A., 2003. The analysis of sustainability indicators as socially constructed policy instruments: benefits and challenges of 'interactive research'. *Local environment*, 8, 627–640.

Attua, E.M. and Fisher, J.B., 2011. Historical and Future Land-Cover Change in a Municipality of Ghana. *Earth Interact.*, 15, 1–26. https://doi.org/10.1175/2010EI304.1

Aubynn, A.K., 2006. Live and let live': The relationship between artisanal/small-scale and large-scale miners at Abosso Goldfield, Ghana. In *Small-Scale Mining, Rural Subsistence and Poverty in West Africa*; Hilson, G.M., Ed.; Practical Action Publishing: Rugby, UK.

Aurand, A. G., 2007. "Is Smart Growth for Low-Income House- holds: A Study of the Impact of Four Smart Growth Principles on the Supply of Affordable Housing," Doctoral Thesis, University of Pittsburgh, Pittsburgh, 2007, 451 p.

Avelino, F., 2009. Empowerment and the challenge of applying transition management to ongoing projects. *Policy Sciences*, 42(4), 369–390.

Awortwi, N., 2015. Ghana National Urban Policy: Exercise fit for purpose or another policy document on the shelf?

https://www.urbangateway.org/icnup/sites/default/files/ICNUP%20Ghana.pdf Accessed: 23/01/2019. Awumbila, M., 2006. Gender equality and poverty in Ghana: Implications for poverty reduction strategies. *GeoJournal*, 67(2), 149-161. Retrieved from http://www.jstor.org/stable/41148110

Azumah, F.D., Baah, E. and Nachinaab, J.O., 2019. Causes and effects of illegal gold mining (Galamsey) activities on school dropout and residents at the Tutuka Central Circuit in Obuasi Municipality in Ashanti Region, Ghana. *International Journal of Research and Innovation in Social Science*, 3 (7), 477-487.

Baabou, W., Grunewald, N., Ouellet-Plamondon, C., Gressot, M. and Galli, A., 2017. The Ecological Footprint of Mediterranean cities: Awareness creation and policy implications. *Environmental Science & Policy*. 69, 94–104. http://dx.doi.org/10.1016/j.envsci.2016.12.013

Bao, C. and Fang C., 2012. Water resources flows related to urbanisation in China: Challenges and perspectives for water management and urban development. *Water Resources Management*, 26 (2), 531-552.

Bargain, O. and Kwenda, P., 2011. Earning structures, informal employment and selfemployment: New evidence from Brazil, Mexico and South Africa. *Review of Income and Wealth*, 57 (2011), 100–122.

Barkemeyer, R., Holt, D., Preuss, L. and Tsang, S., 2014. What happened to development in sustainable development? *Sustainable Development*, 22(1), 15-32.

Barrera-Roldan, A. and Saldivar-Valdes, A., 2002. Proposal and application of a sustainable development index. *Ecol. Indic.*, 2 (2002), 251-256.

Barrios, S., Bertinelli, L. and Strobl, E., 2006. Climatic Change and Rural-Urban Migration: The Case of Sub-Saharan Africa. *Journal of Urban Economics*, 60 (3), 357–371.

Barry, J., 1999. Rethinking Green Politics, Sage, London, GB.

Bastianoni, S., Coscieme, L. and Pulselli, F.M., 2016. The input-state-output model and related indicators to investigate the relationships among environment, society and economy. *Ecol. Model.*, 325 (2016), 84-88.

Bates, R.H., 1984. *Markets and states in tropical Africa: the political basis of agricultural policies*, [new ed.] ed. Univ. of California Press, Berkeley.

Baur, A. H., Thess, M., Kleinschmit, B. and Creutzig, F., 2013. Urban climate change mitigation in Europe: looking at and beyond the role of population density. *J. Urban Plann. Dev.* 140 (1), 04013003.

Batlle-Montserrat, J., Blat, J. and Abadal, E., 2016. Local e-government bench learning: Impact analysis and applicability to smart cities benchmarking. *Inf. Polity*, 21 (1), 43–59.

Becker, C. M., Mendelsohn, S. J. and Benderskaya, K. A., 2012. Russian urbanization in the Soviet and post-Soviet eras. International Institute for Environment and Development, 1–128.

Bekker, S. and Therborn, G., 2012. Capital Cities in Africa eds. Cape Town: HSRC Press.

Bell, D. N. F. and Blanchflower, D. G., 2010. UK Unemployment in the Great Recession, *National Institute Economic Review*, 214, 3-25.

Bell, S. and Morse, S., 1999. Sustainability Indicators: Measuring the Immeasurable? Routledge, GB.

Bell, S. and Morse, S., 2001. Measuring Sustainability: Learning from doing. Routledge, GB.

Bellinger, G., 2004. "Modeling & Simulation: An Introduction," in Mental Model Musings. Available at http://www.systems-thinking.org/modsim/modsim.htm Accessed:05/11/2017.

Bengtsson, M., Shen Y.J. and Oki, T. A., 2006. SRES-based gridded global population dataset for 1990–2100. *Popul Environ* 28 (2), 113–131.
Bergek, A., Jacobsson, S., Carlsson, B., Lindmark, S. and Rickne, A., 2008. Analyzing the functional dynamics of technological innovation systems: a scheme of analysis. *Research Policy*, 37, 407–429.

Berkes, F. and Folke, C., 1998. Linking social and ecological systems for resilience and sustainability. In F. Berkes & C. Folke (Eds.), *Linking social and ecological systems* Cambridge: Cambridge University Press, GB.

Berliner, J., 1977. Internal migration: A comparative disciplinary view. In Alan Brown & EgonNeuberger (Eds.), *Internal migration: A comparative perspective*. New York: Academic Press.

Bezemer, D. and Headey D., 2008. Agriculture development and urban bias. *World Development*, 36 (8), 1342-64.

Birago, D., Opoku, S.M. and Sharma, S., 2017. Level of service delivery of public transport and mode choice in Accra, Ghana. *Transportation Research Part F: Traffic Psychology and Behaviour*, 46 (Part B), 284-300. <u>http://dx.doi.org/10.1016/j.trf.2016.09.033</u>

Bird, S.J. and Spier, R.E., 2008. A Conflict of Interest Disclosure Policy for Science and Engineering Ethics. *Sci Eng Ethics*, 14 (149). https://doi.org/10.1007/s11948-008-9073-6

Biswas, A.K., 2006. Water Management for Major Urban Centres, *Water Resources Development*, 22(2), 183-197. DOI: 10.1080/07900620600690789

Bloch, E., 1986. *The Principle of Hope*, Edited by: Plaice, N., Plaice, S. and Wright, P. Vol.3, Cambridge: MIT.

Bloch, R. and Owusu, G., 2012. Linkages in Ghana's gold mining industry: Challenging the enclave thesis. *Resources Policy*, 37 (4), 434-442.

Boamah, N.A., 2010. Housing affordability in Ghana: a focus on Kumasi and Tamale. *Ethiopian Journal of Environmental Studies and Management*, 3 (3), 1–11.

Boamah, N.A., Gyimah, C. and Nelson, J.K.B., 2012. Challenges to the enforcement of development controls in Wa municipality. *Habitat International*, 36, 136–142.

Bogaert, K., 2011. "The Problem of Slums: Shifting Methods of Neoliberal Urban Government in Morocco". *Development and Change*, 42 (3), 709–731.

Bolis, I., Morioka, S. N. and Sznelwar L.I., 2014. "When sustainable development risks losing its meaning. Delimiting the concept with a comprehensive literature review and a conceptual model". *Journal of Cleaner Production*, 83, 7-20. https://doi.org/10.1016/j.jclepro.2014.06.041

Boon, E. K., 2009. *Area Studies - Regional Sustainable Development:* Africa - Volume 1. SBN: 978-1-84826-066-5 (eBook).

Bornschier, V. and Chase-Dunn, C., 1985. *Transnational Corporations and Underdevelopment*. Praeger Press, New York.

Borucke, M., Moore, D., Cranston, G., Gracey, K., Iha, K., Larson, J., Lazarus, E., Morales, J.C., Wackernagel, M. and Galli, A., 2013. Accounting for demand and supply of the biosphere's regenerative capacity: the National Footprint Accounts' underlying methodology and framework. *Ecol. Indic*, 24, 518-533.

Bosehans, G. and Walker, I., 2020. My sustainable city – Exploring lay people's conception of sustainable urban design. *The Social Science Journal*, 1-17. https://doi.org/10.1016/j.soscij.2019.08.004

Bossel, H., 1999. Indicators for Sustainable Development: Theory, Method, Applications. A Report to the Balaton Group International Institute for Sustainable Development (1999) (The International Institute for Sustainable Development).

Bott, H. and Grassl, G., 2013. Nachhaltige Stadtplanung, München Institut f. intern Architektur-Dok (2013).

Bowen, G., 2009. Document Analysis as a Qualitative Research Method, *Qualitative Research Journal*, 9 (2), 27-40. https://doi.org/10.3316/QRJ0902027

Bradley., P. and Yee, S., 2015. Using the DPSIR Framework to Develop a Conceptual Model: Technical Support Document. US Environmental Protection Agency, Office of Research and Development, Atlantic Ecology Division, Narragansett, RI. EPA/600/R-15/154. https://cfpub.epa.gov/si/si_public_record_report.cfm?dirEntryId=311236&Lab=NHEERL Accessed: 09/01/2019.

Bradshaw, Y. W., 1987. Urbanization and underdevelopment: A global study of modernization, urban bias, and economic dependency. *American Sociological Review*, 52 (2), 224-239.

Bradshaw, Y.W. and Noonan, R., 1997. Urbanization, economic growth, and women's labour-force participation: a theoretical and empirical reassessment J. Gulger (Ed.), *Cities in the Developing World: Issues, Theory, and Policy*, Oxford University Press, Oxford, GB.

Brauer, M., Amann, M., Burnett, R.T., Cohen, A., Dentener, F. and Ezzati, M., 2012. Exposure assessment for estimation of the global burden of disease attributable to outdoor air pollution. *Environ. Sci. Technol.*, 46 (2), 652–660.

Brenner, N. and Theodore, N., 2002. *Spaces of Neoliberalism: Urban Restructuring in North America and Western Europe*, Oxford: Blackwell, GB.

Brunner, P.H., 2007. Reshaping urban metabolism. *Journal of Industrial Ecology*, 11 (2007), 11-13.

Bruton, G., Ireland, D. and Ketchen, D., 2012. "Toward a research agenda on the informal economy". *Academy of Management Perspectives*, 26 (3), 1-11.

Bryceson, D.F., Gough, K.V., Rigg, J. and Agergaard, J., 2009. Critical commentary. The World Development Report 2009. *Urban Studies*, 46 (4), 723–738.

Bryceson, D. and MacKinnon, D., 2012. Eureka and beyond: mining's impact on African urbanisation. *Journal of Contemporary African Studies*, 30 (4). 513-537.

Bryman, A., 1989. Research methods and organization studies, Unwin Hyman: London, GB.

Buckley, R. and Kallergis, A., 2014. "Does African Urban Policy Provide a Platform for Sustained Economic Growth?" In *The Routledge Handbook on Cities of the Global South*, edited by S. Parnell and S. Oldfield, 173–190. London: Routledge, GB.

Bugliarello, G., 2006. Urban sustainability: Dilemmas, challenges and paradigms. *Technol. Soc.*, 28, 19–26.

Bulkeley, H. and Betsill, M., 2005. Rethinking sustainable cities: multi-level governance and the urban politics of climate change. *Environmental politics*, 14(1), 42–63.

Bull, B. and Bøås, M., 2012. Between Ruptures and Continuity: Modernisation, Dependency and the Evolution of Development Theory, *Forum for Development Studies*, 39(3), 319-336, DOI: 10.1080/08039410.2012.688860

Burke-Johnson, R. and Onwuegbuzie, A., 2004. Mixed methods research: A research paradigm whose time has come. *Educational Researcher*, 33 (7), 14-36.

Byres, T.J., 1979. Of neopulist pipe dreams. Journal of Peasant Studies, 6, 210-44.

Caprotti, F., 2014. Critical research on eco-cities? A walk through the Sino-Singapore Tianjin Eco-City, China. *Cities*, 36 (2014), 10–17.

Caprotti, F., Springer, C. and Harmer, N., 2015. 'Eco' For Whom? Envisioning Ecourbanism in the Sino-Singapore Tianjin Eco-city, China. *International Journal of Urban and Regional Research*, 39, 495-517. doi:10.1111/1468-2427.12233 Caprotti, F., Cowley, R., Datta, A., Broto, V.C., Gao, E., Georgeson, L., Herrick, C., Odendaal, N., Joss, S., 2017. The New Urban Agenda: key opportunities and challenges for policy and practice. Urban research & practice, 10 (30), 367-378. https://doi.org/10.1080/17535069.2016.1275618

Caprotti, F., 2018. Future cities: moving from technical to human needs. *Palgrave Communications*, 4 (35), 1-4. DOI: 10.1057/s41599-018-0089-5

Caprotti, F. and Liu, D., 2019. Emerging platform urbanism in China: Reconfigurations of data, citizenship and materialities. *Technological Forecasting and Social Change*, Article in Press, https://doi.org/10.1016/j.techfore.2019.06.016

Carter, M. R., 2000. Land ownership inequality and the income distribution consequences of economic growth. UN-WIDER Working Paper No. 201. Helsinki: World Institute for Development Economics Research.

Cavalcanti T., Da Mata D. and Toscani, F., 2014. Winning the Oil Lottery: The Impact of Natural Resource Extraction on Growth. Unpublished Manuscript, Department of Economics, University of Cambridge.

Cavaye, A.L.M., 1996. Case study research: a multi-faceted research approach for IS. *Information Systems Journal*, 6, 227-242.

Cervero, R., 2013. Transport Infrastructure and the Environment: Sustainable Mobility and Urbanism. Working Paper 2013–03), University of California: Berkeley Institute of Urban and Regional Development. http://iurd.berkeley.edu/publications/wp/2013-03.pdf> Accessed: 27/04/2016.

Chang, S.C., 2009. Productivity and territorial specialization. *Applied Economics*, 41, 941–945.

Chang, I.C.C. and Sheppard, E., 2013. China's Eco-Cities as Variegated1 Urban Sustainability: Dongtan Eco-City and Chongming Eco-Island. *Journal of Urban Technology*, 20(1), 57-75. DOI: 10.1080/10630732.2012.735104

Chen, D. and Stow, D., 2002. The effect of training strategies on supervised classification at different spatial resolutions. *Photogrammetric Engineering & Remote Sensing*, 68(11), 1155-1162.

Cheru, F., 2014. *Structural Transformation in Ethiopia: The Urban Dimension*. Nairobi: UN-Habitat.

Childers, D.L, Pickett, S.T.A, Grove, J.M., Ogden, L. and Whitmer A., 2014. Advancing urban sustainability theory and action: Challenges and opportunities. *Landscape and Urban Planning*, 125, 320-328. https://doi.org/10.1016/j.landurbplan.2014.01.022

Cities Alliance, 2008. Guide to City Development Strategies: Improving Urban Performance. (Washington D.C.).

Clark, W.C., 2003. Urban environments: battlegrounds for global sustainability. *Environment*, 45 (7), 1.

Clark M.C. and Sharf, B.F., 2007. The dark side of truth(s): ethical dilemmas in researching the personal. *Qualitative Inquiry*, 13(3), 399-416.

Cloete, F., 2015. Measuring progress towards sustainable development in Africa. African *Journal of Public Affairs*, 8 (3), 1-24.

Cobbinah, P.B., Erdiaw-Kwasie, M.O. and Amoateng, P., 2015a. Africa's urbanisation: Implications for sustainable development. *Cities*, 47 (2015), 62–72. http://dx.doi.org/10.1016/j.cities.2015.03.013

Cobbinah, P., Gaisie, E., and Owusu-Amponsah, L., 2015b. Peri-urban morphology and indigenous livelihoods in Ghana. *Habitat International*, 50, 120-129. DOI: 10.1016/j.habitatint.2015.08.002

Cobbinah, P. B., and Aboagye, H. N., 2017. A Ghanaian twist to urban sprawl. *Land Use Policy*, 61, 231-241. DOI: 10.1016/j.landusepol.2016.10.047

Cobbinah, P. B. and Niminga-Beka, R., 2017. Urbanisation in Ghana: Residential land use under siege in Kumasi central. *Cities*, 60(A), 388-401. DOI: 10.1016/j.cities.2016.10.011

Coenen, L., Benneworth, P. and Truffer, B., 2012. Toward a spatial perspective on sustainability transitions. *Res. Policy*, 41, 968–979.

Cohn, E.S. and Lyons, K.D., 2003. The perils of power in interpretive research. *Am. J. Occup. Ther.*, 57(1), 40-8.

Collard, D., 2003. Research on wellbeing: Some advice from Jeremy Bentham. Wellbeing in Developing Countries, ESRC Research Group Working Paper No. 2. University of Bath, Bath, GB.

Commission for Africa, 2005. Our Common Interest: Report of the Commission for Africa, London, GB.

Cooke, E., Hague, S. and Mckay, A., 2016. The Ghana Poverty and Inequality Report: Using the 6th Ghana Living Standards Survey 2016.

Coquery-Vidrovitch, C., 1991. The process of urbanization in Africa (from the origins to the beginning of independence). *African Studies Review*, 34 (1), 1–98.

Corbin, J. and Strauss, A., 2008. *Basics of qualitative research: Techniques and procedures for developing grounded theory* (3rd ed.). Thousand Oaks, CA: Sage.

Costanza, R., Fisher, B., Ali, S., Beer, C., Bond, L. and Boumans, R., 2007. Quality of life: An approach integrating opportunities, human needs, and subjective well-being. *Ecological Economics*, 61(2–3), 267–276.

Costanza, R., Hart, M., Posner, S. and Talberth, J., 2009. Beyond GDP: The Need for New Measures of Progress. Technical Report 4, The Frederick S. Pardee Center for the Study of the Longer-Range Future, Boston University, Boston, MA.

Coulombe, H. and Lanjouw, P., 2013. Poverty, access to services and city size in a selection of African countries. Mimeo, the World Bank.

Coulter, L.L., Stow, D.A., Tsai, Y., Ibanez, N., Shih, H., Kerr, A., Benza, M., Weeks, J.R. and Mensah, F., 2016. Classification and assessment of land cover and land use change in southern Ghana using dense stacks of Landsat 7 ETM+ imagery. *Remote Sensing of Environment*, 184, 396-409. https://doi.org/10.1016/j.rse.2016.07.016

Cowburn, M., 2005. Confidentiality and public protection: ethical dilemmas in qualitative research with adult male sex offenders. *J. Sex Aggress.*, 11(1), 49-63.

Creswell, J.W., 2015. *A concise introduction to mixed methods research*. Los Angeles, CA: SAGE.

Creswell, J.W. and Plano-Clark, V.L., 2007. *Designing and Conducting Mixed Methods Research*. Thousand Oaks, CA: Sage Publications.

da Mata, D., Deichmann, U., Henderson, J.V., Lall, S.V. and Wang, H.G. 2007. Determinants of City Growth in Brazil. *Journal of Urban Economics*, 62(2), 252–272.

Dahl, A.L., 2012. Achievements and gaps in indicators for sustainability. *Ecol Indic*, 17, 14–19 doi:10.1016/j.ecolind.2011.04.032.

Daily Graphic Online, 2018. Ghana faces housing glut - Despite deficit. [Online] Available: https://www.graphic.com.gh/features/features/ghana-faces-housing-glut-despite-deficit.html Accessed: 15/01/2018.

Dalal, S., Beunza, J.J., Volmink, J., Adebamowo, C., Bajunirwe, F. and Njelekela, M., 2011. Non-communicable diseases in sub-Saharan Africa: what we know now. *Int. J. Epidemiol.*, 40 (4), 885–901.

Dalkmann, H. and Sakamoto, K., 2011. Transport: Investing in energy and resource efficiency. Green economy report UNEP (2011).

Daly, H., 1991. Steady State Economics (Second ed.). Island Press, Washington DC.

Daly, H.E., 1995. On Wilfred Beckerman's critique of sustainable development. *Environ Values*, 4, 49–55.

Danquah, M. and Ohemeng, W., 2017. "Unmasking the factors behind income inequalities in Ghana". *International Journal of Social Economics*, 44(7), 884-899. https://doi.org/10.1108/IJSE-09-2015-0250

Darke, P., Shanks, G. and Broadbent, M., 1998. Successfully completing case study research: combining rigour, relevance and pragmatism. *Info Systems J.*, 8, 273-289.

Darkwah, R. M., and Cobbinah, P., 2014. Stewardship of Urban Greenery in an Era of Global Urbanisation. *International Journal of Environmental, Ecological, Geological and Mining Engineering*, 8(10), 676-679.

Davidson, B., 1970. The Lost Cities of Africa. Little Brown & Company, Boston.

Davidson, K. and Arman, M., 2014. Planning for sustainability: an assessment of recent metropolitan planning strategies and urban policy in Australia. *Australian Planner*, 51(4), 296-306. DOI: 10.1080/07293682.2013.877508

Davis, M., 2010. Who will build the ark? New Left Rev., 61 (2010), 29-46.

Davis, K. and Golden, H., 1954. Urbanization and the development of pre-industrial areas *Econ. Dev. Cult. Change*, 3 (1), 6–26.

Davis, J.C. and Henderson, J.V., 2003. Evidence on the political economy of the urbanization process. *J. Urban Econ.*, 53 (1), 98–125.

De Haan, J. and Rotmans, J., 2011. Patterns in transition: understanding complex chains of change. *Technol. Forecast. Soc. Change*, 78, 90-102.

de Jong, M., Joss, S., Schraven, D., Zhan, C. and Weijnen, M., 2015. Sustainable-smartresilient-low carbon-eco-knowledge-cities: making sense of a multitude of concepts promoting sustainable urbanization. *Journal of Cleaner Production*, 109, 25-38.

de Stefano, L., 2010. International initiatives for water policy assessment: a review. *Water Resources Management*, 24 (11), 2449-2466. http://dx.doi.org/10.1007/s11269-009-9562-7

Delmas, M. and Toffel, M.W., 2004. Stakeholders and environmental management practices: an institutional framework. *Bus. Strate.g Environ*, (13), 209–222. doi: 10.1002/bse.409

Deng, J.S., Wang, K., Hong, Y. and Qi, J., G., 2009. Spatio-temporal dynamics and evolution of land use change and landscape pattern in response to rapid urbanization. *Landscape and Urban Planning*, 92 (34), 187-198. doi: http://dx.doi.org/10.1016/j.landurbplan.2009.05.001.

Depietri, Y., Renaud, F.G. and Kallis, G., 2012. Heat waves and floods in urban areas: a policy-oriented review of ecosystem services. *Sustain Sci.*, 7, 95-107.

De Soto, H., 1989. The other path. Harper and Row, New York (1989).

DETR, 2000. Public Participation in Making Local Environmental Decisions: Good Practice Handbook. Department of the Environment, Transport and the Regions, UK (2000).

Diener, E., Lucas, R., Schimmack, U. and Helliwell, J., 2009. *Well-Being for Public Policy*. Oxford University Press.

Dionisio, K., Arku, R.E., Hughes, A.F., Vallarino, J., Carmichael, H., Spengler, J.D., Agyei-Mensah, S. and Ezzati, M., 2010. Air pollution in Accra neighborhoods: spatial, socioeconomic and temporal patterns. *Environmental Science and Technology*, 44 (7), 2270-2276.

Dizdaroglu, D. and Yigitcanlar, T., 2014. A parcel-scale assessment tool to measure sustainability through urban ecosystem components: the MUSIX model. *Ecol. Indic.*, 41, 115–130. http://dx.doi.org/10.1016/j.ecolind.2014.01.037

312

Dodman, D., Brown, D., Francis, K., Hardoy, J., Johnson, C. and Satterthwaite, D., 2013. Understanding the nature and scale of urban risk in low- and middle-income countries and its implications for humanitarian preparedness, planning and response. Available from: http://pubs.iied.org/pdfs/10624IIED.pdf? Accessed: 24/10 2016.

Domfeh, K. A., Ahenkan, A., Bawole, J.N., 2012. Is sustainable development achievable in Ghana? An analysis of Ghana's development policy achievements and challenges. *Int. J. Environment and Sustainable Development*, 11 (3). Available from: https://www.researchgate.net/publication/256120956_Is_sustainable_development_achievabl e_in_Ghana_An_analysis_of_Ghana%27s_development_policy_achievements_and_challeng es Accessed:10/4/2017.

Dorosh, P. and Thurlow, J., 2012. "Can cities or towns drive African development? Economy-wide analysis for Ethiopia and Uganda", Working Paper, WIDER, Helsinki, 24 pages.

Duh, J-D, Shandas, V., Chang, H. and George, L.A. 2008. Rates of urbanisation and the resiliency of air and water quality. *Science of The Total Env*ironment, 400 (1-3), 238-256. https://doi.org/10.1016/j.scitotenv.2008.05.002

Du Pisani, J.A., 2006. Sustainable development – historical roots of the concept. *Environmental Sciences*, 3(2), 83-96. DOI: 10.1080/15693430600688831

Duranton, G., 2014. Growing through cities in developing countries. Policy Research Working Paper Series 6818, The World Bank, Washington DC.

Dutt, A. K., 2001. Global urbanization: Trends, form and density gradients. Allahabad: Professor R. N. Dubey Foundation.

Earle, L., 2016. Urban crises and the new urban agenda. *Environment and Urbanization*, 28(1), 77–86. https://doi.org/10.1177/0956247815620335

Easterly, W., 2007. Inequality does cause underdevelopment: insights from a new instrument. *Journal of Development Economics*, 84, 755-776.

Eastwood, R. and Lipton, M., 2000. Pro-poor growth and pro-growth poverty reduction: Meaning, evidence, and policy implications. *Asian Development Review* 18, 22–58.

EEA, 1995. Europe's Environment: the Dobris Assessment. European Environment Agency, Copenhagen.

Ekins, P., 2011. Environmental sustainability: from environmental valuation to the sustainability gap. *Prog Phys Geog*, 35(5), 629–651.

Ekins, P., Simon, S., Deutsch, L., Folke C. and De Groot, R., 2003. A framework for the practical application of the concepts of critical natural capital and strong sustainability. *Ecol Econ*, 44, 165–185.

Elzen, B., Geels, F.W. and Green, K., 2004. *System Innovation and the Transition to Sustainability: Theory, Evidence and Policy* (Eds.), Edward Elgar, Cheltenham, GB.

Energy Commission, 2016. National Energy Statistics 2006-2015. The Ghana Energy Commission. http://energycom.gov.gh/files/National%20Energy%20Statistics_2016.pdf Accessed: 10/2/2019.

Energy Commission, 2018. National Energy Statistics 2006-2015. The Ghana Energy Commission. http://www.energycom.gov.gh/files/ENERGY_STATISTICS_2018_FINAL.pdf Accessed: 10/2/2019.

Energy Commission, 2019. Ghana's renewable energy masterplan. http://www.energycom.gov.gh/files/Renewable-Energy-Masterplan-February-2019.pdf Accessed: 10/8/2019.

Ensign, J., 2003. Ethical issues in qualitative health research with homeless youths. *J. Adv. Nurs.*, 43(1), 43-50.

EPA, 2006. Strategic Environmental Assessment of the National Transport Policy. Accra, Ghana. Ernst, L., de-Graaf-Van Dinther, R.E., Peek, G.J. and Loorbach, D.A., 2016. Sustainable urban transformation and sustainability transitions; conceptual framework and case study. *Journal of Cleaner Production*, 112 (4), 2988-2999. https://doi.org/10.1016/j.jclepro.2015.10.136

Ernstson, H., Leeuw, S. E., Redman, C. L., Meffert, D. J., Davis, G.and Alfsen, C., 2010. Urban Transitions: On Urban Resilience and Human-Dominated Ecosystems. *AMBIO*, 2010 (39), 531-545.

Eum, J-H., Scherer, D., Fehrenbach, U. and Woo, J-H., 2011. Development of an urban land cover classification scheme suitable for representing climatic conditions in a densely built-up Asian megacity. *Landscape and Urban Planning*, 103(3-4), 362-371. https://doi.org/10.1016/j.landurbplan.2011.08.010

Ewing, R. and Cervero, R., 2010. Travel and the built environment. *Journal of the American Planning Association*, 76(3), 265–294.

Fajnzylber, P., Maloney, W. and Rojas, G., 2006. Microenterprise dynamics in developing countries: how similar are they to those in the industrialized world? Evidence from Mexico. *World Bank Economic Review*, 20 (2006), 389–419.

Falco, P., Kerr, A., Rankin, N., Sandefur, J. and Teal, F., 2011. The returns to formality and informality in urban Africa. *Labour Economics*, 18, Supplement 1, 23–3. http://doi.org/10.1016/j.labeco.2011.09.002

Fang, P., Chen, J. and John, R., 2016. Urbanization and environmental change during the economic transition on the Mongolian Plateau: Hohhot and Ulaanbaatar. *Environmental Research*, 144 (Part B), 96–112.

Fangjun, C., 2009. Modernization Theory and China's Road to Modernization. *Chinese Studies in History*, 43 (1), 7-16.

Fairfield, K. D., Harmon, J., and Behson, S. J., 2011. Influences on the organizational implementation of sustainability: An integrative model. *Organization Management Journal*, 8, 4–20.

Fält, L., 2019. New cities and the emergence of 'privatized urbanism' in Ghana, *Built Environment*, 44(4), 438-460.

Fält, L., 2020. New urban horizons in Africa. A critical analysis of changing land uses in the Greater Accra Region, Ghana. Doctoral thesis. <u>https://su.diva-portal.org/smash/get/diva2:1387866/FULLTEXT01.pdf</u> Accessed: 11/06/2020.

Farvacque-Vitkovic, C., Raghunath, M., Eghoff, C. and Boakye, C., 2008. Development of the Cities of Ghana Challenges, Priorities and Tools, Africa Region Working Paper Series Number 110. Washington, DC: World Bank.

Feldstein, A. C. and Glasgow, R. E., 2008. A practical, robust implementation and sustainability model (PRISM) for integrating research findings into practice. *The Joint Commission Journal on Quality and Patient Safety*, 34(4), 228–243.

Ferguson, J., 1999. *Expectation of modernity: myths and meanings of urban life on the Zambian Copperbelt.* University of California Press, Berkeley, Los Angeles, London.

Ferraro, V., 2008. Dependency theory: An introduction, in Secondi Giorgio, ed, *The Development Economics Reader*, London: Routledge, GB.

Fielding, A. J., 1989. Inter-regional migration and social-change—a study of South East England based upon data from the longitudinal-study. *Transactions of the Institute of British Geographers*, 14(1), 24–36.

Fincher, R. and Iveson, K., 2008. *Planning and Diversity in the City: Redistribution, Recognition and Encounter* (Palgrave Macmillan: Basingstoke and New York).

Fincher, R., Iveson, K., Leitner, H. and Preston, V., 2014. Planning in the multicultural city: celebrating diversity or reinforcing difference? *Progress in Planning*, 92, 1-55.

Firman, T., Kombaitan, B. and Pradono, P., 2007. The Dynamics of Indonesia's Urbanisation, 1980-2006. *Urban Policy and Research*, 25 (4), 433-454.

Fischer, J., Manning, A.D., Steffen, W., Rose, D.B., Daniell, K., Felton, A., Garnett, S., Gilna, B., Heinsohn, .R, Lindenmayer, D.B., MacDonald, B., Mills, F., Newell, B., Reid, J., Robin, L., Sherren, K. and Wade, A., 2007. Mind the sustainability gap. *Trends Ecol Evol*, 22(12), 621–624.

Flinders, D., 1992. In search of ethical guidance: Constructing a basis for dialogue. *Qualitative Studies in Education*, 5(2), 101–115.

Florczak K.L., 2014. Purists need not apply: The Case for Pragmatism in Mixed Methods. *Nursing Science Quarterly*, 27 (4), 278-282. https://doi.org/10.1177/0894318414546419

Foli, G., Nide, P.M., Amdejoe, C.G. and Kyei, L., 2012. Arsenic Leaching in Mill Tailings at the AngloGold Ashanti Mine, Ghana: Management of contamination in the related water environment. *West Africa Journal of Applied Ecology*, 20 (1), 11-23.

Foliente, G., Rodgers, A., Blustein, H. and Wang, X., 2007. Urban Sustainability Transition.
A tipping point approach. Conference Paper.
https://pdfs.semanticscholar.org/eef1/39db2043b6fd1d312dcc112baaec12028748.pdf
Accessed: 03/06/2016.

Fox, T., 2004. 'Corporate Social Responsibility and Development: In quest of an agenda'. *Development*, 47(3), 29-36.

Fox, S., 2011. Understanding the origins and pace of Africa's urban transition. Crisis States Working Papers Series No.2. pp. 1-32. Crisis States Research Centre September 2011.

Fox, S., 2012. Urbanization as a global historical process: Theory and evidence from sub-Saharan Africa. *Population and Development Review*, 38 (2), 285–310.

Fox, S., 2014. The Political Economy of Slums: Theory and Evidence from Sub-Saharan Africa. *World Development*, 54, 191–203. http://dx.doi.org/10.1016/j.worlddev.2013.08.005

Frank, C.R., 1968. Urban Unemployment and Economic Growth in Africa. *Oxford Economic Papers*, New Series, 20 (2), 250-274.

Frankema, E., 2005. The colonial origins of inequality: A global investigation of land distribution. GGDC Working Paper. Groningen: Groningen Growth and Development Centre.

Frantzeskaki, N. and de Haan, H., 2009. Transitions: two steps from theory to policy. *Futures*, 41 (2009), 593–606.

Fraser, E., Dougill, A.J., Mabee, W., and Reed, M., 2006. Bottom up and top down: analysis of participatory processes for sustainability indicator identification as a pathway to community empowerment and sustainable environmental management. *Journal of Environmental Management*, 78(2), 114–127.

Freitas, A., 2013. *Water as a stress factor in sub-Saharan Africa*. European Union Institute for Security Studies, Paris.

Friedmann, T. L., 2005. *The world is flat: a brief history of the globalized world in the twenty-first century*, London: Allen Lane, GB.

Fuseini, I. and Kemp, J., 2015. A review of spatial planning in Ghana's soco-economic development trajectory: A sustainable development perspective. *Land Use Policy*, 4, 309-320.

Garforth, L., 2009. No Intentions? Utopian Theory After the Future. *Journal for Cultural Research*, 13(1), 5-27. DOI: 10.1080/14797580802674787

Geels, F. W., 2001. Technological transitions as evolutionary reconfiguration processes: A multi-level perspective and a case-study. Presented at the conference "The Future of

Innovation Studies", organised by ECIS (Eindhoven Centre of innovation Studies) Eindhoven.

Geels, F.W., 2004. From sectoral systems of innovation to socio-technical systems: insights about dynamics and change from sociology and institutional theory. *Research Policy*, 33, 897–920.

Geels, F. W., 2007. Analysing the breakthrough of rock 'n' roll (1930-1970). Multi-regime interaction and reconfiguration in the multi-level perspective. *Technological Forecasting & Social Change*, 74, 1411-1431. http://dx.doi.org/10.1016/j.techfore.2006.07.008

Geri, F., Amici, V. and Rocchini, D., 2010. Human activity impact on the heterogeneity of a Mediterranean landscape. *Applied Geography*, 30 (2010), 370–379.

Ghana Chamber of Mines, 2019. Obuasi redevelopment project launched. http://ghanachamberofmines.org/news-updates/anglogold-ashanti-obuasi-mineredevelopment-project-launched/ Accessed: 11/5/2019.

Ghana Statistical Service, 2002. 2000 Population and Housing Census: Special Report on 20 Largest Localities. Ghana: GSS.

Ghana Statistical Service, 2007. Pattern and trends of poverty in Ghana 1991–2006. Accra.

Ghana Statistical Service, 2014a. 2010 Population and Housing report: urbanisation. http://www.statsghana.gov.gh/gssmain/fileUpload/pressrelease/Urbanisation%20in%20Ghan a.pdf Accessed: 10/1/2017.

Ghana Statistical Service, 2014b. District Analytical Report: Kumasi Metropolitan. http://www2.statsghana.gov.gh/docfiles/2010_District_Report/Ashanti/KMA.pdf Accessed: 10/1/2017.

Ghana Statistical Service, 2014c. District Analytical Report: Obuasi Municipality. http://www2.statsghana.gov.gh/docfiles/2010_District_Report/Ashanti/OBUASI.pdf Accessed: 10/1/2017.

319

Ghana Statistical Service, 2019. Population projections.

http://www.statsghana.gov.gh/nationalaccount_macros.php?Stats=MTA1NTY1NjgxLjUwNg ==/webstats/s679n2sn87 Accessed: 13/3/2019.

Ghansah, B., Asare, Y.M., Tchao, E.R. and Forkuo, E.K., 2016. Mapping the spatial changes in Lake Volta using multitemporal remote sensing approach. *Lakes Reserv. Res. Manage*, 21, 206-215. doi:10.1111/lre.12138

Giffinger, R., Fertner, C., Kramar, H., Kalasek, R., Pichler-Milanovic, N. and Meijers, E., 2007. Smart cities-ranking of European medium-sized cities, Tech. Rep. Vienna University of Technology (2007) Smart-cities EU, 2007. Smart cities: Ranking of European medium-sized cities. Final report 2007.

Gillespie, T., 2016. Accumulation by urban dispossession: struggles over urban space in Accra, Ghana. *Trans. Inst. Br. Geogr.*, 41, 66–77. doi:10.1111/tran.12105

Girardet, H., 2000. Cities and the culture of sustainability. In F. Dodds (Ed). *Earth Summit 2002*, London: Earthscan.

Gisladottir, R. and Stocking, M., 2005. Land degradation control and its global environmental benefits. *Land Degrad. Dev.*, 16 (2005), 99-112.

Global Footprint Network, 2019. http://data.footprintnetwork.org/#/ Accessed: 5/3/2019.

Godard, X., 2013. Comparisons of urban transport sustainability: Lessons from West and North Africa. *Urban Transport in Developing Countries:* CODATU Special Issue. 40 (1), 96–103. <u>http://dx.doi.org/10.1016/j.retrec.2012.06.033</u>

GOG, 2010. Medium-Term National Development Policy Framework: Ghana Shared Growth and Development Agenda (GSGDA), 2010–2013, vol. I.

Gollin D., Jedwab R. and Vollrath D., 2013. Urbanization with and without Industrialization. Unpublished manuscript, Oxford University, Department of International Development.

Gong, P., Liang, S., Carlton, E.J., Jiang, Q., Wu, J., Wang, L., and Remais, J.V., 2012. Urbanisation and health in China. *The Lancet.* 379 (9818), 843–852.

Gonzalez, A., Donnelly, A., Jones, M., Klostermann, J., Groot, A., and Breil, M., 2011. Community of practice approach to developing urban sustainability indicators. *Journal of Environmental Assessment Policy and Management*, 13(04), 591–617.

Gordon, M., Lockwood, M., Vanclay, F., Hanson, D. and Schirmer, J., 2012. Divergent stakeholder views of corporate social responsibility in the Australian forest plantation sector. *Journal of Environmental Management*, 113, 390 – 398. https://doi.org/10.1016/j.jenvman.2012.09.016

Gosling, M., 2006. Battle to Save Earth will be Won or Lost in our Cities, Klaus Topfer, Cape Times, Cape Town.

Gough, K.V. and Yankson, P.W.K., 2012. Exploring the connections: mining and urbanisation in Ghana, *Journal of Contemporary African Studies*, 30:4, 651-668, DOI: 10.1080/02589001.2012.724867

Gough, K. V., Esson, J., Andreasen, M.H., Singirankabo, A. U., Yankson, P. W. K., Yemmafouo, A., Mainet H. and Ninot, O., 2013. 'City Dynamics' RurbanAfrica Briefing No. 3, 2013, Department of Geosciences and Natural Resource Management, University of Copenhagen, Denmark.

Gould, W.T.S., 1998. African Mortality and the New 'Urban Penalty'. *Health and Place*, 4 (2), 171-181.

Government of India, 2011. Report on urban infrastructure and services GOI (2011) http://www.niua.org/projects/hpec/FinalReport-hpec.pdf Accessed: 05.03.16.

Gozgor, G. and Kablamaci, B., 2015. What happened to urbanization in the globalization era? An empirical examination for poor emerging countries. *The Annals of Regional Science*, 55 (2), 533–553. Graham, S. and Marvin, S., 2001. Splintering urbanism: *Networked infrastructures, technological mobilities and the urban condition*. Routledge, New York.

Grant, R., 2015. Africa: Geographies of Change. Oxford: Oxford University Press.

Grant, R. and Nijman, J., 2002. Globalization and the Corporate Geography of Cities in the Less-Developed World. *Annals of the Association of American Geographers*, 92(2), 320-340. DOI: 10.1111/1467-8306.00293

Greasley, D., Hanley, N., Kunnas, J., McLaughlin, E., Oxley, L. and Warde, P., 2014. Testing genuine savings as a forward-looking indicator of future well-being over the (very) longrun. *J. Environ. Econ. Manag.*, 67(2), 171–188.

Greca, I.M. and Moreira, M.A., 2000. Mental models, conceptual models, and modelling, *International Journal of Science Education*, 22(1), 1-11, DOI:10.1080/095006900289976

Gregory, A.J., Atkins, J.P., Burdon, D. and Elliott, M., 2013. A problem structuring method for ecosystem-based management: the DPSIR modelling process. *Eur. J. Oper. Res.*, 227 (3), 558-569. http://dx.doi.org/10.1016/j.ejor.2012.11.020

Graymore, M. L. M., 2014. Sustainability reporting: An approach to get the right mix of theory and practicality for local actors. *Sustainability*, 6(6), 3145–3170.

Grimmond, S., 2007. Urbanization and global environmental change: local effects of urban warming. *Geogr. J.*, 173,83–88.

Grin, J, Rotmans, J., Schot, J., Geels, F.W. and Loorbach, D., 2010.*Transitions to sustainable development: new directions in the study of long-term transformative change*. Routledge, London.

Groves, R. M., 1989. Survey errors and survey costs, New York: Wiley.

Grundey, D., 2008. Applying sustainability principles in the economy. *Technological and Economic Development of Economy*, 14(2), 101–106.

Gruneberg, S., 2006. Effective regeneration action: Finance and regeneration. Report of the LFIG Regeneration Group, Appendix 'B' (pp. 57–61).

Hagerty, M. R., Cummins, A. R., Ferriss, A. L., Land, K., Michalos, A. C. and Peterson, M., 2001. Quality of life indexes for national policy: Review and agenda for research. *Social Indicators Research*, 55, 1–96.

Halfani, M., 1996. Marginality and dynamism: prospects for the sub-Saharan African city. M. Cohen, B. Ruble, J. Tulchin, A. Garland (Eds.), *Preparing for the urban future: global pressures and local forces*, Woodrow Wilson Center Press, Washington DC.

Hall, R.E., Bowerman, B., Braverman, J., Taylor, J., Todosow, H. and Von Wimmersperg, U., 2000. *The Vision of a Smart City*; Brookhaven National Lab: Upton, NY, USA.

Hamiduddin, I., 2018. Journey to work travel outcomes from 'city of short distances' compact city planning in Tübingen, Germany. *Planning Practice & Research*, 33 (4), 372-391.

Hamilton, A., Mitchell, G. and Yli-Karjanmaa, S., 2002. The BEQUEST toolkit: a decision support system for urban sustainability. *Building Research & Information*, 30 (2), 109-115. DOI: 10.1080/096132102753436486

Hammer, S., Kamal-Chaoui, L., Robert, A. and Plouin M., 2011. Cities and green growth: A conceptual framework. OECD Regional Development Working Papers, 2011/08 OECD Publishing (2011). Available at: <u>http://dx.doi.org/10.1787/5kg0tflmzx34-en</u> Accessed: 23/01/2017.

Handy, S.L., Boarnet, M.G., Ewing, R. and Killingsworth, R.E., 2002. How the built environment affects physical activity: views from urban planning. *American Journal of Preventive Medicine*, 23, 64–73.

Hansen, A.J., Knight, R.L., Marzluff, J.M., Powell, S., Brown, K., Gude, P.H. and Jones, A., 2005. Effects of exurban development on biodiversity: patterns, mechanisms and research needs. *Ecological Applications*, 15, 1893–1905.

Haq, M., 1976. *The Poverty Curtain: Choices for the Third World*, New York: Columbia University Press.

Hartig, T., Mitchell, R., de Vries, S. and Frumkin, H., 2014. Nature and health. *Annual Review of Public Health*, 35, 207-228.

Harvey, D., 1989. From Managerialism to Entrepreneurialism: The Transformation in Urban Governance in Late Capitalism. *Geografiska Annaler*, 71 (B-1), 3-17.

Heath, G. W., Brownson, R. C., Kruger, J., Miles, R., Powell, K. E. and Ramsey, L. T., 2006. The effectiveness of urban design and land use and transport policies and practices to increase physical activity: A systematic review. *Journal of Physical Activity & Health*, 3, S55-S76.

Held, D., 1992. The decline of the nation state. In Hall, S & Jacques, S (Eds), *New times: the changing face of politics in the 1990s*. London: Lawrence & Wishart.

Henderson, V., Roberts, M. and Storeygard, A., 2013. Is Urbanization in Sub-Saharan Africa Different? Unpublished manuscript, Department of Geography, LSE.

Heng, Y. T., Wagner, D. T., Barnes, C. M., and Guarana, C. L., 2018. Archival research: Expanding the methodological toolkit in social psychology. *Journal of Experimental Social Psychology*, 78, 14–22. https://doi.org/10.1016/j.jesp.2018.04.012

Hertzler, J.O., 1922. The History of Utopian Thought. London: George Allen.

Herzi, A.A., 2004. Sustainability indicator system and policy processes in Malaysia: a framework for utilisation and learning. *J. Environ. Manage.*, 73 (4) (2004), 357–371. http://dx.doi.org/10.1016/j.jenvman.2004.07.010

Herzi, A.A., 2006. Connecting Sustainability Indicators to Policy Statements. Australian National University (2006).

Herzi, A. A. and Hasan, N., 2004. Management framework for sustainable development indicators in the Stated of Selangor, Malaysia. *Ecological Indicators*, 4, 287–304.

Hicks, D., 1997. The Inequality-Adjusted Human Development Index: A Constructive Proposal. *World Development*, 25 (8), 1283 – 1298. https://doi.org/10.1016/S0305-750X(97)00034-X

Hidalgo, D. and Huizenga, C., 2013. Implementation of sustainable urban transport in Latin America. *Research in Transportation Economics*, 40 (2013), 66-77.

Hildén, M. and Rosenström, U., 2008. The use of indicators for sustainable development. *Sustainable Development*, 16, 237–240. DOI: 10.1002/sd.375

Hilson, G., 2002a. Harvesting minerals riches: 1000 years of gold mining in Ghana, *Resource Policy*, 28 (2002), 13-26.

Hilson, G., 2002b. An overview of land use conflicts in mining communities. *Land Use Policy*, 19 (1), 65-73.

Hilson, G., 2012. Corporate social responsibility in the extractive industries: experiences from developing countries, *Resources Policy*, 37, 131-137.

Hilson, G., Hilson, A. and Adu-Darko, E., 2014. Chinese participation in Ghana's informal gold mining economy: Drivers, implications and clarifications. *Journal of Rural Studies*, 34 (2014), 292-303.

Hodson, M. and Marvin, S., 2010. Can cities shape socio-technical transitions and how would we know if they were? *Research Policy*, 39 (2010), 477–485. doi:10.1016/j.respol.2010.01.020

Höjer, M. and Wangel, S., 2015. Smart sustainable cities: Definition and challenges. L. Hilty,B. Aebischer (Eds.), ICT innovations for sustainability, Springer-verlag, Berlin (2015), 333–349.

Holden, E., Linnerud, K. and Banister, D., 2014. Sustainable Development: Our Common Future revisited. *Global Environmental Change*, 26, 130 – 139. https://doi.org/10.1016/j.gloenvcha.2014.04.006

Holden, M., Roseland, M., Ferguson, K. and Perl, A., 2008. Seeking urban sustainability on the world stage. *Habitat International*, 32, (3), 305-317.

Hollands, R.G., 2008. Will the real smart city please stand up? Intelligent, progressive or entrepreneurial? *City*, 12 (3), 303–320.

Holloway, S. R., Wright, R. and Ellis, M., 2012. The racially fragmented city? Neighborhood racial segregation and diversity jointly considered. *The Professional Geographer*, 64(1), 63-82.

Hong, S. and Vicdan, H., 2016. Re-imagining the utopian: Transformation of a sustainable lifestyle in ecovillages. *Journal of Business Research*, 69 (1), 120-136. https://doi.org/10.1016/j.jbusres.2015.07.026

Hoogma, R., Kemp, R., Schot, J. and Truffer, B. 2002. *Experimenting for Sustainable Transport. The Approach of Strategic Niche Management.* Spon Press, London/New York.

Hoornweg, D, Sugar, L. and Trejos G'omez, C.L., 2011. Cities and greenhouse gas emissions: moving forward. *Environ. Urbanization*, 23 (1), 207–227.

Hopwood, B., Mellor, M. and O'Brian, G. 2005. Sustainable development: Mapping different approaches. *Sustainable Development*, 13, 38–52. DOI:10.1002/sd.244.

Hormeku, T., 1998. The Transformation and Development of the Informal Sector and the Role of Trade Unions, Paper prepared for an OATUU/ILO/ETUF Seminar on "Trade unions and the informal sector", Cairo, Egypt, 4-6 May.

Hou, J., Walsh, P. P. and Zhang, J., 2015. The dynamics of Human Development Index. *The Social Science Journal*, 52(3), 331–347.

Hsieh, C-M., 2012. Incorporating perceived importance of service elements into client satisfaction measures. *Research on Social Work Practice*, 22(1), 93–99.

Hubert-Moy, L., Cotonnec, A, Du, L.L., Chardin, A. and Perez, A., 2001. Comparison of Parametric Classification Procedures of Remotely Sensed Data Applied on Different Landscape Units. *Remote Sensing of Environment*, 75 (2), 174-187.

Huete A.R., 2004. Remote sensing for environmental monitoring. *Environmental Monitoring and Characterization*, 183-206. https://doi.org/10.1016/B978-012064477-3/50013-8

Hull, R.W., 1976. *African Cities and Towns before the European Conquest*. W.W. Norton & Company, New York/London.

ICMM, 2007. Ghana Country Case Study: The Challenge of Mineral Wealth: Using Resource Endowments to Foster Sustainable Development. International Council on Mining and Metals, London, UK.

IEHIAS, 2019. The DPSIR Framework. http://www.integrated assessment.eu/eu/indexba0d.html?q=guidebook/dpsir_framework Assessed: 10/3/2019.

IIED, 2002. Breaking New Ground: Mining Minerals and Sustainable Development, International Institute of Environment and Development with support from the World Business Council on Sustainable Development, London.

Iliffe, J., 2007. Africans: The History of a Continent, 2007, Cambridge University Press; Cambridge. Population Studies: Key Issues and Contemporary Trends in Ghana.

Ingle, M., 2012. Unbalanced growth and dependency theory in Zambia: A post-independence survey. *African Journal of Business Management*, 6(16), 5467-5471.

Inkoom, D.K.B., 2009. Planning education in Ghana, 2009, 1-22.

IPCC, 2011. Renewable Energy Sources and Climate Change Mitigation. Special Report of the Intergovernmental Panel on Climate Change

Cambridge University Press, New York.

Irvine, J.R. and Riddell. B.E., 2007. Salmon as status indicators for North Pacific ecosystems. *N. Pac. Anadr. Fish Comm. Bull.*, 4, 285–287.

Isaksen, L., 2019. Nyliberal transformasjon av landskap i appolonia, Ghana: meningsskapende fremtider og produksjonen av livsverdener. Master's thesis. <u>http://bora.uib.no/bitstream/handle/1956/20383/Nyliberal-Transformasjon-av-Landskap--</u> <u>Ghana.pdf?sequence=1</u> Accessed: 14/5/2020.

Jabareen, Y., 2006. Sustainable urban forms: Their typologies, models, and concepts. *Journal of Planning Education and Research*, 26(1), 38–52.

Jabareen, Y., 2008. A new conceptual framework for sustainable Development. *Environ Dev Sustain*, 10, 179–192. DOI 10.1007/s10668-006-9058-z

Jaeger, J.A.G., Bertiller, R., Schwick, C. and Kienast, F., 2010. Suitability criteria for measures of urban sprawl. *Ecol. Indicators*, 10 (2), 397-406. https://doi.org/10.1016/j.ecolind.2009.07.007

Jedwab, R. 2013. Urbanization without Industrialization: Evidence from Consumption Cities in Africa. Unpublished manuscript, Department of Economics, George Washington University.

Jedwab, R., Christiaensen, L. and Gindelsky, M., 2017. Demography, urbanisation and Rural push, urban pull and urban push? *Journal of Urban Economics*, 98, 6-16. http://dx.doi.org/10.1016/j.jue.2015.09.002

Jeronen, E., 2013. Sustainability and Sustainable Development. Reference Work Entry Encyclopedia of Corporate Social Responsibility. 2370-2378. DOI: 10.1007/978-3-642-28036-8_662

Jessop, B., 2002. Liberalism, neoliberalism, and urban governance: A state-theoretical perspective. *Antipode*, 34(3), 452–472.

Jnr, S.D.; Cieem, G.; Ayensu-Ntim, A.; Twumasi-Ankrah, B.; Barimah, P.T. Effects of Loss of Agricultural Land Due to Large-Scale Gold Mining on Agriculture in Ghana: The Case of the Western Region. Br. J. Res. 2016, 2, 196–221.

Johnson, R. B. and Turner, L. A., 2003. Data collection strategies in mixed methods research. In A. Tashakkori & C. Teddlie (Eds.), *Handbook of mixed methods in social & behavioral research*. Thousand Oaks, CA: Sage.

Jones, E., 2010. The Chicago School, Hayek and the Mont Pélerin Society. *Journal of Australian Political Economy*, 35, 139–155.

Jones, G. A. and Corbridge, S., 2010. The continuing debate about urban bias. *Progress in Development Studies*, 10 (1), 1–18.

Joss, S., 2010. Eco-cities: a global survey 2009. WIT Trans. *Ecol. Environ.*, 129 (2010), 239–250.

Joss S., 2011a. Eco-cities. Modus (Royal Institute of Chartered Surveyors, 6, 32-35.

Joss, S., 2011b. Eco-City Governance: A Case Study of Treasure Island and Sonoma Mountain Village, *Journal of Environmental Policy & Planning*, 13(4), 331-348. DOI: 10.1080/1523908X.2011.611288

Joss, S., Tomozeiu, D. and Cowley, R., 2011. Eco-cities: a Global Survey 2011. University of Westminster, International Eco-cities Initiatives, London (2011).

Joumard, R. and Gudmundsson, H., 2010. Indicators of environmental sustainability in transport (Eds.). Les collections de l'INRETS.

Joung, C.B., Carrell, J., Sarkar, P. and Feng S.C., 2012. Categorization of indicators for sustainable manufacturing. *Ecol. Indic.*, 24 (2012), 148–157.

Kalsoom, Q., 2019. Attitude Change to Sustainable Development. In: Leal Filho W. (eds) *Encyclopedia of Sustainability in Higher Education*. Springer, Cham.

Kamete, A., Y., Tostensen, A. and Tvedten, I., 2001. "From global village to urban globe: urbanization and poverty in Africa: implications for Norwegian aid policy", Chr. Michelsen Institute, Bergen, quoted in Njoh, Ambeh J (2003), "Urbanization and development in sub-Saharan Africa", Cities Vol 20, No 3, p. 169.

Karlsson, S., 2008. Closing the technospheric flows of toxic metals: modelling lead losses from a lead-acid battery system for Sweden. *Journal of Industrial Ecology*, 3 (2008), 23–40.

Kates, R.W., Parris, T.M. and Leiserowitz, A.A., 2005. What is sustainable development? *Environment*, 47 (3), 8–21.

Keirstead, J., 2007. Selecting sustainability indicators for urban energy systems. International Conference on Whole Life Urban Sustainability and its Assessment Glasgow; 2007.

Kemp, D., 2009. Mining and community development: problems and possibilities of locallevel practice. *Community Development Journal*, 45(2), 198–218.

Kemp, R. 1994. Technology and the transition to environmental sustainability. The problem of technological regime shifts. *Futures*, 26, 1023–1046.

Kemp, R., Schot, J. and Hoogma, R., 1998. Regime shifts to sustainability through processes of niche formation: the approach of strategic niche management. *Technol. Anal. Strategic Manage*. 10, 175–196.

Kemp, R. and Loorbach, D., 2003. Governance for Sustainability Through Transition Management. Paper for EAEPE 2003 Conference, Maastricht. Kennedy, C., Cuddihy, J., Engel-Yan, J., 2007. The changing metabolism of cities. *Journal of Industrial Ecology*, 11 (2007), pp. 43-59.

Kerselaers, E., Rogge, E., Vanempten, E., Lauwers, L. and Van Huylenbroeck, G., 2013. Changing land use in the countryside: stakeholders' perception of the ongoing rural planning processes in Flanders. *Land Use Policy*, 32, 197-206. http://dx.doi.org/10.1016/j.landusepol.2012.10.016

Kessides, C., 2007. "The urban transition in sub-Saharan Africa: challenges and opportunities". *Environment and Planning C – Government and Policy*, 24 (4), 466–485.

Kidd, J. and Finlayson, M., 2006. Navigating uncharted water: research ethics and emotional engagement in human inquiry. *J. Psychiatr. Ment. Hlth. Nurs.*, 13(4), 423-8.

Kim, S. and Rowe, P.G., 2012. Does large-sized cities' urbanisation predominantly degrade environmental resources in China? Relationships between urbanisation and resources in the Changjiang Delta Region. *International Journal of Sustainable Development & World Ecology*, 19 (4), 321-329. DOI: 10.1080/13504509.2011.647108

King, M.F., Renó, V.F. and Novo, E.M.L.M., 2014. The Concept, Dimensions and Methods of Assessment of Human Well-Being within a Socioecological Context: A Literature Review. *Soc. Indic. Res.*, 116 (3), 681-698. https://doi.org/10.1007/s11205-013-0320-0

Kirkby, J., O'Keefe, P. and Timberlake, L., 1995. *The Earthscan Reader in Sustainable Development*. Earthscan: London.

Klijn, J.A., 2004. Driving Forces behind Landscape Transformation in Europe, from a Conceptual Approach to Policy Options, Springer, Dordrecht.

Knack, S., 2001. Aid Dependence and the Quality of Governance: Cross-Country Empirical Tests. *Southern Economic Journal*, 68 (2), 310-329.

Knox, P.L. and Taylor, P.J., 1995. *World Cities in a World System* (eds). Cambridge: Cambridge University Press.

Konadu-Agyeman, K., 2001. A survey of housing conditions and characteristics in Accra, an African city. *Habitat International (2001)*, 15–34.

Koziatek, O., Dragicevic, S. and Li, S., 2016. Geospatial modelling approach for 3d Urban densification developments. *ISPRS-international archives of the photogrammetry, remote sensing and spatial information sciences*, XLI-B2 (2016), 349-352.

Kramers, A., Höjer, M., Lövehagen, and N., Wangel, J., 2014. Smart sustainable cities: Exploring ICT solutions for reduced energy use in cities'. *Environmental Modelling & Software*, 56 (2014), 52–62.

Kupiszewska, D., 1997. Modelling for sustainable cities: Conceptual approach and an audit of existing sectoral models for transport, air pollution, land use, and population modelling. Working Paper, Institute of Transport Studies, University of Leeds, UK.

Kuo, F. E., Bacaicoa, M. and Sullivan, W. C., 1998. Transforming inner-city landscapes: Trees, sense of safety, and preference. *Environment and Behavior*, 30, 28–59.

Kverndokk, S., Nvdal, E. and Nstbakken, L., 2014. The trade-off between intra- and intergenerational equity in climate policy. *Eur. Econ. Rev.*, 69 (2014), 40-58.

Kwakye, E. A. and Fouracre, P. R., 1998. Urban transport policy reform in Ghana', paper presented at the CODATU VIII Conference, Cape Town, South Africa Mabogunje A.L., 1968, Urbanisation in Nigeria, University of London Press.

Lachman, D. A., 2013. A survey and review of approaches to study transitions. *Energy Policy*, 58 (2013), 269–276. http://dx.doi.org/10.1016/j.enpol.2013.03.013

Lafferty, W. M., 2004. *Governance for Sustainable Development. The Challenge of Adapting Form to Function*, Elgar, Cheltenham.

Lambin, E.F., Turner, B.I. and Geist, H.J., 2001. The causes of land use and land cover change: moving beyond the myths. *Global Environmental Change*, 11, 261-269.

Landscape Toolbox, 2019a. Landscape Thematic Mapper 4 and 5. https://wiki.landscapetoolbox.org/doku.php/remote_sensor_types:landsat_tm_5 Accessed: 11/09/2019.

Landscape Toolbox, 2019b. Landscape Enhanced Thematic Mapper 7. https://wiki.landscapetoolbox.org/doku.php/remote_sensor_types:landsat_etm_7 Accessed: 11/09/2019.

Landscape Toolbox, 2019c. Landsat Data Continuity Mission (Landsat 8) https://wiki.landscapetoolbox.org/doku.php/remote_sensor_types:landsat_8 Accessed: 11/09/2019.

Laurian, L and Crawford, J., 2016. Organizational factors of environmental sustainability implementation: An empirical analysis of US cities and counties. *Journal of Environmental Policy & Planning*, 18(4), 482–506.

Lauermann, J., 2017. Urban Managerialism / Entrepreneurialism. The Wiley-Blackwell Encyclopedia of Social Theory. https://doi.org/10.1002/9781118430873.est0546 Accessed: 02/02/2018.

Lee. J. and Zhu, Y., 2006. Urban governance, neoliberalism and housing reform in China, *The Pacific Review*, 19 (1), 39-61. DOI: 10.1080/09512740500417657

Le Galès, P., 2016. Neoliberalism and Urban Change: Stretching a Good Idea Too Far? *Territory, Politics, Governance*, 4(2), 154-172. DOI: 10.1080/21622671.2016.1165143

Lehtonen, M., 2012. Indicators as an appraisal technology: framework for analysing the policy influence of the UK Energy Sector. Sustainable Development, Evaluation and Policy (2012).

Leith, J.C., 1974. Growth factors. J.C. Leith (Ed.), Foreign Trade Regimes and Economic Development: Ghana, vol. 2, UMI, New York (1974, 391-401. <u>https://www.nber.org/books-</u>

and-chapters/foreign-trade-regimes-and-economic-development-ghana/growth-factors Accessed: 5/6/2020.

Lemmen, C., 2010. The Social Tenure Domain Model: A Pro-Poor Land Tool; The International Federation of Surveyors: Copenhagen, Denmark, 2010.

Leon, J.K., 2015. The role of global cities in land grabs. *Third World Quarterly*, 36(2), 257-273. DOI: 10.1080/01436597.2015.1015787

Levitas, R. 1990. The Concept of Utopia. Hertfordshire: Philip Allan.

Levitas, R., 2007. Looking for the blue: The necessity of utopia. *Journal of Political Ideologies*, 12:3, 289-306, DOI: 10.1080/13569310701622184

Lewison, R.L., Rudd, M.A., Al-Hayek, W., Baldwin, C., Beger, M., Lieske, S.N., Jones, C., Satumanatpan, S., Junchompoo, C. and Hines, E., 2016. How the DPSIR framework can be used for structuring problems and facilitating empirical research in coastal systems. *Environ. Sci. Policy*, 56 (2016), 110-119.

Li, F., Liu, X., Hu, D., Wang, R. and Yang, W., 2009. Measurement indicators and an evaluation approach for assessing urban sustainable development: A case study for China's Jining city. *Landscape and Urban Planning*, 90 (2009), 134-142.

Li, H., Long, R. and Chen, H., 2013. Economic transition policies in Chinese resource-based cities: an overview of government efforts. *Energy Policy*, 55 (2013), 251–260.

Li, Y., Beeton, R.J.S., Sigler T., and Halog, A., 2016. Modelling the transition toward urban sustainability: a case study of the industrial city of Jinchang, China. *Journal of Cleaner Production*, 134, 22-30. DOI: 10.1016/j.jclepro.2015.10.053

Lipton, M., 1977. *Why poor people stay poor: a study of urban bias in world development.* Temple Smith, London. Lipton, M., 2005. Urban bias. In Forsyth, T., *Encyclopedia of international development*. Routledge.

Locke, A. and Henley, G., 2016. Urbanisation, land and property rights: the need to refocus attention. Overseas Development Institute report.

London, B. and Ross, R. J. S., 1995. The Political Sociology of Foreign Direct Investment: Global Capitalism and Capital Mobility, 1965–1980. *International Journal of Comparative Sociology*, 36, 198–219.

Loorbach, D. and Rotmans, J., 2006. Managing transitions for sustainable development Book Chapter in: *Understanding Industrial Transformation. Views from different disciplines*. X. Olshoorn, Wieczorek, A. J. Dordrecht, Springer, 2006.

Loorbach, D. and Rotmans, J., 2010. The practice of transition management; examples and lessons from four distinct cases. *Futures*, 42: 237–246.

Lu, D. and Weng, Q., 2007. A survey of image classification methods and techniques for improving classification performance. *International journal of Remote Sensing*, 28 (5), 823-870.

Lubell, M., Feiock, R., and Handy, S., 2009. City Adoption of Environmentally Sustainable Policies in California's Central Valley. *Journal of the American Planning Association*, 75 (3), 293-308. DOI: 10.1080/01944360902952295

Lucas, R. E., 2004. Life Earnings and Rural-Urban Migration. *Journal of Political Economy*, 112 (S1), 29–59.

Lucy, W.H. and Phillips, D.L., 2000. Suburban decline: The next urban crisis. *Issues in Science and Technology*, 27 (2000), 55-62.

Lunetta, R.S. and Lyon, J.G., 2004. *Remote sensing and GIS Accuracy Assessment* (Eds). CRC Press, Boca Raton, FL.

Lwasa, S., Mugagga, F., Wahab, B., Simon, D., Connors, J. and Griffith, C., 2014. Urban and peri-urban agriculture and forestry: Transcending poverty alleviation to climate change mitigation and adaptation. *Urban Climate*, 7 (2014), 92–106.

Lynch, Amy, J. and Mosbah, S.M., 2017. Improving local measures of sustainability: A study of built-environment indicators in the United States. *Cities*, 60 (2017), 301–313.

Macke, J., Casagrande, R.M., Sarate, J.A.R. and Silva, K.A., 2018. Smart city and quality of life: citizens' perception in a Brazilian case study. *J. of Clean. Prod.*, 182, 717-726.

Magee, L. and Scerri, A., 2012. From issues to indicators: Developing robust community sustainability measures. *Local Environment*, 17(8), 915–933.

Majumdar, S., Mani, A. and Mukand, S.W., 2004. Politics, information and the urban bias. *Journal of Development Economics*, 75 (1), 137–165.

Mancini, M.S., Galli, A., Niccolucci, V., Lin, D., Hanscom, L., Wackernagel, M., Bastianoni, S. and Marchettini, N., 2017. Stocks and flows of natural capital: Implications for Ecological Footprint. *Ecological Indicators*, 77, 123-128. https://doi.org/10.1016/j.ecolind.2017.01.033

Mantey, J. and Sakyi, E.K., 2019. A Study of Energy Related Greenhouse Gas Emissions of High-Income Urban Residents in the City of Accra, Ghana. *OIDA International Journal of Sustainable Development*, 12 (2), 41-60.

Marcotullio, P.J., 2001. Asian sustainability in the era of globalization. *Habitat International*, 25 (4), 577-598.

Markard, J., 2011. Transformation of infrastructures: sector characteristics and implications for fundamental change. *Journal of Infrastructure Systems (ASCE)*, 17, 107–117.

Markard, J., Raven, R., and Truffer, B., 2012. Sustainability transitions: An emerging field of research and its prospects. *Research Policy*, 41(6), 955–967.

Mather, P.M., 2004. *Computer processing of remotely-sensed images: an introduction*. Third Edition. Wiley and Sons, Chichester.

Mathers N, Fox, N. and Hunn, A., 2007. Surveys and Questionnaires. The NIHR RDS for the East Midlands / Yorkshire & the Humber, 2007. https://www.rds-yh.nihr.ac.uk/wp-content/uploads/2013/05/12_Surveys_and_Questionnaires_Revision_2009.pdf Accessed: 09/03/ 2018.

Matschoss, K., Repo, P. and Timonen, P., 2019. Embedding European citizen visions in sustainability transition: Comparative analysis across 30 European countries. *Futures*, 111, 102437. https://doi.org/10.1016/j.futures.2019.102437

Matthews, B. and Ross, L., 2010. Research Methods: A Practical Guide for the Social Sciences, Pearson.

Matthews, Z., Channon, A., Neal, S., Osrin, D., Madise, N. and Stones, W., 2010. Examining the "Urban advantage" in maternal health care in developing countries. *PLoS Med*, *7*, e1000327.

Mattingly, M., 1999. The role of government of urban areas in the creation of urban poverty. In *Urban Poverty in Africa: From Understanding to Alleviation*, (eds.) S Jones, N Nelson. Intermediate Technology Publication, Southampton Row, London.

Matunhu, J., 2011. A critique of Modernization and Development Theories in Africa: A Critical Assessment. *African Journal of History and Culture*. 3(5), 65–72.

McCall, G. and Simmons, J., 1969. *Issues in participant observation: a text and reader*. Reading, MA: Addison-Wesley.

McCann, P. and Shefer, S., 2004. Location, agglomeration and infrastructure. *Papers in Regional Science*, 83(1), 177–196.

McCormick, K., Anderberg, S., Coenen, L., Neij, L., 2013. Advancing sustainable urban transformation. *J. Clean. Prod.*, 50, 1–11.

McGillivray, M. and Clarke, M., 2006. *Understanding human well-being*. United Nations University Press ISBN 9280811304 (pbk.).

McGranahan, G., Mitlin, D., Satterthwaite, D., Tacoli, C. and Turok, I., 2009. Africa's Urban Transition and the Role of Regional Collaboration, Human Settlements. Working Paper Series Theme: Urban Change-5. IIED, London, 58.

McGregor, J. A., 2008. Wellbeing, Poverty and Conflict. WeD Policy Briefing 01/08. http://www.bath.ac.uk/soc-pol/welldev/research/bp/bp1-08.pdf Accessed: 19/06/2017.

McGuirk, P., 2012. "Geographies of Urban Politics: Pathways, Intersections, Intervention". *Geographical Research*, 50, 256–268. doi:10.1111/j.1745-5871.2011.00726.x.

Mckinsey and Company, 2010. What's driving Africa's growth? https://www.mckinsey.com/featured-insights/middle-east-and-africa/whats-driving-africasgrowth Accessed: 12/03/2016.

Mckinsey Global Institute, 2012. Urban World: Cities and the Rise of the Consuming Class. http://www.mckinsey.com/global-themes/urbanization/urban-world-cities-and-the-rise-of-the-consuming-class Accessed: 03/03/2016.

Mee-Udon, F., 2009. The contribution of universal health insurance coverage scheme to villagers' wellbeing in Northeast Thailand. Unpublished doctoral dissertation (2009). University of Bath, Bath SES, UK.

Mejía, M.R., Franco-Garcia, M-L. and Jauregui-Becker, J.M., 2018. Sustainability transitions in the developing world: Challenges of sociotechnical transformations unfolding in contexts of poverty. *Environmental Science and Policy*, 84 (2018), 217-223. http://dx.doi.org/10.1016/j.envsci.2017.03.010

Merriam, S., 1988. *Case Study Research In Education: A Qualitative Approach*; Jossey-Bass: San Francisco, CA, USA.
Mieg, H. A., 2012. Sustainability and innovation in urban development: concept and case. *Sustainable Development*, 20 (4), 251-263.

Miller, H., 2014. *What are the features of urbanisation and cities that promote productivity, employment and salaries?* In E. PEAKs (Ed.), Helpdesk Request. London, UK.

Mishra, R., 1999. Globalization and the Welfare State, Cheltenham, Edward Elgar.

Missemer, A., 2018. Natural Capital as an Economic Concept, History and Contemporary Issues. *Ecological Economics*, 143, 90-96. https://doi.org/10.1016/j.ecolecon.2017.07.011

Misilu, M.N.E., Shouyu, C. and Qin, Z.L., 2010. Sustainable urbanisation's challenge in Democratic Republic of Congo. *Journal of Sustainable Development*, 3 (2), 242-254.

Mitchell, G., 1996. Problems and fundamentals of sustainable development indicators. *Sustain. Dev.*, 4 (1), 1–11. http://dx.doi.org/10.1002/(SICI)1099-1719(199603)4:1<1::AID-SD24>3.0.CO;2-N

MLGRD, 2013. National Urban Policy-Action Plan. http://www.ghanaiandiaspora.com/wp/wp-content/uploads/2014/05/ghana-national-urban-policy-action-plan-2012.pdf Accessed:3/5/2016.

MLGRD, 2015. Habitat III National Report for Ghana. Third united nations conference on housing and sustainable urban development. http://habitat3.org/wp-content/uploads/Approved-Final-Ghana-Habitat3-Report.pdf Accessed: 05/06/2016.

Moganadas, S.R., Corral-Verdugo, V. and Ramanathan, S., 2013. Toward systemic campus sustainability: gauging dimensions of sustainable development via a motivational and perception-based approach. *Environ Dev Sustain*, 15(6), 1443–1464.

Mondal, A., 2011. Gandhi, utopianism and the construction of colonial difference. *Interventions*, 3(3), 419-438. DOI: 10.1080/713769071

Moss, M. and Grunkemeyer, T., 2007. Using resident formulated multi-dimensional indicators to assess urban communities, progress toward meeting sustainability goals. Proceedings of the International Conference on Whole Life Urban Sustainability and its Assessment, Glasgow Caledonian University, Scotland, UK (2007).

MRH, MoT and GSS, 2011. Statistical and Analytical Report (2000–2009); Transport Indicator Database. Ministry of Roads and Highways, Ministry of Transport and Ghana Statistical Service, Accra.

Mulder, P., Reschke, C. H., and Kemp, R., 1999. Evolutionary theorising on technological change and sustainable development. Presented at the conference "European Meeting on Applied Evolutionary Economics", organised by Institut for Energy Politics and Economics and INRA-Unit of Sociology and Economics of Research and Development 7.6 1999. Grenoble.

Munier, N., 2007. Handbook on urban sustainability. Springer, Dordrecht.

Myers, N. and Kent, J., 2003. New consumers: the influence of affluence on the environment. *Proceedings of the National Academy of Sciences of the United States of America*, 100(2003), 4963-4968.

Nader, M. R., Salloum, B.A. and Karam, N., 2008. Environment and sustainable development indicators in Lebanon: A practical municipal level approach. *Ecological Indicators*, 8 (5), 771-777. https://doi.org/10.1016/j.ecolind.2007.09.001

Namkoong, Y., 1999. Dependency Theory: Concepts, Classifications, and Criticisms. *The International Area Studies Review*, 2.1 (1999), 2-31.

Narducci, J., Quintas-Soriano, C., Castro, A., Som-Castellano, R. and Brandt, J.S., 2019. Implications of urban growth and farmland loss for ecosystem services in the Western United States. *Land Use Policy*, 86 (2019), 1-11. Nassauer, J.I. and Raskin, J., 2014. Urban vacancy and land use legacies: A frontier for urban ecological research, design, and planning. *Landscape and Urban Planning*, 125 (2014), 245-253.

National Development Planning Commission, 2019. Annual Progress Reports for Districts, 2009-2017. http://ndpc-cms.herokuapp.com/downloads/34/ Accessed: 4/02/2019.

National Research Council, 2011. Sustainability and the USEPA. Washington, DC: National Academic Press.

NDC, 2008. NDC Manifesto. National Democratic Congress, Accra (2008).

Neirotti, P., De Marco, A., Cagliano, A.C., Mangano, G. and Scorrano, F., 2014. Current trends in Smart City initiatives: Some stylised facts. *Cities*, 38, 25-36.

Nelson, R. and Winter, S., 1982. *An Evolutionary Theory of Economic Change*. Cambridge, Massachusetts: The Belknap Press of Harvard University Press.

Neumayer, E., 2013. Weak versus strong sustainability: exploring the limits of two opposing paradigms 4th, Edward Elgar Publishing, Cheltenham, UK. ISBN 9781781007075

Nevens, F, Frantzeskaki, N., Gorissen, L. and Loorbach, D., 2013. Urban Transition Labs: co-creating transformative action for sustainable cities. *Journal of Cleaner Production*, 50, 111–122.

Njoh, A.J., 2003. Urbanization and development in sub-Saharan Africa. *Cities*, 20 (3), 167-174.

Njoh, A., 2005. African cities and regional trade in historical perspective: Implications for contemporary globalization trends. *Cities*. 23(1), 18–29.

Nilsson, M., Wiklund, H., Finnveden, G., Jonsson, D., Lundberg, K., Tyskeng, S., and Wallgren, O., 2009. Analytical framework and tool kit for SEA follow-up. *Environmental Impact Assessment Review*, 29(3), 186-199.

Nitsch, V., 2006. Trade Openness and Urban Concentration: New Evidence. *Journal of Economic Integration*, 21, 340-362.

Niyonkuru, F., 2016. Failure of Foreign Aid in Developing Countries: A Quest for Alternatives. *Bus. Eco. J.*, 7 (231), 1-9. doi:10.4172/2151-6219.1000231

NPP, 2008. NPP Manifesto. New Patriotic Party, Accra (2008).

NUP, 2012. Ghana National Urban Policy Framework. <u>https://www.nalag-ghana.org/wp-</u> <u>content/uploads/2019/03/national_urban_policy_framework_ghana_2012.pdf</u> Accessed: 1/5/2016.

Nyame, F.K. and Blocher, J., 2010. Influence of land tenure practices on artisanal mining activity in Ghana. *Resources Policy*, 35, 47–53.

Obeng-Odoom, F., 2009a. The future of our cities. Cities, 26 (2009), 49-53.

Obeng-Odoom, F., 2009b. An urban twist to politics in Ghana. *Habitat International*, 34(4), 392-399.

Obeng-Odoom, F., 2012. Neoliberalism and the Urban Economy in Ghana: Urban Employment, Inequality, and Poverty. *Growth and Change*, 43, 85-109. doi:10.1111/j.1468-2257.2011.00578.x

Obeng-Odoom, F., 2013a. Regeneration for some; degeneration for others in Leary M and McCarthy J eds, *The Routledge companion to urban regeneration*, Routledge, London.

Obeng-Odoom, F., 2013b. Governance for pro-poor urban development: lessons from Ghana. Routledge, London.

Oduro, C. Y., Ocloo, K. and Peprah, C., 2014. Analysing growth patterns of Greater Kumasi Metropolitan Area using GIS and Multiple Regression Techniques. *Journal of Sustainable Development*, 7 (5), 13-31. doi:10.5539/jsd.v7n5p13

OECD, 1994. OECD core set of indicators for environmental performance reviews. OECD Environment Monographs No. 83. OECD, Paris. http://www.oecd.org/env/indicators-modelling-outlooks/31558547.pdf Accessed: 03/04/2018.

OECD, 2005. Glossary of Statistical Terms. OECD Statistics Portal. [Created July 06, 2005; Last updated July 19, 2005] http://stats.oecd.org/glossary/index.htm Accessed: 06/01/2017.

OECD, 2007. Competitive cities: a new entrepreneurial paradigm in spatial development. ISBN-92-64- 022597. https://www.oecd.org/cfe/regional-policy/38747575.pdf Accessed:3/01/2019.

OECD, 2013. Tackling Long-term unemployment amongst vulnerable groups. http://www.oecd.org/cfe/leed/tackling-vulnerable-report.htm Accessed: 10/01/2017.

OECD and CDRF, 2010. Trends in Urbanisation and Urban Policies in OECD Countries: What Lessons for China? Available from: http://www.oecd.org/urban/roundtable/45159707.pdf. Accessed: 03/02/2016.

Ofori, E. G., 2009. Taxation of the informal sector in Ghana: a critical examination. A Dissertation presented to the Institute of Distance Learning, Kwame Nkrumah University of Science and Technology, Kumasi in partial fulfillment of the requirements for the degree of Commonwealth Executive Master of Business Administration (CEMBA). Institute Of Distance Learning. KNUST, Kumasi. May, 2009.

Okeke, I.N., 2014. Towards a New Growth Path in Africa: A Study of National Urban Policy Responses to Urbanisation. Working Paper 2014, South African Cities Network.

Okyere, C.Y., Yacouba, Y. and Gilgenbach, D. ,2012. The problem of annual occurrences of Floods in Accra: An integration of hydrological, economic and political perspectives.

nterdisciplinary term paper. ZEF Doctoral Studies Programme. https://www.zef.de/fileadmin/downloads/.../2012_2_Gilgenbach_Okyere_Yacouba.pdf Accessed: 20/04/2016.

Olsson, P., Galaz, V. and Boonstra, W., J., 2014. Sustainability transformations: A resilience perspective. *Ecology and Society*, 19(4), 1. doi:10.5751/ES-06799-190401.

Olsson, D. and Gericke, N., 2016. The adolescent dip in students' sustainability consciousness – implications for education for sustainable development. *J Environ Educ*, 47(1), 35–51.

Opoku-Mensah, S. and Asare-Okyere, S., 2014. Mining, Environment and Community Conflicts: A Study of Company-Community Conflicts over Gold Mining in the Obuasi Municipality of Ghana. Journal of Sustainable Development Studies, 5 (1), 64-99.

O'riordan, 2016. Editorial - The Transition to Sustainability Is By No Means Smooth. Envronment: Science and Policy for sustainable development. http://www.environmentmagazine.org/Archives/Back%20Issues/2016/May-June%202016/editorial.html Accessed: 04/04/2017.

Osei-Boateng, C. and Ampratwum, E. 2011. The Informal Sector in Ghana. https://library.fes.de/pdf-files/bueros/ghana/10496.pdf Accessed: 29/7/2018.

Ostrom, E., 2009. A general framework for analysing sustainability of social-ecological systems. *Science*, 352, 419–422.

Ouyang, X. and Lin, B., 2017. Carbon dioxide (CO₂) emissions during urbanization: A comparative study between China and Japan. *Journal of Cleaner Production*, 143, 356–368. http://dx.doi.org/10.1016/j.jclepro.2016.12.102.

Owoo, N.S. and Boakye-Yiadom, L., 2015. The gender dimension of land tenure security on agricultural productivity: Some evidence from two districts in Kenya. *J. Int. Dev.*, 27, 917–928.

Owusu, G., 2004. Small towns and decentralised development in Ghana. *Theory Pract.*, 39 (2), 165-195.

Pacione, M., 2007. Sustainable Urban Development in the UK: Rhetoric or reality? *Geography*, 92 (3), 248-265.

Page, J., 2012. Can Africa Industrialise? Journal of African Economies, 21(suppl 2), 86-124.

Pappas, J.B. and Pappas, E.C., 2014. The sustainable personality: Values and behaviors in individual sustainability. *Int. J. Higher Educ.*, 4, 12–21.

Parnell, S., Simon, D. and Vogel, C., 2007. Global environmental change: conceptualizing the growing challenge for cities in poor countries. *Area*, 39 (3), 357–369.

Parnell, S., Pieterse, E. and Watson, V., 2009. "Planning for Cities in the Global South." *Progress in Planning*, 72, 233–248.

Parnell, S. and Simon, D., 2010. National urbanisation and urban policies: necessary but absent policy instruments in Africa. *In*: Pieterse, E. (Ed.), *Urbanization Imperatives for Africa: transcending Policy Inertia*. Cape Town: African Centre for Cities.

Partelow, S., 2018. A review of the social-ecological systems framework: applications, methods, modifications, and challenges. *Ecology and Society*, 23(4), 36. https://doi.org/10.5751/ES-10594-230436

Pastille Consortium, 2002. Indicators into Action: A Practitioners Guide. Department of Geography, The London School of Economics and Political Science: London.

Pauleit, S., Lindley S., Cilliers, S. and Shackleton, C., 2018. Urbanisation and ecosystem services in sub-Saharan Africa: Current status and scenarios. *Landscape and Urban Planning*, 180, 247-248. https://doi.org/10.1016/j.landurbplan.2018.09.008

Pearce, D., Barbier, E. and Markandya, A., 1990. *Sustainable Development: Economics and environment in the Third World*, London: Earthscan.

Pearce, D.G., 2008. A needs-functions model of tourism distribution. *Annals of Tourism Research*, 35 (1), 148–168.

Pelletier, N., 2010. Environmental sustainability as the first principle of distributive justice: Towards an ecological communitarian normative foundation for ecological economics. *Ecological Economics*, 69, 1887-1894.

Peter, C. and Swilling, M., 2012. Sustainable, Resource Efficient Cities: Making it Happen! (United Nations Environment Programme, Paris, 2012).

Peters, G. B., 2018. The challenge of policy coordination. *Policy Design and Practice*, 1(1), 1-11. DOI: 10.1080/25741292.2018.1437946

Petrosillo I., Aretano, R. and Zurlini, G., 2015. Socioecological Systems, *Reference Module in Earth Systems and Environmental Sciences*, Elsevier. doi: 10.1016/B978-0-12-409548-9.09518-X.

Pieterse, E., 2008. City Futures: Confronting the Crisis of Urban Development. London: Zed Books, 206.

Pigato, M. A., 2000. Foreign direct investment in Africa: old tales and new evidence. Africa region working paper series; no. 8. Washington, D.C.: The World Bank. http://documents.worldbank.org/curated/en/381251468768676709/Foreign-direct-investment-in-Africa-old-tales-and-new-evidence Accessed: 13/02/2016.

Pigozzi, M., 2007. 'Quality in Education Defines ESD'. *Journal of Education for Sustainable Development*, 1 (1), 27–35.

Pillay, U., 2008. Urban Policy in Post-Apartheid South Africa: Context, Evolution and Future Directions. *Urban Forum* (2008), 19, 109-132.

Pillarisetti, J.R., 2005. The World Bank's 'genuine savings' measure and sustainability. *Ecol Econ*, 55(4), 599–609.

Pintér, L., Hardi, P., Bartelmus, P., 2005. Sustainable Development Indicators Proposal for a Way Forward – Prepared for the United Nations Division for Sustainable Development UN-DSD. 41 pp., http://www. iisd. org/pdf/2005/measure_indicators_sd_way_forward. pdf. Accessed: 24/05/2017.

Pires, S.M.M., 2011. Sustainability Indicators and Local Governance in Portugal. Universidade de Aveiro (2011).

Pissourios, I. A., 2013. An interdisciplinary study on indicators: A comparative review of quality-of-life, macroeconomic, environmental, welfare and sustainability indicators. *Ecological Indicators*, 34, 420–427.

Polidoro, M., Lollo, D.A.L. and Barros, F.V.M., 2012. Urban sprawl and the challenges for urban planning. *Journal of Environmental Protection*, 3 (2012), 1010-1019. http://dx.doi.org/10.4236/jep.2012.39117

Poortman, C.L. and Schildkamp, K., 2012. Alternative quality standards in qualitative research? *Quality and Quantity*, 46 (6), 1727 – 1751. https://doi.org/10.1007/s11135-011-9555-5

Portney, K. E., 2003. *Taking sustainable cities seriously: Economic development, the environment, and quality of life in American cities.* Cambridge: MIT Press.

Portney, K. E., and Berry, J. F., 2010. Participation and the Pursuit of Sustainability in U.S. Cities. *Urban Affairs Review*, 46(1), 119–39.

Portney, K. E. and Cutler, Z., 2010. The local non-profit sector and the pursuit of sustainability in American cities: A preliminary exploration. *Local Environment*, 15, 323-339.

Poston, D. L. and Bouvier, L.F., 2010. *Population and society: an introduction to demography*, Cambridge: Cambridge University Press.

Potts, D., 2016. Debates about African urbanisation, migration and economic growth. *Geogr. J.*, 182, 251-264. doi:10.1111/geoj.12139

Pow, C. P. and Neo, H., 2013. Seeing red over green: Contesting urban sustainabilities in China. *Urban Studies*, 50 (11), 2256–2274.

Premalatha, M., Tauseef, S. M. and Abbasi, T., 2013. The promise and the performance of the world's first two zero carbon eco-cities. *Renewable and Sustainable Energy Reviews*, 25, 660–669.

Prescott-Allen, R., 2001. *The Wellbeing of Nations: A Country-by-Country Index of Quality of Life and the Environment*. Washington, DC: Island Press.

Pupphachai, U. and Zuidema, C., 2017. Sustainability indicators: A tool to generate learning and adaptation in sustainable urban development. *Ecological Indicators*, 72 (2017), 784–793. https://doi.org/10.1016/j.ecolind.2016.09.016

Ramos, T. B., 2009. Development of regional sustainability indicators and the role of academia in this process: the Portuguese practice. *Journal of Cleaner Production*, 17(12), 1101-1115.

Ravallion, M., Chen, S. and Sangraula, P., 2007. "New evidence on the urbanization of global poverty", World Bank Policy Research Working Paper, Washington DC, page 27.

Raven, R., 2012. Analysing emerging sustainable energy niches in Europe: a strategic niche management perspective. In: Verbong, G., Loorbach, D. (Eds.), *Governing the Energy Transition: Reality, Illusion or Necessity?* Routledge: Abingdon.

Raven, R. and Verbong, G., 2007. Multi-regime interactions in the Dutch energy sector: the case of combined heat and power technologies in the Netherlands 1970–2000. *Technology Analysis & Strategic Management*, 19 (4), 491–507.

Raven, R., Van den Bosch, S. and Weterings, R., 2010. Transitions and strategic niche management: towards a competence kit for practitioners. *Int. J. Technol. Manag.*, 51 (2010), 57–74.

Redclift, M., 2005. An oxymoron comes of age. *Sustain. Dev.*, 13 (2005), 212-227. Reed, M.S, Evely, A.C., Cundill, G., Fazey, I., Glass, J., Laing, A. and Stringer, L.C., 2010. What is social learning? *Ecology and Society*, 15 (4), 1-10.

Register, R., 1987. Ecocity. Berkeley, CA: North Atlantic Books.

Richards, H.M. and Schwartz, L.J., 2002. Ethics of qualitative research: are there special issues for health services research? *Family Practice*, 19(2), 135-139.

Ridd, M., 1995. Exploring a V–I–S (vegetation–impervious–surface-soil) model for urban ecosystem analysis through remote sensing: Comparative anatomy for cities. *International Journal of Remote Sensing*, 16 (1995), 2165-2185.

Rodrik, D., 2000. How Far Will International Economic Integration Go? *The Journal of Economic Perspectives*. 14 (1), 177-186.

Roseland, M., 1997. Dimensions of the eco-city. Cities, 14 (1997), 197-202.

Roseland, M., 2001. *How Green is the City? Sustainability assessment and the management of urban environments.* Columbia University Press, New York.

Rosenström, U., 2006. Exploring the policy use of sustainable development indicators: interviews with Finnish politicians. *Journal of Transdisciplinary Environmental Studies*, 5 (1), 1-13.

Ross, R. J. S. and Trachte, K., 1990. *Global Capitalism: The New Leviathan State*. University of New York Press, New York.

Rotmans, J. and van Vries, H. J. M., 1997. *Perspectives on global change: The TARGETS approach*. Cambridge: Cambridge University Press.

Rotmans, J., Kemp, R., van Asselt, M., Geels, F., Verbong, G., and Molendijk, K. 2000. Transitions and Transition Management, The Case of an Emission-free Energy Supply, ICIS/MERIT, Maastricht, 2000.

Rotmans, J. and Kemp, R., 2003. Managing societal transitions: dilemmas and uncertainties: The Dutch energy case-study. <u>http://www.oecd.org/environment/cc/2483769.pdf</u> Accessed: 08/12/2020.

Rotmans, J., 2005. Societal innovation: Between dream and reality stands complexity. Rotterdam: Inaugural Lecture, Erasmus University.

Rotmans, J. and Loorbach, D., 2009. Complexity and transition management. J. Ind. Ecol., 13 (2009), 184–196.

Rowan, J., 2000. Research ethics. Int. J. Psychother., 5(2), 103-111.

Sakalauskas, L., 2010. Sustainability models and indicators. *Technological and Economic Development of Economy*, 16(4), 567–577.

Samal, D. and Gedam, S., 2015. Monitoring land use changes associated with urbanization: An object-based image analysis approach. *European Journal of Remote Sensing* 48, 85–99. doi:10.5721/EuJRS20154806.

Sandelowski, M., 1994. The use of quotes in qualitative research. *Res. Nurs. Hlth.*, 17(6), 479-82.

Sarakemsky, I., 1993. Utopia as a political theory. *South African Journal of Political Studies*, 2(3), 111-125.

Satterthwaite, D., McGranahan, G. and Tacoli, C., 2010. Urbanization and its implications for food and farming. *Phil. Trans. R. Soc. B*, (2010) 365, 2809–2820. doi:10.1098/rstb.2010.0136

Sassen, S., 2000. Cities in a world economy. 2nd ed. Thousand Oaks, CA: Pine Forge Press.

Sassen, S., 2011. Cities in a world economy. 4th ed. Sage Publications.

Saunders, M., Lewis, P. and Thornhill, A., 2009. *Research Methods for Business Students* 5th ed., Essex, England: Pearson Education Limited.

Saunders, B., Kitzinger, J. and Kitzinger, C., 2015. Anonymising interview data: challenges and compromise in practice. *Qualitative research: QR*, 15(5), 616–632. Doi: 10.1177/1468794114550439

Schelbe, L., Randolph, K.A, Yelick, A., Cheatham, L.P. and Groton, D.B., 2018. Systems theory as a framework for examining a college campus-based support program for the former foster youth. *Journal of Evidence-Informed Social Work*, 15(3), 277-295, DOI: 10.1080/23761407.2018.1436110

Schnore, L., 1961. The statistical measurement of urbanization and economic development. *Land Econ.*, 37 (3) (1961), 229–245.

Schot, J., W. and Geels, F. W., 2008. Strategic niche management and sustainable innovation journeys: theory, findings, research agenda and policy. *Technology Analysis and Strategic Management*, 20, 537–554.

Schwerhoff, G. and Sy, M., 2017. Financing Renewable Energy in Africa - Key Challenge of the Sustainable Development Goals. *Renewable and Sustainable Energy Reviews*, 75, 393-401. http://dx.doi.org/10.1016/j.rser.2016.11.004

Scipioni, A., Mazzi, A., Mason, M., and Manzardo, A., 2009. The Dashboard of Sustainability to measure local urban sustainable development: The case study of Padua Municipality. *Ecological Indicators*, 9 (2), 364-380.

Scott, A. J. and Storper, M., 2003. Regions, globalization, development. *Regional Studies*, 37(6–7), 579–593.

Scott, K., 2012. Measuring Wellbeing: Towards Sustainability? Routledge, London.

Seabrooke, W., Yeung, C.W.S. and Ma, M.F.F., 2004. Implementing sustainable urban development at the operational level (with reference to Hong Kong and Guangzhou). *Habitat International*, 28 (2004), 443-466.

Selman, P., 1999. Three decades of environmental planning: what have we really learned? M. Kenny, J. Meadowcroft (Eds.), *Planning sustainability*, Routledge, London.

Sen, A., 1999. Development as Freedom. Oxford, Oxford University Press).

Serra, P., Pons, X. and Saurì, D., 2008. Land-cover and land-use change in a Mediterranean landscape: a spatial analysis of driving forces integrating biophysical and human factors. *Applied Geography*, 28 (2008), 189–209.

Seto, K., Sánchez-Rodríguez, R. and Fragkias, M., 2010. The new geography of contemporary urbanization and the environment. *Ann. Rev. Environ. Resour*, 35 (2010), 167–194.

Seto, K.C., Parnell, S. and Elmqvist, T., 2013. A Global Outlook on Urbanization. In: Elmqvist T. et al. (eds) *Urbanization, Biodiversity and Ecosystem Services: Challenges and Opportunities*. Springer, Dordrecht

Seyfang, G. and Longhurst, N., 2014. Understanding the Diffusion of Grassroots Innovations for Sustainability: An International Study of Community Currency Niches. Science, Society and Sustainability Research Group. Norwich 35 Working Paper 2014–25.

Shaw, I.F., 2003. Ethics in qualitative research and evaluation. J. Soc. Work, 3(1), 9-29.

Shen, L., Ochoa, J., Shah, M.N. and Zhang, X., 2011.The application of urban sustainability indicators — A comparison between various practices. *Habitat International*, 35 (1) (2011), 17–29. <u>http://dx.doi.org/10.1016/j.habitatint.2010.03.006</u>

Short, J.R. and Kim, Y., 1999. Globalization and the City. New York: Addison Wesley Longman.

Short, J.R., Breitbach, C., Buckman, S. and Essex, J., 2000. From world cities to gateway cities: Extending the boundaries of globalization theory, *City*, 4(3), 317-340. DOI: 10.1080/713657031

Siiba, A., Adams, E.A. and Cobbinah, P.B., 2018. Chieftaincy and sustainable urban land use planning in Yendi, Ghana: Towards congruence. Cities, 73, 96–105.

Simon, D., 2010. The Challenges of Global Environmental Change for Urban Africa. UNU-WIDER, Working Paper No. 2010/51.

Singh, R.K., Murty, H.R., Gupta, S.K. and Dikshit, A.K., 2012. An overview of sustainability assessment methodologies. *Ecol. Indic.*, 15 (2012), 281–299.

Smart Cities, 2013. Trace analysis and mining for smart cities: issues, methods, and applications. IEEE Commun. Mag., 121 (2013).

Smeets, E. and Weterings, R., 1999. Environmental indicators: typology and overview. European Environment Agency, Copenhagen. http://www.eea.europa.eu/publications/TEC25 Accessed: 03/04/2018.

Smit, W. and Parnell, S., 2012. Urban sustainability and human health: an African perspective. *Current Opinion in Environmental Sustainability*, 4 (4), 443-450.

Smith, A. and Raven, R., 2012. What is protective space? Reconsidering niches in transitions to sustainability. *Res. Policy*, 41 (6), 1025–1036.

Sokoloff, K. L. and Engerman, S. L., 2000. Institutions, factor endowments, and paths of development in the New World. *Journal of Economic Perspectives*, 14 (3), 217 - 232.

Solarin, S.A. and Bello, M.O., 2018. Persistence of policy shocks to an environmental degradation index: the case of ecological footprint in 128 developed and developing countries. *Ecol. Indicat.*, 89, 35-44.

Songsore, J., 2003. Towards a Better Understanding of Urban Change: Urbanization, National Development and Inequality in Ghana, Accra: Ghana Universities Press.

Songsore, J., 2009. The urban transition in Ghana: urbanisation, national development and poverty reduction (2009). pubs.iied.org/pdfs/G02540.pdf. Accessed: 17/08/2016.

Spangenberg, J.H., 2002. Environmental space and the prism of sustainability: frameworks for indicators measuring sustainable development. *Ecol. Indic.*, 2 (3) (2002), 259–309. http://dx.doi.org/10.1016/S1470-160X(02)00065-1

Spiekermann, K. and Wegener, M., 2003. Modelling urban sustainability. *International Journal of Urban Sciences*, 7(1), 47-64.

Spindler, E. A., 2011. Geschichte der Nachhaltigkeit: Vom Werden und Wirken eines beliebten Begriffes (2011) [Online]. Available at https://www.nachhaltigkeit.info/media/1326279587phpeJPyvC.pdf Accessed: 04/02/2016.

Sporrek, A., 1985. Food Marketing and Urban Growth in Dar es Salaam. Lund Studies in Geography, Series B, *Human Geography*, vol. 51. Malmö: Gleerup.

Springett, D., 2013. Editorial: Critical Perspectives on Sustainable Development. *Sust. Dev.* 21, 73–82. DOI: 10.1002/sd.1556

Sta, H.B., 2017. Quality and the efficiency of data in "Smart-Cities". *Future Generation Computer Systems*, 74, 409–416. https://doi.org/10.1016/j.future.2016.12.021

Stahl, C. and Bridges, T.S., 2013. "Fully baked" sustainability using decision analytic principles and ecosystem services. *Integrated Environmental Assessment and Management*, 9, 551–553.

Stake, R. E., 1995. The art of case study research. London: Sage Publications.

Standing, A., 2014. Ghana's extractive industries and community benefit sharing: the case for cash transfers. *Resources Policy*, 40 (2014), 74-82.

Statistics Solutions, 2019. Kruskal-Wallis test. <u>https://www.statisticssolutions.com/kruskal-wallis-test/</u> Accessed: 14/02/2019.

Steiner, F., 2014. Frontiers in urban ecological design and planning research. *Landscape and Urban Planning*, 125 (2014), 304-311.

Stiglitz, J.E., Sen, A. and Fitoussi, J.P., 2010. *Mismeasuring Our Lives: Why GDP Doesn't Add Up.* The New Press, New York.

Stirling, A., 2007. Deliberate futures: precaution and progress in social choice of sustainable technology. *Sustain. Dev.*, 15 (2007), 286–295.

Stock, T., Obenaus, M., Slaymaker, A. and Seliger, G., 2017. A model for the development of sustainable innovations for the early phase of the innovation process. 14th Global Conference on Sustainable Manufacturing, GCSM 3-5 October 2016, Stellenbosch, South Africa. *Procedia Manufacturing*, 8 (2017), 215–222. doi: 10.1016/j.promfg.2017.02.027

Strohbach, M.W., Haase, D. and Kabisch, N., 2009. Birds and the city: urban biodiversity, land use, and socioeconomics. *Ecology and society*, 14 (2), 31.

Struss, P., 2004. "Artificial intelligence methods for environmental decision support". In e-Environment: progress and challenge, Edited by: Prastacos, P., Cortés, U., Díaz de León, J.L. and Murillo, M. 1–14. Mexico City, Mexico: Instituto Politéchnico Nacional.

Sujata, J., Saksham, S., and Shreya G.T., 2016. Developing Smart Cities: An Integrated Framework. *Procedia Computer Science*, 93 (2016), 902 – 909. doi:10.1016/j.procs.2016.07.258

Summers, J.K. and Smith, L.M., 2014. The Role of Social and Intergenerational Equity in Making Changes in Human Well-Being Sustainable. *AMBIO*, 43 (3), 718-728. https://doi.org/10.1007/s13280-013-0483-6

Sustainable Measures Group, 2009. Sustainable Measures, 2009. http://sustainablemeasures.com. Accessed: 19/02/2017. Sustainable Seattle, 1998. Indicators of Sustainable Community (ed.). Sustainable Seattle: Seattle.

Suzuki, H., Dastur, A., Moffat, S. and Yabuki, N., 2010. Eco2 Cities: Ecological Cities as Economic Cities, Washington DC: World Bank.

Svarstad, H., Petersen, L.K., Rothman, D., Siepel, H. and Wätzold, F., 2008. Discursive biases of the environmental research framework DPSIR. *Land Use Policy*, 25 (1), 116-125.

Swann, W.L., 2017. Examining the impact of local collaborative tools on urban sustainability efforts: Does the managerial environment matter? *The American Review of Public Administration*, 47(4), 455–468.

Swyngedouw, E. and Heynen, N. C., 2003. Urban Political Ecology, Justice and the Politics of Scale. *Antipode*, 898-918.

Syms, P., 2002. Land, development and design. Blackwell, Oxford.

Tanguay, G.A., Rojaoson, J., Lefebvre, J.F. and Lanoie, P., 2010. Measuring the sustainability of cities: an analysis of the use of local indicators. *Ecological Indicators*, 10 (2010), 407-418.

Teddlie, C. and Yu, F., 2007. Mixed methods sampling: A typology with examples. *Journal of Mixed Methods Research*, 1(1), 77-100. DOI 10.1177/1558689806292430

The New York Times, 1964. Nkrumah presents 7-year growth plan. <u>https://www.nytimes.com/1964/03/12/archives/nkrumah-presents-7year-growth-plan.html</u> Accessed: 6/6/2020.

Thomas, W.I., and Thomas, D.S., 1928. The Child in America: Behavior Problems and introduction by Morris Janowitz. University of Chicago Press Programs.

Todaro, M., and Smith., S. 2006. *Economic development*, 9th ed. London: Pearson Addison Wesley.

Todorov, V. and Marinova, D., 2011. 'Modelling sustainability', *Mathematics and Computers in Simulation*, 81 (7), 1397-1408.

Toth-Szabo, Z. and Várhelyi, A., 2012. Indicator framework for measuring sustainability of transport in the city. *Proc. – Soc. Behav. Sci.*, 48 (2012), 2035–2047.

Turcu, C., 2013. Re-thinking sustainability indicators: local perspectives of urban sustainability. *Journal of Environmental Planning and Management*, 56 (5), 695-719. http://dx.doi.org/10.1080/09640568.2012.698984

Turner, B.L. II., Lambin, E.F. and Reenberg, A., 2007. The emergence of land change science for global environmental change and sustainability. *Proc Natl Acad Sci*, 104(52), 20666–20671.

Turnheim, B., Berkhout, F., Geels, F., Hof, A., McMeekin, A., Nykvist, B. and van Vuuren, D., 2015. Evaluating sustainability transitions pathways: Bridging analytical approaches to address governance challenges. *Global Environmental Change*, 35, 239–253. http://dx.doi.org/10.1016/j.gloenvcha.2015.08.010

Turok, I., 2013. Securing the resurgence of African cities. *Local Economy*, 28(2), 142–157. https://doi.org/10.1177/0269094212469920

Turok, I., 2015. Turning the tide? The emergence of national urban policies in Africa. *Journal of Contemporary African Studies*, 33(3), 348-369.

Turok, I., 2016. Housing and the urban premium. Habitat International, 54 (3), 234-240.

Turok, I., and McGranahan, G., 2013. Urbanization and economic growth: the arguments and evidence for Africa and Asia. *Environment and Urbanization*, 25(2), 465–482. https://doi.org/10.1177/0956247813490908

Turok, I. and Parnell, S., 2009. Reshaping Cities, Rebuilding Nations: The Role of National Urban Policies. *Urban Forum*, 20 (2), 157–174.

Turok, I. and Borel-Saladin, J., 2014. Is urbanisation in South Africa on a sustainable trajectory? *Development Southern Africa*, 31 (5), 675-691. http://dx.doi.org/10.1080/0376835X.2014.937524

Tselios, V., 2014. Urbanization and Socioeconomic Status in the European Regions: The Role of Population Ageing and Capital City Regions. *European Planning Studies*, 22(9), 1879-1901. DOI: 10.1080/09654313.2013.812063

Tscherning, K., Helming, K., Krippner, B., Sieber, S. and Gomez y Paloma, S., 2012. Does research applying the DPSIR framework support decision making? *Land Use Policy*, 29 (1), 102-110. http://dx.doi.org/10.1016/j.landusepol.2011.05.009

Tuncer, G., 2008. University students' perception on sustainable development: a case study from Turkey. *Int. Res. Geogr. Environ. Educ.*, 17(3), 212–226.

Trudeau, D., 2018. Integrating social equity in sustainable development practice: Institutional commitments and patient capital. *Sustainable Cities and Society*, 41, 601-610. https://doi.org/10.1016/j.scs.2018.05.007

Tsikata, D. and Yaro, J., 2011. Land Market Liberalization and Trans-National Commercial Land Deals in Ghana since the 1990s. http://citeseerx.ist.psu.edu/viewdoc/download?doi=10.1.1.469.1604&rep=rep1&type=pdf Accessed: 3/22/2018.

Uitermark, J., 2005. The genesis and evolution of urban policy: A confrontation of regulationist and governmentality approaches. *Political Geography*, 24,137–163.

Ulucak, R. and Bilgili, F., 2018. A reinvestigation of EKC model by ecological footprint measurement for high, middle and low-income countries. *Journal of Cleaner Production*, 188, 144-157. https://doi.org/10.1016/j.jclepro.2018.03.191

UNDESA / PD, 2012. World Urbanisation Prospects: The 2011 Revision, United Nations, New York.

UNDP, 1990. United Nations Development Programme. *Human Development Report*, New York: UNDP.

UNDP, 2011. Sustainability and Equity: A Better Future for All. Human Development Report 2011. United Nations Development Programme, New York.

UNDP, 2015. Ghana Millennium Development Goals 2015 Report. http://www.gh.undp.org/content/dam/ghana/docs/Doc/Inclgro/UNDP_GH_2015%20Ghana% 20MDGs%20Report.pdf Accessed: 29/05/2016.

UNDP, 2018. Human Development Indices and Indicators: 2018 Statistical Update. Briefing note for countries on the 2018 Statistical Update.

http://hdr.undp.org/sites/all/themes/hdr_theme/country-notes/GHA.pdf Accessed: 10/2/2019.

UNDP, 2019a. Human development report 2019. Beyond income, beyond averages, beyond today: inequalities in human development in the 21st century. http://hdr.undp.org/sites/default/files/hdr_2019_overview_-_english.pdf Accessed: 20/06/2020.

UNDP, 2019b. Human development report. Inequalities in human development in the 21st century. Briefing notes for countries on the 2019 Human Development Report: Ghana. http://hdr.undp.org/sites/all/themes/hdr_theme/country-notes/GHA.pdf_Accessed: 15/5/2020.

UNECA, 2012a. Unleashing Africa's Potential as a Pole of Growth: Economic Report on Africa, ECA, Addis Ababa, 186 pages.

UNECA, 2012b. Progress towards Sustainable Development in Africa: Summary Report. Addis Ababa: Economic Commission on Africa.

UNEP, 2016. United Nations Environmental Programme: About Green Economy Initiative. http://www.unep.org/greeneconomy/AboutGEI/WhatisGEI/tabid/29784/Default.aspx Accessed: 22/02/2016. UN-Habitat / DFID, 2002. UN-Habitat / DFID Sustainable urbanisation: achieving Agenda 21. United Nations Human Settlement Programme/Department for International Development, Nairobi (2002).

UN-Habitat 2008. The State of African Cities: A framework for addressing urban challenges in Africa, Nairobi: UN-Habitat & UNEP, Addis Ababa: ECA-CEA, Canada: IDRC-CDRI.

UN-Habitat 2012. Urbanisation in Africa, African Development Bank, 13 December 2012.

UN-Habitat, 2014. State of African Cities 2014: Re-imagining Sustainable urban transitions. Nairobi, Kenya: UNHABITAT.

UN-Habitat, 2016a. National Urban Policies. <u>http://unhabitat.org/urban-initiatives/initiatives-programmes/national-urban-policies/</u> Accessed: 20/03/2016.

UN-Habitat, 2016b. City prosperity Initiative. https://unhabitat.org/wp-content/uploads/2019/02/CPI-METADATA.2016.pdf Accessed: 3/4/2018.

UNICEF, 2012. Understanding the specifics of urban poverty: A case study from Accra, Ghana. The state of the world's children 2012.

United Nations, 2008. World Urbanization Prospects: The 2007 Revision. United Nations, New York.

United Nations, 1992. United Nations Conference on Environment & Development. Rio de Janeiro, 3-14 June 1992. AGENDA 21, CHAPTER 40. INFORMATION FOR DECISION-MAKING. http://www.un.org/earthwatch/about/docs/a21ch40.htm Accessed: 3/3/2017.

United Nations, 2005. In remarks to 'united cities and local governments', Secretary-General says local action needed to achieve global goals. https://www.un.org/press/en/2005/sgsm10081.doc.htm Accessed: 12/03/2016.

United Nations, 2010. State of the World's cities 2012/2011. UN-HABITAT (2010). http://www.unhabitat.org/documents/SOWC10/R8.pdf Accessed: 20/4/ 2017.

United Nations, 2014. World urbanization prospects: the 2014 revision highlights. New York: United Nations Department of Economic and Social Affairs, Population Division; 2014 https://population.un.org/wup/Publications/Files/WUP2014-Report.pdf Accessed: 13/02/2016.

United Nations, 2015. Sustainable Development Goals. Goal 11. Para.1. https://www.un.org/sustainabledevelopment/cities/ Assessed: 5/3/2017.

University of Bergen, 2019. Urban Enclaving Futures consists of two principal subprojects: Financing and Inhabiting. https://www.uib.no/en/urbanenclavingfutures/119683/subprojects-and-case-studies#jason-sumich Accessed:15/09/2019.

University of Sheffield, 2019. Mann-Whitney U test. https://www.sheffield.ac.uk/polopoly_fs/1.714552!/file/stcp-marshall-MannWhitS.pdf Accessed: 05/09/2019.

University of Rhode Island, 2019. How remote sensing works. http://www.edc.uri.edu/nrs/classes/NRS409/RS/Lectures/HowRemoteSensonWork.pdf Accessed:11/9/2019.

Urban Ecology, 1996. 'Mission Statement and Accomplishments', World Wide Web site (1996).

USGS, 2019. Landsat Levels of Processing. https://www.usgs.gov/land-resources/nli/landsat/landsat-levels-processing Accessed: 25/09/2019.

Valencia, S.C., Simon, D., Croese, S., Nordqvist, J., Oloko, M., Sharma, T., Buck, N.T. and Versace, I., 2019. Adapting the Sustainable Development Goals and the New Urban Agenda to the city level. *International Journal of Urban Sustainable Development*, 11, 4-23. https://doi.org/10.1080/19463138.2019.1573172

Vallati, A., Vollaro, A. D. L., Golasi, I., Barchiesi, E. and Caranese, C., 2015. On the impact of urban micro climate on the energy consumption of buildings. *Energy Procedia*, 82, 506-511.

Van Alstine, J. and Barkemeyer, R., 2014. Business and development: Changing discourses in the extractive industries, *Resources Policy*, 40, 4-16. doi: 10.1016/j.resourpol.2014.01.006

Van Arkadie, B., 1978. Review article: Town versus country. *Development and Change*, 8, 409–15.

Van den Bergh J.C.J.M. and Hofkes M. W., 1998. A survey of economic modelling of sustainable development. In Theory and Implementation of Economic Models for Sustainable Development. *Economy and Environment*, 15, 11-37.

Van den Bergh, J.C.J.M., Truffer, B. and Kallis, G., 2011. Environmental innovation and societal transitions: introduction and overview. *Environmental Innovation and Societal Transitions*, 1 (1), 1–23.

Van der Merwe, I.J., 2004. The Global Cities of Sub-Saharan Africa: Fact or Fiction? Urban Forum, 15 (1), 36-47.

Van Geenhuizen, M. and Thissen, W., 2002. Uncertainty in intelligent transport systems, a policy view. *J. Technol. Policy Manag.*, 2 (2002), 5–19.

Van Geenhuizen, M. and Ye, Q., 2014. Responsible innovators: open networks on the way to sustainability transitions. *Technological Forecasting and Social Change*, 87, 28-40.

Van Meerhaeghe, M. A. G., 2012. Globalisation: concept, outcome, future—a continental view. *European Journal of Law and Economics*, 33 (2), 239–306.

Van Noorloos, F. and Kloosterboer, M., 2018. Africa's new cities: The contested future of urbanisation. *Urban Studies*, *55*(6), 1223–1241. https://doi.org/10.1177/0042098017700574

Vanolo, A., 2013. Smart mentality: The smart city as disciplinary strategy *Urban Stud.*, (2013), 0042098013494427

Van Vliet, W., 2002. Cities in a globalizing world: from engines of growth to agents of change. *Environment & Urbanization*, 14(t), 31-40.

Velazquez, L., Munguia, N. and Sanchez, M., 2005. Deterring sustainability in higher education institutions: An appraisal of the factors which influence sustainability in higher education institutions. *International Journal of Sustainability in Higher Education*, 6(4), 383–391.

Verbrugge, B., 2019. Best Practice, Model, Framework, Method, Guidance, Standard: towards a consistent use of terminology – revised. https://www.vanharen.net/blog/bestpractice-model-framework-method-guidance-standard-towards-consistent-use-terminology/ Accessed: 19/11/2019.

Vlahov. D., Galea, S. and Freudenberg, N., 2005. The urban health "advantage". J. Urban Health, 82, 1–4.

Voß, J. P., Smith, A. and Grin, J., 2009. Designing long-term policy: rethinking transition management. *J. Policy Sci*, (2009) 42, 275. https://doi.org/10.1007/s11077-009-9103-5

Vojnovic, I., 2014. Urban sustainability: Research, politics, policy and practice. *Cities*, 41, Supplement 1, S30-S44. https://doi.org/10.1016/j.cities.2014.06.002

Voon, B.H., 2012. Role of service environment for restaurants: the youth customers' perspective. *Procedia - Social and Behavioral Sciences*, 38, 388 – 395.

Walker, W., 2007. Ethical considerations in phenomenological research. *Nurse Res.*, 14(3), 36-45.

Walsh, E., Babakina, O., Pennok, A., Shi, H., Chi, Y., Wang, T. and Graedel, T.E. 2006. Quantitative guidelines for urban sustainability. *Technol. Soc.*, 28, 45–61.

Wan, P.M.J., 2014. Environmental justices and injustices of large-scale gold mining in Ghana: A study of three mining communities near Obuasi. *The Extractive Industries and Society*, 1(1), 38-47.

Wang, X. and Tomaney, J., 2019. Zhengzhou – Political economy of an emerging Chinese megacity. *Cities*, 84, 104-111. https://doi.org/10.1016/j.cities.2018.07.011

Wang, X., Shi, R. and Zhou, Y., 2020. Dynamics of urban sprawl and sustainable development in China. *Socio-Economic Planning Sciences*, 70, 100736. https://doi.org/10.1016/j.seps.2019.100736

Ward, D., Phinn, S.R. and Murray, A. T., 2000. Monitoring growth in rapidly urbanizing areas using remotely sensed data. *Prof. Geogr.*, 52, 371–386.

Weststrate, J., Dijkstra, G., Eshuis, J., 2019. The Sustainable Development Goal on Water and Sanitation: Learning from the Millennium Development Goals. *Social Indicators Research*, 143 (2), 795-810. https://doi.org/10.1007/s11205-018-1965-5

Wheaton, W. C., 2004. Commuting, congestion, and employment dispersal in cities with mixed land Use. *Journal of Urban Economics*, 55(3), 417-438.

Wheeler, S. M., 2008. State and municipal climate change plans: The first generation. *Journal of the American Planning Association*, 74, 481-496.

Whitehead, M., 2003. (Re)Analyzing the Sustainable City: Nature, Urbanization and the Regulation of Socio-environmental Relations in the UK. *Urban Studies*, 40, 1183–1206.

WHO, 2013. A Global Brief on Hypertension WHO, Geneva (2013) Available http://apps.who.int/iris/bitstream/10665/79059/1/WHO_DCO_WHD_2013.2_eng.pd accessed 05.10.15.

Wiegleb, V., 2016. A Literature Review on Wetlands in Accra. Water Power Working Paper, No. 5. Governance and Sustainability Lab. Trier. http://www.waterpower.science/uploads/5/3/1/1/53111533/5_wiegleb-viviana_a-literaturereview-on-wetlands-in-accra.pdf Accessed: 03/11/2018.

Wilkinson, R. and Pickett, K., 2009. *The Spirit Level. Why Equality is Better for Everyone*. Penguin, London.

Williams, K., 2010. Sustainable cities: research and practice challenges. *International Journal of Urban Sustainable Development*. 1 (1-2), 128-132.

Wilsdon, J., Allen, L., Belfiore, E., Campbell, P., Curry, S., Hill, S., Jones, R., Kain, R., Kerridge, S., Thelwall, M., Tinkler, J., Viney, I., Wouters, P., Jude, H. and Johnson, B., 2015. The Metric Tide: Report of the Independent Review of the Role of Metrics in Research Assessment and Management. DOI: 10.13140/RG.2.1.4929.1363

Winters, C., 1983. The classification of traditional African cities. *Journal of Urban History*, 10 (1) (1983), 3–31.

Wolch J., Byrne, J. and Newell J.P., 2014. Urban green space, public health, and environmental justice: The challenge of making cities 'just green enough'. *Landscape and Urban Planning*, 125 (2014), 234–244. doi: 10.1016/j.landurbplan.2014.01.017.

Wolfe, A.K., Kerchner, N. and Wilbanks, T., 2001. Public involvement on a regional scale. *Environmental Impact Assessment Review*, 21 (5), 431-448.

World Bank, 2009. World Development Report: Reshaping Economic Geography. The World Bank, Washington, DC.

World Bank 2015. Rising through cities in Ghana: urbanization review- overview report. Washington D.C.: World Bank Group.

World Bank, 2018. Global Platform for Sustainable Cities, "Urban Sustainability Framework." 1st ed. Washington, DC: World Bank. http://documents.worldbank.org/curated/en/339851517836894370/pdf/123149-Urban-Sustainability-Framework.pdf Accessed: 3/4/2018.

World Bank, 2019a. GINI index (World Bank estimate) https://data.worldbank.org/indicator/SI.POV.GINI?locations=GH Assessed: 10/2/2019.

World Bank, 2019b. World Bank country data Ghana: Gross National Income. https://data.worldbank.org/country/ghana Accessed: 10/2/2019. World Bank Report, 2000. World Development Report: Attacking Poverty. Oxford Univ. Press, New York.

World Economic and Social Survey, 2013. Sustainable development challenges, Department of Economic and Social Affairs. United Nations, New York (2013).

World Commission on Environment and Development, 1987. *Our common future*. London: Oxford University Press.

World Conservation Union, 2006. The future of sustainability: re-thinking environment and development in the twenty-first century.

http://cmsdata.iucn.org/downloads/iucn_future_of_sustanability.pdf, 2006. Accessed: 08.02.2017.

World Population Review, 2019. Human Development Index (HDI) by Country 2019. http://worldpopulationreview.com/countries/hdi-by-country/ Accessed: 19/11/2019.

Wu, F., 2001. China's recent urban development in the process of land and housing marketisation and economic globalisation. *Habitat International*, 25(3), 273-289. https://doi.org/10.1016/S0197-3975(00)00034-5

Wu, F., 2012. China's eco-cities. Geoforum, 43 (2012), 169-171.

Wu, J.G., 2013. Landscape sustainability science: ecosystem services and human well-being in changing landscapes. *Landscape Ecol.*, 28(6), 999–1023.

Wu, J. 2014. Urban ecology and sustainability: The state-of-the-science and future directions. *Landscape and Urban Planning*, 125 (2014), 209–221. http://dx.doi.org/10.1016/j.landurbplan.2014.01.018

Xiao, Y.I., Coppin, and Van Bavel, J. J., 2016. Perceiving the World Through Group-Colored Glasses: A Perceptual Model of Intergroup Relations. *Psychological Inquiry*, 27(4), 255-274, DOI: 10.1080/1047840X.2016.1199221

Yakubu, I., Akaateba M.W. and Akanbang B.A.A., 2014. *Habitat International*, 44, 394–402. http://dx.doi.org/10.1016/j.habitatint.2014.08.003

Yang, Y., 2010. Sustainable urban transformation: driving forces, indicators and processes. Doctoral Dissertation. ETH ZURICH.

Yankson, P.W.K. and Gough, K.V., 1999. The environmental impact of rapid urbanization in the peri-urban area of Accra, Ghana, *Geografisk Tidsskrift,-Danish Journal of Geography*, 99 (1), 89-100. DOI: 10.1080/00167223.1999.10649426

Yaro, J.A., Codjoe, S.N.A., Agyei-Mensah, S., Darkwah, A. and Kwankye, S.O., 2011. Migration and population dynamics: Changing community formations in Ghana. Center for Migration Studies, University of Ghana, Migrations Studies Technical Paper Series, Technical Paper no. 2.

Ye, H., He, X., Song, Y., Li, X., Zhang, G., Lin, T. and Xiao, L., 2015. A sustainable urban form: The challenges of compactness from the view point of energy consumption and carbon emission. *Energy and Buildings*, 93, 90–98.

Yeboah, E. and Obeng-Odoom, F., 2010. We are not the only ones to blame': District Assemblies' perspectives on the state of planning in Ghana. *Commonwealth Journal of Local Governance*, 7 (2010), 78-98. 10.5130/cjlg.v0i7.1893

Yin, R.K., 1994. *Case Study Research: Design and Methods*, 2nd edn. Sage Publications, Thousand Oaks.

Yin, R.K., 2003. *Case study research: design and methods* (3rd edition), Sage Publications, Thousand Oaks, CA.

Young, C., Diep, M. and Drabble, S., 2006. Living with Difference? The "Cosmopolitan City" and Urban Reimaging in Manchester, UK. *Urban Studies*, 43(10), 1687–1714. https://doi.org/10.1080/00420980600888486. York, J.G. and Venkatamaran, S., 2010. The entrepreneur–environment nexus: uncertainty, innovation and allocation. *J. Bus. Ventur.*, 25 (2010), 449–463.

Yuan, P., 2009. Modernization Theory. Chinese Studies in History, 43(1), 37-45.

Zeng, H., Sui, D.Z. and Li, S., 2005. Linking Urban Field Theory with GIS and Remote Sensing to Detect Signatures of Rapid Urbanization on the Landscape: Toward a New Approach for Characterizing Urban Sprawl. *Urban Geography*, 26(5), 410-434. https://doi.org/10.2747/0272-3638.26.5.410

Zhang, X. Q., 2016. The trends, promises and challenges of urbanisation in the world. *Habitat International*, 54 (3), 241-252. https://doi.org/10.1016/j.habitatint.2015.11.018

Zhang, P., Yuan, H. and Tian, X., 2018. Sustainable development in China: Trends, patterns, and determinants of the "Five Modernizations" in Chinese cities. *Journal of Cleaner Production*, 214, 685-695. https://doi.org/10.1016/j.jclepro.2018.12.307

Zhao, J., Xiao, L., Tang, L., Shi, L., Su, X., Wang, H. and Shao, G., 2014. Effects of spatial form on urban commute for major cities in China. *International Journal of Sustainable Development & World Ecology*, 21(4), 361–368.

Zhao, Y. and Wang, S., 2015. The Relationship between Urbanization, Economic Growth and Energy Consumption in China: An Econometric Perspective Analysis. *Sustainability*, 7, 5609-5627.

Zhou, J., Shen, L., Song, Z. and Zhang, X. 2015. Selection and modelling sustainable urbanization indicators: A responsibility-based method. *Ecological Indicators*, 56, 87-95.

Appendices

Appendix 1: Summary of key studies consulted in establishing indicators for sustainability assessment.

Authors and	Title	Journal	Issue	No. of
year			and page	Indicators
Mutisya and Yarime 2014	Moving towards urban sustainability in Kenya: a framework for integration of environmental, economic, social and governance dimensions	Sustainability Science	9; p. 205- 215	19 environmental, social, economic, and governance dimensions indicators
Turcu 2013	Re-thinking sustainability indicators: Local perspectives of urban sustainability.	Journal of Environmental Planning and Management	56 (5); p. 695-719	24 indicators covering institutional, environmental, economic, and social sustainability.
Yang et al. 2016	Urban sustainability assessment of Taiwan based on data envelope analysis	Renewable and Sustainable Energy Reviews	61 (2016); p. 341-353.	22 indicators based on resource inputs, settlement dynamics, liveability, and waste outputs
Shen et al. 2011	The application of urban sustainability indicators – A comparison between various practices	Habitat International	35 (1); p. 17-29	37 indicators covering environmental, economic, social, and governance dimensions of sustainability
Hosseini and	Dynamic sustainability	Ecological	11 (3); p.	29 indicators covering

Kaneko 2011	assessment of countries at the macro level: A principal component analysis	Indicators	811-823	environmental, economic, social, and institutional dimensions of sustainability
Michael et al.	Review of urban	Habitat	44; p.	42 sustainability
2014 (citing	sustainability	International	491-500	indicators
Urban	indicators- Case			
Indicators	study between Asian			
Guidelines	countries			
2013)				
,				
Wu 2014	Urban ecology and	Landscape and	125; p.	15 indicators
	sustainability: The	Urban Planning	209-221	representing the
	state-of-the-science			environment-
	and future directions			economy-
				society nexus
				5

Appendix 2: Questionnaire for field survey.

Appendix 2A. Information and consent sheet for survey participants.

CITY
ID:

LOCAL:....



Researcher: Kwasi Peprah Anarfi Bournemouth University Fern Barrow, Poole BH12 5BB, UK

kanarfi@bournemouth.ac.uk

INFORMATION AND CONSENT SHEET FOR SURVEY PARTICIPANTS

PROJECT TOPIC: 'An African perspective on urban sustainability: a comparative study of two urban areas in Ghana'.

This information sheet provides information about a survey for a research project that you are invited to participate in, on urbanisation and the transition towards sustainable cities. This survey is being undertaken Kwasi Peprah Anarfi, a PhD student at Bournemouth University, UK. Your participation will help throw light on the nature of urbanisation in Sub-Saharan Africa and help make informed contributions which will potentially steer the urbanisation process towards a desired status of urban sustainability. The information from this survey will be confidential and used solely for academic purposes. Participation in this survey is entirely voluntary and the respondent may withdraw at any time without giving a reason for doing so.

1. PURPOSE OF STUDY

The study examines the nature of urbanisation in the study locations and to explore ways of transforming urban systems towards sustainability.

2. CHOICE OF PARTICIPANTS

The study involves seeking views from urban residents on issues in concerning urbanisation, the dimensions of sustainable development and urban sustainability.

3. CONSENT

a) The investigator has explained the purpose and nature of the research to me.

b) My doubts and questions have been answered satisfactorily.

c) I agree to participate based on assurances that I can voluntarily withdraw from the research without any problems.

Date:

Appendix 2B. Questionnaire.

01	
1. Gender	Male [] Female []
2. Age (in years)	18-30 [] 31-45 [] 46-60 [] >60 []
3. Length of Residence (yrs)	<2[] 2-5[] 6-10[] > 10[]
4. Level of Education	None [] Primary [] Secondary [] Tertiary []
5. Employment Status	Employed with wage [] Self-employed [] Unemployed / Retired []
6. Housing	Home Owner [] Tenant [] Family / Friends Housing []
7. Household Size	

Part A: Demographic data

Part B: Perspectives on Dimensions of sustainability

Please indicate your level of agreement with the following statements:

Q.	Economic Sustainability	Strongly Agree	Agree	Undecided	Disagree	Strongly Disagree
8	It is easy to find a job.	1	2	3	4	5
9	The cost of goods and services are acceptable (compared to elsewhere in Ashanti)	1	2	3	4	5
10	Income levels are adequate (compared to elsewhere in Ashanti)	1	2	3	4	5

Q.	Social Sustainability	Strongly Agree	Agree	Undecided	Disagree	Strongly Disagree
11	Access to education is adequate	1	2	3	4	5

12	Access to healthcare is adequate	1	2	3	4	5
13	Access to housing is adequate	1	2	3	4	5
14	Public transport is widely available	1	2	3	4	5
15	Public safety levels are acceptable	1	2	3	4	5

Q.	Environmental Sustainability	Strongly Agree	Agree	Undecided	Disagree	Strongly Disagree
16	The level of air quality/ pollution is acceptable.	1	2	3	4	5
17	The amount of accessible green and public spaces is adequate.	1	2	3	4	5
18	The level of sanitation service provision and general cleanliness are adequate	1	2	3	4	5
19	There is good access to potable water	1	2	3	4	5
20	Recycling/ re-use of waste materials are important to save the environment.	1	2	3	4	5

Part C: Perspectives on Urbanisation

Please indicate your level of agreement with the following statements:

Q.		Strongly Agree	Agree	Undecided	Disagree	Strongly Disagree
21	The city's population and landscape are changing rapidly.	1	2	3	4	5
22	The rate of urbanisation is commensurate with the level of development.	1	2	3	4	5

23	There is great access to economic opportunities, essential infrastructure and services from where I live in this city.	1	2	3	4	5
24	I am happy to live in this city and I personally do not seek to migrate to another place in this country.	1	2	3	4	5

Part D: Attitudes towards urban sustainability

Please indicate your level of agreement with the following statements:

Q.		Yes	Maybe	No
25	Would you willingly participate in a waste recycling scheme in order to promote sustainability in this city?	1	2	3
26	Would you willingly patronise means of transport like cycling and public transport provided they are safe, easily accessible and convenient, in order to promote sustainability in this city?	1	2	3
27	Would you willingly regulate direct water use in your household in order to promote sustainability in this city?	1	2	3
28	Would you willingly regulate direct energy use in your household in order to promote sustainability in this city?	1	2	3

Further comments:

Thank you.
Appendix 3: Selected responses from 'further comments' section of questionnaire.

Location	ID	Comment
	0107	"The government must come to our aid. We have no jobs and the
		future looks bleak"
	0304	"I used to engage in 'galamsey' but there has been a clamp down
		by the government and the company (AngloGold Ashanti) due to
		environmental concerns and the fact that the land belongs to the
		company. But what should we do since there are no jobs? I know
		'galamsey' destroys the land and water (environment) but what
		choice do have when 'galamsey' is the only means through which
		we can survive" (Author's field study).
	0341	"I am currently studying, but in the future, I think, I would love to
Obuasi		move to Kumasi or Accra where I can find a good paying job to
		secure my future"
	0349	"I am not happy with my quality of life. Things have been hard
		since company halted operations. As a trader, people are not
		purchasing my items because there is no money"
	0425	"They (government and AngloGold Ashanti)) say they will not
		allow us to engage in "galamsey" because it destroys the land. But
		I was born here and within my lifetime I have seen the company's
		mining operation destroy our river. What did they do about it then?
		Galamsey is risky, but people still want to do it because that's how
		they can get money to cater for their families".
	0511	"If I understand you well, you want to know whether we are happy
		with the level of development we are experiencing. I will ask you to
		look around, just look at the pot holes on the roads, the population
		is growing but the infrastructure has not changed significantly".
		·
	K115	"I think Kumasi is the most cosmopolitan city in Ghana as many
		people from different tribes of the country have come here. But
		where are the jobs?

Table A3.1 Summary of selected responses from further comments section of questionnaire.

	K321	"The last few years were uncomfortable as the energy (electricity)
		rationing was going on. So, the government must find a way to
		educate citizens to save energy"
	K325	"Kumasi has changed considerably over the years. I think it the city
		has exhausted the capacity of the people it can accommodate"
	K727	"I understand the fact that you are sensitising us on the things we
		could do to promote good development. However, if we want to
Kumasi		recycle our waste where do we send it? We can only do it if
		adequate provisions are made. These days, in order to be able to go
		to work you must buy a car if you can afford it. The 'trotro' (public
		transport) is unreliable and you spend a lot time waiting it".
	K406	"The population has increased but I do not think the government is
		doing enough. I am moving houses almost every two years because
		landlords are increasing rent at every tenancy renewal period"
	K 519	"If the government does not put in place good policies to enhance
		the performance of the economy and create jobs for the up and
		coming youth, then our future will be hopeless"
	K702	"The number of public recreational places in a big city like Kumasi
		is woefully inadequate. In fact,
	K322	"The government must help us else we are going to die poor. It's
		been three years since I completed SSS (senior high school), yet I
		cannot find a job, neither can I afford to continue my school".
	K118	"People (researchers, government etc) are always soliciting for
		information on our development circumstances and almost every
		time I have to narrate the same story. Our economic circumstances
		are not great and we are fighting to make ends meet"

Appendix 4: Confusion matrices for accuracy assessment of classified land cover maps.

Mapped Class	Reference points								
	Water	Urban	Mixed	Forest	Agricultural	Total	User		
		/ Built-	Vegetation		land		Accuracy		
		up					(%)		
Water	9	0	0	0	1	10	90		
Urban /	0	16	2	0	3	21	76		
Built-up									
Mixed	0	0	16	2	2	20	80		
Vegetation									
Forest	1	0	0	9	1	11	82		
Agricultural	0	1	0	0	6	7	86		
Land									
Total	10	17	18	11	13	69			
Producer	90	94	89	82	46.1	Overal	l Accuracy		
Accuracy						= 82%			
(%)									

Table A4.1. Confusion matrix for land cover classification in Kumasi for 1986. Kappa=0.78.

Mapped Class	Reference points								
	Water	Urban	Mixed	Forest	Agricultural	Total	User		
		/ Built-	Vegetation		land		Accuracy		
		up					(%)		
Water	5	0	0	0	0	5	100		
Urban /	0	22	2	0	1	25	88		
Built-up									
Mixed	0	1	7	1	0	9	78		
Vegetation									
Forest	0	0	1	4	0	5	80		
Agricultural	0	2	0	2	14	18	78		
Land									
Total	5	25	10	7	15	62			
Producer	100	88	70	57	93	Overall Accuracy			
Accuracy						= 84%			
(%)									

Mapped	Reference points								
Class		T	1	[
	Water	Urban	Mixed	Forest	Agricultural	Total	User		
		/ Built-	Vegetation		land		Accuracy		
		up					(%)		
Water	5	0	0	0	0	5	100		
Urban /	0	20	0	0	2	22	91		
Built-up									
Mixed	0	0	7	0	0	7	100		
Vegetation									
Forest	0	0	0	8	1	9	89		
Agricultural	0	1	0	0	9	10	90		
Land									
Total	5	21	8	8	11	53			
Producer	100	91	88	100	82	Overall	Accuracy		
Accuracy						= 92%			
(%)									

Table 4.3. Confusion matrix for land cover classification in Kumasi for 2010. Kappa=0.86.

Table A4.4. Confusion matrix for land cover classification in Kumasi for 2018. Kappa=0.83.

Mapped Class	Reference points								
Ciuss	Water	Urban	Mixed	Forest	Agricultural	Total	User		
		/ Built-	Vegetation		land		Accuracy		
		up					(%)		
Water	1	0	0	0	0	1	100		
Urban /	0	8	1	0	0	9	89		
Built-up									
Mixed	0	0	10	0	0	10	100		
Vegetation									
Forest	0	0	0	5	0	5	100		
Agricultural	0	1	1	0	3	5	60		
Land									
Total	1	9	12	5	3	30			
Producer Accuracy (%)	100	89	83	100	100	Overal = 90%	Accuracy		

Mapped	Reference points									
Class	Water	Urban / Built- up	Mixed Vegeta tion	Forest	Agricult ural land	Bare ground	Mine site	Total	User Accuracy (%)	
Water	-	-	-	-	-	-	-	-	-	
Urban / Built-up	-	9	0	0	1	-	0	10	90	
Mixed Vegetation	-	0	11	1	1	-	0	13	85	
Forest	0	0	1	9	0	-	0	10	90	
Agricultural Land	-	1	1	0	8	-	0	10	80	
Bare ground	-	-	-	-	-	-	-	-	-	
Mine site	-	1	0	0	0	-	4	5	80	
Total	-	11	13	10	10	-	4	48		
Producer Accuracy (%)	-	82	85	90	80	-	100	Overal = 85%	ll Accuracy	

Table A4.5. Confusion matrix for land cover classification in Obuasi for 1986. Kappa=0.81.

Table A4.6. Confusion matrix for land cover classification in Obuasi for 2000. Kappa=0.76.

Mapped	Reference points								
Class	Water	Urban	Mixed	Forest	Agricultu	Bare	Mine	Total	User
		1	Vegetation		ral land	ground	site		Accuracy
		Built-							(70)
		up							
Water	1	0	0	0	0	-	0	1	100
Urban /	0	13	0	0	2	-	0	15	87
Built-up									
Mixed	0	1	10	2	2	-	0	15	67
Vegetation									
Forest	0	0	0	5	0	-	0	5	100
Agricultural	0	0	1	0	4	-	0	5	80
Land									
Bare ground	-	-	-	-	-	-	-	-	-
Mine site	0	1	0	0	0	-	4	5	80
Total	1	15	11	7	8	-	4	46	
Producer	100	87	91	71	50	-	100	Overa	11
Accuracy								Accur	acy = 80%
(%)									

Mapped		Reference points									
Class	Water	Built-	Mixed	Forest	Agriculture	Bare	Mining	Total	User		
		up	Vegetation		land	ground			Accuracy (%)		
Water	1	0	0	0	0	0	0	1	100		
Built-up	0	13	0	0	2	0	0	15	87		
Mixed	0	0	8	5	2	0	0	15	53		
Vegetation											
Forest	0	0	1	4	0	0		5	80		
Agriculture	0	0	1	1	8	0	0	10	80		
Land											
Bare	0	0	0	0	0	2	0	2	100		
ground											
Mining	0	0	0	0	0	0	5	5	100		
Total	1	13	10	10	12	2	5	53			
Producer	100	100	80	40	67	100	100	Overa	11		
Accuracy								Accur	acy = 77%		
(%)											

Table A4.7. Confusion matrix for land cover classification in Obuasi for 2010. Kappa=0.68

Table A4.8.	Confusion ma	trix for land cov	er classification in	Obuasi for 2018.	<i>Kappa</i> =0.79.
	./			./	

Mapped	Reference points								
Class	Water	Urban	Mixed	Forest	Agricultu	Bare	Mine	Total	User
		/	Vegetation		ral land	ground	site		Accuracy
		Built-							(%)
		up							
Water	2	0	0	0	0	-	0	2	100
Urban /	0	8	0	0	1	-	0	9	89
Built-up									
Mixed	0	0	4	1	0	-	0	5	80
Vegetation									
Forest	0	0	1	4	0	-	0	5	80
Agricultural		1	0	0	4	-	0	5	80
Land									
Bare	-	-	-	-	-	-	-	-	-
ground									
Mine site	1	0	0	0	0	-	3	4	75
Total	3	9	5	5	5	-	3	30	
Producer	67	89	80	80	80	-	100	Overa	11
Accuracy								Accur	acy = 83%
(%)									

Appendix 5: LULC change matrices for Kumasi and Obuasi, 1986–2018.

	LULC			Kumasi 2000) (Area in	km ²)						
	Class	Water	Urban / Built-	Mixed Woody	Forest	Agricultural land	Grand Total					
			up	Vegetation			1986					
	Water	0.50	0.00	0.00	0.00	0.00	0.50					
	Urban /	0.00	42.76	0	0	0.47	43.23					
	Built-up											
Kumasi	Mixed	0.21	125.25	106.88	1.15	395.27	628.76					
1986	Woody											
	Vegetation											
	Forest	0	2.51	69.00	3.27	3.33	78.11					
	Agricultural	0.25	62.40	0.46	0	1.64	64.75					
	Land											
	Grand	0.96	232.92	176.34	4.42	400.71	815.35					
	<i>Total 2000</i>											

Table A5.1. LULC change matrix for Kumasi 1986–2000.

Table A5.2. LULC change matrix for Kumasi 2000–2010.

	LULC	Kumasi 2010 (Area in km ²)						
	Class	Water	Urban	Mixed	Forest	Agricultural	Grand	
			/ Built-	Woody		land	Total	
			up	Vegetation			2000	
	Water	0.96	0	0	0	0	0.96	
	Urban /	0	232.81	0	0	0.11	232.92	
	Built-up							
Kumasi	Mixed	0	39.50	44.11	1.61	91.12	176.34	
2000	Woody							
	Vegetation							
	Forest	0	0.13	1.24	1.84	1.21	4.42	
	Agricultural	0	117.79	1.22	0.18	281.52	400.71	
	Land							
	Grand	0.96	390.23	46.57	3.63	373.96	815.35	
	Total 2010							

	LULC	Kumasi 2018 (Area in km ²)					
	Class	Water	Urban	Mixed	Forest	Agricultural	Grand
			/ Built-	Woody		land	Total
			up	Vegetation			2010
	Water	0.67	0.27	0	0	0.02	0.96
	Urban /	0	390.23	0	0	0	390.23
	Built-up						
Kumasi	Mixed	0	5.89	30.98	2.36	7.34	46.57
2010	Woody						
	Vegetation						
	Forest	0	0	0.38	2.91	0.34	3.63
	Agricultural	0	112.81	81.99	68.31	110.85	373.96
	Land						
	Grand	0.67	509.20	113.35	73.58	118.55	815.35
	<i>Total 2018</i>						

Table A5.3. LULC change matrix for Kumasi 2010–2018.

Table A5.4. LULC change matrix for Obuasi 1986–2000.

		Obuasi 2000 (Area in km ²)							
	LULC Class	Water	Urban / Built- up	Mixed Woody Vegetation	Forest	Agricultu ral land	Bare ground	Mine site	Grand Total 1986
	Water	-	-	-	-	-	-	-	-
	Urban / Built-up	0	23.51	0	0	0.09	0	0	23.60
Obuasi 1986	Mixed Woody Vegetation	0	2.17	147.87	6.94	3.72	-	0.86	161.56
	Forest	0	1.61	8.01	45.09	3.43	-	0.02	58.16
	Agricultural Land	0.07	8.42	27.10	18.05	72.35	-	0.65	126.64
	Bare ground	-	-	-	-	-	-	-	-
	Mining	0	0.91	6.43	0	0.54	-	2.67	10.55
	Grand Total 2000	0.07	36.62	189.41	70.08	80.13	-	4.20	380.51

		Obuasi 2010 (Area in km ²)								
LULC Class	LULC Class	Water	Urban / Built- up	Mixed Woody Vegetation	Forest	Agricultu ral land	Bare ground	Mine site	Grand Total 2000	
	Water	0.07	0	0	0	0	0	0	0.07	
	Urban /	0	36.62	0	0	0	0	0	36.62	
	Built-up									
	Mixed	0.92	2.51	71.03	64.31	39.38	2.75	8.51	189.41	
	Woody									
Obuasi	Vegetation									
2000	Forest	0	1.10	3.56	51.98	11.47	0	1.97	70.08	
	Agricultural	0.21	6.27	1.25	2.75	58.45	7.85	2.57	80.13	
	Land									
	Bare	-	-	-	-	-	-	-	-	
	ground									
	Mining	0	0	0.22	0	0	0.50	3.48	4.20	
	Grand	1.20	46.50	76.06	119.04	110.03	11.15	16.53	380.51	
	Total 2010									

Table A5.5. LULC change matrix for Obuasi 2000–2010.

Table A5.6. LULC change matrix for Obuasi 2010–2018.

			Obuasi 2018 (Area in km ²)								
	LULC Class	Water	Urban / Built- up	Mixed Woody Vegetation	Forest	Agricultu ral land	Bare ground	Mine site	Grand Total 2010		
	Water	0.07	0.92	0	0	0.21	0	0	1.20		
	Urban /	0	46.48	0	0	0.02	-	0	46.50		
	Built-up										
	Mixed	0	2.91	65.29	2.21	5.27	-	0.38	76.06		
	Woody										
Obuasi	Vegetation										
2010	Forest	0	0.36	43.25	68. 7	5.51	-	1.22	119.04		
	Agricultural	0	2.95	5.98	0.64	99.30	-	1.16	110.03		
	Land										
	Bare	0	4.06	0.66	0	6.31	-	0.12	11.15		
	ground										
	Mining	0	0	2.34	0	10.21	-	3.98	16.53		
	Grand	0.07	57.68	117.52	71.55	126.83	-	6.86	380.51		
	<i>Total 2018</i>										





Figure A6.1: Built-up vs Non-Built-up land cover in Kumasi, 1986.



Figure A6.2: Built-up vs Non-Built-up land cover in Kumasi, 2000.



Figure 6.3: Built-up vs Non-Built-up land cover in Kumasi, 2010.



Figure A6.4: Built-up vs Non-Built-up land cover in Kumasi, 2018.



Figure A6.5: Built-up vs Non-Built-up land cover in Obuasi, 1986.



Figure A6.6: Built-up vs Non-Built-up land cover in Obuasi, 2000.



Figure A6.7: Built-up vs Non-Built-up land cover in Obuasi, 2010.



Figure A6.8: Built-up vs Non-Built-up land cover in Obuasi, 2018.

Appendix 7: Statistics showing which economic groups were causing group differences in Kruskal Wallis H test for Kumasi.

Table A7.1. Table showing ranks between	'Employed with wage' a	and 'Self-employed' in
Kumasi		

Ranks			
	Employment		Mean
	Status	Ν	Rank
It is easy to find a	Employed	47	99.24
job.	with wage		
	Self-	145	95.61
	employed		
	Total	192	
The cost of goods	Employed	47	86.13
and services are	with wage		
acceptable	Self-	145	99.86
(compared to	employed		
elsewhere in	Total	192	
Ashanti)			

Table A7.2. Table showing test statistics between 'Employed with wage' and 'Self-employed' in Kumasi.

Test Statistics ^{a,b}		
	It is easy to find a job.	The cost of goods and services are acceptable (compared to elsewhere in Ashanti)
Kruskal-Wallis H	.209	2.463
df	1	1
Asymp. Sig.	.648	.117
a. Kruskal Wallis Test		
b. Grouping Variable: Employment Status		

Table A7.3. Table showing ranks between 'Self-employed' and 'Unemployed in Kumasi.

Ranks			
	Employment Status	N	Mean Rank
It is easy to find a job.	Self-employed	145	158.40
	Unemployed/Retired	126	110.22
	Total	271	
The cost of goods and	Self-employed	145	149.47
services are	Unemployed/Retired	126	120.50
(compared to elsewhere in Ashanti)	Total	271	

Table A7.4. Table showing test statistics between 'Self-employed' and 'Unemployed' in Kumasi.

Test Statistics ^{a,b}		
	It is easy to find a job.	The cost of goods and services are acceptable (compared to elsewhere in Ashanti)
Kruskal-Wallis H	32.688	10.918
df	1	1
Asymp. Sig.	.000	.001
a. Kruskal Wallis Test		
b. Grouping Variable: Employment Status		

Table A7.5. Table showing ranks between 'Employed with wage'' and 'Unemployed in Kumasi.

Ranks			
	Employment Status	N	Mean Rank
It is easy to find a job.	Employed with wage	47	112.83
	Unemployed/Retired	126	77.37
	Total	173	
The cost of goods and services are	Employed with wage	47	91.10
acceptable	Unemployed/Retired	126	85.47
elsewhere in Ashanti)	Total	173	

Table A7.6. Table showing test statistics between 'Employed with wage' and 'Unemployed' in Kumasi.

Test Statistics ^{a,b}		
	It is easy to find a job.	The cost of goods and services are acceptable (compared to elsewhere in Ashanti)
Kruskal-Wallis H	22.393	.565
df	1	1
Asymp. Sig.	.000	.452
a. Kruskal Wallis Test		·
b. Grouping Variable: Employment Status		

Appendix 8: Statistics showing which economic groups were causing group differences in Kruskal Wallis H test for Obuasi.

Ranks			
	Employment Status	N	Mean Rank
It is easy to find a job.	Employed with wage	51	114.09
	Self- employed	131	82.71
	Total	182	
There is great access to economic opportunities	Employed with wage	51	90.38
essential infrastructure and	Self- employed	131	91.94
I live in this city.	Total	182	

r.

Table A8.1. Table showing ranks between 'Employed with wage' and 'Self-employed' in Obuasi.

Table A8.2. Table showing test statistics between 'Employed with wage' and 'Self-employed' in Obuasi.

Test Statistics ^{a,b}		
	It is easy to find a job.	There is great access to economic opportunities, essential infrastructure and services from where I live in this city.
Kruskal-Wallis H	19.400	.037
df	1	1
Asymp. Sig.	.000	.847
a. Kruskal Wallis Test		
b. Grouping Variable: Employment Status		

Table A8.3. Table showing ranks between 'Self-employed' and 'Unemployed in Obuasi.

Ranks			
	Employment Status	N	Mean Rank
It is easy to find a job	Self-employed	131	136.35
inia a joo.	Unemployed/Retired	120	114.70
	Total	251	
There is great	Self-employed	131	138.25
economic	Unemployed/Retired	120	112.63
essential infrastructure and services from where I live in this city.	Total	251	

Table A8.4. Table showing test statistics between 'Self-employed' and 'Unemployed' in Obuasi.

Test Statistics ^{a,b}		
	It is easy to find a job.	There is great access to economic opportunities, essential infrastructure and services from where I live in this city.
Kruskal-Wallis H	9.083	8.948
df	1	1
Asymp. Sig.	.003	.003
a. Kruskal Wallis Test		
b. Grouping Variable: Employment Status		

Table A8.5. Table showing ranks between 'Employed with wage'' and 'Unemployed in Obuasi.

Ranks			
	Employment Status	N	Mean Rank
It is easy to find a job.	Employed with wage	51	114.29
2	Unemployed/Retired	120	73.98
	Total	171	
There is great access to economic	Employed with wage	51	98.51
opportunities, essential	Unemployed/Retired	120	80.68
infrastructure and services from where I live in this city.	Total	171	

Table A8.6. Table showing test statistics between 'Employed with wage' and 'Unemployed' in Obuasi.

Test Statistics ^{a,b}		
	It is easy to find a job.	There is great access to economic opportunities, essential infrastructure and services from where I live in this city.
Kruskal-Wallis H	31.772	5.504
df	1	1
Asymp. Sig.	.000	.019
a. Kruskal Wallis Test		
b. Grouping Variable: Employment Status		

Appendix 9: Statistics showing which age groups were causing significant interactions in Kruskal Wallis H test for Kumasi.

Ranks			
	Age (in years)	N	Mean Rank
I am happy to live in this city and I	18 - 30	139	114.10
personally do not	31 - 45	113	141.75
seek to migrate to another place in this country.	Total	252	

Table A9.1. Table showing ranks between '18-30' and '31-45' in Kumasi

Table A9.2	. Table showi	ng test statistic	s between	<i>'18-30'</i>	and	<i>'31-45</i>	' in Kumasi.
------------	---------------	-------------------	-----------	----------------	-----	---------------	--------------

Test Statistics ^{a,b}	
	I am happy to live in this
	city and I personally do
	another place in this
	country.
Kruskal-Wallis H	9.707
46	1
	1
Asymp. Sig.	.002
a. Kruskal Wallis Test	
b. Grouping Variable: Age (in years)	

Table A9.3. Table showing ranks between '18-30' and '46-60' in Kumasi

Ranks			
	Age (in years)	N	Mean Rank
I am happy to live in this city and I	18 - 30	139	82.83
personally do not	46 - 60	51	130.04
another place in this country.	Total	190	

Table A9.4. Table showing test statistics between '18-30' and '46-60' in Kumasi.

Test Statistics ^{a,b}	
	I am happy to live in this
	city and I personally do not
	seek to migrate to another
	place in this country.
Kruskal-Wallis H	29.515
df	1
Asymp. Sig.	.000
a. Kruskal Wallis Test	
b. Grouping Variable: Age (in years)	

Table A9.5. Table showing ranks between '18-30' and '>60' in Kumasi

Ranks			
	Age (in years)	N	Mean Rank
I am happy to live in	18 - 30	139	71.51
personally do not	>60	15	133.03
another place in this country.	Total	154	

Table A9.6. Table showing test statistics between '18-30' and '>60' in Kumasi.

Test Statistics ^{a,b}	
	I am happy to live in this city and I personally do not seek to migrate to another place in this country.
Kruskal-Wallis H	27.813
df	1
Asymp. Sig.	.000
a. Kruskal Wallis Test	
b. Grouping Variable: Age (in years)	

Table A9.7. Table showing ranks between '31-45' and '46-60' in Kumasi.

Ranks			
	Age (in years)	N	Mean Rank
I am happy to live in	31 - 45	113	74.70
this city and I personally do not seek	46 - 60	51	99.78
place in this country.	Total	164	

Table A9.8. Table showing test statistics between '31-45' and '46-60' in Kumasi.

Test Statistics ^{a,b}	
	I am happy to live in this city and I personally do not seek to migrate to another place in this country.
Kruskal-Wallis H	10.708
df	1
Asymp. Sig.	.001
a. Kruskal Wallis Test	
b. Grouping Variable: Age (in years)	-

Table A9.9. Table showing ranks between '31-45' and '>60' in Kumasi.

Ranks			
	Age (in years)	N	Mean Rank
I am happy to live in this city and I	31 - 45	113	59.35
personally do not seek to migrate to another place in this country.	>60	15	103.33
	Total	128	

Table A9.10. Table showing test statistics between '31-45' and '>60' in Kumasi.

Test Statistics ^{a,b}	
	I am happy to live in this city and I personally do not seek to migrate to another place in this country.
Kruskal-Wallis H	20.184
df	1
Asymp. Sig.	.000
a. Kruskal Wallis Test	
b. Grouping Variable: Age (in years)	-

Table A9.11. Table showing ranks between '46	6-60' and	'>60' in Kumasi.
--	-----------	------------------

Ranks			
	Age (in years)	N	Mean Rank
I am happy to live in	46 - 60	51	30.20
personally do not seek	>60	15	44.73
place in this country.	Total	66	

Table A9.12. Table showing test statistics between '46-60' and '>60' in Kumasi.

Test Statistics ^{a,b}	
	I am happy to live in this city and I
	personally do not seek to migrate to
	another place in this country.
Kruskal-Wallis H	7.744
df	1
Asymp. Sig.	.005
a. Kruskal Wallis Test	
b. Grouping Variable: Age (in years)	